# **United States Bankruptcy Court**

			District Of	
In re	Debtor	,		
				Case No Chapter

## GENERAL POWER OF ATTORNEY

[Abrogated]

The form is abrogated. Former Official Form 11A, although abrogated as an Official Form, continues to be available as a Director's Procedural Form.

Parties routinely modify the General Power of Attorney form to conform to state law, the needs of the case, or local practice. The exact language of the form is not needed. The proposed amendment to Rule 9009, however, restricts alteration of the Official Forms, except as provided in the rules or in a particular Official Form.

The Director's Procedural Forms are issued by the Director of the Administrative Office pursuant to Rule 9009 as an accommodation for the courts and parties. The procedural forms may be altered as needed and their use is not mandatory, unless required by local rule.

# **United States Bankruptcy Court**

			_ District Of _		
In re	Debtor	,			
				Case No	

## SPECIAL POWER OF ATTORNEY

[Abrogated]

The form is abrogated. Former Official Form 11B, although abrogated as an Official Form, continues to be available as a Director's Procedural Form.

Parties routinely modify the Special Power of Attorney form to conform to state law, the needs of the case, or local practice. The exact language of the form is not needed. The proposed amendment to Rule 9009, however, restricts alteration of the Official Forms, except as provided in the rules or in a particular Official Form.

The Director's Procedural Forms are issued by the Director of the Administrative Office pursuant to Rule 9009 as an accommodation for the courts and parties. The procedural forms may be altered as needed and their use is not mandatory, unless required by local rule.

Fill in this information to identify your case:	
United States Bankruptcy Court for the:	
Case number (If known): Chapter you are filing under: Chapter 7	☐ Check if this is an amended filing

## Official Form 101

# Voluntary Petition for Individuals Filing for Bankruptcy

12/15

The bankruptcy forms use you and Debtor 1 to refer to a debtor filing alone. A married couple may file a bankruptcy case together—called a joint case—and in joint cases, these forms use you to ask for information from both debtors. For example, if a form asks, "Do you own a car," the answer would be yes if either debtor owns a car. When information is needed about the spouses separately, the form uses Debtor 1 and Debtor 2 to distinguish between them. In joint cases, one of the spouses must report information as Debtor 1 and the other as Debtor 2. The same person must be Debtor 1 in all of the forms.

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write your name and case number (if known). Answer every question.

Part 1: Identify Yourself

		About Debtor 1:	About Debtor 2 (Spouse Only in a Joint Case):
1.	Your full name		
	Write the name that is on your government-issued picture identification (for example, your driver's license or	First name	First name
	passport).	Middle name	Middle name
	Bring your picture identification to your meeting with the trustee.	Last name	Last name
		Suffix (Sr., Jr., II, III)	Suffix (Sr., Jr., II, III)
2.	All other names you have used in the last 8	First name	First name
	years	Middle	Middle name
	Include your married or maiden names.	Middle name	Middle name
		Last name	Last name
		First name	First name
		Middle name	Middle name
		Last name	Last name
3.	Only the last 4 digits of your Social Security	xxx - xx	xxx - xx
	number or federal	OR	OR
	Individual Taxpayer Identification number (ITIN)	9 xx - xx	9 xx - xx

П	0	ht	h	r	1

First Name	Middle Name	Lact Name

Case number (# #	known)
------------------	--------

		About Debtor 1:	About Debtor 2 (Spouse Only in a Joint Case):
a 	Any business names and Employer dentification Numbers (EIN) you have used in the last 8 years notude trade names and doing business as names	☐ I have not used any business names or EINs.  Business name  Business name	☐ I have not used any business names or EINs.  Business name  Business name
		EIN	EIN — - — — — — — —
		EIN	EIN
5. <b>V</b>	Where you live		If Debtor 2 lives at a different address:
		Number Street	Number Street
		City State ZIP Code	City State ZIP Code
		If your mailing address is different from the one above, fill it in here. Note that the court will send any notices to you at this mailing address.	County  If Debtor 2's mailing address is different from yours, fill it in here. Note that the court will send any notices to this mailing address.
		Number Street	Number Street
		P.O. Box	P.O. Box
		City State ZIP Code	City State ZIP Code
t	Why you are choosing this district to file for pankruptcy	Check one:  Over the last 180 days before filing this petition, I have lived in this district longer than in any other district.  I have another reason. Explain. (See 28 U.S.C. § 1408.)	Check one:  Over the last 180 days before filing this petition, I have lived in this district longer than in any other district.  I have another reason. Explain. (See 28 U.S.C. § 1408.)

-	h.	tο	4

First Name Middle Name Last Name

Case number (if known)\_\_\_\_\_

Dort 1	`	į
Part∠	_	ı

Tell the Court About Your Bankruptcy Case

7.	The chapter of the Bankruptcy Code you are choosing to file under		ruptcy (Foter 7 oter 11 oter 12	a brief description of each, see form 2010)). Also, go to the top			U.S.C. § 342(b) for Individuals Filing e appropriate box.
8.	How you will pay the fee	local your subm with  I nee Appl  I req By la less pay to	court for self, you intring you a pre-p d to partication uest the w, a just than 15 he fee	or more details about how you may pay with cash, cashid rour payment on your behald rinted address.  The second of the second of the second of the second of the official poverty line in more detailed.	you mer's classiff, you lift, you lift you may may do to, when the lose the	ay pay. Typically heck, or money for attorney may pure choose this operate in Installment request this optivative your fee, and at applies to your is option, you mis	order. If your attorney is pay with a credit card or check tion, sign and attach the pats (Official Form 103A).  on only if you are filing for Chapter 7. Ind may do so only if your income is a family size and you are unable to just fill out the Application to Have the
9.	Have you filed for bankruptcy within the last 8 years?	☐ No☐ Yes.	District	\	When	MM / DD / YYYY	Case number  Case number  Case number
10.	Are any bankruptcy cases pending or being filed by a spouse who is not filing this case with you, or by a business partner, or by an affiliate?	☐ No☐ Yes.	District  Debtor	\	When	MM / DD / YYYY	Relationship to you  Case number, if known  Relationship to you  Case number, if known
11.	Do you rent your residence?	☐ No. ☐ Yes.	residen  No.	ur landlord obtained an eviction nee?  Go to line 12.			and do you want to stay in your  Against You (Form 101A) and file it with

bt	tor 1 First Name Middle Nam		Last Name		Case number (if know	vn)		
	riist waine - Middle Nan	ie	Last Name					
ar	rt 3: Report About Any E	Business	ses You Own as a So	le Proprietor				
	Are you a sole proprietor	□ N-	O- t- D-+ 1					
	of any full- or part-time	_	Go to Part 4.					
	business?	☐ Yes.	Name and location of bu	usiness				
	A sole proprietorship is a business you operate as an							
	individual, and is not a		Name of business, if any					
	separate legal entity such as a corporation, partnership, or							
	LLC.		Number Street					
	If you have more than one sole proprietorship, use a							
	separate sheet and attach it							
	to this petition.		City		State	ZIP Code		
			Check the appropriate b	•				
			Health Care Busines	•	. ,,,			
			☐ Single Asset Real E			;))		
			☐ Stockbroker (as defi					
			Commodity Broker (as defined in 11 U.S.C. § 101(6))					
			■ None of the above					
	are you a small business debtor? For a definition of small business debtor, see	_	I am not filing under Cha		small business deb	otor according to	the definition in	
	11 U.S.C. § 101(51D).		the Bankruptcy Code.					
		☐ Yes.	s. I am filing under Chapter 11 and I am a small business debtor according to the definition in the Bankruptcy Code.					
ar	t 4: Report if You Own	or Have	Any Hazardous Prop	erty or Any Prop	erty That Needs	Immediate A	ittention	
	_							
	Do you own or have any property that poses or is	☐ No						
	alleged to pose a threat	☐ Yes.	What is the hazard?					
	of imminent and identifiable hazard to							
ı	public health or safety?							
	Or do you own any property that needs							
	immediate attention?		If immediate attention i	is needed, why is it n	eeded?			
1	For example, do you own perishable goods, or livestock that must be fed, or a building that needs urgent repairs?							
			Where is the property?	<b>?</b>				
			r -r9-	Number Stree	et			
				City		State	ZIP Code	

Case number (if known)

#### Part 5:

Explain Your Efforts to Receive a Briefing About Credit Counseling

15. Tell the court whether you have received a briefing about credit counseling.

The law requires that you receive a briefing about credit counseling before you file for bankruptcy. You must truthfully check one of the following choices. If you cannot do so, you are not eligible to file.

If you file anyway, the court can dismiss your case, you will lose whatever filing fee you paid, and your creditors can begin collection activities again.

#### **About Debtor 1:**

You must check one:

☐ I received a briefing from an approved credit counseling agency within the 180 days before I filed this bankruptcy petition, and I received a certificate of completion.

Attach a copy of the certificate and the payment plan, if any, that you developed with the agency.

□ I received a briefing from an approved credit counseling agency within the 180 days before I filed this bankruptcy petition, but I do not have a certificate of completion.

Within 14 days after you file this bankruptcy petition, you MUST file a copy of the certificate and payment plan, if any.

□ I certify that I asked for credit counseling services from an approved agency, but was unable to obtain those services during the 7 days after I made my request, and exigent circumstances merit a 30-day temporary waiver of the requirement.

To ask for a 30-day temporary waiver of the requirement, attach a separate sheet explaining what efforts you made to obtain the briefing, why you were unable to obtain it before you filed for bankruptcy, and what exigent circumstances required you to file this case.

Your case may be dismissed if the court is dissatisfied with your reasons for not receiving a briefing before you filed for bankruptcy.

If the court is satisfied with your reasons, you must still receive a briefing within 30 days after you file. You must file a certificate from the approved agency, along with a copy of the payment plan you developed, if any. If you do not do so, your case may be dismissed.

Any extension of the 30-day deadline is granted only for cause and is limited to a maximum of 15 days.

I am not required	o receive	a briefing	about
credit counseling	because o	of:	

☐ Incapacity. I have a mental illness or a mental deficiency that makes me

incapable of realizing or making rational decisions about finances.

☐ **Disability.** My physical disability causes me to be unable to participate in a

briefing in person, by phone, or through the internet, even after I reasonably tried to do so.

Active duty. I am currently on active military duty in a military combat zone.

If you believe you are not required to receive a briefing about credit counseling, you must file a motion for waiver of credit counseling with the court.

#### About Debtor 2 (Spouse Only in a Joint Case):

You must check one:

 ☐ I received a briefing from an approved credit counseling agency within the 180 days before I filed this bankruptcy petition, and I received a certificate of completion.

Attach a copy of the certificate and the payment plan, if any, that you developed with the agency.

☐ I received a briefing from an approved credit counseling agency within the 180 days before I filed this bankruptcy petition, but I do not have a certificate of completion.

Within 14 days after you file this bankruptcy petition, you MUST file a copy of the certificate and payment plan, if any.

■ I certify that I asked for credit counseling services from an approved agency, but was unable to obtain those services during the 7 days after I made my request, and exigent circumstances merit a 30-day temporary waiver of the requirement.

To ask for a 30-day temporary waiver of the requirement, attach a separate sheet explaining what efforts you made to obtain the briefing, why you were unable to obtain it before you filed for bankruptcy, and what exigent circumstances required you to file this case.

Your case may be dismissed if the court is dissatisfied with your reasons for not receiving a briefing before you filed for bankruptcy.

If the court is satisfied with your reasons, you must still receive a briefing within 30 days after you file. You must file a certificate from the approved agency, along with a copy of the payment plan you developed, if any. If you do not do so, your case may be dismissed.

Any extension of the 30-day deadline is granted only for cause and is limited to a maximum of 15 days.

I am not required to receive a briefing a	bout
credit counseling because of:	

☐ Incapacity. I have a mental illness or a mental

deficiency that makes me incapable of realizing or making rational decisions about finances.

Disability. My physical disability causes me to be unable to participate in a briefing in person, by phone, or

briefing in person, by phone, or through the internet, even after I reasonably tried to do so.

Active duty. I am currently on active military duty in a military combat zone.

If you believe you are not required to receive a briefing about credit counseling, you must file a motion for waiver of credit counseling with the court.

1	4	 tr	L	-	

First Name Middle Name Last Name

Case number (if known)\_\_\_\_\_

Part 6: Answer These Que:	stions for Reporting Purpos	ses	
16. What kind of debts do you have?	as "incurred by an individu No. Go to line 16b. Yes. Go to line 17.	ual primarily for a personal, family, c	er debts are defined in 11 U.S.C. § 101(8) or household purpose."  debts are debts that you incurred to obtain
		nvestment or through the operation	
	Yes. Go to line 17.		
	16c. State the type of debts yo	u owe that are not consumer debts	or business debts.
17. Are you filing under Chapter 7?	☐ No. I am not filing under C	chapter 7. Go to line 18.	
Do you estimate that after any exempt property is		ter 7. Do you estimate that after an es are paid that funds will be availa	y exempt property is excluded and ble to distribute to unsecured creditors?
excluded and administrative expenses	□ No		
are paid that funds will be available for distribution to unsecured creditors?	☐ Yes		
18. How many creditors do	<b>1</b> -49	1,000-5,000	25,001-50,000
you estimate that you owe?	☐ 50-99 ☐ 100-199	5,001-10,000 10,001-25,000	☐ 50,001-100,000 ☐ More than 100,000
	200-999	- 10,001 20,000	<u> </u>
19. How much do you	\$0-\$50,000	□ \$1,000,001-\$10 million	<b>\$500,000,001-\$1</b> billion
estimate your assets to be worth?	\$50,001-\$100,000	\$10,000,001-\$50 million	\$1,000,000,001-\$10 billion
20 11011111	□ \$100,001-\$500,000 □ \$500,001-\$1 million	\$50,000,001-\$100 million \$100,000,001-\$500 millio	
20. How much do you	\$0-\$50,000	□ \$1,000,001-\$10 million	\$500,000,001-\$1 billion
estimate your liabilities	\$50,001-\$100,000	☐ \$10,000,001-\$50 million	□ \$1,000,000,001-\$10 billion
to be?	□ \$100,001-\$500,000 □ \$500,001-\$1 million	\$50,000,001-\$100 million \$100,000,001-\$500 millio	_ · · · · ·
Part 7: Sign Below	<b>4</b> \$500,001-\$1 million	<b>4</b> 100,000,001-4000 millio	III I I I I I I I I I I I I I I I I I
For you	I have examined this petition, a correct.	and I declare under penalty of perjui	ry that the information provided is true and
			ceed, if eligible, under Chapter 7, 11,12, or 13 der each chapter, and I choose to proceed
		nd I did not pay or agree to pay som and read the notice required by 11	neone who is not an attorney to help me fill out U.S.C. § 342(b).
	I request relief in accordance w	vith the chapter of title 11, United St	tates Code, specified in this petition.
		sult in fines up to \$250,000, or impri	taining money or property by fraud in connection sonment for up to 20 years, or both.
	×	×	
	Signature of Debtor 1	Siç	gnature of Debtor 2
	Executed on MM / DD /		tecuted on

ebtor 1				Case number (if known)
	Circl Name	Middle None	Loot None	

For your attorney, if you are represented by one

If you are not represented by an attorney, you do not need to file this page. I, the attorney for the debtor(s) named in this petition, declare that I have informed the debtor(s) about eligibility to proceed under Chapter 7, 11, 12, or 13 of title 11, United States Code, and have explained the relief available under each chapter for which the person is eligible. I also certify that I have delivered to the debtor(s) the notice required by 11 U.S.C. § 342(b) and, in a case in which § 707(b)(4)(D) applies, certify that I have no knowledge after an inquiry that the information in the schedules filed with the petition is incorrect.

	Date	
Signature of Attorney for Debtor		MM / DD /YYYY
Printed name		
Firm name		
Number Street		
City	State	ZIP Code
Contact phone	Email addre	ss

For you if you are filing this bankruptcy without an attorney

If you are represented by an attorney, you do not need to file this page. The law allows you, as an individual, to represent yourself in bankruptcy court, but you should understand that many people find it extremely difficult to represent themselves successfully. Because bankruptcy has long-term financial and legal consequences, you are strongly urged to hire a qualified attorney.

To be successful, you must correctly file and handle your bankruptcy case. The rules are very technical, and a mistake or inaction may affect your rights. For example, your case may be dismissed because you did not file a required document, pay a fee on time, attend a meeting or hearing, or cooperate with the court, case trustee, U.S. trustee, bankruptcy administrator, or audit firm if your case is selected for audit. If that happens, you could lose your right to file another case, or you may lose protections, including the benefit of the automatic stay.

You must list all your property and debts in the schedules that you are required to file with the court. Even if you plan to pay a particular debt outside of your bankruptcy, you must list that debt in your schedules. If you do not list a debt, the debt may not be discharged. If you do not list property or properly claim it as exempt, you may not be able to keep the property. The judge can also deny you a discharge of all your debts if you do something dishonest in your bankruptcy case, such as destroying or hiding property, falsifying records, or lying. Individual bankruptcy cases are randomly audited to determine if debtors have been accurate, truthful, and complete. Bankruptcy fraud is a serious crime; you could be fined and imprisoned.

If you decide to file without an attorney, the court expects you to follow the rules as if you had hired an attorney. The court will not treat you differently because you are filing for yourself. To be successful, you must be familiar with the United States Bankruptcy Code, the Federal Rules of Bankruptcy Procedure, and the local rules of the court in which your case is filed. You must also be familiar with any state exemption laws that apply.

Are you aware that filing for bankruptcy is a serious consequences?	action with long-te	rm financial and legal
□ No □ Yes		
Are you aware that bankruptcy fraud is a serious cr inaccurate or incomplete, you could be fined or imp		bankruptcy forms are
☐ No ☐ Yes		
Did you pay or agree to pay someone who is not ar ☐ No	n attorney to help yo	ou fill out your bankruptcy forms?
Yes. Name of PersonAttach Bankruptcy Petition Preparer's Notice,	Declaration, and Sig	gnature (Official Form 119).
By signing here, I acknowledge that I understand the have read and understood this notice, and I am awa attorney may cause me to lose my rights or propert	are that filing a banl	kruptcy case without an
,	×	
Signature of Debtor 1	Signature of De	btor 2
Date MM / DD / YYYY	Date	MM / DD / YYYY
Contact phone	Contact phone	
Cell phone	_ Cell phone	
Email address	Email address	

Official Form 101, Voluntary Petition for Individuals Filing for Bankruptcy, applies only in cases of individual debtors. Form 101 replaces Official Form 1, Voluntary Petition. It is renumbered to distinguish it from the forms used by non-individual debtors, such as corporations, and includes stylistic changes throughout the It is revised as part of the Forms Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. Because the goals of the Forms Modernization Project include improving the interface between technology and the forms so as to increase efficiency and reduce the need to produce the same information in multiple formats, many of the open-ended questions and multiple-part instructions have been replaced with more specific questions.

Official Form 101 has been substantially reorganized. References to Exhibits A, B, C, and D, and the exhibits themselves, have been eliminated because the requested information is now asked in the form or is not applicable to individual debtors.

Part 1, *Identify Yourself*, line 6, replaces the venue box from page 2 of Official Form 1 and deletes venue questions that pertain only to non-individuals.

Part 2, Tell the Court About Your Bankruptcy Case, line 7, removes choices for chapters 9 and 15 filings because they do not pertain to individuals. The status of "being filed" is added to the question regarding bankruptcy cases pending or filed by a spouse, business partner, or affiliate (line 10). Lastly, the question "Do you rent your residence?" (line 11) and Official Forms 101A, Initial Statement About an Eviction Judgment Against You, and 101B, Statement About Payment of an Eviction Judgment Against You, replace "certification by a debtor who resides as a tenant of residential property," on page 2 of Official Form 1.

Part 3, Report About Any Businesses You Own as a Sole Proprietor, line 12, incorporates options from the "nature of business" box from page 1 of Official Form 1 that would apply to individual debtors, thus eliminating checkboxes for railroads and clearing banks. Part 3, line 13, also eliminates a checkbox to report whether a plan was filed with the petition, or if plan acceptances were solicited prepetition. Additionally, line 13 rephrases the question relating to whether a debtor filing under Chapter 11 is a small business debtor.

Part 4, Report if You Own or Have Any Hazardous Property or Any Property That Needs Immediate Attention, line 14, replaces Exhibit C from Official Form 1 and adds the category of "property that needs immediate attention."

Part 5, Explain Your Efforts to Receive a Briefing About Credit Counseling (line 15), replaces Exhibit D from Official Form 1. Additionally, this part describes incapacity and disability using a simplified definition, tells the debtor of the ability to file a motion for a waiver, and eliminates statutory reference about districts where credit counseling does not apply because such districts are rare.

Part 6, Answer These Questions for Reporting Purposes (line 16c), provides a text field for the debtor to describe the type of debts owed if the debtor believes they are neither primarily consumer nor business debts.

Part 7, Sign Below, deletes from the debtor's declaration the phrase "to the best of my knowledge, information, and belief" in order to conform to the language of 28 U.S.C. § 1746. See Rule 1008. This part combines the two attorney signature blocks into one certification and eliminates signature lines for corporations/partnerships and chapter 15 Foreign Representative. The declaration and signature section for a non-attorney bankruptcy petition preparer (BPP) has also been removed as unnecessary. The same declaration, required under 11 U.S.C. § 110, is contained in Official Form 119. That form must be completed and signed by the BPP and filed with each document prepared by a BPP.

Fill in this in	formation to identify	your case:	
Debtor 1			
_	First Name	Middle Name	Last Name
Debtor 2			
(Spouse, if filing)	First Name	Middle Name	Last Name
United States B	Bankruptcy Court for the:		District of(State)
Case number (If known)			

## Official Form 101A

# Initial Statement About an Eviction Judgment Against You

12/15

File this form with the court and serve a copy on your landlord when you first file bankruptcy only if:

- you rent your residence; and
- your landlord has obtained a judgment for possession in an eviction, unlawful detainer action, or similar proceeding (called eviction judgment) against you to possess your residence.

Landlord's name			 <del> </del>	
Landlord's address				
	Number	Street		
	Citv		 State	ZIP Code

If you want to stay in your rented residence after you file your case for bankruptcy, also complete the certification below.

Certification About	Applicable Law and Deposit of Re	nt
I certify under penalty of	perjury that:	
	ther nonbankruptcy law that applies to the gay in my residence by paying my landlord t	udgment for possession (eviction judgment), he entire delinquent amount.
-	kruptcy court clerk a deposit for the rent the rent the nert then to the for Individuals Filing for Bankruptcy (Office)	at would be due during the 30 days after I file sial Form 101).
×		*
Signature of Debt	or 1	Signature of Debtor 2
Date	/YYYY	Date
Stay of Eviction: (a)	and served your landlord with a copy of the	necked both boxes above, signed the form to certify that both apply, his statement, the automatic stay under 11 U.S.C. § 362(a)(3) will gainst you for 30 days after you file your <i>Voluntary Petition for</i> Form 101).
(b)	receive the protection of the automatic sta amount to your landlord as stated in the cout Statement About Payment of an Evic	to stay in your residence after that 30-day period and continue to ay under 11 U.S.C. § 362(a)(3), you must pay the entire delinquent exiction judgment before the 30-day period ends. You must also fill the before the 30-day period ends. You file it with the la copy of it before the 30-day period ends.

Check the Bankruptcy Rules (<a href="http://www.uscourts.gov/rules-policies/current-rules-practice-procedure">http://www.uscourts.gov/rules-policies/current-rules-practice-procedure</a>) and the local court's website (to find your court's website go to <a href="http://www.uscourts.gov/court-locator">http://www.uscourts.gov/court-locator</a>) for any specific requirements that you might have to meet to serve this statement. 11 U.S.C. §§ 362(b)(22) and 362(l)

Fill in this in	formation to ide	ntify your case:	
Debtor 1	FintNews	No. 10 and 10 an	
	First Name	Middle Name	Last Name
Debtor 2			
(Spouse, if filing)	First Name	Middle Name	Last Name
United States E	Bankruptcy Court for	the:	District of (State)
Case number			<del></del>

## Official Form 101B

## Statement About Payment of an Eviction Judgment Against You

12/15

Fill out this form only if:

- you filed Initial Statement About an Eviction Judgment Against You (Official Form 101A); and
- you served a copy of Form 101A on your landlord; and
- you want to stay in your rented residence for more than 30 days after you file your Voluntary Petition for Individuals Filing for Bankruptcy (Official Form 101).

File this form within 30 days after you file your *Voluntary Petition for Individuals Filing for Bankruptcy* (Official Form 101). Also serve a copy on your landlord within that same time period.

Certification About Applicable Law and Paymo	
I certify under penalty of perjury that (Check all that apply	):
☐ Under the state or other nonbankruptcy law that applies	to the judgment for possession (eviction
judgment), I have the right to stay in my residence by pa	
☐ Within 30 days after I filed my Voluntary Petition for Indi	ividuals Filing for Bankruptov (Official
Form 101), I have paid my landlord the entire amount I	• , , ,
	• , , ,
Form 101), I have paid my landlord the entire amount I	• , , ,
Form 101), I have paid my landlord the entire amount I (eviction judgment).	owe as stated in the judgment for possession
Form 101), I have paid my landlord the entire amount I (eviction judgment).	owe as stated in the judgment for possession
Form 101), I have paid my landlord the entire amount I (eviction judgment).  Signature of Debtor 1  Date	owe as stated in the judgment for possession  Signature of Debtor 2  Date

### You must serve your landlord with a copy of this form.

Check the Bankruptcy Rules (<a href="http://www.uscourts.gov/rules-policies/current-rules-practice-procedure">http://www.uscourts.gov/court-locator to find</a> your court's website) for any specific requirements that you might have to meet to serve this statement.

Official Form 101A, *Initial Statement About an Eviction Judgment Against You*, and Official Form 101B, *Statement About Payment of an Eviction Judgment Against You*, are new forms promulgated as part of the Forms Modernization Project. They replace the "Certification by a Debtor Who Resides as a Tenant of Residential Property" section on Official Form 1, Voluntary Petition. The forms apply only in cases of individual debtors.

Official Form 101A explains that debtors need to complete and file the form only if their landlord has a judgment for possession or an eviction judgment against them and they rent their residence. The form further explains that if the debtor wishes to stay in their residence for 30 days after filing their bankruptcy petition, the certification must be completed. The form adds references to the provisions in the Bankruptcy Code that specify when debtor-tenants subject to eviction may remain in their residence after filing for bankruptcy.

The form eliminates the checkboxes that the debtor has served the landlord with the certification and paid the court the rent that would be due during the 30 days after the filing of the bankruptcy petition. Instead, debtors are required to certify under penalty of perjury that the rent has been paid to the court, and the instructions direct debtors to serve a copy of the statement on the landlord.

The form eliminates the checkbox that the debtor claims there are circumstances under applicable nonbankruptcy law under which the debtor would be permitted to cure the monetary default that gave rise to the judgment for possession (or eviction judgment) and remain in residence. Instead, debtors are required to certify under penalty of perjury that they have the right to stay in their residence under state law or other nonbankruptcy law by paying their landlord the entire delinquent amount.

Official Form 101B is new. If debtors wish to stay in their residence for more than 30 days after filing the petition, they must complete, file, and serve the form within 30 days after the petition is filed. Under Official Form 101B, debtors certify under penalty of perjury that they have the right to stay in their residence under state law or other nonbankruptcy law by paying their landlord the entire delinquent amount and that they have paid their landlord the entire amount owed as stated in the judgment for possession or in the eviction judgment.

Fill in this information to identify your case:		
Debtor 1		
First Name Middle Name	Last Name	_
Debtor 2 (Spouse, if filing) First Name Middle Name	Last Name	_
United States Bankruptcy Court for the:	District of(State)	_
Case number (If known)		
		Check if this is an amended filing
Official Form 103A		
Application for Individuals	s to Pay the I	Filing Fee in Installments 12/15
Be as complete and accurate as possible. If two mainformation.	rried people are filing tog	ether, both are equally responsible for supplying correct
Part 1: Specify Your Proposed Payment	Timetable	
Which chapter of the Bankruptcy Code	☐ Chapter 7	
are you choosing to file under?	☐ Chapter 11	
	☐ Chapter 12	
	☐ Chapter 13	
2. You may apply to pay the filing fee in up to four installments. Fill in the amounts you	Vou propose to pay	
propose to pay and the dates you plan to	You propose to pay	
pay them. Be sure all dates are business days. Then add the payments you propose	\$	☐ With the filing of the petition
to pay.  You must propose to pay the entire fee no	· · · · · · · · · · · · · · · · · · ·	On or before this date MM / DD / YYYY
later than 120 days after you file this bankruptcy case. If the court approves your	\$	On or before this date
application, the court will set your final payment timetable.		MM / DD / YYYY
payment unletable.	\$	On or before this date
+	\$	On or before this date
Total	\$	■ Your total must equal the entire fee for the chapter you checked in line 1.
Part 2: Sign Below		
By signing here, you state that you are unable to understand that:	pay the full filing fee at o	nce, that you want to pay the fee in installments, and that you
You must pay your entire filing fee before you m preparer, or anyone else for services in connect		transfer any more property to an attorney, bankruptcy petition ase.
You must pay the entire fee no later than 120 debts will not be discharged until your entire fee		nkruptcy, unless the court later extends your deadline. Your
If you do not make any payment when it is due, may be affected.	your bankruptcy case may	be dismissed, and your rights in other bankruptcy proceedings
×		<b>x</b>
Signature of Debtor 1 Sig	nature of Debtor 2	Your attorney's name and signature, if you used one
Date Dat	e MM / DD / YYYY	Date
		20

Fill in this information to identify the case:							
Debtor 1							
	First Name	Middle Name	Last Name				
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name				
(Opouse, it illing)	i iist Name	Middle Name	Last Name				
United States I	Bankruptcy Court fo	or the:	District of				
			(State)				
Case number (If known)							
Chapter filing	under:						
	,		☐ Chapter 7				
			Chapter 11				
			☐ Chapter 12				
			☐ Chapter 13				

# Order Approving Payment of Filing Fee in Installments

After considering the *Application for Individuals to Pay the Filing Fee in Installments* (Official Form 103A), the court orders that:

- [ ] The debtor(s) may pay the filing fee in installments on the terms proposed in the application.
- [ ] The debtor(s) must pay the filing fee according to the following terms:

		You must pay	On or before this date
		\$	Month / day / year
		\$	Month / day / year
		\$	Month / day / year
	+	\$	Month / day / year
Total		\$	

Until the filing fee is paid in full, the debtor(s) must not make any additional payment or transfer any additional property to an attorney or to anyone else for services in connection with this case.

	By the court:	
Month / day / year	•	United States Bankruptcy Judge

The form number is updated to comport with the form numbering style developed as part of the Forms Modernization Project. Other stylistic changes were made throughout the form.

Fill in this in	formation to iden	tify your case:		
Debtor 1	First Name	Middle Name	Last Name	
Debtor 2				
(Spouse, if filing)		Middle Name	Last Name	
	Bankruptcy Court for t	the:	District of	State)
Case number (If known)				

## Official Form 103B

# Application to Have the Chapter 7 Filing Fee Waived

12/15

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct

	ormation. If more space is needed, at known).	tach a separate sheet to	this form. On the top of any a	dditional pages, w	vrite your name and case numb
F	Part 1: Tell the Court About Y	our Family and Your F	Family's Income		
1.	What is the size of your family? Your family includes you, your spouse, and any dependents listed on Schedule J: Your Expenses (Official Form 106J).	Check all that apply:  You Your spouse Your dependents	How many dependents?	Total number of p	people
2.	Fill in your family's average monthly income.  Include your spouse's income if your spouse is living with you, even if your spouse is not filing.  Do not include your spouse's income if you are separated and your spouse is not filing with you.	value (if known) of any non that you receive, such as fo Supplemental Nutrition Ass subsidies. If you have already filled of line 10 of that schedule.	spouse's income. Include the n-cash governmental assistance ood stamps (benefits under the sistance Program) or housing but Schedule I: Your Income, see wernmental assistance that you monthly net income	You  Your spouse  Subtotal	That person's average monthly net income (take-home pay)  \$  + \$  \$  - \$  \$
3.	Do you receive non-cash governmental assistance?	☐ No ☐ Yes. Describe	Type of assistance		
4.	Do you expect your family's average monthly net income to increase or decrease by more than 10% during the next 6 months?	☐ No ☐ Yes. Explain			
5.	Tell the court why you are unable to installments within 120 days. If you h circumstances that cause you to not be fee in installments, explain them.	ave some additional			

		 Case
=:	14:11: 11	

	First Name Middle Name	Last Name					
P	Part 2: Tell the Court About Yo	our Monthly Expense	es				
6.	Estimate your average monthly exp Include amounts paid by any governm reported on line 2.		\$				
	If you have already filled out Schedule line 22 from that form.	J, Your Expenses, copy					
7.	Do these expenses cover anyone who is not included in your family as reported in line 1?	□ No □ Yes. Identify who					
8.	Does anyone other than you regularly pay any of these expenses?  If you have already filled out Schedule I: Your Income, copy the total from line 11.	☐ No☐ Yes. How much do	you regu	ularly receive	e as contributions	? \$ mont	hly
9.	Do you expect your average monthly expenses to increase or decrease by more than 10% during the next 6 months?	☐ No☐ Yes. Explain					
Pa	Tell the Court About Yo	our Property					
If	you have already filled out Schedule	e A/B: Property (Official	Form 10	06A/B) attac	h copies to this	application and go	to Part 4.
10.	. How much cash do you have?						
	Examples: Money you have in your wallet, in your home, and on hand when you file this application	Cash:	\$		_		
11.	Bank accounts and other deposits of money?		Institu	tion name:			Amount:
	Examples: Checking, savings, money market, or other financial	Checking account:	Checking account:				\$
	accounts; certificates of deposit; shares in banks, credit unions,	Savings account:					\$
	brokerage houses, and other similar institutions. If you have	Other financial accounts:					\$
	nore than one account with the same institution, list each. Do not enclude 401(k) and IRA accounts.	Other financial accounts:					\$
12	. Your home? (if you own it outright or						
	are purchasing it)  Examples: House, condominium,	Number Street				Current value:	\$
	manufactured home, or mobile home	City		State	ZIP Code	Amount you owe on mortgage and liens:	\$
13.	Other real estate?					Current value:	\$
		Number Street				Amount you owe	\$
		City		State	ZIP Code	on mortgage and liens:	\$
14	The vehicles you own?	Make:		_			
	Examples: Cars, vans, trucks,	Model:		-		Current value:	\$
	sports utility vehicles, motorcycles, tractors, boats	Year:				Amount you owe on liens:	\$
		Mileage		-			
		Make: Model:		-		Current value:	\$
		Year:		-		Amount you owe	· · · · · · · · · · · · · · · · · · ·
		Mileage		_		on liens:	\$

Debtor 1 First Name Middle Name	Last Nan	ne	Case number (#	known)		
15. Other assets?  Do not include household items and clothing.	Describe	e the other assets:		Current va		\$ \$
16. Money or property due you?  Examples: Tax refunds, past due or lump sum alimony, spousal support, child support, maintenance, divorce or property settlements, Social Security benefits, workers' compensation, personal injury recovery		es you the money or property?	How much   \$\$		paymen  No	believe you will likely receive t in the next 180 days? Explain:
Part 4: Answer These Addition	al Ques	tions				
17. Have you paid anyone for services for this case, including filling out this application, the bankruptcy filing package, or the schedules?	☐ No☐ Yes.	. Whom did you pay? Check all that ap  An attorney  A bankruptcy petition preparer, par  Someone else	ralegal, or typin			How much did you pay?
18. Have you promised to pay or do you expect to pay someone for services for your bankruptcy case?	☐ No☐ Yes.	. Whom do you expect to pay? Check  ☐ An attorney ☐ A bankruptcy petition preparer, par ☐ Someone else	ralegal, or typin	-	_	How much do you expect to pay?
19. Has anyone paid someone on your behalf for services for this case?	☐ No☐ Yes.	. Who was paid on your behalf? Check all that apply:  ☐ An attorney ☐ A bankruptcy petition preparer, paralegal, or typing service ☐ Someone else	Who paid? Check all th Parent Brother Friend Pastor of Someon	or sister		How much did someone else pay?
20. Have you filed for bankruptcy within the last 8 years?	□ No □ Yes.	District  District  District	_ When MM/ DI	O/ YYYY	ase numb	er

Signature of Debtor 1

Date MM / DD / YYYY

Signature of Debtor 2

Date MM / DD / YYYY

Fill in this in	formation to i	dentify the case:		
	First Name	Middle Name	Last Name	
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name	
United States	Bankruptcy Court	for the:	District of	
Case number			(State)	
(If known)				
Ordor	on the	Application to	allovo tho C	hantar 7 Filing Foo Waiyad
Order (	Jii tile i	Аррпсаноп и	nave the C	hapter 7 Filing Fee Waived
	dering the de the applicati		ave the Chapter 7 Filin	g Fee Waived (Official Form 103B), the court
[ ] Granted		the court may order the ring the bankruptcy cas		in the future if developments in was unwarranted.
[ ] Denied	The debto	r must pay the filing fee	e according to the follo	wing terms:
		You must pay	On or before this date	-
		\$	Month / day / year	
		\$	Month / day / year	
		¢		

Month / day / year Month / day / year **Total** 

If the debtor would like to propose a different payment timetable, the debtor must file a motion promptly with a payment proposal. The debtor may use Application for Individuals to Pay the Filing Fee in Installments (Official Form 103A) for this purpose. The court will consider it.

The debtor must pay the entire filing fee before making any more payments or transferring any more property to an attorney, bankruptcy petition preparer, or anyone else in connection with the bankruptcy case. The debtor must also pay the entire filing fee to receive a discharge. If the debtor does not make any payment when it is due, the bankruptcy case may be dismissed and the debtor's rights in future bankruptcy cases may be affected.

[ ] Scheduled for hearing	hearing	for h	led	chedu	] S	[ ]	
---------------------------	---------	-------	-----	-------	-----	-----	--

A hearing to conside	r the debto	or's application wil	ll be held	
on	_ at	AM / PM at		
Month / day / year			Address of courthouse	
If the debtor does no	t appear a	t this hearing, the	court may deny the application.	
		By the court:		
Month / day / year			United States Bankruptcy Judge	26

The form number is updated to comport with the form numbering style developed as part of the Forms Modernization Project. Other stylistic changes were made throughout the form.

Fill in this in	formation to identify y	our case:	
Debtor 1			
_	First Name	Middle Name	Last Name
Debtor 2			
(Spouse, if filing)	First Name	Middle Name	Last Name
United States E	Bankruptcy Court for the: _		District of(State)
Case number (If known)			

# ☐ Check if this is an amended filing

### Official Form 104

For Individual Chapter 11 Cases: List of Creditors Who Have the 20 Largest Unsecured Claims Against You and Are Not Insiders

12/15

If you are an individual filing for bankruptcy under Chapter 11, you must fill out this form. If you are filing under Chapter 7, Chapter 12, or Chapter 13, do not fill out this form. Do not include claims by anyone who is an *insider*. Insiders include your relatives; any general partners; relatives of any general partners; partnerships of which you are a general partner; corporations of which you are an officer, director, person in control, or owner of 20 percent or more of their voting securities; and any managing agent, including one for a business you operate as a sole proprietor. 11 U.S.C. § 101. Also, do not include claims by secured creditors unless the unsecured claim resulting from inadequate collateral value places the creditor among the holders of the 20 largest unsecured claims.

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information.

Part 1

List the 20 Unsecured Claims in Order from Largest to Smallest. Do Not Include Claims by Insiders.

		Unsecured claim
1	What is the nature of the claim?	<del></del> \$
Creditor's Name	As of the date you file, the claim is: Check all that apply.	
Creditor's Name	Contingent	
Number Street	Unliquidated	
	Disputed	
	☐ None of the above apply	
City State ZIP Code	Does the creditor have a lien on your property? ☐ No	
Contact	Yes. Total claim (secured and unsecured): \$	
	Value of security:	
Contact phone	Unsecured claim \$	
2	What is the nature of the claim?	
Creditor's Name	As of the date you file, the claim is: Check all that apply.	
	☐ Contingent	
Number Street	☐ Unliquidated	
	☐ Disputed	
	☐ None of the above apply	
City State ZIP Code	Does the creditor have a lien on your property? ☐ No	
Contact	Yes. Total claim (secured and unsecured): \$	
	Value of security:	
Contact phone	Unsecured claim \$	

9	h	١,	1

						Unsecured claim
3				What is the nature of the claim?		
	Creditor's Name			What is the nature of the claim.		\$
				As of the date you file, the claim is: Check	all that apply.	
	Number Street			☐ Contingent		
				☐ Unliquidated		
				☐ Disputed		
				■ None of the above apply		
	City	State	ZIP Code	Does the creditor have a lien on your prop	perty?	
				☐ No		
	Contact			Yes. Total claim (secured and unsecured):	\$	_
				Value of security:	\$	_
	Contact phone			Unsecured claim	\$	_
1						_
	Out if the de Name			What is the nature of the claim?		\$
	Creditor's Name			As of the date you file, the claim is: Check	all that apply.	
	Number Street			Contingent		
	Tunis.			Unliquidated		
				☐ Disputed		
				☐ None of the above apply		
	City	State	ZIP Code	Does the creditor have a lien on your prop	perty?	
				□ No		
	Contact			☐ Yes. Total claim (secured and unsecured):	\$	_
				Value of security:	\$	_
	Contact phone			Unsecured claim	\$	_
7						
5	Creditor's Name			What is the nature of the claim?		\$
	Creditor's Name			As of the date you file, the claim is: Check	all that apply.	
	Number Street			☐ Contingent		
				Unliquidated		
				☐ Disputed		
				■ None of the above apply		
	City	State	ZIP Code	Does the creditor have a lien on your pro	perty?	
				No		
	Contact			☐ Yes. Total claim (secured and unsecured):	\$	_
				Value of security:	\$	_
	Contact phone			Unsecured claim	\$	
6				What is the nature of the claim?		\$
	Creditor's Name			As of the date you file, the claim is: Check		•
				Contingent	. ан тпасарргу.	
	Number Street			☐ Unliquidated		
				☐ Disputed		
				☐ None of the above apply		
	City	State	ZIP Code	Does the creditor have a lien on your pro	perty?	
				□ No	•	
	Overland			☐ Yes. Total claim (secured and unsecured):	\$	_
	Contact			Value of security:	\$	_
	Contact phone			Unsecured claim	\$	_
7						
				What is the nature of the claim?		\$
	Creditor's Name			As of the date you file, the claim is: Check	all that apply.	
	Number Street			Contingent		
				Unliquidated		
				Disputed		
				☐ None of the above apply		
	City	State	ZIP Code	Does the creditor have a lien on your prop	perty?	
				□ No		
	Contact			☐ Yes. Total claim (secured and unsecured):	\$	_
				Value of security:	\$	_
	Contact phone			Unsecured claim	\$	_

_					
П	10	ht	· ^	r	1

First Name	Middle Name	Last Name	

					Unsecured claim
8			What is the nature of the slaim?		\$
	Creditor's Name		What is the nature of the claim?		Ψ
			As of the date you file, the claim is: Check a  Contingent	all that apply.	
	Number Street		☐ Unliquidated		
			Disputed		
			☐ None of the above apply		
	City	State ZIP Code	Does the creditor have a lien on your prop	erty?	
	,		□ No	<b></b>	
			Yes. Total claim (secured and unsecured):	\$	_
	Contact		Value of security:	\$	
	Contact phone		Unsecured claim	\$	_
9					
J	Creditor's Name		What is the nature of the claim?		\$
			As of the date you file, the claim is: Check a	all that apply.	
	Number Street		- Contingent		
			Unliquidated		
			Disputed		
			None of the above apply		
	City	State ZIP Code	Does the creditor have a lien on your prop	erty?	
			□ No		
	Contact		Yes. Total claim (secured and unsecured):	\$	_
			Value of security:	\$	_
	Contact phone		Unsecured claim	\$	_
10			Mile at the discount of the colline O		\$
. •	Creditor's Name		What is the nature of the claim?		Ψ
	oroanor o riamo		As of the date you file, the claim is: Check a	all that apply.	
	Number Street		_ Contingent		
			Unliquidated		
			Disputed		
			☐ None of the above apply		
	City	State ZIP Code	<ul> <li>Does the creditor have a lien on your prop</li> <li>No</li> </ul>	erty?	
			Yes. Total claim (secured and unsecured):	•	
	Contact			<b>5</b>	_
			Value of security:	\$	_
	Contact phone		Unsecured claim	\$	_
11					\$
ш	Creditor's Name		What is the nature of the claim?		\$
			As of the date you file, the claim is: Check a	all that apply.	
	Number Street		Contingent		
			Unliquidated		
			<ul><li>☐ Disputed</li><li>☐ None of the above apply</li></ul>		
	City	State ZIP Code		out.c2	
	Опу	State ZIF COde	Does the creditor have a lien on your prop  ☐ No	erty f	
			<ul> <li>Yes. Total claim (secured and unsecured):</li> </ul>	\$	
	Contact			\$	
			Value of security:	\$	_
	Contact phone		Unsecured claim	\$	_
12			What is the nature of the claim?		\$
	Creditor's Name		As of the date you file, the claim is: Check a	all that apply.	
			Contingent		
	Number Street		Unliquidated		
			Disputed		
			□ None of the above apply		
	Cit.	Chata	Does the creditor have a lien on your prop	erty?	
	City	State ZIP Code	□ No		
			Yes. Total claim (secured and unsecured):	\$	_
	Contact		<ul> <li>Value of security:</li> <li>Unsecured claim</li> </ul>	\$ \$	_
			Shoodarda didiiri	Ψ	_
	Contact phone				

$\Box$	_	hi	· ^	4

Case number (if known	)	
-----------------------	---	--

						Unsecured claim
13				What is the nature of the claim?		 \$
	Creditor's Name			As of the date you file, the claim is: Chec		
	Number Street			☐ Contingent	m an arat appriy.	
				Unliquidated		
				Disputed		
	City	State	ZIP Code	None of the above apply	anartu?	
	City	State	ZIP Code	Does the creditor have a lien on your property No	operty?	
	<del></del>			☐ Yes. Total claim (secured and unsecured):	\$	
	Contact			Value of security:	<b>-</b> \$	
	Contact phone			Unsecured claim	\$	
77						
14	Creditor's Name			What is the nature of the claim?		\$
	Creditor's Name			As of the date you file, the claim is: Chec	ck all that apply.	
	Number Street			Contingent		
				☐ Unliquidated☐ Disputed☐		
				☐ None of the above apply		
	City	State	ZIP Code	Does the creditor have a lien on your pro	operty?	
				☐ No		
	Contact			Yes. Total claim (secured and unsecured):	\$	
				Value of security:	- \$	
	Contact phone			Unsecured claim	\$	
15				What is the nature of the claim?		\$
	Creditor's Name			As of the date you file, the claim is: Chec		
				Contingent	ok all that apply.	
	Number Street			☐ Unliquidated		
				Disputed		
				None of the above apply		
	City	State	ZIP Code	Does the creditor have a lien on your property No	operty?	
				Yes. Total claim (secured and unsecured):	\$	
	Contact			Value of security:	<b>-</b> \$	
	Contact phone			Unsecured claim	\$	
	Contact phone					
16				What is the nature of the claim?	·	<u> </u>
	Creditor's Name			As of the date you file, the claim is: Chec	ck all that apply.	
	Number Street			Contingent		
				Unliquidated		
				☐ Disputed ☐ None of the above apply		
	City	State	ZIP Code	Does the creditor have a lien on your pro	operty?	
				□ No	оролу.	
	Contact			☐ Yes. Total claim (secured and unsecured):	\$	
	Someon			Value of security:	- \$	
	Contact phone			Unsecured claim	\$	
17				What is the nature of the claim?		\$
	Creditor's Name			As of the date you file, the claim is: Chec		
				☐ Contingent		
	Number Street			Unliquidated		
				Disputed		
				☐ None of the above apply  Does the creditor have a lien on your property.	onorty?	
	City	State	ZIP Code	□ No	operty:	
				Yes. Total claim (secured and unsecured):	\$	
	Contact			Value of security:	<b>-</b> \$	
				Unsecured claim	\$	
	Contact phone			Onscouled Gailli	Ψ	

മ	htor	1	

First Name	Middle Name	Last Name	

					Unsecured claim
18				What is the nature of the claim?	- \$
	Creditor's Name			As of the date you file, the claim is: Check all that apply.	
	Number Street			Contingent	
				☐ Unliquidated☐ Disputed	
				☐ None of the above apply	
	City	State	ZIP Code	Does the creditor have a lien on your property?	
	Oity	Oldic	Zii Oodc	□ No	
	Contact			Yes. Total claim (secured and unsecured): \$	
	Contact			Value of security:	_
	Contact phone			Unsecured claim \$	_
19				What is the nature of the claim?	
	Creditor's Name				\$
	Cieutoi s ivanie			As of the date you file, the claim is: Check all that apply.	
	Number Street			☐ Contingent	
				☐ Unliquidated☐ Disputed	
				☐ None of the above apply	
	City	State	ZIP Code	Does the creditor have a lien on your property?	
				□ No	
	Contact			Yes. Total claim (secured and unsecured): \$	
				Value of security:	
	Contact phone			Unsecured claim \$	_
20				What is the nature of the claim?	\$
				As of the date you file, the claim is: Check all that apply.  Gontingent	
	Creditor's Name			☐ Unliquidated	
				☐ Disputed	
	Number Street			☐ None of the above apply	
				Does the creditor have a lien on your property? ☐ No	
	City	State	ZIP Code	Yes. Total claim (secured and unsecured): \$	
				Value of security:	
	Contact			Unsecured claim \$	_
	Contact phone				
	Contact priorie				
Pa	rt 2: Sign Below				
ı	Jnder penalty of perjury,	I declare	that the inf	formation provided in this form is true and correct.	
×	•			×	
	Signature of Debtor 1			Signature of Debtor 2	
				•	
	Date MM / DD / YYYY	-		Date	

Official Form 104, For Individual Chapter 11 Cases: List of Creditors Who Have the 20 Largest Unsecured Claims Against You and Are Not Insiders, is revised as part of the Forms Modernization Project. It replaces Official Form 4, List of Creditors Holding 20 Largest Unsecured Claims in chapter 11 cases filed by individuals or joint debtors. The form is renumbered to distinguish it from the version to be used in chapter 11 cases filed by non-individuals, such as corporations and partnerships, and in chapter 9 cases.

Form 104 is reformatted to make it easier to complete and understand and to be more visually appealing. Blanks and checkboxes are provided for specific information about each claim, replacing columns for listing information. A separate, numbered section is provided for each of the 20 claims.

The instruction not to include fully secured claims is restated in less technical terms. Debtors are instructed to include a secured creditor only if the creditor has an unsecured claim resulting from inadequate collateral value that is among the 20 largest unsecured claims. Blanks are provided to calculate the value of the unsecured portion of a partially secured claim.

Examples of "insiders" are provided in addition to the statutory reference. The form adds an explicit instruction not to file the form in a chapter 7, chapter 12, or chapter 13 case. An instruction to be as complete and accurate as possible is added, along with a warning that, if two married people are filing jointly, both are equally responsible for supplying correct information.

With respect to children who may be creditors, the direction to state only the initials of a minor child and the name and address of the child's parent or guardian, rather than the child's full name, is moved to the general

instruction booklet for the forms because it applies to all of the forms.

F	ill in this information to identif	the case:
U	nited States Bankruptcy Court for	
_	District	(State)
	ase number (If known):	Chapter
•	€: al Farra 405	☐ Check if this is ar amended filing
	fficial Form 105	
lr	nvoluntary Pe	tition Against an Individual 12/15
cas pos	se against a non-individual, us	cy case against an individual you allege to be a debtor subject to an involuntary case. If you want to begin a the <i>Involuntary Petition Against a Non-individual</i> (Official Form 205). Be as complete and accurate as , attach a separate sheet to this form. On the top of any additional pages, write name and case number (if
Pa	rt 1: Identify the Chapter	of the Bankruptcy Code Under Which Petition Is Filed
1.	Chapter of the	Check one:
	Bankruptcy Code	☐ Chapter 7
		Chapter 11
Pa	rt 2: Identify the Debtor	
2.	Debtor's full name	
		First name
		Middle name
		Last name
		Suffix (Sr., Jr., II, III)
	Other names you know the debtor has used in the last 8 years	
	Include any assumed, married, maiden, or trade names, or doing business as names.	
	Only the last 4 digits of debtor's Social Security	☐ Unknown
	Number or federal Individual Taxpayer Identification Number (ITIN)	xxx - xx OR <b>9</b> xx - xx
	Any Employer Identification Numbers (EINs) used in the last 8	Unknown
	years	EIN

 $\overline{\text{EIN}}$ 

Debtor	Case number (if known)

6.	Debtor's address	Pri	ncipal residence		Mailing addr	ess, if different from residence
		Num	nber Street		Number S	Street
		City	State ZIP	Code	City	State ZIP Code
		Cou	nty			
		Prin	ncipal place of business			
		Num	nber Street			
		City	State ZIP	Code		
		Cou	nty			
7.	Type of business		Debtor does not operate a business			
		_	eck one if the debtor operates a business:			
			Health Care Business (as defined in 11 U.S			
			Single Asset Real Estate (as defined in 11		(101(51B))	
			Stockbroker (as defined in 11 U.S.C. § 101			
		Ц	Commodity Broker (as defined in 11 U.S.C.	. § 101(6	5))	
			None of the above			
8.	Type of debt	Eac	ch petitioner believes:			
			<b>Debts are primarily consumer debts</b> "incurred by an individual primarily for a per			
			<b>Debts are primarily business debts.</b> for a business or investment or through the			
9.	Do you know of any		No			
٠.	bankruptcy cases	_	Yes. Debtor			Deletionship
	pending by or against any partner, spouse, or					
	affiliate of this debtor?			M	M / DD / YYYY	Case number, if known
			Debtor			Relationship
				M	M / DD / YYYY	Case number, if known

Debtor		Case number (if known)		
Part 3: Report About the	Case			
10. Venue  Reason for filing in this court.	business, or had principal asset	ne filing of this bankruptcy, the debtor has resided, is in this district longer than in any other district.  The lebtor's affiliates, general partner, or partnership is the lebtor's 1408.)	s pending in	this district.
11. Allegations	<ul> <li>At least one box must be checked.</li> <li>☐ The debtor is generally not paying bona fide dispute as to liability of a Within 120 days before the filing</li> </ul>	n involuntary case under 11 U.S.C. § 303(a).  Ing such debtor's debts as they become due, unless ar amount.  I of this petition, a custodian, other than a trustee, is than substantially all of the property of the debto	, receiver, or	agent appointed or
12. Has there been a transfer of any claim against the debtor by or to any petitioner?	☐ No ☐ Yes. Attach all documents that end of the second o	evidence the transfer and any statements required	d under Banl	kruptcy Rule
13. Each petitioner's claim	Name of petitioner	Nature of petitioner's claim		Amount of the claim above the value of any lien
				\$ 
			Total	\$

If more than 3 petitioners, attach additional sheets with the statement under penalty of perjury, each petitioner's (or representative's) signature under the statement, along with the signature of the petitioner's attorney, and the information on the petitioning creditor, the petitioner's claim, the petitioner's representative, and the attorney following the format on this form.

Debtor	Case number (if known)	
Denioi	Case Hullibel (# known)	

#### Part 4:

Request for Relief

Petitioners request that an order for relief be entered against the debtor under the chapter specified in Part 1 of this petition. If a petitioning creditor is a corporation, attach the corporate ownership statement required by Bankruptcy Rule 1010(b). If any petitioner is a foreign representative appointed in a foreign proceeding, a certified copy of the order of the court granting recognition is attached.

Petitioners declare under penalty of perjury that the information provided in this petition is true and correct. Petitioners understand that if they make a false statement, they could be fined up to \$250,000 or imprisoned for up to 5 years, or both.

18 U.S.C. §§ 152 and 3571. If relief is not ordered, the court may award attorneys' fees, costs, damages, and punitive damages. 11 U.S.C. § 303(i).

Petitioners or Petitioners' Representative	Attorneys
×	×
Signature of petitioner or representative, including representative's title	Signature of attorney
Printed name of petitioner	Printed name
Date signed MM / DD / YYYY	Firm name, if any
Mailing address of petitioner	Number Street
Number Street	City State ZIP Code
City State ZIP Code	Date signed MM / DD / YYYY
·	Contact phone Email
If petitioner is an individual and is not represented by an attorney:	
Contact phone	
Email	
Name and mailing address of petitioner's representative, if any	
Name	
Number Street	
City State ZIP Code	

Signature of p	petitioner or representa	ative, including represe	entative's title
Printed name	of petitioner		
Date signed	MM / DD /YY	ΥΥ	
Mailing add	dress of petitioner		
	reet		
Number St	1661		
City St	icet	State	ZIP Cod
City	mailing address of		
City  Name and			
City			
City  Name and I			

Signature of At	torney		
Printed name			
Firm name, if a	ny		
Number Stre	et		
City		State	ZIP Code
Date signed	MM / DD / YYYY		
Contact phone		Email	

Signature of petition	er or representative, including rep	resentative's title
Printed name of pet	itioner	
Date signed MI	M / DD /YYYY	
Mailing address	of petitioner	
Number Street		
Number Street  City	State	ZIP Cod
	State	ZIP Cod
City	State ng address of petitioner's re	
City Name and mailir		
City		

Signature of Attorney		
Printed name		
Firm name, if any		
Number Street		
City	State	ZIP Code
Date signed MM / DD / YYYY		
Contact phone	Email	

State ZIP Code

City

#### **COMMITTEE NOTE**

Official Form 105, Involuntary Petition Against an Individual, which is used only in cases of individual debtors, is revised in its entirety as part of the Forms Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. In addition, the form is renumbered to distinguish it from the version to be used in non-individual cases, and stylistic changes were made throughout the form.

The form is derived from Official Form 5, *Involuntary Petition*. The new form separates questions into four parts likely to be more familiar to non-lawyers, groups questions of a similar nature together, and eliminates questions unrelated to individual debtors.

Part 1, *Identify the Chapter of the Bankruptcy Code Under Which Petition is Filed*, moves to the beginning of the form the question regarding the chapter of the Bankruptcy Code under which the petition is filed.

Part 2, *Identify the Debtor*, includes the questions regarding the debtor's name, prior names, Social Security Number, Individual Taxpayer Identification Number, and Employer Identification Number. Petitioners must list the address for the debtor's principal residence, mailing address (if different), and principal place of business. Petitioners must indicate whether the debtor operates a business, and, if so, use checkboxes to indicate whether the business falls into certain categories. The statutory definition of "consumer debts" is provided, as well as a definition of "business debts."

Part 3, Report About the Case, amends the question regarding venue to advise that venue is "the reason for filing in this court" and amends the choices for venue. The first option is revised to read: "Over the last 180 days before the filing of this bankruptcy, the debtor has resided, had the principal place of business, or had principal assets in this district longer than in any other district." Also, the form adds an option for "Other reason. Explain," with a statutory reference. In the question for Allegations, the exact citation to the Bankruptcy Code is provided for the second allegation, and checkboxes are provided for the last

allegation. Petitioners must check "yes" or "no" to answer whether there has been a transfer of any claim against the debtor by or to a petitioner. The information regarding the petitioner's claims is moved to this part of the form, and the portion listing the amount of the claim is amended to ask about the amount of the claim that exceeds the value of the lien, if any.

Part 4, *Request for Relief*, amends the instructions to include a warning about making a false statement, and adds a separate requirement for each petitioner's mailing address. Also, petitioners' attorneys must provide their email addresses, or if a petitioner is an individual and not represented by an attorney, the contact phone and email address of that petitioner must be provided.

Fill in this information to identify your case and this filing:				
Debtor 1				
	First Name	Middle Name	Last Name	
Debtor 2				
(Spouse, if filing)	First Name	Middle Name	Last Name	
United States I	Bankruptcy Court for the	:	District of (State)	
Case number				

# Check if this is an amended filing

### Official Form 106A/B

## Schedule A/B: Property

12/15

In each category, separately list and describe items. List an asset only once. If an asset fits in more than one category, list the asset in the category where you think it fits best. Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write your name and case number (if known). Answer every question.

Part 1: Describe Each Residence, Building, Land, or Other Real Estate You Own or Have an Interest In 1. Do you own or have any legal or equitable interest in any residence, building, land, or similar property? ■ No. Go to Part 2. ☐ Yes. Where is the property? What is the property? Check all that apply. Do not deduct secured claims or exemptions. Put the amount of any secured claims on Schedule D: Single-family home Creditors Who Have Claims Secured by Property. ☐ Duplex or multi-unit building Street address, if available, or other description ☐ Condominium or cooperative Current value of the Current value of the ■ Manufactured or mobile home entire property? portion you own? ☐ Land ■ Investment property Describe the nature of your ownership ■ Timeshare City State ZIP Code interest (such as fee simple, tenancy by Other the entireties, or a life estate), if known. Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only County Check if this is community property Debtor 1 and Debtor 2 only (see instructions) At least one of the debtors and another Other information you wish to add about this item, such as local property identification number: If you own or have more than one, list here: What is the property? Check all that apply. Do not deduct secured claims or exemptions. Put Single-family home the amount of any secured claims on Schedule D: Creditors Who Have Claims Secured by Property. ☐ Duplex or multi-unit building Street address, if available, or other description Condominium or cooperative Current value of the Current value of the ■ Manufactured or mobile home entire property? portion you own? Land ■ Investment property Describe the nature of your ownership ☐ Timeshare City State ZIP Code interest (such as fee simple, tenancy by ☐ Other the entireties, or a life estate), if known. Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only County ☐ Debtor 1 and Debtor 2 only Check if this is community property (see instructions) At least one of the debtors and another Other information you wish to add about this item, such as local property identification number:

	First Name Mid	dle Name Last Na	Case number (# k		
1.3.	Street address, if availab	ole, or other description	What is the property? Check all that apply.  ☐ Single-family home ☐ Duplex or multi-unit building ☐ Condominium or cooperative ☐ Manufactured or mobile home	Do not deduct secured clathe amount of any secure Creditors Who Have Clair  Current value of the entire property?	d claims on Schedule D: ms Secured by Property.
	City	State ZIP Cod	Land Investment property	Describe the nature of interest (such as fee the entireties, or a life	simple, tenancy by
	County		Who has an interest in the property? Check one.  Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another	Check if this is co	ommunity property
			Other information you wish to add about this ite property identification number:		
			all of your entries from Part 1, including any entries		\$
	have attached for Par	t 1. Write that number			\$
you Part 2: Do you ou own	Describe Your  own, lease, or have lead that someone else drives, vans, trucks, tractor	Vehicles  egal or equitable interes. If you lease a veh	rest in any vehicles, whether they are registered or icle, also report it on Schedule G: Executory Contracts	not? Include any vehicle:	\$s
you Part 2: Do you you own	Describe Your  own, lease, or have lead that someone else drives, vans, trucks, tractor	Vehicles  egal or equitable interes. If you lease a veh	rest in any vehicles, whether they are registered or icle, also report it on Schedule G: Executory Contracts	not? Include any vehicle:	\$S
you Part 2: Do you You own Cars	Describe Your  own, lease, or have lead that someone else drives, vans, trucks, tractor	Vehicles  egal or equitable interes. If you lease a veh	rest in any vehicles, whether they are registered or icle, also report it on Schedule G: Executory Contracts es, motorcycles  Who has an interest in the property? Check one.	not? Include any vehicle:	aims or exemptions. Put d claims on <i>Schedule D</i> :
you  Part 2:  Do you own  Cars	Describe Your  own, lease, or have lead that someone else drives, vans, trucks, tractor lookes  Make:	Vehicles  gal or equitable interes. If you lease a vehicles, sport utility vehicles.	rest in any vehicles, whether they are registered or icle, also report it on Schedule G: Executory Contracts es, motorcycles  Who has an interest in the property? Check one.	not? Include any vehicles and Unexpired Leases.  Do not deduct secured clathe amount of any secure	aims or exemptions. Put d claims on <i>Schedule D</i> :

If you own or have more than one, describe here:

3.2.	Make:	
	Model:	
	Year:	

Approximate mileage:	
Other information:	

Who has an interest in the property? Check one.

Debtor 1 only Debtor 2 only

☐ Debtor 1 and Debtor 2 only

lacktriangle At least one of the debtors and another

☐ Check if this is community property (see instructions)

Do not deduct secured claims or exemptions. Put the amount of any secured claims on Schedule D: Creditors Who Have Claims Secured by Property.

Current value of the entire property? Current value of the portion you own?

Tiret Nieres	Middle Nones	Look Nome	

3.3.	Make:	Who has an interest in the property? Check one.	Do not deduct secured cla	
	Model:	Debtor 1 only	the amount of any secured claims on Schedule D: Creditors Who Have Claims Secured by Property.	
	Year:	☐ Debtor 2 only☐ Debtor 1 and Debtor 2 only	Current value of the	Current value of the
	Approximate mileage:	At least one of the debtors and another	entire property?	portion you own?
	Other information:	At least one of the deptors and another		
		Check if this is community property (see instructions)	\$	\$
3.4.	Make:	Who has an interest in the property? Check one.	Do not deduct secured cla	
	Model:	Debtor 1 only	the amount of any secured Creditors Who Have Claim	
	Year:	Debtor 2 only	Current value of the	Current value of the
	Approximate mileage:	Debtor 1 and Debtor 2 only	entire property?	portion you own?
		At least one of the debtors and another		
	Other information:	☐ Check if this is community property (see instructions)	\$	\$
□ N	No	atercraft, fishing vessels, snowmobiles, motorcycle accesso		
4.1.	Make:	Who has an interest in the property? Check one.	Do not deduct secured cla	ims or exemptions. Put
	Model:	Debtor 1 only	the amount of any secured Creditors Who Have Claim	
	Year:	Debtor 2 only		
	Other information:	Debtor 1 and Debtor 2 only	Current value of the	Current value of the
		At least one of the debtors and another	entire property?	portion you own?
		☐ Check if this is community property (see instructions)	\$	\$
If you	u own or have more than one, list here:			
4.2.	Make:	Who has an interest in the property? Check one.	Do not deduct secured cla	
	Model:	Debtor 1 only	the amount of any secured Creditors Who Have Claim	
	Year:	Debtor 2 only	Current value of the	Current value of the
	Other information:	☐ Debtor 1 and Debtor 2 only☐ At least one of the debtors and another	entire property?	portion you own?
		At least one of the debtors and another		
			\$	\$
		☐ Check if this is community property (see instructions)		

Part 3: Describe Your Personal and Household Items

Do you own or have any legal or equitable interest in any of the following items?	Current value of the portion you own?  Do not deduct secured claims or exemptions.			
6. Household goods and furnishings	2. 2. 2			
Examples: Major appliances, furniture, linens, china, kitchenware				
□ No	$\neg$			
☐ Yes. Describe	\$			
7. Electronics				
Examples: Televisions and radios; audio, video, stereo, and digital equipment; computers, printers, scanners; music collections; electronic devices including cell phones, cameras, media players, games				
□ No				
Yes. Describe	\$			
8. Collectibles of value				
Examples: Antiques and figurines; paintings, prints, or other artwork; books, pictures, or other art objects; stamp, coin, or baseball card collections; other collections, memorabilia, collectibles				
☐ Yes. Describe	\$			
9. Equipment for sports and hobbies				
Examples: Sports, photographic, exercise, and other hobby equipment; bicycles, pool tables, golf clubs, skis; canoes and kayaks; carpentry tools; musical instruments				
☐ No ☐ Yes. Describe				
Tes. Describe	\$			
10. Firearms				
Examples: Pistols, rifles, shotguns, ammunition, and related equipment				
☐ No ☐ Yes. Describe	┐.			
Tes. Describe	\$			
11. Clothes				
Examples: Everyday clothes, furs, leather coats, designer wear, shoes, accessories				
☐ Yes. Describe	\$			
_ , , , , , , , , , , , , , , , , , , ,	Φ			
12. Jewelry				
Examples: Everyday jewelry, costume jewelry, engagement rings, wedding rings, heirloom jewelry, watches, gems, gold, silver				
□ No				
Yes. Describe	\$			
13. Non-farm animals  Examples: Dogs, cats, birds, horses				
□ No				
Yes. Describe	\$			
14. Any other personal and household items you did not already list, including any health aids you did not list				
□ No				
Yes. Give specific information	\$			
15. Add the dollar value of all of your entries from Part 3, including any entries for pages you have attached				
for Part 3. Write that number here	\$			

Eiret Name	Middle Name	Last Name	

Da	rt	1	
Рα	ıι	-	

Describe Your Financial Assets

20 you om of haro any logar of equitable interest in any of the following !			Current value of the portion you own?  Do not deduct secured claims or exemptions.	
16. Cash  Examples: Money you h	ave in your wallet, in your hom	ne, in a safe deposit box, and on hand when you	file your petition	
			Cash:	\$
		nts; certificates of deposit; shares in credit unior ultiple accounts with the same institution, list eac		
□ No □ Yes		Institution name:		
	17.1. Checking account:			\$
	17.2. Checking account:			\$
	17.3. Savings account:			\$
	17.4. Savings account:			\$
	17.5. Certificates of deposit:			\$
	17.6. Other financial account:			\$
	17.7. Other financial account:			\$
	17.8. Other financial account:			\$
	17.9. Other financial account:		<del> </del>	\$
18. Bonds, mutual funds, of Examples: Bond funds, i	-	erage firms, money market accounts		
Yes	Institution or issuer name:			
				\$
19. Non-publicly traded st an LLC, partnership, a		rated and unincorporated businesses, includ	ing an interest in	
□ No	Name of entity:		% of ownership:	
☐ Yes. Give specific information about			%	\$
them			%	\$
			%	\$

First Name	Middle Name	Last Name

20.	Government and corpo	rate bonds and other negotiable and non-negotiable instruments	
		nclude personal checks, cashiers' checks, promissory notes, and money orders. nts are those you cannot transfer to someone by signing or delivering them.	
	☐ No☐ Yes. Give specific	Issuer name:	
	information about them		
			<b></b> \$
			\$
21.	Retirement or pension Examples: Interests in IR	accounts A, ERISA, Keogh, 401(k), 403(b), thrift savings accounts, or other pension or profit-sharin	ng plans
	☐ No☐ Yes. List each		
	account separately	Type of account: Institution name:	
		401(k) or similar plan:	
		Pension plan:	\$
		IRA:	\$
		Retirement account:	\$
		Keogh:	\$
		Additional account:	
		Additional account:	 \$
22.	Examples: Agreements v companies, or others	deposits you have made so that you may continue service or use from a company with landlords, prepaid rent, public utilities (electric, gas, water), telecommunications	
	□ No		
	☐ Yes	Institution name or individual:  Electric:	
		Gas:	<u> </u>
		Heating oil:	\$
		Security deposit on rental unit:	\$
		Prepaid rent:	\$ \$
		Telephone:	\$
		Water:	
		Rented furniture:	
		Other:	<b>\$</b>
23.	Annuities (A contract for	a periodic payment of money to you, either for life or for a number of years)	
	☐ No		
	☐ Yes	Issuer name and description:	
			<b>\$</b>
			\$ \$
			Ψ

24. Interests in an education IRA, in an account in a qualified ABLE program, or under a qualified st 26 U.S.C. §§ 530(b)(1), 529A(b), and 529(b)(1).	tate tuition program.	
Yes	rests.11 U.S.C. § 521(c	):
		\$
		\$
		\$
25. Trusts, equitable or future interests in property (other than anything listed in line 1), and rights exercisable for your benefit	or powers	
□ No		-
Yes. Give specific		Φ.
information about them		\$
26. Patents, copyrights, trademarks, trade secrets, and other intellectual property  Examples: Internet domain names, websites, proceeds from royalties and licensing agreements		
□ No		7
Yes. Give specific information about them		\$
information about them		Ψ
27. Licenses, franchises, and other general intangibles  Examples: Building permits, exclusive licenses, cooperative association holdings, liquor licenses, profes	essional licenses	
□ No		7
Yes. Give specific information about them		\$
information about them		Ψ
Money or property owed to you?		Current value of the portion you own? Do not deduct secured claims or exemptions.
28. Tax refunds owed to you		
□ No		
☐ Yes. Give specific information	Federal:	\$
about them, including whether you already filed the returns	State:	Ψ \$
and the tax years		\$
	Local	Ψ
29. Family support  Examples: Past due or lump sum alimony, spousal support, child support, maintenance, divorce settler	ment property settleme	nt
□ No	5 p 2 p 2 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	
☐ Yes. Give specific information	]	
	Alimony:	\$
	Maintenance:	\$
	Support:	\$
	Divorce settlement:	\$ \$
	Property settlement:	φ
30. Other amounts someone owes you Examples: Unpaid wages, disability insurance payments, disability benefits, sick pay, vacation pay, wo Social Security benefits; unpaid loans you made to someone else	orkers' compensation,	
□ No		7
Yes. Give specific information		\$

E:	 	

31. Interests in insurance policies  Examples: Health, disability, or life insurance	ee; health savings account (HSA); credit, h	nomeowner's, or renter's insurance	
☐ No ☐ Yes. Name the insurance company of each policy and list its value	Company name:	Beneficiary:	Surrender or refund value:
			\$
		<del></del>	\$
32. Any interest in property that is due you f If you are the beneficiary of a living trust, ex	rom someone who has died	y, or are currently entitled to receive	Φ
property because someone has died.  D No			
☐ Yes. Give specific information			
			\$
<ul><li>33. Claims against third parties, whether or Examples: Accidents, employment disputes</li><li>No</li></ul>	_	demand for payment	
☐ Yes. Describe each claim			\$
34. Other contingent and unliquidated claims to set off claims	s of every nature, including countercla	ims of the debtor and rights	Φ
☐ No☐ Yes. Describe each claim			$\neg$
L			\$
OF Any financial access you did not already.	lia4		
35. Any financial assets you did not already  No	list		
☐ Yes. Give specific information			\$
36. Add the dollar value of all of your entries for Part 4. Write that number here			\$
Part 5: Describe Any Business-R	Related Property You Own or H	lave an Interest In. List any r	eal estate in Part 1.
37. Do you own or have any legal or equitable No. Go to Part 6.	le interest in any business-related prop	perty?	
☐ Yes. Go to line 38.			
			Current value of the portion you own?  Do not deduct secured claims or exemptions.
38. Accounts receivable or commissions you	u already earned		
☐ No☐ Yes. Describe			\$
39. Office equipment, furnishings, and supp	lies		
Examples: Business-related computers, software,  No		, telephones, desks, chairs, electronic devices	
☐ Yes. Describe			\$

40. Machinery, fixtures, equipment, supplies y	ou use in business, and tools of your trade				
Yes. Describe		\$			
41. Inventory  No					
Yes. Describe		\$			
42. Interests in partnerships or joint ventures					
☐ No☐ Yes. Describe Name of entity:	0/	of ownership:			
		% \$			
		% \$			
		% \$			
43. Customer lists, mailing lists, or other com	pilations				
☐ No					
	dentifiable information (as defined in 11 U.S.C. § 101(41A))?				
Yes. Describe					
		\$			
44. Any business-related property you did no	already list				
☐ No☐ Yes. Give specific					
information					
		\$			
	from Part 5, including any entries for pages you have attach	1 \$			
Part 6: Describe Any Farm- and Com  If you own or have an interest in the second secon	mercial Fishing-Related Property You Own or Have farmland, list it in Part 1.	an Interest In.			
46. Do you own or have any legal or equitable  No. Go to Part 7.	interest in any farm- or commercial fishing-related propert	y?			
□ No. Go to Part 7. □ Yes. Go to line 47.					
		Current value of the			
		portion you own?  Do not deduct secured claims			
47. Farm animals		or exemptions.			
Examples: Livestock, poultry, farm-raised fish	n				
□ No					
☐ Yes					
		<u> </u>			

Debtor 1	As Crops—either growing or harvested   No   Yes. Give specific   Information					
	First Name	Middle Name	Last Name	_		
48. <b>Crops—e</b>	ither growing	g or harvested				
	Γ					
						\$
	l fishing equi					_
☐ Yes						•
50 Farm and	l fishing succ					Ψ
	i nsinig supp	mes, chemicals,	aliu leeu			
☐ Yes						
						\$
	- and comme	rcial fishing-rela	ted property you did not	already list		
Yes.						•
	L	£ -11 -£	David S. J. II.			Ψ
						\$
Part 7:	Describe A	All Property Y	ou Own or Have ar	n Interest in That	You Did Not List Above	
				1?		
☐ No						\$
	Give specific nation					\$
						\$
- ماما داده -		f all of variance	on from Dout 7 Militar 41-	4 mumbau barra		s
54. Add the C	ioliar value o	i ali of your entri	es from Part 7. Write tha	number nere		Ψ
Part 8:	List the To	otals of Each	Part of this Form			
55. <b>Part 1: To</b>	otal real estat	e, line 2				\$
56. <b>Part 2: To</b>	otal vehicles,	line 5		\$	-	
57. Part 3: To	otal personal	and household it	tems, line 15	\$	-	
58. <b>Part 4: T</b> c	otal financial	assets, line 36		\$	-	
59. <b>Part 5: T</b> c	otal business	-related property	, line 45	\$	-	
60. <b>Part 6: To</b>	otal farm- and	fishing-related p	property, line 52	\$	-	
61. <b>Part 7: To</b>	otal other pro	perty not listed, l	line 54	+\$	-	
62. Total pers	sonal propert	y. Add lines 56 th	rough 61	\$	Copy personal property total 🗲	+\$

63. Total of all property on Schedule A/B. Add line 55 + line 62.

Fill in this in	formation to ide	nation to identify your case:		
Debtor 1				
	First Name	Middle Name	Last Name	
Debtor 2				
(Spouse, if filing)	First Name	Middle Name	Last Name	
United States E	Bankruptcy Court fo	or the:	District of	
			(State)	
Case number (If known)			<del></del>	

☐ Check if this is an amended filing

## Official Form 106C

## Schedule C: The Property You Claim as Exempt

12/15

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. Using the property you listed on *Schedule A/B: Property* (Official Form 106A/B) as your source, list the property that you claim as exempt. If more space is needed, fill out and attach to this page as many copies of *Part 2: Additional Page* as necessary. On the top of any additional pages, write your name and case number (if known).

For each item of property you claim as exempt, you must specify the amount of the exemption you claim. One way of doing so is to state a specific dollar amount as exempt. Alternatively, you may claim the full fair market value of the property being exempted up to the amount of any applicable statutory limit. Some exemptions—such as those for health aids, rights to receive certain benefits, and tax-exempt retirement funds—may be unlimited in dollar amount. However, if you claim an exemption of 100% of fair market value under a law that limits the exemption to a particular dollar amount and the value of the property is determined to exceed that amount, your exemption would be limited to the applicable statutory amount.

2. <b>F</b>	For any property you list on Schedule A/B to	hat you claim as exem	pt, fill in the information below.	
	Brief description of the property and line on Schedule A/B that lists this property	Current value of the portion you own	Amount of the exemption you claim	Specific laws that allow exemption
		Copy the value from Schedule A/B	Check only one box for each exemption.	
	Brief description: Line from Schedule A/B:	\$	□ \$ □ 100% of fair market value, up to any applicable statutory limit	
	Brief description: Line from Schedule A/B:	\$	\$ 100% of fair market value, up to any applicable statutory limit	
	Brief description: Line from Schedule A/B:	\$	\$ 100% of fair market value, up to any applicable statutory limit	

Last Name

#### Case number (if known)\_

### Part 2:

## Additional Page

Brief description of the property and line on Schedule A/B that lists this property	Current value of the portion you own	Amount of the exemption you claim	Specific laws that allow exemption
	Copy the value from Schedule A/B	Check only one box for each exemption	
Brief description:  Line from Schedule A/B:	\$	□ \$ □ 100% of fair market value, up to any applicable statutory limit	
Brief description:  Line from Schedule A/B:	\$	\$ 100% of fair market value, up to any applicable statutory limit	
Brief description: Line from Schedule A/B:	\$	\$ \$ 100% of fair market value, up to any applicable statutory limit	
Brief description: Line from Schedule A/B:	\$	\$ \$ 100% of fair market value, up to any applicable statutory limit	
Brief description:  Line from Schedule A/B:	\$	\$ 100% of fair market value, up to any applicable statutory limit	
Brief description: Line from Schedule A/B:	\$	\$ \$ 100% of fair market value, up to any applicable statutory limit	
Brief description:  Line from Schedule A/B:	\$	\$ \$ 100% of fair market value, up to any applicable statutory limit	
Brief description:  Line from Schedule A/B:	\$	□ \$ □ 100% of fair market value, up to any applicable statutory limit	
Brief description: Line from Schedule A/B:	\$	□ \$ □ 100% of fair market value, up to any applicable statutory limit	
Brief description:  Line from Schedule A/B:	\$	□ \$ □ 100% of fair market value, up to any applicable statutory limit	
Brief description: Line from Schedule A/B:	\$	□ \$ □ 100% of fair market value, up to any applicable statutory limit	
Brief description: Line from Schedule A/B:	\$	\$ 100% of fair market value, up to any applicable statutory limit	

Debtirs 1   The Name   Debtirs 2   Debtirs 2   Debtirs 3   Debtirs 4   Debtirs 4   Debtirs 4   Debtirs 5   Debtirs 5   Debtirs 6   Debtirs	biblior 2 genus. If files a large state st				
Column   C	I m m and mornianen te menning year case				
Check if this is an amended filing   Check if this chair in a check if this is an amended filing   Check if this chair in a check if this chair in a check if this is an amended filing   Check if this chair in a check if this check if this chair in a check if this check if this chair in a check if this check if this chair in a check if this check if t		ame Last Name			
Cause number		ame Last Name			
Cause number	United States Bankruptcy Court for the:	District of			
Official Form 106D  Schedule D: Creditors Who Have Claims Secured by Property  12/15  Be as complete and accurate as possible, if two married people are filing together, both are equally responsible for supplying correct information more appeals enserted. Copy the Additional Pages, fill it out, number the entries, and attach it to this form. On the top of any additional pages, write your name and case number (if known)  1. Do any creditors have claims secured by your property?  1. No. Check this box and submit this form to the court with your other schedules. You have nothing else to report on this form.  1. List All Secured Claims  2. List all secured claims. If a creditor has a more than one secured claim, list the other creditor's spannible. As much as possible, list the claims in alphabetical order according to the creditor's name.  2. List all secured claims. If more than one content has a particular dain, list the other creditor's name.  3. Secretion of the delay of the creditor spannible. The content of the creditor's name.  3. Secretion of the delay of the creditor's name.  3. Secretion of the delay of the creditor's name.  3. Secretion of the delay of the creditor's name.  4. Secretion of the delay of the creditor's name.  3. Secretion of the delay of the creditor's name.  4. Secretion of the delay of the creditor's name.  5. S.					
Schedule D: Creditors Who Have Claims Secured by Property    Schedule D: Creditors Who Have Claims Secured by Property   12/15					
Schedule D: Creditors Who Have Claims Secured by Property				amende	ea tiling
Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. If more space is needed, copy the Additional Page, fill it out, number the entries, and attach it to this form. On the top of any additional pages, write your names and case number (if known).  1. Do any creditors have claims secured by your property?    No. Check this box and submit this form to the court with your other schedules. You have nothing else to report on this form.    Port 10	Official Form 106D				
Information. If more space is needed, copy the Additional Page, fill it out, number the entries, and attach it to this form. On the top of any additional pages, write your name and case number (if known).  1. Do any creditors have claims secured by your property?    No. Check this box and submit this form to the court with your other schedules. You have nothing else to report on this form.    Part 15	Schedule D: Creditors	s Who Have Claims Secu	ed by Prop	perty	12/15
1. Do any creditors have claims secured by your property?	information. If more space is needed, copy	the Additional Page, fill it out, number the entries			
No. Check this box and submit this form to the court with your other schedules. You have nothing else to report on this form.   Yes. Fill in all of the information below.   Column S	, , ,	,			
Yes. Fill in all of the information below.		, , , ,	hing also to report on t	this form	
2. List all secured claims. If a creditor has more than one secured claim, list the creditor separately for each claim. If more than one creditor has a particular claim, list the other creditors in Part 2. As much as possible, list the claims in alphabetical order according to the creditor's name.    Describe the property that secures the claim:   \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$		n to the court with your other schedules. You have no	ning eise to report on t	inis form.	
2. List all secured claims. If a creditor has more than one secured claim, list the creditor separately for each claim. If more than one creditor has a particular claim, list the other creditors in Part 2. As much as possible, list the claims in alphabetical order according to the creditor's name.    2-1					
2. List all secured claims. If a creditor has more than one secured claim, list the creditor separately for each claim. If more than one creditor has a particular claim, list the other creditor's name.  2.1  Describe the property that secures the claim:    Describe the property that secures the claim:   S	Part 1: List All Secured Claims				
for each claim. If more than one creditor has a particular claim, list the other creditors in Part 2. As much as possible, list the claims in alphabetical order according to the creditor's name.    2.1	2. List all secured claims. If a creditor has m	ore than one secured claim, list the creditor separatel			
Creditor's Name	for each claim. If more than one creditor ha	as a particular claim, list the other creditors in Part 2.	Do not deduct the	that supports this	
Creditor's Name Number Street  As of the date you file, the claim is: Check all that apply. Contingent Uniquidated Disputed  Who owes the debt? Check one. Debtor 1 and Debtor 2 only Debtor 1 and Debtor 3 only Date debt was incurred Last 4 digits of account number  Creditor's Name Number Street  As of the date you file, the claim is: Check all that apply.  As a greement you made (such as mortgage or secured car loan) Statutory lien (such as tax lien, mechanic's lien) Judgment lien from a lawsuit Other (including a right to offset) Creditor's Name Number Street  As of the date you file, the claim is: Check all that apply.  Creditor's Name  As of the date you file, the claim is: Check all that apply.  Contingent Uniquidated Disputed  Nature of lien. Check all that apply.  As of the date you file, the claim is: Check all that apply.  Contingent Uniquidated Disputed Nature of lien. Check all that apply.  An agreement you made (such as mortgage or secured car loan) Statutory lien (such as tax lien, mechanic's lien) Judgment lien from a lawsuit Other (including a right to offset) Other (including a right to offset)  Check if this claim relates to a community debt Date debt was incurred Last 4 digits of account number		abelical order according to the creditor's name.	value of collateral.	ciaim	If any
As of the date you file, the claim is: Check all that apply.    Contingent   Uniquidated   Disputed		Describe the property that secures the claim:	\$	\$	\$
As of the date you file, the claim is: Check all that apply.    Contingent   Unliquidated   Disputed	Creditor's Name				
Contingent Unliquidated Disputed  Who owes the debt? Check one. Debtor 1 only Debtor 2 only At least one of the debtors and another Creditor's Name Number Street  As of the date you file, the claim is: Check all that apply. Debtor 1 only Debtor 1 only Describe the property that secures the claim: S \$ \$  Creditor's Name Number Street  As of the date you file, the claim is: Check all that apply. Debtor 1 only Debtor 1 only Debtor 1 only Debtor 1 only City State ZIP Code Who owes the debt? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 3 only Debtor 1 and Debtor 4 debtors and another Check if this claim relates to a community debt Date debt was incurred Last 4 digits of account number  Check if this claim relates to a community debt Date debt was incurred Last 4 digits of account number  Last 4 digits of account number	Number Street				
Unliquidated   Disputed			ly.		
Who owes the debt? Check one.  Debtor 1 only Debtor 2 only An agreement you made (such as mortgage or secured car loan) Statutory lien (such as tax lien, mechanic's lien) Date debt was incurred  Last 4 digits of account number  2.2.2  Describe the property that secures the claim: Number Street  As of the date you file, the claim is: Check all that apply. Contingent Unliquidated Disputed  Who owes the debt? Check one. Debtor 1 only Debtor 2 only Debtor 1 only Debtor 2 only Debtor 1 only Debtor 1 only Debtor 1 only Debtor 1 only Debtor 2 only Debtor 1 only Debtor 3 only Debtor 1 only Debtor 4 only Debtor 3 only Debtor 4 only Debtor 5 only Debtor 6 only Debtor 6 only Debtor 6 only Debtor 6 only Debtor 7 only Debtor 1 only D					
Debtor 1 only	City State ZIP Code	☐ Disputed			
Debtor 2 only	_	****			
Debtor 1 and Debtor 2 only			I		
Check if this claim relates to a community debt  Date debt was incurred  Last 4 digits of account number  2.2  Describe the property that secures the claim:  S:  Creditor's Name  Number Street  As of the date you file, the claim is: Check all that apply.  Contingent Unliquidated Disputed  Who owes the debt? Check one. Debtor 1 only Debtor 2 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this claim relates to a community debt Date debt was incurred  Last 4 digits of account number					
Check if this claim relates to a community debt  Date debt was incurred  Last 4 digits of account number  2.2  Describe the property that secures the claim:  Street  As of the date you file, the claim is: Check all that apply.  City State ZIP Code  Who owes the debt? Check one.  Debtor 1 only Debtor 2 only Debtor 2 and Debtor 2 only At least one of the debtors and another Check if this claim relates to a community debt  Date debt was incurred  Last 4 digits of account number	At least one of the debtors and another	_			
Date debt was incurred Last 4 digits of account number		Other (including a right to offset)			
Describe the property that secures the claim:  Creditor's Name  Number Street  As of the date you file, the claim is: Check all that apply.  Contingent Unliquidated Disputed  Who owes the debt? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this claim relates to a community debt  Date debt was incurred  Last 4 digits of account number  Last 4 digits of account number    Street   As of the date you file, the claim is: Check all that apply.   Contingent Unliquidated Disputed   Disputed   Nature of lien. Check all that apply.   An agreement you made (such as mortgage or secured car loan)   Statutory lien (such as tax lien, mechanic's lien)   Judgment lien from a lawsuit   Other (including a right to offset)   Check if this claim relates to a community debt   Date debt was incurred Last 4 digits of account number		Last 4 digits of account number			
Creditor's Name  Number Street  As of the date you file, the claim is: Check all that apply.  City State ZIP Code  Who owes the debt? Check one.  Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another  Check if this claim relates to a community debt  Date debt was incurred  Last 4 digits of account number  Last 4 digits of account number  As of the date you file, the claim is: Check all that apply.  Check all that apply.  An agreement you made (such as mortgage or secured car loan)  Statutory lien (such as tax lien, mechanic's lien)  Judgment lien from a lawsuit  Other (including a right to offset)  Last 4 digits of account number			\$		<b>\$</b>
As of the date you file, the claim is: Check all that apply.  City State ZIP Code Unliquidated Disputed  Who owes the debt? Check one. Debtor 1 only Debtor 2 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this claim relates to a community debt  Date debt was incurred Last 4 digits of account number	Creditor's Name			· ·	*
As of the date you file, the claim is: Check all that apply.  City State ZIP Code Unliquidated Disputed  Who owes the debt? Check one. Debtor 1 only Debtor 2 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this claim relates to a community debt  Date debt was incurred Last 4 digits of account number	Number Street				
City State ZIP Code □ Disputed □ Disputed □ Disputed □ Disputed □ Disputed □ Debtor 1 only □ Debtor 2 only □ Debtor 1 and Debtor 2 only □ At least one of the debtors and another □ Check if this claim relates to a community debt □ Date debt was incurred □ Last 4 digits of account number □ Unliquidated □ Disputed □ Dis	Number Street	As of the date you file, the claim is: Check all that app	ly.		
City		•			
Who owes the debt? Check one.  Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this claim relates to a community debt  Date debt was incurred  Nature of lien. Check all that apply. An agreement you made (such as mortgage or secured car loan) Statutory lien (such as tax lien, mechanic's lien) Judgment lien from a lawsuit Other (including a right to offset)  Last 4 digits of account number  Last 4 digits of account number	City State ZIP Code				
□ Debtor 1 only □ Debtor 2 only □ Debtor 1 and Debtor 2 only □ At least one of the debtors and another □ Check if this claim relates to a community debt □ Date debt was incurred Last 4 digits of account number	Who owes the debt? Check one.	'			
□ Debtor 2 only □ Debtor 1 and Debtor 2 only □ At least one of the debtors and another □ Check if this claim relates to a community debt □ Date debt was incurred Last 4 digits of account number			I		
☐ At least one of the debtors and another ☐ Check if this claim relates to a community debt ☐ Date debt was incurred Last 4 digits of account number	Debtor 2 only	car loan)			
Other (including a right to offset)  Check if this claim relates to a community debt  Date debt was incurred Last 4 digits of account number					
community debt  Date debt was incurred Last 4 digits of account number		<del>-</del>	<u> </u>		
	community debt				
			<b>\$</b>	ı	

$\square$	htor	1

irot Nomo	Middle Nome	Loot Namo	

Creditor's Name    Creditor's Name   Creditor's Name   Creditor's Name   Creditor's Name   Contingent   Conti	Part 1: Additional Page  After listing any entries on this p by 2.4, and so forth.	page, number them beginning with 2.3, followed	Column A  Amount of claim  Do not deduct the value of collateral.	Column B  Value of collateral that supports this claim	Column C Unsecured portion If any
As of the date you file, the claim is: Check all that apply.    Contingent   Uniquicated		Describe the property that secures the claim:	\$	\$	•
As of the date you file, the claim is: Check all that apply.    Contingent	Creditor's Name	Describe the property that secures the damin	1	Ψ	Ψ
Centingent	Number Street				
Uniquidated   Disputed   Disput		_			
Disputed	City State 7IP Code	•			
Debtor 2 only   Debtor 2 only   Debtor 2 only   Debtor 3 nonly   Debtor 2 only   Debtor 3 nonly   Debtor 3 nonly   Debtor 3 nonly   Debtor 3 nonly   Debtor 4 nan Debtor 3 only   Debtor 4 nonly   Debtor 4 nonly   Debtor 4 only   Debtor 5 only   Debtor 4 only   Debtor 5 only   Debtor 6 only   Debtor 7 only   Debtor 8 only   Debtor 8 only   Debtor 8 only   Debtor 8 only   Debtor 9 only   Debtor 1 only   Debt	Only Glate Zii Gode				
Debtor 1 only   Debtor 2 only   Debtor 2 only   Debtor 2 only   Debtor 3 only   Debtor 3 only   Debtor 3 only   Debtor 4 and Debtor 2 only   Debtor 4 and Debtor 2 only   Debtor 5 and Debtor 5 and Debtor 2 only   Debtor 5 and Debtor 6 only   Debtor 7 only   Debtor 7 only   Debtor 8 only   Debtor 9 on	Who owes the debt? Check one.	Nature of lien. Check all that apply.			
Debtor 2 only   Debtor 1 and Debtor 2 only   Debtor 1 only   Debtor 2 only   Debtor 1 only   Debtor 1 only   Debtor 1 only   Debtor 1 only   Debtor 2 only   Debtor 1 only   Debtor 1 only   Debtor 2 only   Debtor 1 only   Debtor 1 only   Debtor 2 only   Debtor 2 only   Debtor 2 only   Debtor 3 only   Debtor 2 only   Debtor 3 only   Debtor 4 only   Debtor 3 only   Debtor 4 only   Debtor 3 only   Debtor 4 only   Debtor 5 only   Debtor 5 only   Debtor 6 only   Debtor 7 only   Debtor 8 only   Debtor 8 only   Debtor 8 only   Debtor 8 only   Debtor 9 only   Debtor 1	☐ Debtor 1 only				
At least one of the debtors and another   Check iff this claim relates to a community debt		car loan)			
Check if this claim relates to a community debt  Date debt was incurred					
Check if this claim relates to a community debt  Date debt was incurred   Describe the property that secures the claim: \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	At least one of the debtors and another				
Creditor's Name  Namber Street  As of the date you file, the claim is: Check all that apply. Cotingent Unliquidated Disputed  Who owes the debt? Check one. Debtor 1 and Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check iff this claim relates to a community debt  Date debt was incurred  Last 4 digits of account number  City State ZIP Code  Who owes the debt? Check one. Debtor 1 and Debtor 2 only Creditor's Name  Number Street  As of the date you file, the claim is: Check all that apply. Debtor 1 and Debtor 2 only Debtor 1 and Debtor 2 only Creditor's Name  Number Street  As of the date you file, the claim is: Check all that apply. Creditor's Name  Number Street  As of the date you file, the claim is: Check all that apply. Contingent Unliquidated Disputed  As of the date you file, the claim is: Check all that apply. Contingent Unliquidated Disputed  As of the date you file, the claim is: Check all that apply. Contingent Unliquidated Disputed  Who owes the debt? Check one. Debtor 1 only Debtor 1 and Debtor 2 only At least one of the debtors and another An agreement you made (such as mortgage or secured carloan) Statutory lien (such as tax lien, mechanic's lien) Undiquidated Disputed  An agreement you made (such as mortgage or secured carloan) Statutory lien (such as tax lien, mechanic's lien) Undiquidated Disputed  At least one of the debtors and another Check if this claim relates to a community debt  Date debt was incurred  Last 4 digits of account number  Add the dollar value of your entries in Column A on this page. Write that number here:  S					
As of the date you file, the claim is: Check all that apply.	Date debt was incurred	Last 4 digits of account number			
As of the date you file, the claim is: Check all that apply.		Describe the property that secures the claim:	\$	\$	\$
As of the date you file, the claim is: Check all that apply.    Contingent   Contin	Creditor's Name		]		
Contingent Unliquidated Unliqui	Number Street				
Who owes the debt? Check one. Debtor 1 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this claim relates to a community debt  Date debt was incurred  Last 4 digits of account number  Creditor's Name  Number Street  As of the date you file, the claim is: Check all that apply. Debtor 1 only Debtor 1 only Debtor 2 only Statutory len (such as tax lien, mechanic's lien)  Unliquidated Disputed  Nature of lien. Check all that apply. Other (including a right to offset)  Street  As of the date you file, the claim is: Check all that apply. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only Debtor 1 and Debtor 2 only Debtor 1 and Debtor 2 only Statutory len (such as tax lien, mechanic's lien) Unliquidated Disputed  Nature of lien. Check all that apply. Statutory len (such as mortgage or secured car loan) Statutory len (such as mortgage or secured car loan) Statutory len (such as tax lien, mechanic's lien) Unliquidated Disputed  An agreement you made (such as mortgage or secured car loan) Statutory len (such as tax lien, mechanic's lien) Undiquidated Disputed  At least one of the debtors and another Check if this claim relates to a community debt  Date debt was incurred Last 4 digits of account number  Check if this claim relates to a community debt  Date debt was incurred Last 4 digits of account number  Last 4 digits of account number  S  S  S  S  S  S  S  S  S  S  S  S  S		As of the date you file, the claim is: Check all that apply.	1		
City State ZIP Code  Who owes the debt? Check one.  Debtor 1 only Debtor 2 only Debtor 2 only Debtor 3 and Debtor 2 only Debtor 4 and Debtor 2 only Debtor 4 least one of the debtors and another Check if this claim relates to a community debt  Date debt was incurred  Describe the property that secures the claim:  Creditor's Name  Describe the property that secures the claim is: Check all that apply. City State ZIP Code  Who owes the debt? Check one. Debtor 1 only Debtor 2 only Debtor 1 only Debtor 1 only Debtor 2 only Debtor 2 only Debtor 3 only Debtor 4 only Debtor 5 only Debtor 5 only Debtor 5 only Debtor 6 only Debtor 7 only Debtor 1 only Debtor 1 only Debtor 1 only Debtor 2 only Debtor 3 only Debtor 4 only Debtor 5 only Debtor 5 only Debtor 6 only Debtor 7 only Debtor 8 only Debtor 9 only Debtor 9 only Debtor 9 only Debtor 1 only Debtor 9 only Debtor 1 only Debtor 9 only Debtor 1 on		Contingent			
Who owes the debt? Check one.  Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only Al least one of the debtors and another Check if this claim relates to a community debt  Date debt was incurred  Last 4 digits of account number  City State ZIP Code Debtor 1 and Debtor 2 only Debtor 2 only Debtor 1 and Debtor 2 only Debtor 2 anly Debtor 3 and Debtor 2 only Debtor 4 least 4 digits of account number Debtor 4 least 4 digits of account number Debtor 4 least 4 digits of account number Debtor 5 least 8	City State ZIP Code				
Debtor 1 only	·	☐ Disputed			
Debtor 2 only	_	Nature of lien. Check all that apply.			
Debtor 1 and Debtor 2 only					
At least one of the debtors and another   Check if this claim relates to a community debt   Check if this claim relates to a community debt   Check if this claim relates to a community debt   Cast 4 digits of account number   Creditor's Name   Creditor's Name   Creditor's Name   Contingent   Contingen	<u> </u>				
Community debt  Date debt was incurred Last 4 digits of account number  Describe the property that secures the claim: \$ \$	At least one of the debtors and another				
Date debt was incurred Last 4 digits of account number  Describe the property that secures the claim: \$ \$	☐ Check if this claim relates to a	Other (including a right to offset)			
Creditor's Name Number Street  As of the date you file, the claim is: Check all that apply. Contingent Unliquidated Disputed  Who owes the debt? Check one. Debtor 1 only Debtor 2 only Debtor 2 only An agreement you made (such as mortgage or secured car loan) Statutory lien (such as tax lien, mechanic's lien) Judgment lien from a lawsuit Disputed  Add the dollar value of your entries in Column A on this page. Write that number here:  \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	community debt				
As of the date you file, the claim is: Check all that apply.  City State ZIP Code Unliquidated Disputed  Who owes the debt? Check one. Nature of lien. Check all that apply.  Debtor 1 only An agreement you made (such as mortgage or secured car loan)  Debtor 2 only Statutory lien (such as tax lien, mechanic's lien)  At least one of the debtors and another Unliquiding a right to offset)  Check if this claim relates to a community debt  Date debt was incurred Last 4 digits of account number —  As of the date you file, the claim is: Check all that apply.  Check all that apply.  An agreement you made (such as mortgage or secured car loan)  Statutory lien (such as tax lien, mechanic's lien)  Judgment lien from a lawsuit  Other (including a right to offset)  Sample of the date you file, the claim is: Check all that apply.  Check all that apply.  As of the date you file, the claim is: Check all that apply.  Check all that apply.  An agreement you made (such as mortgage or secured car loan)  Car loan)  Check if this claim relates to a community debt  Date debt was incurred Last 4 digits of account number —	Date debt was incurred	Last 4 digits of account number			
As of the date you file, the claim is: Check all that apply.    Contingent   Unliquidated   Disputed		Describe the property that secures the claim:	\$	\$	\$
As of the date you file, the claim is: Check all that apply.    Contingent   Unliquidated   Disputed	Creditor's Name				
City State ZIP Code Unliquidated Disputed  Who owes the debt? Check one. Nature of lien. Check all that apply.  Debtor 1 only An agreement you made (such as mortgage or secured car loan)  Debtor 2 only Statutory lien (such as tax lien, mechanic's lien)  At least one of the debtors and another Other (including a right to offset)  Check if this claim relates to a community debt  Date debt was incurred Last 4 digits of account number Add the dollar value of your entries in Column A on this page. Write that number here:  Add the dollar value of your entries in Column A on this page. Write that number here:	Number Street				
City State ZIP Code Unliquidated Disputed  Who owes the debt? Check one. Nature of lien. Check all that apply.  Debtor 1 only An agreement you made (such as mortgage or secured car loan)  Debtor 2 only Statutory lien (such as tax lien, mechanic's lien)  At least one of the debtors and another Other (including a right to offset)  Check if this claim relates to a community debt  Date debt was incurred Last 4 digits of account number Add the dollar value of your entries in Column A on this page. Write that number here:  Add the dollar value of your entries in Column A on this page. Write that number here:		As of the date you file, the claim is: Check all that apply	I		
Who owes the debt? Check one.  Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this claim relates to a community debt  Date debt was incurred Last 4 digits of account number  Add the dollar value of your entries in Column A on this page. Write that number here:  Disputed  Nature of lien. Check all that apply. An agreement you made (such as mortgage or secured car loan) Statutory lien (such as tax lien, mechanic's lien) Judgment lien from a lawsuit Other (including a right to offset)  Last 4 digits of account number  \$		<u> </u>			
Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this claim relates to a community debt  Date debt was incurred Last 4 digits of account number  An agreement you made (such as mortgage or secured car loan) Statutory lien (such as tax lien, mechanic's lien) Judgment lien from a lawsuit Other (including a right to offset)  Last 4 digits of account number  Add the dollar value of your entries in Column A on this page. Write that number here:  \$	City State ZIP Code				
Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this claim relates to a community debt  Date debt was incurred  Last 4 digits of account number  Add the dollar value of your entries in Column A on this page. Write that number here:  Statutory lien (such as tax lien, mechanic's lien)  Judgment lien from a lawsuit Other (including a right to offset)  Last 4 digits of account number  S  Add the dollar value of your entries in Column A on this page. Write that number here:	Who owes the debt? Check one.	Nature of lien. Check all that apply.			
Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this claim relates to a community debt  Date debt was incurred Last 4 digits of account number  Add the dollar value of your entries in Column A on this page. Write that number here:  Statutory lien (such as tax lien, mechanic's lien)  Judgment lien from a lawsuit Other (including a right to offset)  Check if this claim relates to a community debt  Last 4 digits of account number	Debtor 1 only	☐ An agreement you made (such as mortgage or secured			
At least one of the debtors and another  Check if this claim relates to a community debt  Date debt was incurred Last 4 digits of account number  Add the dollar value of your entries in Column A on this page. Write that number here:  \$ \$					
Check if this claim relates to a community debt  Date debt was incurred Last 4 digits of account number  Add the dollar value of your entries in Column A on this page. Write that number here:  \$					
Check if this claim relates to a community debt  Date debt was incurred Last 4 digits of account number  Add the dollar value of your entries in Column A on this page. Write that number here:					
Add the dollar value of your entries in Column A on this page. Write that number here:	community debt				
<del></del>	Date debt was incurred	Last 4 digits of account number	Г	ı	
If this is the last page of your form, add the dollar value totals, from all pages			\$		
Write that number here:		, add the dollar value totals from all pages.	\$		

#### Part 2:

#### List Others to Be Notified for a Debt That You Already Listed

Use this page only if you have others to be notified about your bankruptcy for a debt that you already listed in Part 1. For example, if a collection agency is trying to collect from you for a debt you owe to someone else, list the creditor in Part 1, and then list the collection agency here. Similarly, if you have more than one creditor for any of the debts that you listed in Part 1, list the additional creditors here. If you do not have additional persons to be notified for any debts in Part 1, do not fill out or submit this page.

be	notified for	any debts in Part 1, d	lo not fill out or submit	this page.	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
					On which line in Part 1 did you enter the creditor?
	Name				Last 4 digits of account number
	Number	Street			-
					-
	City		State	ZIP Code	-
					On which line in Part 1 did you enter the creditor?
	Name				Last 4 digits of account number
	Number	Street			-
					-
	City		State	ZIP Code	-
					On which line in Part 1 did you enter the creditor?
	Name				Last 4 digits of account number
	Number	Street			-
					-
	City		State	ZIP Code	-
					On which line in Part 1 did you enter the creditor?
	Name				Last 4 digits of account number
	Number	Street			-
					-
	City		State	ZIP Code	-
					On which line in Part 1 did you enter the creditor?
	Name				Last 4 digits of account number
	Number	Street			-
					-
	City		State	ZIP Code	-
					On which line in Part 1 did you enter the creditor?
	Name				Last 4 digits of account number
	Number	Street			-
					-
	City		State	ZIP Code	-

Fi	Il in this information to identify your case:				
De	ebtor 1				
De	First Name Middle Name	Last Name			
(S	pouse, if filing) First Name Middle Name	Last Name			
Ur	nited States Bankruptcy Court for the:	District of (State)		☐ Chec	k if this is an
	ase number				nded filing
Of	fficial Form 106E/F				
	chedule E/F: Creditors W	/ho Have Unsecured CI	aims		12/15
List A/E cre- nee any	as complete and accurate as possible. Use Part the other party to any executory contracts or use. Property (Official Form 106A/B) and on Schede ditors with partially secured claims that are listeded, copy the Part you need, fill it out, number to additional pages, write your name and case number to additional pages.	nexpired leases that could result in a claim. A ule G: Executory Contracts and Unexpired Lead d in Schedule D: Creditors Who Have Claims he entries in the boxes on the left. Attach the mber (if known).	Also list executory co ases (Official Form 1 Secured by Property	ontracts on So 06G). Do not i v. If more space	chedule include any ce is
	rt 1: List All of Your PRIORITY Unsecure				
	Do any creditors have priority unsecured claims  No. Go to Part 2.  Yes.	s against you?			
	List all of your priority unsecured claims. If a creach claim listed, identify what type of claim it is. If nonpriority amounts. As much as possible, list the cunsecured claims, fill out the Continuation Page of (For an explanation of each type of claim, see the in	a claim has both priority and nonpriority amounts laims in alphabetical order according to the credi Part 1. If more than one creditor holds a particula	, list that claim here ar tor's name. If you have r claim, list the other c	nd show both p e more than tw	oriority and o priority
	(1 of all explanation of each type of claim, see the	istractions for this form in the instruction bookiet.	Total claim	Priority	Nonpriority
2.1				amount	amount
	Priority Creditor's Name	Last 4 digits of account number	\$	\$	\$
	Number Street	When was the debt incurred?			
	Number Sheet	As of the date you file, the claim is: Check all the	at apply.		
	City State ZIP Code	☐ Contingent	11.7		
	Who incurred the debt? Check one.	☐ Unliquidated			
	Debtor 1 only	☐ Disputed			
	Debtor 2 only	Type of PRIORITY unsecured claim:			
	Debtor 1 and Debtor 2 only	☐ Domestic support obligations			
	At least one of the debtors and another	☐ Taxes and certain other debts you owe the gover	nment		
	☐ Check if this claim is for a community debt	Claims for death or personal injury while you wer	е		
	Is the claim subject to offset?	intoxicated  Other. Specify			
	☐ Yes	_ oneoposity			
2.2		Last 4 digits of account number	Ф.		r
	Priority Creditor's Name	When was the debt incurred?	—		_ ⊅
	Number Street		_		
		As of the date you file, the claim is: Check all the	at apply.		
	City State ZIP Code	☐ Contingent☐ Unliquidated			
	Who incurred the debt? Check one.	Disputed			
	Debtor 1 only	·			
	Debtor 2 only	Type of PRIORITY unsecured claim:  Domestic support obligations			
	Debtor 1 and Debtor 2 only	☐ Taxes and certain other debts you owe the gover	nment		
	At least one of the debtors and another	☐ Claims for death or personal injury while you wer			
	☐ Check if this claim is for a community debt	intoxicated			
	Is the claim subject to offset?	Other. Specify			
	☐ No☐ Yes				

Last Name

r listing any entries on this page, number then	n beginning with 2.3, followed by 2.4, and so forth.	Total claim	Priority amount	Nonprio amount
Dringitu Candidada Nama	Last 4 digits of account number	\$	\$	\$
Priority Creditor's Name	When was the debt incurred?			
Number Street				
	As of the date you file, the claim is: Check all that apply.			
City State ZIP Code	☐ Contingent ☐ Unliquidated			
•	☐ Disputed			
Who incurred the debt? Check one.	Type of PRIORITY unsecured claim:			
Debtor 1 only Debtor 2 only				
Debtor 1 and Debtor 2 only	<ul><li>Domestic support obligations</li><li>Taxes and certain other debts you owe the government</li></ul>			
At least one of the debtors and another	☐ Claims for death or personal injury while you were			
☐ Check if this claim is for a community debt	intoxicated  Other. Specify			
Is the claim subject to offset?				
□ No				
Yes				
	Last 4 digits of account number	\$	\$	\$
Priority Creditor's Name	When was the debt incurred?			
Number Street	valien was the dept incurred?			
	As of the date you file, the claim is: Check all that apply.			
	☐ Contingent			
City State ZIP Code	Unliquidated			
Who incurred the debt? Check one.	☐ Disputed			
☐ Debtor 1 only	Type of PRIORITY unsecured claim:			
Debtor 2 only	☐ Domestic support obligations			
<ul><li>□ Debtor 1 and Debtor 2 only</li><li>□ At least one of the debtors and another</li></ul>	☐ Taxes and certain other debts you owe the government			
_	☐ Claims for death or personal injury while you were intoxicated			
☐ Check if this claim is for a community debt	Other. Specify			
Is the claim subject to offset?				
□ No				
Yes				
Priority Creditor's Name	Last 4 digits of account number	\$	\$	\$
Stocker o Harrie	When was the debt incurred?			
Number Street				
	As of the date you file, the claim is: Check all that apply.			
	☐ Contingent			
City State ZIP Code	☐ Unliquidated☐ Disputed			
Who incurred the debt? Check one.	■ Disputed			
☐ Debtor 1 only	Type of PRIORITY unsecured claim:			
Debtor 2 only	☐ Domestic support obligations			
Debtor 1 and Debtor 2 only	☐ Taxes and certain other debts you owe the government			
☐ At least one of the debtors and another	Claims for death or personal injury while you were			
☐ Check if this claim is for a community debt	intoxicated  Other. Specify			
Is the claim subject to offset?				
□ No				
□ Ves				

#### Part 2:

### List All of Your NONPRIORITY Unsecured Claims

3.	B. Do any creditors have nonpriority unsecured claims against you?  ☐ No. You have nothing to report in this part. Submit this form to the court with your other schedules. ☐ Yes						
4.	List all of your nonpriority unsecured claims in the alphabetical of nonpriority unsecured claim, list the creditor separately for each claim included in Part 1. If more than one creditor holds a particular claim, li claims fill out the Continuation Page of Part 2.	. For each claim listed, identify what type of claim it is. Do not	list claims already				
			Total claim				
4.1		look dedicite of account mountain					
	Nonpriority Creditor's Name	Last 4 digits of account number	\$				
		When was the debt incurred?					
	Number Street						
	City State ZIP Code	As of the date you file, the claim is: Check all that apply.					
		☐ Contingent					
	Who incurred the debt? Check one.	Unliquidated					
	Debtor 1 only	Disputed					
	Debtor 2 only	Type of NONPRIORITY unsecured claim:					
	☐ Debtor 1 and Debtor 2 only ☐ At least one of the debtors and another						
	_	<ul><li>☑ Student loans</li><li>☑ Obligations arising out of a separation agreement or divorce</li></ul>					
	☐ Check if this claim is for a community debt	that you did not report as priority claims					
	Is the claim subject to offset?	Debts to pension or profit-sharing plans, and other similar debts	:				
	□ No	Other. Specify					
	☐ Yes						
4.2		Last 4 digits of account number	\$				
	Nonpriority Creditor's Name	When was the debt incurred?					
	Number Street	As of the date you file, the claim is: Check all that apply.					
	City State ZIP Code						
		Contingent					
	Who incurred the debt? Check one.	☐ Unliquidated ☐ Disputed					
	Debtor 1 only	■ Disputeu					
	☐ Debtor 2 only ☐ Debtor 1 and Debtor 2 only	Type of NONPRIORITY unsecured claim:					
	☐ At least one of the debtors and another	☐ Student loans					
		Obligations arising out of a separation agreement or divorce					
	☐ Check if this claim is for a community debt	that you did not report as priority claims					
	Is the claim subject to offset?	<ul> <li>□ Debts to pension or profit-sharing plans, and other similar debts</li> <li>□ Other. Specify</li> </ul>	;				
	☐ No ☐ Yes	- Galer. Opeony					
	■ res						
4.3		Last 4 digits of account number	\$				
	Nonpriority Creditor's Name	When was the debt incurred?	Ψ				
	Number Street						
		As of the date you file, the claim is: Check all that apply.					
	City State ZIP Code	_					
	Who incurred the debt? Check one.	☐ Contingent ☐ Unliquidated					
	☐ Debtor 1 only	☐ Disputed					
	Debtor 2 only	_ Disputed					
	☐ Debtor 1 and Debtor 2 only	Type of NONPRIORITY unsecured claim:					
	☐ At least one of the debtors and another	☐ Student loans					
	☐ Check if this claim is for a community debt	Obligations arising out of a separation agreement or divorce					
	Is the claim subject to offset?	that you did not report as priority claims					
	□ No	Debts to pension or profit-sharing plans, and other similar debts					
	☐ Yes	Other. Specify					

Part 2:

Your NONPRIORITY Unsecured Claims - Continuation Page

Last Name

After listing any entries on this page, number them beginning wit	th 4.5, followed by 4.6, and so forth.	Total claim
	Last 4 digits of account number	\$
Nonpriority Creditor's Name	When was the debt incurred?	
Number Street	As of the date you file, the claim is: Check all that apply.	
City State ZIP Code	Contingent Unliquidated	
Who incurred the debt? Check one.  Debtor 1 only	☐ Disputed	
☐ Debtor 2 only ☐ Debtor 1 and Debtor 2 only	Type of <b>NONPRIORITY</b> unsecured claim:	
At least one of the debtors and another	☐ Student loans ☐ Obligations arising out of a separation agreement or divorce that	
☐ Check if this claim is for a community debt	you did not report as priority claims  Debts to pension or profit-sharing plans, and other similar debts	
Is the claim subject to offset?  No Yes	Other. Specify	
	Last 4 digits of account number	\$
Nonpriority Creditor's Name	When was the debt incurred?	
Number Street	As of the date you file, the claim is: Check all that apply.	
City State ZIP Code	Contingent Unliquidated	
Who incurred the debt? Check one.  Debtor 1 only	☐ Disputed	
Debtor 2 only	Type of <b>NONPRIORITY</b> unsecured claim:	
<ul><li>Debtor 1 and Debtor 2 only</li><li>At least one of the debtors and another</li></ul>	<ul><li>Student loans</li><li>Obligations arising out of a separation agreement or divorce that</li></ul>	
☐ Check if this claim is for a community debt	you did not report as priority claims  Debts to pension or profit-sharing plans, and other similar debts	
Is the claim subject to offset?	Other. Specify	
□ No □ Yes		
	Last 4 digits of account number	\$
Nonpriority Creditor's Name	When was the debt incurred?	
Number Street	As of the date you file, the claim is: Check all that apply.	
City State ZIP Code	Contingent Unliquidated	
Who incurred the debt? Check one.	☐ Disputed	
☐ Debtor 1 only ☐ Debtor 2 only	Type of NONPRIORITY unsecured claim:	
☐ Debtor 1 and Debtor 2 only ☐ At least one of the debtors and another	Student loans	
☐ Check if this claim is for a community debt	Obligations arising out of a separation agreement or divorce that you did not report as priority claims	
Is the claim subject to offset?	<ul><li>Debts to pension or profit-sharing plans, and other similar debts</li><li>Other. Specify</li></ul>	
□ No □ Yes	,	

Part 3:

List Others to Be Notified About a Debt That You Already Listed

			On which entry in Part 1 or Part 2 did you list the original creditor?
			Line of (Check one):  Part 1: Creditors with Priority Unsecured Claims
Street			□ Part 2: Creditors with Nonpriority Unsecured Cla
			Last 4 digits of account number
	State	ZIP Code	
			On which entry in Part 1 or Part 2 did you list the original creditor?
			Franchis Charles and D. Dart A. Oarditary with Drivita Unaccount Claire
Street			Line of (Check one): Part 1: Creditors with Priority Unsecured Claims
0001			Part 2: Creditors with Nonpriority Unsecured Claims
	State	ZIP Code	Last 4 digits of account number
			On which entry in Part 1 or Part 2 did you liet the existing exeditor?
			On which entry in Part 1 or Part 2 did you list the original creditor?
			Line of (Check one):  Part 1: Creditors with Priority Unsecured Claims
Street			☐ Part 2: Creditors with Nonpriority Unsecured
			Claims
			Last 4 digits of account number
	State	ZIP Code	
			On which entry in Part 1 or Part 2 did you list the original creditor?
			Line of (Check one):  Part 1: Creditors with Priority Unsecured Claims
Street			Part 2: Creditors with Nonpriority Unsecured
			Claims
			Look 4 digita of account number
	State	ZIP Code	Last 4 digits of account number
			On which entry in Part 1 or Part 2 did you list the original creditor?
<del></del>			
			Line of (Check one):  Part 1: Creditors with Priority Unsecured Claims
Street			Part 2: Creditors with Nonpriority Unsecured Claims
			Olains
			Look A divite of account number
	State	ZIP Code	Last 4 digits of account number
	State	ZIP Code	
	State	ZIP Code	On which entry in Part 1 or Part 2 did you list the original creditor?
	State	ZIP Code	
Street	State	ZIP Code	On which entry in Part 1 or Part 2 did you list the original creditor?  Line of (Check one):  Part 1: Creditors with Priority Unsecured Claims
Street	State	ZIP Code	On which entry in Part 1 or Part 2 did you list the original creditor?  Line of (Check one):  Part 1: Creditors with Priority Unsecured Claims
Street			On which entry in Part 1 or Part 2 did you list the original creditor?  Line of (Check one):  Part 1: Creditors with Priority Unsecured Claims
Street	State	ZIP Code	On which entry in Part 1 or Part 2 did you list the original creditor?  Line of (Check one):  Part 1: Creditors with Priority Unsecured Claims Part 2: Creditors with Nonpriority Unsecured Claims
Street			On which entry in Part 1 or Part 2 did you list the original creditor?  Line of (Check one):  Part 1: Creditors with Priority Unsecured Claims Part 2: Creditors with Nonpriority Unsecured Claims
Street			On which entry in Part 1 or Part 2 did you list the original creditor?  Line of (Check one):  Part 1: Creditors with Priority Unsecured Claims Part 2: Creditors with Nonpriority Unsecured Claims  Last 4 digits of account number On which entry in Part 1 or Part 2 did you list the original creditor?
Street			On which entry in Part 1 or Part 2 did you list the original creditor?  Line of (Check one):  Part 1: Creditors with Priority Unsecured Claims Part 2: Creditors with Nonpriority Unsecured Claims  Last 4 digits of account number On which entry in Part 1 or Part 2 did you list the original creditor?  Line of (Check one):  Part 1: Creditors with Priority Unsecured Claims
			On which entry in Part 1 or Part 2 did you list the original creditor?  Line of (Check one):  Part 1: Creditors with Priority Unsecured Claims Part 2: Creditors with Nonpriority Unsecured Claims  Last 4 digits of account number On which entry in Part 1 or Part 2 did you list the original creditor?
	Street	Street  Street  Street  State	Street  State ZIP Code  Street  State ZIP Code  Street  State ZIP Code

Part 4:

Middle Name

Last Name

Add the Amounts for Each Type of Unsecured Claim

6. Total the amounts of certain types of unsecured claims. This information is for statistical reporting purposes only. 28 U.S.C. § 159. Add the amounts for each type of unsecured claim.

# Total claims from Part 1

- 6a. Domestic support obligations
- 6b. Taxes and certain other debts you owe the government
- 6c. Claims for death or personal injury while you were intoxicated
- Other. Add all other priority unsecured claims.
   Write that amount here.
- 6e. Total. Add lines 6a through 6d.

## Total claims from Part 2

- 6f. Student loans
- 6g. Obligations arising out of a separation agreement or divorce that you did not report as priority claims
- 6h. Debts to pension or profit-sharing plans, and other similar debts
- 6i. **Other.** Add all other nonpriority unsecured claims. Write that amount here.
- 6j. Total. Add lines 6f through 6i.

#### Total claim

- 6a. <sub>\$</sub>
- 6b. <sub>\$</sub>
- 6c.
- 6d. + s
- 6e. \$\_\_\_\_\_

#### **Total claim**

- 6f. \$\_\_\_\_\_
- 6g. \$\_\_\_\_\_
- 6h. s
- 6i **+** ¢
- 6j. \$\_\_\_\_\_

Fill in this information to identify your case:				
Debtor _	First Name	Middle Name	Last Name	
Debtor 2 (Spouse If filing)	First Name	Middle Name	Last Name	-
United States E	Bankruptcy Court fo	or the:	District of	
Case number (If known)			(State)	

☐ Check if this is an amended filing

## Official Form 106G

## Schedule G: Executory Contracts and Unexpired Leases

12/15

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. If more space is needed, copy the additional page, fill it out, number the entries, and attach it to this page. On the top of any additional pages, write your name and case number (if known).

- 1. Do you have any executory contracts or unexpired leases?
  - ☐ No. Check this box and file this form with the court with your other schedules. You have nothing else to report on this form.
  - Tyes. Fill in all of the information below even if the contracts or leases are listed on Schedule A/B: Property (Official Form 106A/B).
- 2. List separately each person or company with whom you have the contract or lease. Then state what each contract or lease is for (for example, rent, vehicle lease, cell phone). See the instructions for this form in the instruction booklet for more examples of executory contracts and unexpired leases.

	Person or	company with who	om you l	nave the contract or lease	State what the contract or lease is for
2.1					
	Name				_
	Number	Street			_
	City		State	ZIP Code	-
2.2	Name				_
	Number	Street			_
	City	Sueet	State	ZIP Code	_
2.3	City		State	ZIP Code	
	Name				_
	Number	Street			_
	City		State	ZIP Code	-
2.4	Name				_
	Number	Street			_
	City		State	ZIP Code	_
2.5	City		State	ZIF Code	
	Name				_
	Number	Street			_
	City		State	ZIP Code	_

First Name	Middle Name	Last Name	

Case number	(if known)

Additional Page if You Have More Contracts or Leases

	Person or	company with v	whom you h	nave the contract or lease	What the contract or lease is for
2					
	Name				-
	Number	Street			-
	City		State	ZIP Code	
2					
	Name				
	Number	Street			-
	City		State	ZIP Code	-
2					
	Name				-
	Number	Street			-
	City		State	ZIP Code	-
2					
	Name				-
	Number	Street			-
	City		State	ZIP Code	-
2					
	Name				
	Number	Street			-
	City		State	ZIP Code	
2					
	Name				-
	Number	Street			-
	City		State	ZIP Code	
2					
	Name				-
	Number	Street			-
	City		State	ZIP Code	-
2					
	Name				-
	Number	Street			-
	City		State	ZIP Code	-

Fill i	n this information to identify your case:		
Debte			
Debte	First Name Middle Name or 2	Last Name	
(Spou	se, if filing) First Name Middle Name	Last Name	
Unite	d States Bankruptcy Court for the:	District of (State)	
Case (If kn	number		☐ Check if this is an
			amended filing
Offi	cial Form 106H		
Scl	nedule H: Your Codebtors	3	12/15
are fil and n	ng together, both are equally responsible for supp	plying correct information. If	e as complete and accurate as possible. If two married people f more space is needed, copy the Additional Page, fill it out, page. On the top of any Additional Pages, write your name and
	o you have any codebtors? (If you are filing a joint o	ase, do not list either spouse	as a codebtor.)
	Yes	it.,	n 2 / Community nyonayty atatag and to with vice include
	ritnin the last 8 years, nave you lived in a commun rizona, California, Idaho, Louisiana, Nevada, New Me		y? (Community property states and territories include ashington, and Wisconsin.)
	No. Go to line 3.		
	Yes. Did your spouse, former spouse, or legal equi	valent live with you at the time	e?
	No	ou livo?	Fill in the name and current address of that person.
	Yes. In which community state or territory did yo	ou live?	Fill in the name and current address of that person.
			_
	Name of your spouse, former spouse, or legal equivalent		
	Number Street		_
			_
	City State	ZIP Code	
s	Column 1, list all of your codebtors. Do not include hown in line 2 again as a codebtor only if that persistence D (Official Form 106D), Schedule E/F (Official Form 206D), Schedule E/F, or Schedule G to fill out Column 2.	son is a guarantor or cosign	
	Column 1: Your codebtor		Column 2: The creditor to whom you owe the debt
	Column 1. 10th Country		Check all schedules that apply:
3.1			Crieck all scriedules that apply.
0.1	Name		Schedule D, line
			Schedule E/F, line
	Number Street		☐ Schedule G, line
	City State	ZIP Code	
3.2			Schedule D, line
_	Name		Schedule E/F, line
	Number Street		Schedule G, line
	Cib.	710.0-1	
3.3	City State	ZIP Code	
5.5	Name		Schedule D, line
			Schedule E/F, line
	Number Street		Schedule G, line
	City State	ZIP Code	<del></del>

65

		-	t More Codebtors		
	Column 1:	Your codebtor			Column 2: The creditor to whom you owe the debt
					Check all schedules that apply:
3					_ Schedule D, line
	Name				Schedule E/F, line
	Number	Street			Schedule G, line
	City		State	ZIP Code	
3					Schedule D, line
	Name				☐ Schedule E/F, line
	Number	Street			Schedule G, line
	City		State	ZIP Code	
3					Schedule D, line
	Name				☐ Schedule E/F, line
	Number	Street			Schedule G, line
	City		State	ZIP Code	_
3					_ Schedule D, line
	Name				Schedule E/F, line
	Number	Street			Schedule G, line
	City		State	ZIP Code	_
3					_ Schedule D, line
	Name				Schedule E/F, line
	Number	Street			Schedule G, line
	City		State	ZIP Code	_
3					_ Schedule D, line
	Name				☐ Schedule E/F, line
	Number	Street			Schedule G, line
	City		State	ZIP Code	<del>-</del>
3					Schedule D, line
	Name				Schedule E/F, line
	Number	Street			Schedule G, line
3.	City		State	ZIP Code	<del>-</del>
·					Schedule D, line
	Name				☐ Schedule E/F, line
	Number	Street			Schedule G, line
	City		State	7IP Code	_

Debtor 1   Debtor 2   Debtor 3   Debtor 4   Debtor 5   Debtor 6   Debtor 6   Debtor 6   Debtor 6   Debtor 7   Debtor 7   Debtor 7   Debtor 7   Debtor 8   Debtor 8   Debtor 9   Debtor 9	Fill in this information to identify	your case:				
Employer's address   Employer's address   Employer's name   Employer's name   Employer's name   Employer's address   Employer's name   E						
Check if this is:   Check   An amended filling   Country	First Name	Middle Name	Last Name			
Case number		Middle Name	Last Name			
Check if this is:    An amended fling	United States Bankruptcy Court for the:					
An amended filing  An amended filing  An ampended filing  An applement showing postpetition chapter 13 income as of the following date:  MM / DD / YYYY  Be as complete and accurate as possible. If two married people are filing together (Debtor 1 and Debtor 2), both are equally responsible for supplying correct information, if you are married and not filing jointly, and your spouse is living with you, include your propose. If more space is needed, attach a separated and your spouse is not filing with you, do not include information about your spouse. If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write your name and case number (if known). Answer every question.  Part 1:  Describe Employment  1. Fill in your employment information. If you have more than one job, attach a separate page with information about additional employers.  Include part-time, seasonal, or self-employed work.  Occupation may include student or homemaker, if it apples.  Employer's name  Employer's address  Number Street  Number Street  Number Street  Number Street  Number Street  Part 2:  Give Details About Monthly Income  Estimate monthly income as of the date you file this form. If you have nothing to report for any line, write \$0 in the space. Include your non-filing spouse unless you are separated.  If you row your non-filing spouse have more than one employer, combine the information for all employers for that person on the lines below. If you need more space, attach a separate sheet to this form.  For Debtor 1 For Debtor 2 or non-filing spouse  2. List monthly gross wages, salary, and commissions (before all payroll deductions), if not paid monthly, calculate what the monthly wage would be.  2. Setimate and list monthly overtime pay.  3. + Setimate and list monthly overtime pay.			(Glate)		Check if the	nis is:
Schedule I: Your Income    Schedule I: Your Income   1/15	(If Known)				An am	ended filing
Be as complete and accurate as possible. If two married people are filing together (Debtor 1 and Debtor 2), both are equally responsible for supplying correct information. If you are married and not filing jointly, and your spouse is living with you, on include information about your spouse, if you are separated and your spouse is not filing with you, do not include information about your spouse. If you are separated and your spouse is not filing with you, do not include information states a separate sheet to this form. On the top of any additional pages, write your name and case number (if known). Answer every question.  Part 1:  Describe Employment  I. Fill in your employment information. If you have more than one job, attach a separate page with information about additional employers.  Include part-time, seasonal, or self-employed work.  Occupation may include student or homemaker, if it applies.  Employer's address  Employer's address  Employer's address    Number   Street   Number   Street						
Be as complete and accurate as possible. If two married people are filing together (Debtor 1 and Debtor 2), both are equally responsible for supplying correct information. If you are married and not filing jointly, and your spouse is Inviting with you, on on tinclude information about your spouse. If you are separated and your spouse is not filing with you, do not include information about your spouse. If you respace is needed, attach a separate sheet to this form. On the top of any additional pages, write your name and case number (if known). Answer every question.    Part 1:	Official Form 106I				MM / D	D / YYYY
supplying correct information. If you are married and not filing jointly, and your spouse is Inling with you, on contributed information about your spouse. In filing with you, do not include information about your spouse. In separate sheet to this form. On the top of any additional pages, write your name and case number (if known). Answer every question.  Part 1: Describe Employment    Fill in your employment   Debtor 1   Debtor 2 or non-filing spouse	Schedule I: You	ır Income				12/15
If you have more than one job, attach a separate page with information about additional employers.  Include part-time, seasonal, or self-employed work.  Occupation may include student or homemaker, if it applies.  Employer's name  Employer's address    City   State   ZIP Code   City   State   ZIP Code	supplying correct information. If yo If you are separated and your spou separate sheet to this form. On the	ou are married and not filing se is not filing with you, of top of any additional pag	ng jointly, and you lo not include info	r spouse is rmation ab	s living with y out your spo	ou, include information about your spouse. use. If more space is needed, attach a
If you have more than one job, attach a separate page with information about additional employers.  Include part-time, seasonal, or self-employed work.  Occupation may include student or homemaker, if it applies.  Employer's address    City	Fill in your employment					
atach a separate page with information about additional employers.  Include part-time, seasonal, or self-employed work.  Occupation may include student or homemaker, if it applies.  Employer's name  Employer's name  Employer's saddress  Number Street  Number Street  City State ZIP Code  City State ZIP Code  How long employed there?  Part 2: Give Details About Monthly Income  Estimate monthly income as of the date you file this form. If you have nothing to report for any line, write \$0 in the space. Include your non-filing spouse have more than one employer, combine the information for all employers for that person on the lines below. If you need more space, attach a separate sheet to this form.  For Debtor 1  For Debtor 2 or non-filing spouse  2. List monthly gross wages, salary, and commissions (before all payroll deductions). If not paid monthly, calculate what the monthly wage would be.  2. \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	information.		Debtor 1			Debtor 2 or non-filing spouse
Self-employed work.  Occupation may include student or homemaker, if it applies.  Employer's name  Employer's address  Number Street	attach a separate page with information about additional	Employment status		d		<u> </u>
Occupation may include student or homemaker, if it applies.  Employer's name  Employer's address  Number Street						
Employer's address    Number   Street   Number   Street	Occupation may include student	Occupation			<del></del>	
Number Street    Number Street   Number Street   Number Street		Employer's name				
Number Street    Number Street   Number Street   Number Street		Employer's address				
How long employed there?  Estimate monthly income as of the date you file this form. If you have nothing to report for any line, write \$0 in the space. Include your non-filing spouse unless you are separated.  If you or your non-filing spouse have more than one employer, combine the information for all employers for that person on the lines below. If you need more space, attach a separate sheet to this form.  For Debtor 1  For Debtor 2 or non-filling spouse  2. List monthly gross wages, salary, and commissions (before all payroll deductions). If not paid monthly, calculate what the monthly wage would be.  3. Estimate and list monthly overtime pay.  3. + \$			Number Street			Number Street
How long employed there?  Estimate monthly income as of the date you file this form. If you have nothing to report for any line, write \$0 in the space. Include your non-filing spouse unless you are separated.  If you or your non-filing spouse have more than one employer, combine the information for all employers for that person on the lines below. If you need more space, attach a separate sheet to this form.  For Debtor 1  For Debtor 2 or non-filling spouse  2. List monthly gross wages, salary, and commissions (before all payroll deductions). If not paid monthly, calculate what the monthly wage would be.  3. Estimate and list monthly overtime pay.  3. + \$						
How long employed there?  Estimate monthly income as of the date you file this form. If you have nothing to report for any line, write \$0 in the space. Include your non-filing spouse unless you are separated.  If you or your non-filing spouse have more than one employer, combine the information for all employers for that person on the lines below. If you need more space, attach a separate sheet to this form.  For Debtor 1  For Debtor 2 or non-filling spouse  2. List monthly gross wages, salary, and commissions (before all payroll deductions). If not paid monthly, calculate what the monthly wage would be.  3. Estimate and list monthly overtime pay.  3. + \$						
How long employed there?  Estimate monthly income as of the date you file this form. If you have nothing to report for any line, write \$0 in the space. Include your non-filing spouse unless you are separated.  If you or your non-filing spouse have more than one employer, combine the information for all employers for that person on the lines below. If you need more space, attach a separate sheet to this form.  For Debtor 1  For Debtor 2 or non-filling spouse  2. List monthly gross wages, salary, and commissions (before all payroll deductions). If not paid monthly, calculate what the monthly wage would be.  3. Estimate and list monthly overtime pay.  3. + \$						
Estimate monthly income as of the date you file this form. If you have nothing to report for any line, write \$0 in the space. Include your non-filing spouse unless you are separated.  If you or your non-filing spouse have more than one employer, combine the information for all employers for that person on the lines below. If you need more space, attach a separate sheet to this form.  For Debtor 1  For Debtor 2 or non-filing spouse  2. List monthly gross wages, salary, and commissions (before all payroll deductions). If not paid monthly, calculate what the monthly wage would be.  3. Estimate and list monthly overtime pay.  3. + \$ + \$			City	State ZIP	Code	City State ZIP Code
Estimate monthly income as of the date you file this form. If you have nothing to report for any line, write \$0 in the space. Include your non-filing spouse unless you are separated.  If you or your non-filing spouse have more than one employer, combine the information for all employers for that person on the lines below. If you need more space, attach a separate sheet to this form.  For Debtor 1  For Debtor 2 or non-filing spouse  2. List monthly gross wages, salary, and commissions (before all payroll deductions). If not paid monthly, calculate what the monthly wage would be.  3. Estimate and list monthly overtime pay.  3. + \$ + \$		How long employed ther	e?			
spouse unless you are separated.  If you or your non-filing spouse have more than one employer, combine the information for all employers for that person on the lines below. If you need more space, attach a separate sheet to this form.  For Debtor 1  For Debtor 2 or non-filing spouse  2. List monthly gross wages, salary, and commissions (before all payroll deductions). If not paid monthly, calculate what the monthly wage would be.  3. Estimate and list monthly overtime pay.  3. + \$ + \$	Part 2: Give Details About	Monthly Income				
If you or your non-filing spouse have more than one employer, combine the information for all employers for that person on the lines below. If you need more space, attach a separate sheet to this form.  For Debtor 1  For Debtor 2 or non-filing spouse  2. List monthly gross wages, salary, and commissions (before all payroll deductions). If not paid monthly, calculate what the monthly wage would be.  2. \$			. If you have nothin	g to report f	for any line, w	rite \$0 in the space. Include your non-filing
2. List monthly gross wages, salary, and commissions (before all payroll deductions). If not paid monthly, calculate what the monthly wage would be.  3. Estimate and list monthly overtime pay.  3. + \$	If you or your non-filing spouse ha	ave more than one employer		mation for a	II employers fo	or that person on the lines
deductions). If not paid monthly, calculate what the monthly wage would be.  2. \$				Fo	r Debtor 1	
				2. \$		\$
4. Calculate gross income. Add line 2 + line 3. 4. \$ \$	3. Estimate and list monthly over	time pay.		3. <b>+</b> \$		+ \$
	4. Calculate gross income. Add lin	ne 2 + line 3.		4. \$		\$

Official Form 106l Schedule I: Your Income 67
page 1

First Name	Middle Name	Last Name

			For Debtor 1		For Debtor 2 o			
Copy line 4 here		<b>→</b> 4.	\$		\$			
5. List all payroll deducti	ons:							
5a. Tax, Medicare, ar	nd Social Security deductions	5a.	\$		\$			
	butions for retirement plans	5b.	\$		\$			
5c. Voluntary contrib	outions for retirement plans	5c.	\$	_	\$			
5d. Required repaym	ents of retirement fund loans	5d.	\$	_	\$			
5e. Insurance		5e.	\$	_	\$			
5f. Domestic suppor	t obligations	5f.	\$	_	\$			
5g. Union dues		5g.	\$	_	\$			
5h. Other deductions	s. Specify:	5h.	+ \$		+ \$			
6. Add the payroll dedu	ctions. Add lines 5a + 5b + 5c + 5d + 5e +5f + 5g + 5h.	6.	\$	-	\$			
7. Calculate total month	nly take-home pay. Subtract line 6 from line 4.	7.	\$	-	\$			
8. List all other income	regularly received:							
8a. Net income from profession, or fa	rental property and from operating a business, m							
	t for each property and business showing gross and necessary business expenses, and the total e.	8a.	\$	_	\$			
8b. Interest and divid		8b.	\$	_	\$			
8c. Family support p regularly receive	ayments that you, a non-filing spouse, or a depende	∍nt						
Include alimony, s settlement, and pr	pousal support, child support, maintenance, divorce operty settlement.	8c.	\$	-	\$			
8d. Unemployment c	ompensation	8d.	\$	_	\$			
8e. Social Security		8e.	\$	_	\$			
Include cash assis that you receive, s Nutrition Assistance	nt assistance that you regularly receive tance and the value (if known) of any non-cash assistar uch as food stamps (benefits under the Supplemental ce Program) or housing subsidies.		\$		\$			
	<del>-</del>	8f.	Ψ	-	Ψ			
8g. Pension or retire	ment income	8g.	\$	-	\$			
8h. Other monthly in	come. Specify:	8h.	+\$		+\$			
9. Add all other income	. Add lines 8a + 8b + 8c + 8d + 8e + 8f +8g + 8h.	9.	\$		\$		_	
10. Calculate monthly inc Add the entries in line	ome. Add line 7 + line 9.  10 for Debtor 1 and Debtor 2 or non-filing spouse.	10.	\$	+	\$		= 5	\$
Include contributions from friends or relatives.	contributions to the expenses that you list in Scheom an unmarried partner, members of your household, younts already included in lines 2-10 or amounts that are	your d	ependents, your ro			ule J.		
Specify:						11. '	+ :	\$
	e last column of line 10 to the amount in line 11. The ne Summary of Your Assets and Liabilities and Certain S				•	12.		\$
	rease or decrease within the year after you file this	form?	,					Combined monthly income
☐ No. ☐ Yes. Explain:								

Fill in this information to identify your case:			
Debtor 1 First Name Middle Name Last Name	Check if this is:		
Debtor 2	——— An amended fi	lina	
(Spouse, if filing) First Name Middle Name Last Name	☐ A supplement	showing postp	
United States Bankruptcy Court for the: District of (State State	·		date:
Case number (If known)	MM / DD / YYYY		
Official Form 106J			
Schedule J: Your Expenses			12/15
Be as complete and accurate as possible. If two married people are filing information. If more space is needed, attach another sheet to this form. (if known). Answer every question.			
Part 1: Describe Your Household			
1. Is this a joint case?			
<ul><li>□ No. Go to line 2.</li><li>□ Yes. Does Debtor 2 live in a separate household?</li></ul>			
□ No			
☐ Yes. Debtor 2 must file Official Form 106J-2, Expenses for Sep	parate Household of Debtor 2.		
	Dependent's relationship to	Dependent's	Does dependent live
Do not list Debtor 1 and Debtor 2. Yes. Fill out this information for each dependent	Debtor 1 or Debtor 2	age	with you?
Do not state the dependents'	<del></del>		□ No □ Yes
names.			□ No
		· · · · · · · · · · · · · · · · · · ·	☐ Yes
	<del></del>		☐ No ☐ Yes
			<ul><li>☑ No</li><li>☑ Yes</li></ul>
			□ No
			Yes
3. Do your expenses include expenses of people other than yourself and your dependents?			
Part 2: Estimate Your Ongoing Monthly Expenses  Estimate your expenses as of your bankruptcy filing date unless you are	using this form as a supplement in	a Chanter 13 c	ase to report
expenses as of a date after the bankruptcy is filed. If this is a supplement applicable date.	•	•	•
Include expenses paid for with non-cash government assistance if you k	now the value of		
such assistance and have included it on Schedule I: Your Income (Official	,	Your exper	nses
<ol> <li>The rental or home ownership expenses for your residence. Include fi any rent for the ground or lot.</li> </ol>	rst mortgage payments and 4.	\$	<del></del>
If not included in line 4:			
4a. Real estate taxes	4a.	\$	<del></del>
4b. Property, homeowner's, or renter's insurance	4b.	\$	<del></del>
4c. Home maintenance, repair, and upkeep expenses	4c.	\$	<del> </del>
4d. Homeowner's association or condominium dues	4d.	\$	

First Name	Middle Name	Last Name	

Case number (if known)\_\_\_\_\_

			Your expenses
5	Additional mortgage payments for your residence, such as home equity loans	5.	\$
		5.	
6.	Utilities:		
	6a. Electricity, heat, natural gas	6a.	\$
	6b. Water, sewer, garbage collection	6b.	\$
	6c. Telephone, cell phone, Internet, satellite, and cable services	6c.	\$
	6d. Other. Specify:	6d.	\$
7.	Food and housekeeping supplies	7.	\$
8.	Childcare and children's education costs	8.	\$
9.	Clothing, laundry, and dry cleaning	9.	\$
10.	Personal care products and services	10.	\$
11.	Medical and dental expenses	11.	\$
12.	<b>Transportation.</b> Include gas, maintenance, bus or train fare.  Do not include car payments.	12.	\$
13.	Entertainment, clubs, recreation, newspapers, magazines, and books	13.	\$
14.	Charitable contributions and religious donations	14.	\$
15.	Insurance.  Do not include insurance deducted from your pay or included in lines 4 or 20.		
	15a. Life insurance	15a.	\$
	15b. Health insurance	15b.	\$
	15c. Vehicle insurance	15c.	\$
	15d. Other insurance. Specify:	15d.	\$
16.	<b>Taxes.</b> Do not include taxes deducted from your pay or included in lines 4 or 20.  Specify:	16.	\$
17.	Installment or lease payments:		
	17a. Car payments for Vehicle 1	17a.	\$
	17b. Car payments for Vehicle 2	17b.	\$
	17c. Other. Specify:	17c.	\$
	17d. Other. Specify:	17d.	\$
18.	Your payments of alimony, maintenance, and support that you did not report as deducted from your pay on line 5, Schedule I, Your Income (Official Form 106I).	18.	\$
19.	Other payments you make to support others who do not live with you.		
	Specify:	19.	\$
20.	Other real property expenses not included in lines 4 or 5 of this form or on Schedule I: Your Incom	ne.	
	20a. Mortgages on other property	20a.	\$
	20b. Real estate taxes	20b.	\$
	20c. Property, homeowner's, or renter's insurance	20c.	\$
	20d. Maintenance, repair, and upkeep expenses	20d.	\$
	20e. Homeowner's association or condominium dues	20e.	\$

Debtor 1 Case First Name Middle Name Last Name	number (if known)	
21. Other. Specify:	21. <b>+</b> \$	
22. Calculate your monthly expenses.		
22a. Add lines 4 through 21.	22a. \$	
22b. Copy line 22 (monthly expenses for Debtor 2), if any, from Official Form 106J-2	22b. \$	
22c. Add line 22a and 22b. The result is your monthly expenses.	22c. \$	
23. Calculate your monthly net income.		
23a. Copy line 12 (your combined monthly income) from Schedule I.	23a. \$	
23b. Copy your monthly expenses from line 22c above.	23b. <b>–</b> \$	<del></del>
23c. Subtract your monthly expenses from your monthly income.  The result is your <i>monthly net income</i> .	23c. \$	
24. Do you expect an increase or decrease in your expenses within the year after you file thi	is form?	
For example, do you expect to finish paying for your car loan within the year or do you expect y mortgage payment to increase or decrease because of a modification to the terms of your mort		
□ No.		

☐ Yes.

Explain here:

Fill in this information to identify	your case:				
Debtor 1 First Name	Middle Name Last Name	Check if thi	s is:		
Debtor 2	Wildle Name Last Name	An ame		ina	
(Spouse, if filing) First Name	Middle Name Last Name			J	petition chapter 13
United States Bankruptcy Court for the:				the following	· ·
Case number (If known)		MM / DD	/ YYYY		
055.15					
Official Form 106J-2	Expenses for Sepa	rata Hausahald	of D	lehter '	2 40/45
	<u> </u>				
Debtor 2 have one or more depend only with respect to expenses for needed, attach another sheet to the question.	ate household expenses ONLY IF D dents in common, list the dependen Debtor 2 that are not reported on So is form. On the top of any additiona	ts on both Schedule J and this t chedule J. Be as complete and a	orm. A	nswer the que as possible.	estions on this form If more space is
Part 1: Describe Your Hou	usehold				
1. Do you and Debtor 1 maintain se	eparate households?				
<ul><li>No. Do not complete this fo</li><li>Yes</li></ul>	rm.				
2. Do you have dependents?	☐ No	Dependent's relationship to		Dependent's	Does dependent live
Do not list Debtor 1 but list all other dependents of Debtor 2	Yes. Fill out this information for each dependent	Debtor 2:		age	with you?
regardless of whether listed as a dependent of Debtor 1 on Schedule J.					☐ Yes
Do not state the dependents'					□ No
names.					☐ Yes☐ No
				<del> </del>	☐ Yes
					□ No
					☐ Yes
					☐ No
					☐ Yes
<ol><li>Do your expenses include expenses of people other than yourself, your dependents, and Debtor 1?</li></ol>	☐ No ☐ Yes				
Part 2: Estimate Your Ongo	ing Monthly Expenses				
	r bankruptcy filing date unless you	are using this form as a suppler	nent in a	a Chanter 13 c	case to report
expenses as of a date after the bar		are doing and form do a cappion		a onaptor to c	acc to roport
·	n-cash government assistance if yo			Your expe	nses
such assistance and have included it on <i>Schedule I: Your Income</i> (Official Form 106I.)  4. The rental or home ownership expenses for your residence. Include first mortgage payments and					
any rent for the ground or lot.  4. \$					<del></del>
If not included in line 4:					
4a. Real estate taxes			4a.		····
4b. Property, homeowner's, or			4b.		<del></del>
4c. Home maintenance, repair,			4c.		<del></del>
4d. Homeowner's association of	r condominium dues		4d.	\$	<del> </del>

First Name Middle Name Last Name

			Your expenses
5.	Additional mortgage payments for your residence, such as home equity loans	5.	\$
6	Utilities:		
0.	6a. Electricity, heat, natural gas	6a.	\$
	6b. Water, sewer, garbage collection	6b.	\$
	6c. Telephone, cell phone, Internet, satellite, and cable services	6c.	\$
	6d. Other. Specify:	6d.	\$
7.	Food and housekeeping supplies	7.	\$
8.	Childcare and children's education costs	8.	\$
9.	Clothing, laundry, and dry cleaning	9.	\$
10.	Personal care products and services	10.	\$
11.	Medical and dental expenses	11.	\$
12.	<b>Transportation.</b> Include gas, maintenance, bus or train fare.  Do not include car payments.	12.	\$
13.	Entertainment, clubs, recreation, newspapers, magazines, and books	13.	\$
14.	Charitable contributions and religious donations	14.	\$
15.	Insurance.		
	Do not include insurance deducted from your pay or included in lines 4 or 20.		
	15a. Life insurance	15a.	\$
	15b. Health insurance	15b.	\$
	15c. Vehicle insurance	15c.	\$
	15d. Other insurance. Specify:	15d.	\$
16.	<b>Taxes.</b> Do not include taxes deducted from your pay or included in lines 4 or 20. Specify:	16.	\$
17.	Installment or lease payments:		
	17a. Car payments for Vehicle 1	17a.	\$
	17b. Car payments for Vehicle 2	17b.	\$
	17c. Other. Specify:	17c.	\$
	17d. Other. Specify:	17d.	\$
18.	Your payments of alimony, maintenance, and support that you did not report as deducted from your pay on line 5, <i>Schedule I, Your Income</i> (Official Form 106I).	18.	\$
19.	Other payments you make to support others who do not live with you.		<b>Y</b>
	Specify:	19.	\$
20.	Other real property expenses not included in lines 4 or 5 of this form or on Schedule I: Your Incom	e.	
	20a. Mortgages on other property	20a.	\$
	20b. Real estate taxes	20b.	\$
	20c. Property, homeowner's, or renter's insurance	20c.	\$
	20d. Maintenance, repair, and upkeep expenses	20d.	\$
	20e. Homeowner's association or condominium dues	20e.	\$

Debtor 1	First Name Middle Name Last Name Case number (if kno	wn)	
21. Other. Sp	ecify:	21.	+\$
The result	is the monthly expenses. Add lines 5 through 21. is the monthly expenses of Debtor 2. Copy the result to line 22b of Schedule J to calculate the uses for Debtor 1 and Debtor 2.	22.	\$
23. Line not us	ed on this form.		
24. Do you ex	pect an increase or decrease in your expenses within the year after you file this form?		
	le, do you expect to finish paying for your car loan within the year or do you expect your payment to increase or decrease because of a modification to the terms of your mortgage?		
☐ No.			
☐ Yes.	Explain here:		

Fill in this information to identify your case:	
Debtor 1	
First Name Middle Name Last Name  Debtor 2	
(Spouse, if filing) First Name Middle Name Last Name	
United States Bankruptcy Court for the: District of (State)	☐ Check if this is an
Case number (If known)	amended filing
Official Form 106Sum	
Summary of Your Assets and Liabilities and Certain Statistical I	nformation 12/15
Be as complete and accurate as possible. If two married people are filing together, both are equally responsible.	
information. Fill out all of your schedules first; then complete the information on this form. If you are filing am your original forms, you must fill out a new <i>Summary</i> and check the box at the top of this page.	
Part 1: Summarize Your Assets	
	Your assets
Schedule A/B: Property (Official Form 106A/B)	Value of what you own
1a. Copy line 55, Total real estate, from Schedule A/B	<b>\$</b>
4) Canadian CO. Tatal managarity from Calculula A/D	•
1b. Copy line 62, Total personal property, from Schedule A/B	\$ <u></u>
1c. Copy line 63, Total of all property on Schedule A/B	······· \$
Part 2: Summarize Your Liabilities	
	Your liabilities
	Amount you owe
<ol> <li>Schedule D: Creditors Who Have Claims Secured by Property (Official Form 106D)</li> <li>Copy the total you listed in Column A, Amount of claim, at the bottom of the last page of Part 1 of Schedule D</li> </ol>	\$
	············
3. Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 106E/F)  3a. Copy the total claims from Part 1 (priority unsecured claims) from line 6e of Schedule E/F	\$
3b. Copy the total claims from Part 2 (nonpriority unsecured claims) from line 6j of Schedule E/F	
,	+ \$
Your total liabi	lities \$
Part 3: Summarize Your Income and Expenses	
4. Schedule I: Your Income (Official Form 106I)	
Copy your combined monthly income from line 12 of Schedule I	
5. Schedule J: Your Expenses (Official Form 106J)	
Copy your monthly expenses from line 22c of Schedule J	\$

$\square$	htor	1

First Name Middle Name Last Name

Case number (if known)\_\_\_\_\_

Da	rt	1	
Рα	ľ	-	

Answer These Questions for Administrative and Statistical Records

6.	Are you	filing for	bankruptcy	under Chap	ters 7,	11,	or 133
----	---------	------------	------------	------------	---------	-----	--------

☐ No. You have nothing to report on this part of the form. Check this box and submit this form to the court with your other schedules.

☐ Yes

#### 7. What kind of debt do you have?

☐ Your debts are primarily consumer debts. Consumer debts are those "incurred by an individual primarily for a personal, family, or household purpose." 11 U.S.C. § 101(8). Fill out lines 8-9g for statistical purposes. 28 U.S.C. § 159.

☐ Your debts are not primarily consumer debts. You have nothing to report on this part of the form. Check this box and submit this form to the court with your other schedules.

8. **From the** *Statement of Your Current Monthly Income*: Copy your total current monthly income from Official Form 122A-1 Line 11; **OR**, Form 122B Line 11; **OR**, Form 122C-1 Line 14.

\$		

9. Copy the following special categories of claims from Part 4, line 6 of Schedule E/F:

	Total claim
From Part 4 on <i>Schedule E/F</i> , copy the following:	
9a. Domestic support obligations (Copy line 6a.)	\$
9b. Taxes and certain other debts you owe the government. (Copy line 6b.)	\$
9c. Claims for death or personal injury while you were intoxicated. (Copy line 6c.)	\$
9d. Student loans. (Copy line 6f.)	\$
9e. Obligations arising out of a separation agreement or divorce that you did not report as priority claims. (Copy line 6g.)	\$
9f. Debts to pension or profit-sharing plans, and other similar debts. (Copy line 6h.)	+ \$
9g. <b>Total.</b> Add lines 9a through 9f.	\$

Fill in this information to identify your case:						
Debtor 1	First Name	Middle Name	Last Name			
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name			
United States E	Bankruptcy Court for the	e:	District of(State)			
Case number (If known)						

☐ Check if this is an amended filing

### Official Form 106Dec

### Declaration About an Individual Debtor's Schedules

12/15

If two married people are filing together, both are equally responsible for supplying correct information.

You must file this form whenever you file bankruptcy schedules or amended schedules. Making a false statement, concealing property, or obtaining money or property by fraud in connection with a bankruptcy case can result in fines up to \$250,000, or imprisonment for up to 20 years, or both. 18 U.S.C. §§ 152, 1341, 1519, and 3571.

Sign Below	
Did you pay or agree to pay someone who	is NOT an attorney to help you fill out bankruptcy forms?
□ No	
Yes. Name of person	. Attach Bankruptcy Petition Preparer's Notice, Declaration, and
	Signature (Official Form 119).
	ave read the summary and schedules filed with this declaration and
Under penalty of perjury, I declare that I hat they are true and correct.	ave read the summary and schedules filed with this declaration and
that they are true and correct.	
	ave read the summary and schedules filed with this declaration and
that they are true and correct.	
that they are true and correct.	<b>x</b>

#### **COMMITTEE NOTE**

The schedules to be used in cases of individual debtors are revised as part of the Forms Modernization Project, making them easier to read and, as a result, likely to generate more complete and accurate responses. The goals of the Forms Modernization Project include improving the interface between technology and the forms so as to increase efficiency and reduce the need to produce the same information in multiple formats. Therefore, many of the open-ended questions and multiple-part instructions have been replaced with more specific questions. The individual debtor schedules are also renumbered, starting with the number 106 and followed by the letter or name of the schedule to distinguish them from the versions to be used in non-individual cases.

Official Form 106Sum, Summary of Your Assets and Liabilities and Certain Statistical Information, replaces Official Form 6, Summary of Schedules and Statistical Summary of Certain Liability and Related Data (28 U.S.C. § 159), in cases of individual debtors.

The form is reformatted and updated with cross-references indicating the line numbers of specific schedules from which the summary information is to be gathered. In addition, because most filings are now done electronically, the form no longer requires the debtor to indicate which schedules are attached or to state the number of sheets of paper used for the schedules.

Official Form 106A/B, Schedule A/B: Property, consolidates information about an individual debtor's real and personal property into a single form. It replaces Official Form 6A, Real Property, and Official Form 6B, Personal Property, in cases of individual debtors. In addition to specific questions about the assets, the form also includes open text fields for providing additional information regarding particular assets when appropriate.

The layout and categories of property on Official Form 106A/B have changed. Instead of dividing property interests into two categories (real or personal property), the new form uses seven categories likely to be more familiar to non-lawyers: real estate, vehicles, personal household items, financial assets, business-related property, farm- and

commercial fishing-related property, and a catch-all category for property that was not listed elsewhere in the form. The new form categories and the examples provided in many of the categories are designed to prompt debtors to be thorough and list all of their interests in property. The debtor may describe generally items of minimal value (such as children's clothes) by adding the value of the items and reporting the total.

Although a particular item of property may fit into more than one category, the instructions for the form explain that it should be listed only once.

In addition, because property that falls within a particular category may not be specifically elicited by the particular line items on the form, the debtor is asked in Parts 3–6 (lines 14, 35, 44, and 51) to specifically identify and value any other property in the category.

In Part 1, Describe Each Residence, Building, Land, or Other Real Estate You Own or Have an Interest In, the debtor is asked to state the "current value of the portion you own," and to also state who has an interest in the property. In addition, the debtor is asked for the nature of the ownership interest, if known by the debtor. Furthermore, instead of asking for an open-ended description of the property, the form guides the debtor in answering the description question by providing eight options from which to choose: single-family home, duplex or multi-unit building, condominium or cooperative, manufactured or mobile home, land, investment property, timeshare, and other.

Part 2, *Describe Your Vehicles*, also guides the debtor in answering the question, asking for the make, model, year, and mileage of the car or other vehicle. Because mileage is just a general indication of vehicle value, the debtor is not required to list the exact mileage, but instead is prompted to provide the approximate mileage.

Part 3, Describe Your Personal and Household Items, simplifies wording, updates categories, and uses more common terms. For example, "Wearing apparel" is changed to "Clothes" and examples include furs, which were previously grouped with jewelry. Firearms, on the

other hand, which were previously grouped with sports and other hobbies, are now set out as a separate category. Additionally, because a new Part 6 has been added to separately describe-farm related property, Part 3 includes a category for "Non-farm animals."

Part 4, *Describe Your Financial Assets*, prompts a listing of the debtor's financial assets through several questions providing separate space, after each listed type of account or deposit, for the institution or issuer name and the value of the debtor's interest in the asset. Two new categories of financial assets are added: "Bonds, mutual funds, or publicly traded stocks" and "Claims against third parties, whether or not you have filed a lawsuit or made a demand for payment." In addition, qualified ABLE accounts, as defined in 26 U.S.C. § 529A(b), are added to the list of accounts in question 24. This change is made in response to the Tax Increase Prevention Act of 2014, Pub. Law No. 113-295, which excludes ABLE account contributions meeting the specified requirements from property of the estate.

Part 5, Describe Any Business-Related Property You Own or Have an Interest In, provides prompts for listing business-related property, such as accounts receivable, inventory, and machinery, and includes a direction to list business-related real estate in Part 1, to avoid listing real estate twice.

Part 6, Describe Any Farm- and Commercial Fishing-Related Property You Own or Have an Interest In, provides prompts for listing farm- or commercial fishing-related property, such as farm animals, crops, and feed. It also includes a direction to list any farm- or commercial fishing-related real estate in Part 1.

Part 7, Describe All Property You Own or Have an Interest in That You Did Not List Above, is a catch-all provision that allows the debtor to report property that is difficult to categorize.

Part 8, List the Totals of Each Part of this Form, tabulates the total value of the debtor's interest in the listed property. The tabulation includes two subtotals, one for real estate, which corresponds to the real property total that was reported on former Official Form 6A. The second

subtotal is of Parts 2-7, which corresponds to the personal property total that was reported on former Official Form 6B

Official Form 106C, Schedule C: The Property You Claim as Exempt, replaces Official Form 6C, Property Claimed as Exempt, in cases of individual debtors.

Part 1, *Identify the Property You Claim as Exempt*, includes a table to list the property the debtor seeks to exempt, the value of the property owned by the debtor, the amount of the claimed exemption, and the law that allows the exemption. The first column asks for a brief description of the exempt property, and it also asks for the line number where the property is listed on Schedule A/B. The second column asks for the value of the portion of the asset owned by the debtor, rather than the entire asset. The third column asks for the amount, rather than the value, of the exemption claim.

The form has also been changed in light of the Supreme Court's ruling in *Schwab v. Reilly*, 560 U.S. 770 (2010). Entries in the "amount of the exemption you claim" column may now be listed as either a dollar limited amount or as 100% of fair market value, up to any applicable statutory limit. For example, a debtor might claim 100% of fair market value for a home covered by an exemption capped at \$15,000, and that limit would be applicable. This choice would impose no dollar limit where the exemption is unlimited in dollar amount, such as some exemptions for health aids, certain governmental benefits, and tax-exempt retirement funds.

Official Form 106D, Schedule D: Creditors Who Have Claims Secured by Property, replaces Official Form 6D, Creditors Holding Secured Claims, in cases of individual debtors.

Part 1, List All Secured Claims, now directs the debtor to list only the last four digits of the account number. Part 1 also adds four checkboxes with which to describe the nature of the lien: an agreement the debtor made (such as mortgage or secured car loan); statutory lien (such as tax lien, mechanic's lien); judgment lien from a lawsuit; and other.

The form adds Part 2, List Others to Be Notified for a Debt That You Already Listed. The debtor is instructed to use Part 2 if there is a need to notify someone about the bankruptcy filing other than the creditor for a debt listed in Part 1. For example, if a collection agency is trying to collect for a creditor listed in Part 1, the collection agency would be listed in Part 2.

Official Form 106E/F, Schedule E/F: Creditors Who Have Unsecured Claims, consolidates information about priority and nonpriority unsecured claims into a single form. It replaces Official Form 6E, Creditors Holding Unsecured Priority Claims, and Official Form 6F, Creditors Holding Unsecured Nonpriority Claims, in cases of individual debtors

Although both priority and nonpriority unsecured claims are reported in Official Form 106E/F, the two types of claims are separately grouped so that the total for each type can be reported for case administration and statistical purposes. The form eliminates the question "consideration for claim" and instructs debtors to list claims in the alphabetical order of creditors as much as possible.

Part 1, List All of Your PRIORITY Unsecured Claims, includes four checkboxes for identifying the type of priority that applies to the claim: domestic support obligations; taxes and certain other debts owed to the government; claims for death or personal injury while intoxicated; and "other." The first three categories are required to be separately reported for statistical purposes. If the debtor selects "other," the debtor must specify the basis of the priority, e.g., wages or employee benefit plan contribution.

Part 2, List All of Your NONPRIORITY Unsecured Claims, contains four checkboxes, including three for types of claims that must be separately reported for statistical purposes: student loans; obligations arising out of a separation agreement or divorce not listed as priority claims; and debts to pension or profit-sharing plans and other similar debts. The remaining "other" checkbox treats claims not subject to separate reporting. If the debtor selects "other," the debtor must specify the basis of the claim.

Part 3, List Others to Be Notified About a Debt That You Already Listed, is new. The debtor is instructed to use Part 3 only if there is a need to give notice of the bankruptcy to someone other than a creditor listed in Parts 1 and 2. For example, if a collection agency is trying to collect for a creditor listed in Part 1, the collection agency would be listed in Part 3.

Finally, Part 4, Add the Amounts for Each Type of Unsecured Claim, requires the debtor to provide the total amounts of particular types of unsecured claims for statistical reporting purposes and the overall totals of the priority and nonpriority unsecured claims reported in this form.

Official Form 106G, Schedule G: Executory Contracts and Unexpired Leases, replaces Official Form 6G, Executory Contracts and Unexpired Leases, in cases of individual debtors.

The form is simplified. Instead of requiring the debtor to make multiple assertions about each potential executory contract or unexpired lease, the form simply requires the debtor to identify the name and address of the other party to the contract or lease, and to state what the contract or lease deals with. Definitions and examples of executory contracts and unexpired leases are included in the separate instructions for the form.

An additional page is provided in case the debtor has so many executory contracts and unexpired leases that the available page is not adequate. If the debtor needs to use the additional page, the debtor is required to fill in the entry number.

**Official Form 106H,** *Schedule H: Your Codebtors*, replaces Official Form 6H, *Codebtors*, in cases of individual debtors.

The form breaks out the questions about whether there are any codebters, and whether the debtor has lived with a spouse, former spouse, or legal equivalent in a community property state in the prior eight years. It also removes Alaska from the listed community property states. Finally, it asks the debtor to indicate where the debt is listed on Schedule D, Schedule E/F, or Schedule G, thereby

eliminating the need to list the name and address of the creditor.

Official Form 106I, Schedule I: Your Income, replaces Official Form 6I, Your Income, in cases of individual debtors.

The form is one of an initial set of forms that were published as part of the Forms Modernization Project in 2012. It is renumbered and internal cross references are updated to conform to the new numbering system now being introduced by the Forms Modernization Project.

Official Form 106J, Schedule J: Your Expenses, replaces Official Form 6J, Your Expenses, in cases of individual debtors.

The form is one of an initial set of forms that were published as part of the Forms Modernization Project in 2012. It is renumbered and internal cross references are updated to conform to the new numbering system now being introduced by the Forms Modernization Project.

The form has been revised to include references to new *Schedule J-2: Expenses for Separate Household of Debtor 2* (Official Form 106J-2) at line 1 and new line 22b. The revisions clarify how to calculate monthly net income in joint cases where Debtor 1 and Debtor 2 maintain separate households. Line 22b is added so Schedule J and Schedule J-2 are easily coordinated.

**Official Form 106J-2** is new. It is used to report the monthly expenses of Debtor 2 in a joint debtor case only if Debtor 1 and Debtor 2 maintain separate households.

Official Form 106Dec, Declaration About an Individual Debtor's Schedules, replaces Official Form 6, Declaration Concerning Debtor's Schedules, in cases of individual debtors.

The form, which is to be signed by the debtor and filed with the debtor's schedules, deletes the Declaration and Signature of Bankruptcy Petition Preparer (BPP). Instead, the debtor is directed to complete and file Official Form 119, Bankruptcy Petition Preparer's Notice,

Declaration, and Signature, if a BPP helped fill out the bankruptcy forms.

Because the form applies only to individual debtors, it no longer contains the Declaration Under Penalty of Perjury on Behalf of a Corporation or Partnership. It also deletes from the declaration the phrase "to the best of my knowledge, information, and belief" in order to conform to the language of 28 U.S.C. § 1746. *See* Rule 1008.

Fill in this information to identify your case:					
Debtor 1	First Name	Middle Name	Last Name		
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name		
United States E	Bankruptcy Court for the:		District of		
Case number(If known)					

# ☐ Check if this is an amended filing

### Official Form 107

### Statement of Financial Affairs for Individuals Filing for Bankruptcy

12/15

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write your name and case number (if known). Answer every question.

2.	During the last 3 years,	have you lived as	nywhere other than	where you live now?

ПМ

☐ Married☐ Not married

Yes. List all of the places you lived in the last 3 years. Do not include where you live now.

res. List all of the places you lived in the last 5 year	ars. Do not include	where you live now.	
Debtor 1:	Dates Debtor 1 lived there	Debtor 2:	Dates Debtor 2 lived there
		☐ Same as Debtor 1	☐ Same as Debtor 1
Number Street	From	Number Street	From To
City State ZIP Code		City State ZIP Code	
		☐ Same as Debtor 1	☐ Same as Debtor 1
Number Street	From To	Number Street	From
City State ZIP Code		City State ZIP Code	

3.	Within the last 8 years, did you ever live with a spouse or legal equivalent in a community property state or territory? (Community property states
	and territories include Arizona, California, Idaho, Louisiana, Nevada, New Mexico, Puerto Rico, Texas, Washington, and Wisconsin.)

☐ No

☐ Yes. Make sure you fill out *Schedule H: Your Codebtors* (Official Form 106H).

Part 2:

Explain the Sources of Your Income

ıal	ht	٦r	1

First Name	Middle Name	Last Name	

Case	number	(if known)		

No	ome that you receive toget	nesses, including part-tir her, list it only once unde		dar years?	
Yes. Fill in the details.					
	Debtor 1	_	Debtor 2		
	Sources of income Check all that apply.	Gross income (before deductions and exclusions)	Sources of income Check all that apply.	Gross income (before deductions and exclusions)	
From January 1 of current year until the date you filed for bankruptcy:	<ul><li>☐ Wages, commissions, bonuses, tips</li><li>☐ Operating a business</li></ul>	\$	<ul><li>☐ Wages, commissions, bonuses, tips</li><li>☐ Operating a business</li></ul>	\$	
For last calendar year:  (January 1 to December 31,)	<ul><li>☐ Wages, commissions, bonuses, tips</li><li>☐ Operating a business</li></ul>	\$	<ul><li>☐ Wages, commissions, bonuses, tips</li><li>☐ Operating a business</li></ul>	\$	
For the calendar year before that:	Wages, commissions, bonuses, tips	\$	Wages, commissions, bonuses, tips	\$	
(January 1 to December 31,)	Operating a business	Ψ	Operating a business	Ψ	
winnings. If you are filing a joint case and you List each source and the gross income from e	have income that you rec	eived together, list it only	d from lawsuits; royalties; ar y once under Debtor 1.		
, , ,	have income that you rec	eived together, list it only	d from lawsuits; royalties; ar y once under Debtor 1.		
List each source and the gross income from e	have income that you rec	eived together, list it only	d from lawsuits; royalties; ar y once under Debtor 1.		
List each source and the gross income from e	have income that you recearch source separately. Do	eived together, list it only	d from lawsuits; royalties; ar y once under Debtor 1. t you listed in line 4.		
List each source and the gross income from e  No Yes. Fill in the details.  From January 1 of current year until	have income that you recearch source separately. Do  Debtor 1  Sources of income	o not include income that  Gross income from each source (before deductions and	d from lawsuits; royalties; ar y once under Debtor 1. It you listed in line 4.  Debtor 2  Sources of income Describe below.	Gross income from each source (before deductions and	
List each source and the gross income from e  No Yes. Fill in the details.	have income that you recearch source separately. Do  Debtor 1  Sources of income	Gross income from each source (before deductions and exclusions)	d from lawsuits; royalties; ar y once under Debtor 1. It you listed in line 4.  Debtor 2  Sources of income Describe below.	Gross income from each source (before deductions and	
List each source and the gross income from e  No Yes. Fill in the details.  From January 1 of current year until	have income that you recearch source separately. Do  Debtor 1  Sources of income	Gross income from each source (before deductions and exclusions)	d from lawsuits; royalties; ar y once under Debtor 1. It you listed in line 4.  Debtor 2  Sources of income Describe below.	Gross income from each source (before deductions and	
List each source and the gross income from e  No Yes. Fill in the details.  From January 1 of current year until the date you filed for bankruptcy:	have income that you recearch source separately. Do  Debtor 1  Sources of income	Gross income from each source (before deductions and exclusions)	d from lawsuits; royalties; ar y once under Debtor 1. It you listed in line 4.  Debtor 2  Sources of income Describe below.	Gross income from each source (before deductions and	
List each source and the gross income from each No No Yes. Fill in the details.  From January 1 of current year until the date you filed for bankruptcy:  For last calendar year: (January 1 to December 31,)	have income that you recearch source separately. Do  Debtor 1  Sources of income	Gross income from each source (before deductions and exclusions)  \$	d from lawsuits; royalties; ar y once under Debtor 1. It you listed in line 4.  Debtor 2  Sources of income Describe below.	Gross income from each source (before deductions and exclusions)  - \$	
List each source and the gross income from each No No Yes. Fill in the details.  From January 1 of current year until the date you filed for bankruptcy:  For last calendar year:	have income that you recearch source separately. Do  Debtor 1  Sources of income	Gross income from each source (before deductions and exclusions)  \$	d from lawsuits; royalties; ar y once under Debtor 1.  t you listed in line 4.  Debtor 2  Sources of income Describe below.	Gross income from each source (before deductions and exclusions)	
List each source and the gross income from each No No Yes. Fill in the details.  From January 1 of current year until the date you filed for bankruptcy:  For last calendar year: (January 1 to December 31,)	have income that you recearch source separately. Do  Debtor 1  Sources of income	Gross income from each source (before deductions and exclusions)  \$	d from lawsuits; royalties; ar y once under Debtor 1.  t you listed in line 4.  Debtor 2  Sources of income Describe below.	Gross income from each source (before deductions and exclusions)  - \$	
List each source and the gross income from each No No Yes. Fill in the details.  From January 1 of current year until the date you filed for bankruptcy:  For last calendar year: (January 1 to December 31, YYYY)  For the calendar year before that:	have income that you recearch source separately. Do  Debtor 1  Sources of income	Gross income from each source (before deductions and exclusions)  \$	d from lawsuits; royalties; ar y once under Debtor 1.  It you listed in line 4.  Debtor 2  Sources of income Describe below.	Gross income from each source (before deductions and exclusions)  - \$	
List each source and the gross income from each No No Yes. Fill in the details.  From January 1 of current year until the date you filed for bankruptcy:  For last calendar year: (January 1 to December 31,)	have income that you recearch source separately. Do  Debtor 1  Sources of income	Gross income from each source (before deductions and exclusions)  \$	d from lawsuits; royalties; ar y once under Debtor 1.  t you listed in line 4.  Debtor 2  Sources of income Describe below.	Gross income from each source (before deductions and exclusions)  - \$	

П	10	h	ta	r	1

First Name Middle Name Last Name Case number (if known)

Are either	Debtor 1's or Debtor 2's debts primar	ily consumer debt	s?		
□ No. <b>N</b> e "ir	either Debtor 1 nor Debtor 2 has prim nourred by an individual primarily for a p	arily consumer de ersonal, family, or h	bts. Consumer debts are nousehold purpose."	e defined in 11 U.S.C. § 101(	(8) as
Di	uring the 90 days before you filed for ba	nkruptcy, did you pa	ay any creditor a total of	\$6,225* or more?	
	No. Go to line 7.				
	Yes. List below each creditor to whom total amount you paid that credit child support and alimony. Also,	or. Do not include p	ayments for domestic su	pport obligations, such as	
* (	Subject to adjustment on 4/01/16 and ev	ery 3 years after th	at for cases filed on or a	fter the date of adjustment.	
Yes. Do	ebtor 1 or Debtor 2 or both have prim	arily consumer de	bts.		
	uring the 90 days before you filed for ba			\$600 or more?	
_	No. Go to line 7.		-		
	Yes. List below each creditor to whom creditor. Do not include payment alimony. Also, do not include pay	s for domestic supp	ort obligations, such as	child support and	
		Dates of payment	Total amount paid	Amount you still owe	Was this payment for.
			\$	\$	☐ Mortgage
	Creditor's Name				☐ Car
	Number Street				☐ Credit card
	Number Street				Loan repayment
					☐ Suppliers or vendo
	City State ZIP C	ode			Other
			\$	\$	☐ Mortgage
	Creditor's Name				☐ Car
	Creditor's Name				
					Credit card
	Number Street				☐ Credit card ☐ Loan repayment
					☐ Loan repayment
	Number Street				
		ode			☐ Loan repayment☐ Suppliers or vendo
	Number Street	ode			☐ Loan repayment ☐ Suppliers or vendo ☐ Other
	Number Street	ode	\$	\$	□ Loan repayment □ Suppliers or vendo □ Other
	Number Street  City State ZIP C	ode	\$	\$	Loan repayment Suppliers or vendo Other Mortgage Car
	Number Street  City State ZIP C	ode	\$	\$	□ Loan repayment □ Suppliers or vendo □ Other □ Mortgage □ Car □ Credit card
	Number Street  City State ZIP C  Creditor's Name	ode	\$	_ \$	Loan repayment  Suppliers or vendo  Other  Mortgage  Car  Credit card  Loan repayment
	Number Street  City State ZIP C  Creditor's Name	ode	\$	\$	□ Loan repayment □ Suppliers or vendo □ Other □ Mortgage □ Car □ Credit card

Debtor 1				Case number (if known)
200.0.	First Name	Middle Name	Last Name	

In co ag	thin 1 year before you filed for bankruptcy, did you siders include your relatives; any general partners; relatives any general partners; relatives any officer, director, person ent, including one for a business you operate as a soluth as child support and alimony.	atives of any ge in control, or o	eneral partners; par owner of 20% or m	rtnerships of which ore of their voting s	you are a general partner; securities; and any managing
	No Yes. List all payments to an insider.				
		Dates of payment	Total amount paid	Amount you still owe	Reason for this payment
			\$	\$	
	Insider's Name		<u> </u>	<b>Y</b>	
	Number Street				
	City State ZIP Code				
			\$	\$	
	Insider's Name				
	Number Street				
	City State ZIP Code				
ar In	thin 1 year before you filed for bankruptcy, did you insider? Clude payments on debts guaranteed or cosigned by a  No  Yes. List all payments that benefited an insider.		yments or transfe	r any property on  Amount you still	account of a debt that benefited  Reason for this payment
		payment	paid	owe	Include creditor's name
	Insider's Name		\$	\$	
	Number Street				
	City State ZIP Code				
	Insider's Name		\$	\$	
	Number Street				
	City State ZIP Code				

$\Box$	0	h	tο	r	1

Case number (if known)
------------------------

Pai	rt	4	

Identify Legal Actions, Repossessions, and Foreclosures

List a	n 1 year before you filed for bankrupto Il such matters, including personal injury contract disputes.			nemity actions, suppo	ort or custody modification
□ N	0				
□ Y	es. Fill in the details.				
		Nature of the case	Court or agency		Status of the case
(	Case title		Court Name		Pending On appeal
-	Case number		Number Street		Concluded
			City	State ZIP Code	
(	Case title		Court Name		Pending On appeal
-	Case number		Number Street		Concluded
			City	State ZIP Code	
Checi	n 1 year before you filed for bankrupton that apply and fill in the details below the control on		epossessed, foreclosed	, garnished, attache	d, seized, or levied?
Chec N	k all that apply and fill in the details belov o. Go to line 11.			, garnished, attached	Value of the property
hec N	k all that apply and fill in the details belov o. Go to line 11.	<i>i.</i>			
hec N	k all that apply and fill in the details below o. Go to line 11. es. Fill in the information below.	<i>i.</i>	у		Value of the property
hec N	k all that apply and fill in the details below  o. Go to line 11. es. Fill in the information below.  Creditor's Name	Describe the propert	y  ned  epossessed. oreclosed.		Value of the property
hec N	k all that apply and fill in the details below  o. Go to line 11. es. Fill in the information below.  Creditor's Name	Explain what happen  Property was rough Property was go	y  ned  epossessed. oreclosed.	Date	Value of the property
hec N	k all that apply and fill in the details below  o. Go to line 11. es. Fill in the information below.  Creditor's Name  Number Street	Explain what happen  Property was rough Property was go	epossessed. oreclosed. garnished. attached, seized, or levied	Date	Value of the property
hec N	k all that apply and fill in the details below  o. Go to line 11. es. Fill in the information below.  Creditor's Name  Number Street	Explain what happen Property was for Property was go	epossessed. oreclosed. garnished. attached, seized, or levied	Date	Value of the property
hec N	k all that apply and fill in the details below  o. Go to line 11. es. Fill in the information below.  Creditor's Name  Number Street	Explain what happen Property was for Property was go	epossessed. oreclosed. garnished. attached, seized, or levied	Date	Value of the property  \$  Value of the property
Chec N	k all that apply and fill in the details below  o. Go to line 11. es. Fill in the information below.  Creditor's Name  Number Street  City State ZIP Co	Explain what happen Property was for Property was go	epossessed. oreclosed. garnished. attached, seized, or levied	Date	Value of the property  \$  Value of the property
Chec N	k all that apply and fill in the details below  o. Go to line 11. es. Fill in the information below.  Creditor's Name  Number Street  City State ZIP Co	Explain what happen Property was roundered Property was good Property was a Describe the propert  Explain what happen  Explain what happen	epossessed. oreclosed. garnished. attached, seized, or levied y	Date	Value of the property  \$  Value of the property
Chec N	k all that apply and fill in the details below  o. Go to line 11. es. Fill in the information below.  Creditor's Name  Number Street  City State ZIP Co	Explain what happen Property was for Property was for Property was a Property was a Describe the propert	ed epossessed. oreclosed. garnished. attached, seized, or levied by epossessed. oreclosed.	Date	Value of the property  \$  Value of the property

ounts or refuse to make a payment be No			
No Yes. Fill in the details.			
	Describe the action the creditor took	Date action	Amount
		was taken	ranount
Creditor's Name			
lumber Street	_	\$	S
City State ZIP Code	Last 4 digits of account number: XXXX		
in 1 year hefore you filed for hankrun	tcy, was any of your property in the possession c	of an assigned for the henofit	of
litors, a court-appointed receiver, a cu			
No			
⁄es			
List Certain Gifts and Contribu	utions		
List Certain Girts and Contribu	utions		
in 2 years hafara yeu filed for head-			
in 2 years before you filed for bankrup		4l 0000 0	
	otcy, did you give any gifts with a total value of m	ore than \$600 per person?	
No	otcy, did you give any gifts with a total value of m	ore than \$600 per person?	
No	otcy, did you give any gifts with a total value of m	ore than \$600 per person?	
No Yes. Fill in the details for each gift.  Gifts with a total value of more than \$600	otcy, did you give any gifts with a total value of m  Describe the gifts	Dates you gave	Value
No ⁄es. Fill in the details for each gift.			Value
No Yes. Fill in the details for each gift.  Gifts with a total value of more than \$600		Dates you gave	Value
No Yes. Fill in the details for each gift.  Gifts with a total value of more than \$600 per person		Dates you gave	Value
No Yes. Fill in the details for each gift.  Gifts with a total value of more than \$600 per person		Dates you gave	Value \$
No Yes. Fill in the details for each gift.  Gifts with a total value of more than \$600 per person		Dates you gave	Value \$\$
Ves. Fill in the details for each gift.  Gifts with a total value of more than \$600 per person  Person to Whom You Gave the Gift		Dates you gave	<b>Value</b> \$\$
Ves. Fill in the details for each gift.  Gifts with a total value of more than \$600 per person  Person to Whom You Gave the Gift		Dates you gave	<b>Value</b> \$\$
Ves. Fill in the details for each gift.  Gifts with a total value of more than \$600 per person  Person to Whom You Gave the Gift		Dates you gave	<b>Value</b> \$\$
Gifts with a total value of more than \$600 per person  Person to Whom You Gave the Gift  Street  Street  State ZIP Code		Dates you gave	\text{Value}  \$  \$
Gifts with a total value of more than \$600 per person  Person to Whom You Gave the Gift  Street  Sity State ZIP Code  Person's relationship to you	Describe the gifts	Dates you gave the gifts	\$ \$
Gifts with a total value of more than \$600 per person  Person to Whom You Gave the Gift  State ZIP Code  Person's relationship to you  Gifts with a total value of more than \$600		Dates you gave the gifts  Dates you gave	Value  \$  \$ Value
Gifts with a total value of more than \$600 per person  Person to Whom You Gave the Gift  State ZIP Code  Person's relationship to you  Gifts with a total value of more than \$600	Describe the gifts	Dates you gave the gifts	\$ \$
Gifts with a total value of more than \$600 per person  Person to Whom You Gave the Gift  Street  State ZIP Code  Person's relationship to you  Gifts with a total value of more than \$600 per person	Describe the gifts	Dates you gave the gifts  Dates you gave	\$ \$
Gifts with a total value of more than \$600 per person  Person to Whom You Gave the Gift  Street  State ZIP Code  Person's relationship to you  Gifts with a total value of more than \$600 per person	Describe the gifts	Dates you gave the gifts  Dates you gave	\$\$ \$
Gifts with a total value of more than \$600 per person  Person to Whom You Gave the Gift  Street  State ZIP Code  Person's relationship to you  Gifts with a total value of more than \$600 per person	Describe the gifts	Dates you gave the gifts  Dates you gave	\$\$ \$
Gifts with a total value of more than \$600 per person  Person to Whom You Gave the Gift  Street  State ZIP Code  Person's relationship to you  Gifts with a total value of more than \$600 per person	Describe the gifts	Dates you gave the gifts  Dates you gave	\$
Gifts with a total value of more than \$600 per person  Person to Whom You Gave the Gift  State ZIP Code  Person's relationship to you  Gifts with a total value of more than \$600 per person	Describe the gifts	Dates you gave the gifts  Dates you gave	\$
Yes. Fill in the details for each gift.  Gifts with a total value of more than \$600 per person  Person to Whom You Gave the Gift	Describe the gifts	Dates you gave the gifts  Dates you gave	\$

Case number (if known)

Debtor 1

Bescribt in the details for each gift or contribution.    Contributions to charities   Describe what you contributed   Date you contributed   Contributed		ptcy, did you give any gifts or contributions with a total value	of more than \$600	to any charity?
List Certain Losses  In 1 year before you filed for bankruptcy or since you filed for bankruptcy, did you lose anything because of theft, fire, other disaste mbling?  Sees Fill in the details.  Describe any insurance coverage for the loss include the amount that insurance has paid. List pending insurance losts occurred  List Certain Payments or Transfers  In 1 year before you filed for bankruptcy, did you or anyone else acting on your behalf pay or transfer any property to anyone you ulted about seeking bankruptcy or preparing a bankruptcy petition?  List Certain Payments or Transfers  In 1 year before you filed for bankruptcy, did you or anyone else acting on your behalf pay or transfer any property to anyone you ulted about seeking bankruptcy petition preparers, or credit counseling agencies for services required in your bankruptcy.  Description and value of any property transferred  Date payment or transfer was made  Sumber Street  Description and value of any property transferred  Date payment or transfer was made  Amount of payments is street.	No Yes. Fill in the details for each gift or cor	ntribution.		
List Certain Losses  In 1 year before you filed for bankruptcy or since you filed for bankruptcy, did you lose anything because of theft, fire, other disaste mbling?  On as, Fill in the details.  Describe the property you lost and how the loss occurred  Describe any insurance coverage for the loss Include the amount that insurance has paid, List pending insurance claims on line 33 of Schedule A/B: Property.  Significant of the property of property of property to anyone else acting on your behalf pay or transfer any property to anyone you ulted about seeking bankruptcy or preparing a bankruptcy petition?  Let any attorneys, bankruptcy petition preparers, or credit counseling agencies for services required in your bankruptcy.  On as, Fill in the details.  Description and value of any property transferred  Date payment or transfer was made  Amount of payments or Transfers  Significant of payments or transfer was made  Amount of payments or transfer was made	Gifts or contributions to charities that total more than \$600	Describe what you contributed		Value
List Certain Losses  In 1 year before you filed for bankruptcy or since you filed for bankruptcy, did you lose anything because of theft, fire, other disaste mbling?  Describe the property you lost and how he loss occurred  Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance claims on line 33 of Schedule A/B: Property.  List Certain Payments or Transfers  In 1 year before you filed for bankruptcy, did you or anyone else acting on your behalf pay or transfer any property to anyone you ulted about seeking bankruptcy or preparing a bankruptcy petition?  Be any attorneys, bankruptcy petition preparers, or credit counseling agencies for services required in your bankruptcy.  Description and value of any property transferred  Date payment or transfer was made  Description and value of any property transferred  Sumber Street  Amount of payments or Street	Charity's Name	-		\$
List Certain Losses  In 1 year before you filed for bankruptcy or since you filed for bankruptcy, did you lose anything because of theft, fire, other disaste mbling?  Describe the property you lost and how he loss occurred  Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance claims on line 33 of Schedule A/B: Property.  List Certain Payments or Transfers  In 1 year before you filed for bankruptcy, did you or anyone else acting on your behalf pay or transfer any property to anyone you ulted about seeking bankruptcy or preparing a bankruptcy petition?  Be any attorneys, bankruptcy petition preparers, or credit counseling agencies for services required in your bankruptcy.  Description and value of any property transferred  Date payment or transfer was made  Description and value of any property transferred  Sumber Street  Amount of payments or Street		-		\$
List Certain Losses  1 year before you filed for bankruptcy or since you filed for bankruptcy, did you lose anything because of theft, fire, other disaste mbling?  2 bescribe the property you lost and how he loss occurred  Describe any insurance coverage for the loss include the amount that insurance has paid. List pending insurance lost include the amount that insurance has paid. List pending insurance claims on line 33 of Schedule A/B. Property.  List Certain Payments or Transfers  1 year before you filed for bankruptcy, did you or anyone else acting on your behalf pay or transfer any property to anyone you ulted about seeking bankruptcy or preparing a bankruptcy petition? de any attorneys, bankruptcy petition preparers, or credit counseling agencies for services required in your bankruptcy.  Description and value of any property transferred  Date payment or transfer was made  Amount of payments Street  \$	Number Street	-		
n 1 year before you filed for bankruptcy or since you filed for bankruptcy, did you lose anything because of theft, fire, other disastembling?  Describe the property you lost and how he loss occurred  Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance claims on line 33 of Schedule A/B: Property.  List Certain Payments or Transfers  1 year before you filed for bankruptcy, did you or anyone else acting on your behalf pay or transfer any property to anyone you ulted about seeking bankruptcy or preparing a bankruptcy petition?  Description and value of any property transferred  Date payment or transfer was made  Amount of payments or Street  Sumber Street  Sumber Street	City State ZIP Code	-		
n 1 year before you filed for bankruptcy or since you filed for bankruptcy, did you lose anything because of theft, fire, other disastembling?  Describe the property you lost and how he loss occurred  Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance claims on line 33 of Schedule A/B: Property.  List Certain Payments or Transfers  1 year before you filed for bankruptcy, did you or anyone else acting on your behalf pay or transfer any property to anyone you ulted about seeking bankruptcy or preparing a bankruptcy petition?  Description and value of any property transferred  Date payment or transfer was made  Amount of payments or Street  Sumber Street  Sumber Street	6: List Certain Losses			
List Certain Payments or Transfers  n 1 year before you filed for bankruptcy, did you or anyone else acting on your behalf pay or transfer any property to anyone you ulted about seeking bankruptcy or preparing a bankruptcy petition?  le any attorneys, bankruptcy petition preparers, or credit counseling agencies for services required in your bankruptcy.  o es. Fill in the details.  Description and value of any property transferred  Date payment or transfer was made  Amount of payment or transfer was made  \$	No Yes. Fill in the details.  Describe the property you lost and how	Describe any insurance coverage for the loss	Date of your loss	
n 1 year before you filed for bankruptcy, did you or anyone else acting on your behalf pay or transfer any property to anyone you ulted about seeking bankruptcy or preparing a bankruptcy petition?  de any attorneys, bankruptcy petition preparers, or credit counseling agencies for services required in your bankruptcy.  Description and value of any property transferred  Date payment or transfer was made  Person Who Was Paid  Number Street  Street  Street  Street  Street	No Yes. Fill in the details.	Include the amount that insurance has paid. List pending insurance	Date of your loss	Value of propert lost
n 1 year before you filed for bankruptcy, did you or anyone else acting on your behalf pay or transfer any property to anyone you ulted about seeking bankruptcy or preparing a bankruptcy petition?  de any attorneys, bankruptcy petition preparers, or credit counseling agencies for services required in your bankruptcy.  Description and value of any property transferred  Date payment or transfer was made  Person Who Was Paid  Number Street  Street  Street  Street  Street	No Yes. Fill in the details.  Describe the property you lost and how	Include the amount that insurance has paid. List pending insurance	Date of your loss	lost
Person Who Was Paid  Number Street  \$ \$ \$	No Yes. Fill in the details.  Describe the property you lost and how the loss occurred	Include the amount that insurance has paid. List pending insurance claims on line 33 of <i>Schedule A/B: Property</i> .	Date of your loss	lost
S	No Yes. Fill in the details.  Describe the property you lost and how the loss occurred  7: List Certain Payments or Transthin 1 year before you filed for bankrup insulted about seeking bankruptcy or p	Include the amount that insurance has paid. List pending insurance claims on line 33 of Schedule A/B: Property.  Insfers  Insters  Insters	sfer any property to	\$
	No Yes. Fill in the details.  Describe the property you lost and how the loss occurred  7: List Certain Payments or Translation 1 year before you filed for bankrup insulted about seeking bankruptcy or polude any attorneys, bankruptcy petition polynomials. No No Yes. Fill in the details.	Include the amount that insurance has paid. List pending insurance claims on line 33 of Schedule A/B: Property.  Insfers  Insters  Interpretation on your behalf pay or transpreparing a bankruptcy petition?  Interpretation on your behalf pay or transpreparing a bankruptcy petition?  Interpretation of the property of the property of the payor transpreparing a bankruptcy petition?  Interpretation of the property o	efer any property to our bankruptcy.	\$  anyone you  Amount of paym
Sity State ZIP Code	No Yes. Fill in the details.  Describe the property you lost and how the loss occurred  7: List Certain Payments or Transtantin 1 year before you filed for bankrup insulted about seeking bankruptcy or polude any attorneys, bankruptcy petition polyes. Fill in the details.  Person Who Was Paid	Include the amount that insurance has paid. List pending insurance claims on line 33 of Schedule A/B: Property.  Insfers  Insters  Interpretation on your behalf pay or transpreparing a bankruptcy petition?  Interpretation on your behalf pay or transpreparing a bankruptcy petition?  Interpretation of the property of the property of the payor transpreparing a bankruptcy petition?  Interpretation of the property o	efer any property to our bankruptcy.	\$  anyone you  Amount of paym
	No Yes. Fill in the details.  Describe the property you lost and how the loss occurred  7: List Certain Payments or Transtantin 1 year before you filed for bankrup insulted about seeking bankruptcy or polude any attorneys, bankruptcy petition polyes. Fill in the details.  Person Who Was Paid	Include the amount that insurance has paid. List pending insurance claims on line 33 of Schedule A/B: Property.  Insfers  Insters  Interpretation on your behalf pay or transpreparing a bankruptcy petition?  Interpretation on your behalf pay or transpreparing a bankruptcy petition?  Interpretation of the property of the property of the payor transpreparing a bankruptcy petition?  Interpretation of the property o	efer any property to our bankruptcy.	s anyone you  Amount of paym
	No Yes. Fill in the details.  Describe the property you lost and how the loss occurred  7: List Certain Payments or Transtantin 1 year before you filed for bankrup insulted about seeking bankruptcy or polude any attorneys, bankruptcy petition polyes. Fill in the details.  Person Who Was Paid	Include the amount that insurance has paid. List pending insurance claims on line 33 of Schedule A/B: Property.  Insfers  Insters  Interpretation on your behalf pay or transpreparing a bankruptcy petition?  Interpretation on your behalf pay or transpreparing a bankruptcy petition?  Interpretation of the property of the property of the payor transpreparing a bankruptcy petition?  Interpretation of the property o	efer any property to our bankruptcy.	\$  Amount of

വം	htor	1

			Case number (if known)
Tinet Minera	Middle Messes	Lest Name	

			Description and value of any proper	ty transferred	Date payment or transfer was made	Amount of payment
Person Who	o Was Paid					\$
Number S	Street					\$
City	State	ZIP Code				
Email or we	bsite address					
Person Who	o Made the Payment, if N	ot You				
promised to h	help you deal with e any payment or tra	your credito	y, did you or anyone else acting o ors or to make payments to your c u listed on line 16.			·
			Description and value of any proper	ty transferred	Date payment or transfer was made	Amount of paymen
Person Who	no Was Paid					\$
Number	Street					\$
City	State	ZIP Code	cy did you call trade or otherwis	o transfor any proporty to	anyone other than	·
City  S. Within 2 years  transferred in Include both o	State s before you filed to the ordinary could butright transfers and grifts and transfers	for bankrupt rse of your b d transfers m	cy, did you sell, trade, or otherwisusiness or financial affairs? ade as security (such as the grantine already listed on this statement.  Description and value of property transferred		ortgage on your property or payments received	n property perty).
City  S. Within 2 years transferred in Include both o Do not include No Yes. Fill in	State s before you filed to the ordinary could butright transfers and grifts and transfers	for bankrupt rse of your b d transfers m	usiness or financial affairs? ade as security (such as the grantin e already listed on this statement.  Description and value of property	g of a security interest or m  Describe any property	ortgage on your property or payments received	n property perty).  Date transfer
City  S. Within 2 years transferred in Include both o Do not include No Yes. Fill in	State  s before you filed to the ordinary couloutright transfers and gifts and transfers the details.	for bankrupt rse of your b d transfers m	usiness or financial affairs? ade as security (such as the grantin e already listed on this statement.  Description and value of property	g of a security interest or m  Describe any property	ortgage on your property or payments received	n property perty).  Date transfer
City  3. Within 2 years transferred in Include both o Do not include  No Yes. Fill in	State  s before you filed to the ordinary country transfers and transfers and transfers the details.  o Received Transfer	for bankrupt rse of your b d transfers m	usiness or financial affairs? ade as security (such as the grantin e already listed on this statement.  Description and value of property	g of a security interest or m  Describe any property	ortgage on your property or payments received	n property perty).  Date transfer
City  Notation 2 years transferred in Include both on Do not include in Notation Yes. Fill in Person Who	State  s before you filed to the ordinary country transfers and transfers and transfers the details.  o Received Transfer	for bankrupt rse of your b d transfers m that you have	usiness or financial affairs? ade as security (such as the grantin e already listed on this statement.  Description and value of property	g of a security interest or m  Describe any property	ortgage on your property or payments received	n property perty).  Date transfer
City  Within 2 years transferred in Include both o Do not include  No Yes. Fill in  Person Who  City  Person's re	State  s before you filed to the ordinary could but ight transfers and gifts and transfers the details.  o Received Transfer  Street	for bankrupt rse of your b d transfers m that you have	usiness or financial affairs? ade as security (such as the grantin e already listed on this statement.  Description and value of property	g of a security interest or m  Describe any property	ortgage on your property or payments received	n property perty).  Date transfer
City  B. Within 2 years transferred in Include both o Do not include  No Yes. Fill in  Person Who  City  Person's re	State  s before you filed to the ordinary country transfers and transfers and transfers at the details.  o Received Transfer  Street  State  relationship to you o Received Transfer	for bankrupt rse of your b d transfers m that you have	usiness or financial affairs? ade as security (such as the grantin e already listed on this statement.  Description and value of property	g of a security interest or m  Describe any property	ortgage on your property or payments received	n property perty).  Date transfer
City  8. Within 2 years transferred in Include both o Do not include  No Yes. Fill in  Person Who  City  Person's re	State  s before you filed to the ordinary country transfers and transfers and transfers at the details.  o Received Transfer  Street  State  relationship to you o Received Transfer	for bankrupt rse of your b d transfers m that you have	usiness or financial affairs? ade as security (such as the grantin e already listed on this statement.  Description and value of property	g of a security interest or m  Describe any property	ortgage on your property or payments received	n property perty).  Date transfer

otor 1	First Name	Middle Name	Last Name		Case i	number (if know	n)	
are	a beneficiary? (	These are often of	bankruptcy, did you t alled asset-protection		ry to a self-se	ttled trust (	or similar device of wl	nich you
			Description	and value of the prope	erty transferred			Date transfer was made
	Name of trust							
rt 8	3: List Certai	n Financial Ac	counts, Instrume	nts, Safe Deposi	t Boxes, an	d Storage	e Units	
Incl bro	lude checking, s kerage houses,	pension funds,					es in banks, credit uni	ons,
			Last 4 digits	of account number	Type of acc instrument	ount or	Date account was closed, sold, moved, or transferred	Last balance before closing or transfer
	Name of Financial  Number Street	Institution	XXXX		☐ Checkin☐ Savings☐ Money r☐ Brokera	narket		\$
	City	State ZIP	Code		Other_			
	Name of Financial	Institution	xxxx		☐ Checkin☐ Savings☐ Money r			\$
	Number Street  City	State ZIP	Code		Brokera Other_	ge		
sec	curities, cash, or	other valuables		ou filed for bankrup	tcy, any safe	deposit bo	ox or other depository	for
J	res. Fill III tile (	icialis.	Who else ha	nd access to it?		Describe the	contents	Do you sti
								□ No
	Name of Financial	Institution	Name					☐ Yes
	Number Street		Number Stre	et				

City

ZIP Code

City

ZIP Code

State

State

No			
Yes. Fill in the details.			
	Who else has or had access to it?	Describe the contents	Do you sti have it?
			□ No
Name of Storage Facility	Name		Yes
Number Street	Number Street		
	City State ZIP Code		
City State Z	IP Code		
		<u>'</u>	<u>'</u>
t 9: Identify Property Yo	u Hold or Control for Someone Else		_
-	rty that someone else owns? Include any pro	perty you borrowed from, are storing fo	or,
r hold in trust for someone.			
No			
Yes. Fill in the details.			
	Where is the property?	Describe the property	Value
Owner's Name	<del></del>		\$
	Number Street		
Number Street	Number Street		
Number Street	Number Street		
	City State ZIP 0	Code	
		Code	
City State Z	City State ZIP 0	Code	
City State Z t 10: Give Details About 6	City State ZIP ( Environmental Information	Code	
City State Z t 10: Give Details About E	City State ZIP ( Environmental Information  ving definitions apply:		
City State Z t 10: Give Details About E the purpose of Part 10, the follow Environmental law means any fec	City State ZIP of Environmental Information ving definitions apply:	cerning pollution, contamination, releas	
city State z t 10: Give Details About E the purpose of Part 10, the follow Environmental law means any fectors azardous or toxic substances, w	City State ZIP of Environmental Information  ving definitions apply: leral, state, or local statute or regulation convastes, or material into the air, land, soil, surfaces.	cerning pollution, contamination, releas ace water, groundwater, or other media	
City State Z  t 10: Give Details About E  the purpose of Part 10, the follow  Environmental law means any fect azardous or toxic substances, we including statutes or regulations	City State ZIP of	cerning pollution, contamination, releas ace water, groundwater, or other medit wastes, or material.	um,
City State Z  Table 10: Give Details About Enterprise of Part 10, the followed according to the purpose of Part 10, the purpos	City State ZIP of	cerning pollution, contamination, releas ace water, groundwater, or other medit wastes, or material.	um,
City State Z to 10: Give Details About Extremely City The purpose of Part 10, the follow City City City City City City City City	City State ZIP of Environmental Information  ving definitions apply:  deral, state, or local statute or regulation convastes, or material into the air, land, soil, surficentrolling the cleanup of these substances, or property as defined under any environmentate it, including disposal sites.	cerning pollution, contamination, releas ace water, groundwater, or other medit wastes, or material. tal law, whether you now own, operate,	um, , or utilize
City State Z to 10: Give Details About Extremely Details About Details Abo	City State ZIP of Environmental Information  ving definitions apply:  deral, state, or local statute or regulation convastes, or material into the air, land, soil, surficentrolling the cleanup of these substances, or property as defined under any environmental into the including disposal sites.	cerning pollution, contamination, releas ace water, groundwater, or other medit wastes, or material. tal law, whether you now own, operate,	um, , or utilize
City State Z  Table 10: Give Details About Extreme purpose of Part 10, the follow invironmental law means any fed azardous or toxic substances, who cluding statutes or regulations site means any location, facility, or used to own, operate, or utilidazardous material means anything ubstance, hazardous material, p	City State ZIP of Environmental Information  ving definitions apply: deral, state, or local statute or regulation convastes, or material into the air, land, soil, surficontrolling the cleanup of these substances, or property as defined under any environmentate it, including disposal sites.	cerning pollution, contamination, releas ace water, groundwater, or other medit wastes, or material. tal law, whether you now own, operate, ous waste, hazardous substance, toxic	um, , or utilize
City State Z t 10: Give Details About Entre purpose of Part 10, the follow Environmental law means any fector azardous or toxic substances, we including statutes or regulations Site means any location, facility, or or used to own, operate, or utilistazardous material means anythicubstance, hazardous material, p	City State ZIP of Environmental Information  ving definitions apply:  deral, state, or local statute or regulation convastes, or material into the air, land, soil, surficentrolling the cleanup of these substances, or property as defined under any environmental into the including disposal sites.	cerning pollution, contamination, releas ace water, groundwater, or other medit wastes, or material. tal law, whether you now own, operate, ous waste, hazardous substance, toxic	um, , or utilize
City State Z t 10: Give Details About Enthe purpose of Part 10, the follow Environmental law means any fector according statutes or regulations Site means any location, facility, or or used to own, operate, or utiling the statutes of the	City State ZIP of Environmental Information  ving definitions apply: deral, state, or local statute or regulation convastes, or material into the air, land, soil, surficontrolling the cleanup of these substances, or property as defined under any environmentate it, including disposal sites.  Ing an environmental law defines as a hazard ollutant, contaminant, or similar term.  Ceedings that you know about, regardless of	cerning pollution, contamination, releas ace water, groundwater, or other medit wastes, or material. tal law, whether you now own, operate, ous waste, hazardous substance, toxic when they occurred.	um, , or utilize
City State Z t 10: Give Details About Enthe purpose of Part 10, the follow Environmental law means any fector according statutes or regulations Site means any location, facility, or or used to own, operate, or utiling the statutes of the	City State ZIP of Environmental Information  ving definitions apply: deral, state, or local statute or regulation convastes, or material into the air, land, soil, surficontrolling the cleanup of these substances, or property as defined under any environmentate it, including disposal sites.	cerning pollution, contamination, releas ace water, groundwater, or other medit wastes, or material. tal law, whether you now own, operate, ous waste, hazardous substance, toxic when they occurred.	um, , or utilize
City State Z t 10: Give Details About Enthe purpose of Part 10, the follow Environmental law means any fector according statutes or regulations Site means any location, facility, or or used to own, operate, or utilistazardous material means anythicubstance, hazardous material, proof all notices, releases, and proof as any governmental unit notifie	City State ZIP of Environmental Information  ving definitions apply: deral, state, or local statute or regulation convastes, or material into the air, land, soil, surficontrolling the cleanup of these substances, or property as defined under any environmentate it, including disposal sites.  Ing an environmental law defines as a hazard ollutant, contaminant, or similar term.  Ceedings that you know about, regardless of	cerning pollution, contamination, releas ace water, groundwater, or other medit wastes, or material. tal law, whether you now own, operate, ous waste, hazardous substance, toxic when they occurred.	um, , or utilize
City State Z the Details About Enthe purpose of Part 10, the follow Environmental law means any fectorizations or toxic substances, who cluding statutes or regulations of the means any location, facility, of the or used to own, operate, or utilised azardous material means anythicubstance, hazardous material, proof all notices, releases, and proof as any governmental unit notified.	City State ZIP of Environmental Information  ving definitions apply: deral, state, or local statute or regulation convastes, or material into the air, land, soil, surficontrolling the cleanup of these substances, or property as defined under any environmentate it, including disposal sites.  Ing an environmental law defines as a hazard ollutant, contaminant, or similar term.  Ceedings that you know about, regardless of	cerning pollution, contamination, releas ace water, groundwater, or other medit wastes, or material. tal law, whether you now own, operate, ous waste, hazardous substance, toxic when they occurred.	um, , or utilize
City State Z  The Details About Extremely City and City a	City State ZIP of Environmental Information  ving definitions apply: deral, state, or local statute or regulation convastes, or material into the air, land, soil, surficontrolling the cleanup of these substances, or property as defined under any environmentate it, including disposal sites.  Ing an environmental law defines as a hazard ollutant, contaminant, or similar term.  Ceedings that you know about, regardless of	cerning pollution, contamination, releastace water, groundwater, or other medit wastes, or material.  tal law, whether you now own, operate, ous waste, hazardous substance, toxic when they occurred.  ble under or in violation of an environm	um, , or utilize
City State Z t 10: Give Details About E the purpose of Part 10, the follow Environmental law means any fec azardous or toxic substances, w including statutes or regulations Site means any location, facility, of tor used to own, operate, or utili dazardous material means anythicubstance, hazardous material, p ort all notices, releases, and pro- das any governmental unit notifie	City State ZIP of Environmental Information  ving definitions apply:  deral, state, or local statute or regulation convastes, or material into the air, land, soil, surficentrolling the cleanup of these substances, or property as defined under any environmental it, including disposal sites.  Ing an environmental law defines as a hazard ollutant, contaminant, or similar term.  Ceedings that you know about, regardless of d you that you may be liable or potentially lia	cerning pollution, contamination, releas ace water, groundwater, or other medit wastes, or material. tal law, whether you now own, operate, ous waste, hazardous substance, toxic when they occurred.	um, , or utilize : nental law?
City State Z to 10: Give Details About Extremely Detai	City State ZIP of Environmental Information  ving definitions apply:  deral, state, or local statute or regulation convastes, or material into the air, land, soil, surficentrolling the cleanup of these substances, or property as defined under any environmental it, including disposal sites.  Ing an environmental law defines as a hazard ollutant, contaminant, or similar term.  Ceedings that you know about, regardless of d you that you may be liable or potentially lia	cerning pollution, contamination, releastace water, groundwater, or other medit wastes, or material.  tal law, whether you now own, operate, ous waste, hazardous substance, toxic when they occurred.  ble under or in violation of an environm	um, , or utilize : nental law?
City State Z t 10: Give Details About E the purpose of Part 10, the follow Environmental law means any fec azardous or toxic substances, w including statutes or regulations Site means any location, facility, of tor used to own, operate, or utili dazardous material means anythicubstance, hazardous material, p ort all notices, releases, and pro- das any governmental unit notifie	City State ZIP of Environmental Information  ving definitions apply:  deral, state, or local statute or regulation convastes, or material into the air, land, soil, surficentrolling the cleanup of these substances, or property as defined under any environmental it, including disposal sites.  Ing an environmental law defines as a hazard ollutant, contaminant, or similar term.  Ceedings that you know about, regardless of d you that you may be liable or potentially lia	cerning pollution, contamination, releastace water, groundwater, or other medit wastes, or material.  tal law, whether you now own, operate, ous waste, hazardous substance, toxic when they occurred.  ble under or in violation of an environm	um, , or utilize : nental law?
City State Z  the purpose of Part 10, the follower invironmental law means any fectorizations or toxic substances, which is the means any location, facility, of the or used to own, operate, or utilisations material means anythicubstance, hazardous material, prort all notices, releases, and professional as any governmental unit notifier No  Yes. Fill in the details.	City State ZIP of Environmental Information  ving definitions apply:  deral, state, or local statute or regulation convastes, or material into the air, land, soil, surficentrolling the cleanup of these substances, or property as defined under any environmentate it, including disposal sites.  Ing an environmental law defines as a hazard ollutant, contaminant, or similar term.  Ceedings that you know about, regardless of d you that you may be liable or potentially liated.  Governmental unit	cerning pollution, contamination, releastace water, groundwater, or other medit wastes, or material.  tal law, whether you now own, operate, ous waste, hazardous substance, toxic when they occurred.  ble under or in violation of an environm	um, , or utilize : nental law?
City State Z  the purpose of Part 10, the follower invironmental law means any fectorizations or toxic substances, which is the means any location, facility, of the or used to own, operate, or utilisations material means anythicubstance, hazardous material, prort all notices, releases, and professional as any governmental unit notifier No  Yes. Fill in the details.	City State ZIP of Environmental Information  ving definitions apply:  deral, state, or local statute or regulation convastes, or material into the air, land, soil, surficentrolling the cleanup of these substances, or property as defined under any environmentate it, including disposal sites.  Ing an environmental law defines as a hazard ollutant, contaminant, or similar term.  Ceedings that you know about, regardless of d you that you may be liable or potentially liated.  Governmental unit	cerning pollution, contamination, releastace water, groundwater, or other medit wastes, or material.  tal law, whether you now own, operate, ous waste, hazardous substance, toxic when they occurred.  ble under or in violation of an environm	um, , or utilize : nental law?

Debtor 1

De	htor	1

First Name	Middle Name	Last Name	

Case number	(if known)			

25. Hav	re you notified any governmental unit of	f any release of hazardous materia	al?		
	Yes. Fill in the details.				
		Governmental unit	Environmental law	, if you know it	Date of notice
	Name of site	Governmental unit			
	Number Street	Number Street			_
		City State ZIP Code			
	City State ZIP Code				
26 Hav	re you been a party in any judicial or adı	ministrative proceeding under any	v environmental law	/2 Include settlements and or	dore
20.114		ministrative proceeding under any	environmentariaw	i include settlements and or	uers.
	Yes. Fill in the details.				
		Court or agency	Nature of the	case	Status of the
		country agons,	1144410 01 4110		case
	Case title	-			Pending
		Court Name			On appeal
		Number Street			☐ Concluded
	Case number	City State ZIP Co	de		
Part 1		siness or Connections to Any			
27. WIL	hin 4 years before you filed for bankrup  A sole proprietor or self-employed  A member of a limited liability comp  A partner in a partnership  An officer, director, or managing ex	in a trade, profession, or other actors on the control of the control of the control of a corporation	tivity, either full-tim ership (LLP)		1622 (
	An owner of at least 5% of the votin	ig or equity securities of a corpora	ation		
	No. None of the above applies. Go to P				
ч	Yes. Check all that apply above and fill	in the details below for each busi Describe the nature of the busines		Fundamental destination assumbase	
		Describe the nature of the busines	5	Employer Identification number  Do not include Social Security n	umber or ITIN.
	Business Name			FINI.	
	Number Street			EIN:	
		Name of accountant or bookkeepe	r	Dates business existed	
				F T.	
	City State ZIP Code			From To	_
	City State ZIP Code	Describe the nature of the busines	s	Employer Identification number	
	Business Name			Do not include Social Security n	
				EIN:	
	Number Street				
		Name of accountant or bookkeepe	r	Dates business existed	
				Erom T-	
	City State ZIP Code			From To	
	July State ZIF Code				

De	htor	1

			Case number (if known)
Tines Minus a	Middle Messes	Loot Name	

	Describe the nature of the business	Employer Identification number  Do not include Social Security number or ITIN.
Business Name		
		EIN:
Number Street	Name of accountant or bookkeeper	Dates business existed
City State ZIP Code		From To
28. Within 2 years before you filed for bankrupt institutions, creditors, or other parties.	cy, did you give a financial statement to anyone ab	out your business? Include all financial
□ No		
Yes. Fill in the details below.		
	Date issued	
Name	MM / DD / YYYY	
Number Street		
City State ZIP Code		
5 <u>5</u> 5		
Part 12: Sign Below		
rait 12. Sign below		
answers are true and correct. I understand in connection with a bankruptcy case can 18 U.S.C. §§ 152, 1341, 1519, and 3571.	t of Financial Affairs and any attachments, and I deed that making a false statement, concealing proper result in fines up to \$250,000, or imprisonment for	ty, or obtaining money or property by fraud
<b>x</b>	<b>x</b>	
Signature of Debtor 1	Signature of Debtor 2	
Date	Date	
Did you attach additional pages to Your Sa	tatement of Financial Affairs for Individuals Filing f	or Bankruptcy (Official Form 107)?
☐ No		
☐ Yes		
Did you pay or agree to pay someone who ☐ No	is not an attorney to help you fill out bankruptcy fo	rms?
	Attacl	n the Bankruptcy Petition Preparer's Notice,
·	Dec	laration, and Signature (Official Form 119).

#### **COMMITTEE NOTE**

Official Form 107, Statement of Financial Affairs for Individuals Filing for Bankruptcy, which applies only in cases of individual debtors, is revised in its entirety as part of the Forms Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. The goals of the Forms Modernization Project include improving the interface between technology and the forms so as to increase efficiency and reduce the need to produce the same information in multiple formats. Therefore, many of the open-ended questions and multiplepart instructions have been replaced with more specific questions. In addition, the form is renumbered to distinguish it from the version to be used in non-individual cases, and stylistic changes were made throughout the form

The form is derived from former Official Form 7, Statement of Financial Affairs. The new form uses eleven sections likely to be more understandable to non-lawyers, groups questions of a similar nature together, and eliminates questions unrelated to individual debtors. The new form deletes the instruction, previously found in many questions, that married debtors filing under chapter 12 or chapter 13 must include information applicable to their spouse, even if their spouse is not filing with them, unless the spouses are separated. This change was made because a non-filing spouse's general financial affairs are not relevant to the debtor's bankruptcy case.

Part 1, Give Details About Your Marital Status and Where You Lived Before, moves the questions regarding the debtor's prior addresses, as well as residences in a community property state, to the beginning of the form. The form eliminates the "name used" question in reference to prior addresses. Also, the debtor is no longer required to list the name of a spouse or former spouse who lived with the debtor in a community property state since that information will be provided in Official Form 106H.

Part 2, Explain the Sources of Your Income, consolidates the questions regarding income, adding "wages, commissions, bonuses, tips" as a category for sources of income, and it eliminates the option to report income on a fiscal year basis. In addition, the form provides examples of types of "other income." The time period is clarified to indicate that the prior two years means two calendar years, plus the portion of the calendar year in which the bankruptcy is filed.

Part 3, List Certain Payments You Made Before You Filed for Bankruptcy, includes questions related to payments made in the 90 days prior to bankruptcy, with a separate question for payments made to insiders within one year before filing for bankruptcy. The statutory definition of consumer debt is provided. The question regarding the nature of the debtor's debts requires the debtor to use checkboxes to indicate whether or not they are primarily consumer debts. The form instructs debtors not to include payments for domestic support obligations in the section regarding insider payments. The form provides a separate question regarding payments or transfers on account of a debt that benefited an insider. For both questions regarding payments to insiders, the debtor is required to provide a reason for the payment. Partnerships of which the debtor is a general partner have been added to the examples of "insiders."

Part 4, *Identify Legal Actions*, *Repossessions*, *and Foreclosures*, consolidates questions regarding actions against the debtor's property. The form provides examples of types of legal actions, and requires the debtor to indicate the status of any action. The form adds the requirements that a debtor include any property levied on within a year of filing for bankruptcy and that the debtor provide the last four digits of any account number for any setoffs. Also, a debtor must list any assignment for the benefit of creditors made within one year of filing for bankruptcy.

Part 5, *List Certain Gifts and Contributions*, changes the reporting threshold to \$600 per person or charity and increases the look-back period from one to two years.

Part 6, *List Certain Losses*, clarifies how to report insurance coverage for losses. It provides that the debtor must include on this form amounts of insurance that have been paid, but must list pending insurance claims on Official Form 106A/B.

Part 7, List Certain Payments or Transfers, includes questions regarding payments or transfers of property by the debtor. The question regarding payments or transfers to anyone who was consulted about seeking bankruptcy or preparing a bankruptcy petition requires the email or website address of the person who was paid, as well as the name of the person who made the payment if it was not the debtor. There is a separate question asked about payments or transfers to anyone who promised to help the debtor deal with creditors or make payments to creditors, reminding the debtor not to include any payments or transfers already listed. Also, the debtor must list any transfers of property, outright or for security purposes, made within two years of filing for bankruptcy, unless the transfer was made in the ordinary course of the debtor's business. There is a reminder not to list gifts or other transfers already included elsewhere on the form. The question regarding self-settled trusts adds an explanation that such trusts are often referred to as asset-protection devices.

Part 8, List Certain Financial Accounts, Safe Deposit Boxes, and Storage Units, adds money market accounts to the examples provided for the question regarding financial accounts or instruments and removes "other instruments" from the examples. Also, the form adds a question about whether the debtor has or had property stored in a storage unit within one year of filing for bankruptcy. The debtor must provide the name and address of the storage facility and anyone who has or had

access to the unit, as well as a description of the contents and whether the debtor still has access to the storage unit. Storage units that are part of the building in which the debtor resides are excluded.

Part 9, *Identify Property You Hold or Control for Someone Else*, instructs that the debtor should include any property that the debtor borrowed from, is storing for, or is holding in trust for someone.

Part 10, Give Details About Environmental Information, requires the debtor to list the case title and nature of the case for any judicial or administrative proceeding under any environmental law and to indicate the status of the case.

Part 11, Give Details About Your Business or Connections to Any Business, eliminates instructions that apply only to corporations and partnerships. The debtor must indicate if, within four years (previously six years) before filing for bankruptcy, the debtor owned a business or had certain connections to a business, with five categories of businesses provided as checkboxes. If the debtor has a connection to a business, the debtor must list the name, address, nature, and Employer Identification number of the business, the dates the business existed, and the name of an accountant or bookkeeper for the business. Accounting information requested is truncated; the debtor is simply required to provide the name of the business bookkeeper or accountant.

Part 12, *Sign Below*, eliminates the signature boxes for a partnership or corporation and a non-attorney bankruptcy petition preparer. Also, the debtor is asked to indicate through checkboxes whether additional pages are attached to the form

Fill in this information to identify your case:		
Debtor 1  First Name Middle Name Last Name		
Debtor 2		
(Spouse, if filing) First Name Middle Name Last Name		
United States Bankruptcy Court for the: District of (State)	)	
Case number		
(If known)		

☐ Check if this is an amended filing

### Official Form 108

## Statement of Intention for Individuals Filing Under Chapter 7

12/15

If you are an individual filing under chapter 7, you must fill out this form if:

- creditors have claims secured by your property, or
- you have leased personal property and the lease has not expired.

You must file this form with the court within 30 days after you file your bankruptcy petition or by the date set for the meeting of creditors, whichever is earlier, unless the court extends the time for cause. You must also send copies to the creditors and lessors you list on the form.

If two married people are filing together in a joint case, both are equally responsible for supplying correct information. Both debtors must sign and date the form.

Be as complete and accurate as possible. If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write your name and case number (if known).

Part 1: List Your Creditors Who Have Secured Claims

Identify the creditor and the property that is collateral	What do you intend to do with the property that secures a debt?	Did you claim the propert as exempt on Schedule C
Creditor's name:	☐ Surrender the property.	☐ No
	Retain the property and redeem it.	☐ Yes
Description of property securing debt:	Retain the property and enter into a Reaffirmation Agreement.	
g	Retain the property and [explain]:	
Creditor's	☐ Surrender the property.	□ No
name:	Retain the property and redeem it.	☐ Yes
Description of property securing debt:	☐ Retain the property and enter into a Reaffirmation Agreement.	
ocaling door.	☐ Retain the property and [explain]:	
Creditor's	☐ Surrender the property.	□ No
name:	Retain the property and redeem it.	☐ Yes
Description of property securing debt:	☐ Retain the property and enter into a Reaffirmation Agreement.	
	☐ Retain the property and [explain]:	
Creditor's	☐ Surrender the property.	□ No
name:	Retain the property and redeem it.	☐ Yes
Description of property securing debt:	☐ Retain the property and enter into a Reaffirmation Agreement.	
	☐ Retain the property and [explain]:	

			Case number (If known)
First Name	Middle Nome	Loot Name	<del></del>

Part 2:	List Your	Unexpired	Personal	Property	/ Leases

For any unexpired personal property lease that you listed in Schedule G: Executory Contracts and Unexpired Leases (Official Form 106G)
fill in the information below. Do not list real estate leases. Unexpired leases are leases that are still in effect; the lease period has not yet
ended. You may assume an unexpired personal property lease if the trustee does not assume it. 11 U.S.C. § 365(p)(2).

Describe your unexpired personal property leases	Will the lease be assumed?
Lessor's name:	□ No
Description of leased property:	☐ Yes
Lessor's name:	□ No
Description of leased property:	☐ Yes
Lessor's name:	□ No
Description of leased property:	☐ Yes
Lessor's name:	☐ No
Description of leased property:	☐ Yes
Lessor's name:	□ No
Description of leased property:	☐ Yes
Lessor's name:	□ No
Description of leased property:	☐ Yes
Lessor's name:	□ No
Description of leased property:	☐ Yes

Part 3:

Sign Below

Under penalty of perjury, I declare that I have indicated my intention about any property of my estate that secures a debt and any personal property that is subject to an unexpired lease.

×	×
Signature of Debtor 1	Signature of Debtor 2
Date	Date

#### **COMMITTEE NOTE**

Official Form 108, Statement of Intention for Individuals Filing Under Chapter 7, is revised in its entirety as part of the Forms Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. In addition, the form is renumbered, and stylistic changes are made throughout the form.

The form is derived from former Official Form 8, Chapter 7 - Individual Debtor's Statement of Intention. The new form uses language likely to be understandable to non-lawyers. In addition, the instructions are more extensive, advising an individual Chapter 7 debtor that the form must be completed and filed within 30 days and that the debtor must deliver copies of the form to creditors and lessors listed on the form.

Part 1, List Your Creditors Who Have Secured Claims, refers to entering into a "Reaffirmation Agreement" rather than asking whether the debtor intends to "reaffirm the debt." In addition, the debtor is asked if the property is claimed as exempt on Schedule C (Official Form 106C).

Part 2, List Your Unexpired Personal Property Leases, defines unexpired leases and explains that a debtor may assume an unexpired personal property lease if the trustee does not assume it.

Fill in this information to identify the case:				
Debtor 1	First Name	Middle Name	Last Name	
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name	
United States E	Bankruptcy Court for the:		District of	
Case number (If known)			Chapter	

#### Official Form 119

### Bankruptcy Petition Preparer's Notice, Declaration, and Signature

12/15

Bankruptcy petition preparers as defined in 11 U.S.C. § 110 must fill out this form every time they help prepare documents that are filed in the case. If more than one bankruptcy petition preparer helps with the documents, each must sign in Part 3. A bankruptcy petition preparer who does not comply with the provisions of title 11 of the United States Code and the Federal Rules of Bankruptcy Procedure may be fined, imprisoned, or both. 11 U.S.C. § 110; 18 U.S.C. § 156.

#### Part 1:

Notice to Debtor

Bankruptcy petition preparers must give the debtor a copy of this form and have the debtor sign it before they prepare any documents for filing or accept any compensation. A signed copy of this form must be filed with any document prepared.

Bankruptcy petition preparers are not attorneys and may not practice law or give you legal advice, including the following:

- whether to file a petition under the Bankruptcy Code (11 U.S.C. § 101 et seq.);
- whether filing a case under chapter 7, 11, 12, or 13 is appropriate;
- whether your debts will be eliminated or discharged in a case under the Bankruptcy Code;
- whether you will be able to keep your home, car, or other property after filing a case under the Bankruptcy Code;
- what tax consequences may arise because a case is filed under the Bankruptcy Code;
- whether any tax claims may be discharged;
- whether you may or should promise to repay debts to a creditor or enter into a reaffirmation agreement;
- how to characterize the nature of your interests in property or your debts; or
- what procedures and rights apply in a bankruptcy case.

The bankruptcy petition preparerName	9		has notified me of
any maximum allowable fee before pre	paring any document for filing or accepting any fe	е.	
Signature of Debtor 1 acknowledging receipt of	this notice	Date MM / DD / YYYY	<del>/</del>
Signature of Debtor 2 acknowledging receipt of	this notice	Date MM / DD / YYYY	<del>/</del>

$\Box$	ام	٦ŧ	_	r	1

Eiret Name	Middle Name	Last Name	

Part 2:

Declaration and Signature of the Bankruptcy Petition Preparer

Ur	der penalty of perjury, I declare that:				
	I am a bankruptcy petition preparer or the office	er, principal, respo	nsible person, or partner of	a ba	nkruptcy petition preparer:
	I or my firm prepared the documents listed bel Preparer as required by 11 U.S.C. §§ 110(b),	ow and gave the de	ebtor a copy of them and the		
	if rules or guidelines are established according preparers may charge, I or my firm notified the accepting any fee from the debtor.	-	· ·		
	Printed name Title, if an	ny	Firm name, if it applies		<del></del>
	Number Street				
	City State	ZIP Code	Contact phone		_
	I or my firm prepared the documents check (Check all that apply.)	ed below and the	completed declaration is	mad	le a part of each document that I check:
	Usuntary Petition (Form 101)  Statement About Your Social Security Numbers (Form 121)  Summary of Your Assets and Liabilities and Certain Statistical Information (Form 106Sum)  Schedule A/B (Form 106A/B)  Schedule C (Form 106C)  Schedule D (Form 106D)  Schedule E/F (Form 106E/F)  Schedule G (Form 106G)  Schedule H (Form 106H)  Bankruptcy petition preparers must sign and give to which this declaration applies, the signature and sign	Schedules (For Statement of Under Chapter 7 Statement of Under Chapter 7 Statement of Information of Abuse Undured (Form 122A-1)  Chapter 7 Mereor (Form 122A-2)  Chapter 7 Mereor (Form 122A-2)	bout an Individual Debtor's orm 106Dec) Financial Affairs (Form 107) Intention for Individuals Filing or 7 (Form 108) Internet of Your Current one (Form 122A-1) Exemption from Presumption or § 707(b)(2) Supp) ans Test Calculation	nkruj	
	Signature of bankruptcy petition preparer or officer, prince person, or partner	cipal, responsible	Social Security number of po	erson	Date who signed MM / DD / YYYY
	Printed name  Signature of bankruptcy petition preparer or officer, prin person, or partner  Printed name	cipal, responsible	Social Security number of po	 erson	Date who signed MM / DD / YYYY

#### **COMMITTEE NOTE**

Official Form 119, Bankruptcy Petition Preparer's Notice, Declaration, and Signature, applies only in cases of individual debtors. It is revised as part of the Forms Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. In addition, the form is renumbered, and stylistic changes are made throughout the form.

The form is derived from former Official Form 19, Declaration and Signature of Non-Attorney Bankruptcy Petition Preparer. An instruction is added to the form that provides statutory citations. Filers are advised that if more than one bankruptcy petition preparer helped with the documents, each must sign the form.

Part 1, *Notice to Debtor*, is moved to the beginning of the form and revised. An instruction is added that bankruptcy petition preparers must give the debtor a copy of the form and have the debtor sign it before they prepare any documents for filing or accept compensation, and that the form must be filed with any document prepared. It warns the debtor that bankruptcy petition preparers are not attorneys and may not practice law or give legal advice, with a list of examples of advice that may not be provided by a bankruptcy petition preparer. The signature line of this part includes a statement that the debtor acknowledges receipt of the notice.

Part 2, Declaration and Signature of the Bankruptcy Petition Preparer, revises the declaration by the bankruptcy petition preparer to include an officer, principal, responsible person, or partner of a bankruptcy petition preparer. The bankruptcy petition preparer must provide a firm name, if applicable, as well as a contact phone, and must indicate which documents the bankruptcy petition preparer prepared from a list of documents. An "other" option is provided for any additional documents. The signature line includes spaces for the bankruptcy petition

preparer to enter a social security number, and language regarding an officer, principal, responsible person, or partner of the bankruptcy petition preparer.

Fill in this information to identify your case:						
United States Bankruptcy Court for the:						
District of						
Case number (If known):						

## Official Form 121

## Statement About Your Social Security Numbers

12/15

Use this form to tell the court about any Social Security or federal Individual Taxpayer Identification numbers you have used. Do not file this form as part of the public case file. This form must be submitted separately and must not be included in the court's public electronic records. Please consult local court procedures for submission requirements.

To protect your privacy, the court will not make this form available to the public. You should not include a full Social Security Number or Individual Taxpayer Number on any other document filed with the court. The court will make only the last four digits of your numbers known to the public. However, the full numbers will be available to your creditors, the U.S. Trustee or bankruptcy administrator, and the trustee assigned to your case.

Making a false statement, concealing property, or obtaining money or property by fraud in connection with a bankruptcy case can result in fines up to \$250,000, or imprisonment for up to 20 years, or both. 18 U.S.C. §§ 152, 1341, 1519, and 3571.

Part 1: Tell the Court	About Yourself and Your spouse if Your Spouse i	s Filing With You
	For Debtor 1:	For Debtor 2 (Only If Spouse Is Filing):
1. Your name		
	First name	First name
	Middle name	Middle name
	Last name	Last name
Part 2: Tell the Court	About all of Your Social Security or Federal Indiv	ridual Taxpayer Identification Numbers
All Social Security     Numbers you have     used		
	☐ You do not have a Social Security number.	☐ You do not have a Social Security number.
3. All federal Individual Taxpayer Identification	9	9
Numbers (ITIN) you have used	9	9
Part 3: Sign Below	☐ You do not have an ITIN.	☐ You do not have an ITIN.
	Under penalty of perjury, I declare that the information I have provided in this form is true and correct.	Under penalty of perjury, I declare that the information I have provided in this form is true and correct.
	Signature of Debtor 1	Signature of Debtor 2
	Date	Date

## **COMMITTEE NOTE**

Official Form 121, *Statement About Your Social Security Numbers*, is revised as part of the Forms Modernization Project. The form, which applies only in cases of individual debtors, replaces former Official Form 21, *Statement of Social Security Number(s)*. It is renumbered to distinguish it from the forms used by non-individual debtors, such as corporations and partnerships.

To make Form 121 easier to understand and complete, the form is divided into three sections, and directions on the form are simplified. The debtor's Employer Tax-Identification number (EIN) is eliminated from the form, and the debtor's name is moved from the caption to the body of the form.

Fill in this information to identify your case:				only as directed in this form and in
Debtor 1		_	Form 122A-1Sup	p:
First Name Middle Name  Debtor 2	Last Name		1. There is no	presumption of abuse.
(Spouse, if filing) First Name Middle Name  United States Bankruptcy Court for the:	Last Name  District of		abuse applie	ion to determine if a presumption of es will be made under <i>Chapter 7</i> <i>Calculation</i> (Official Form 122A–2).
Case number	(State)			Test does not apply now because of
(If known)	-			itary service but it could apply later.
			☐ Check if this	is an amended filing
Official Form 122A—1				
Chapter 7 Statement of Your	Current Mo	nth	ly Income	12/15
Be as complete and accurate as possible. If two married p space is needed, attach a separate sheet to this form. Including additional pages, write your name and case number (if known on the have primarily consumer debts or because of quality Abuse Under § 707(b)(2) (Official Form 122A-1Supp) with the Part 1: Calculate Your Current Monthly Income	ude the line number to own). If you believe tha fying military service, othis form.	which t t you ar	he additional informate exempted from a	mation applies. On the top of any presumption of abuse because you
1. What is your marital and filling status? Check one only				
<ol> <li>What is your marital and filing status? Check one only.</li> <li>Not married. Fill out Column A, lines 2-11.</li> </ol>	•			
☐ Married and your spouse is filing with you. Fill out	both Columns A and B,	lines 2-1	11.	
☐ Married and your spouse is NOT filing with you. Y	ou and your spouse ar	e:		
☐ Living in the same household and are not leg	jally separated. Fill out b	both Col	umns A and B, lines	2-11.
Living separately or are legally separated. Fill under penalty of perjury that you and your spous spouse are living apart for reasons that do not in	se are legally separated ι	under no	onbankruptcy law tha	at applies or that you and your
Fill in the average monthly income that you received bankruptcy case. 11 U.S.C. § 101(10A). For example, if August 31. If the amount of your monthly income varied of Fill in the result. Do not include any income amount more income from that property in one column only. If you have	you are filing on Septem luring the 6 months, add than once. For example	nber 15, the inco , if both	the 6-month period ome for all 6 months spouses own the sa	would be March 1 through and divide the total by 6.
			Column A Debtor 1	Column B Debtor 2 or non-filing spouse
Your gross wages, salary, tips, bonuses, overtime, ar (before all payroll deductions).	nd commissions		\$	\$
Alimony and maintenance payments. Do not include particular column B is filled in.	ayments from a spouse i	f	\$	\$
4. All amounts from any source which are regularly paid of you or your dependents, including child support. In from an unmarried partner, members of your household, and roommates. Include regular contributions from a spot filled in. Do not include payments you listed on line 3.	nclude regular contribution your dependents, parent	ons s,	\$	\$
5. Net income from operating a business, profession,	Debtor 1 Debtor 2			
or farm Gross receipts (before all deductions)	\$			
Ordinary and necessary operating expenses	- \$ \$			
Net monthly income from a business, profession, or farm	\$	Copy here	\$	\$
Net income from rental and other real property     Gross receipts (before all deductions)	Debtor 1 Debtor 2			_
Ordinary and necessary operating expenses	<b>-</b> \$ \$			
Net monthly income from rental or other real property	\$ \$	Copy here	· \$	\$
7. Interest, dividends, and royalties	Ψ		\$	\$

as a victim of a war crime, a crime against humanity, or international or domestic terrorism. If necessary, list other sources on a separate page and put the total below.

Total amounts from separate pages, if any.

11. Calculate your total current monthly income. Add lines 2 through 10 for each column. Then add the total for Column A to the total for Column B.

+ \$		+ \$		
\$	+	\$	=	\$

Part 2: Determine Whether the Means Test Applies to You

12. Calculate your current monthly income for the year. Follow these steps:

x 12

12b. The result is your annual income for this part of the form.

12b. \$\_\_\_\_\_

monthly income

13. Calculate the median family income that applies to you. Follow these steps:

Fill in the state in which you live.

Fill in the number of people in your household.

MM / DD / YYYY

instructions for this form. This list may also be available at the bankruptcy clerk's office.

14. How do the lines compare?

14a. Line 12b is less than or equal to line 13. On the top of page 1, check box 1, *There is no presumption of abuse.*Go to Part 3.

14b. Line 12b is more than line 13. On the top of page 1, check box 2, *The presumption of abuse is determined by Form 122A-2*. Go to Part 3 and fill out Form 122A-2.

Part 3: Sign Below

By signing here, I declare under penalty of perjury that the information on this statement and in any attachments is true and correct.

Signature of Debtor 1
Signature of Debtor 2

Date

If you checked line 14a, do NOT fill out or file Form 122A-2.

MM / DD / YYYY

If you checked line 14b, fill out Form 122A-2 and file it with this form.

				-
Fill in this in	nformation to identify y	our case:		
Debtor 1				
Debtor 2	First Name	Middle Name	Last Name	
(Spouse, if filing)	First Name	Middle Name	Last Name	
United States I	Bankruptcy Court for the: _		District of(State)	
Case number			(Glate)	
(If known)				☐ Check if this is an amended filing
		_		
Official F	orm 122A—1	Supp		
Statem	ent of Exen	nption from	Presumption	n of Abuse Under § 707(b)(2) 12/19
exempted from	m a presumption of ab	use. Be as complete a to only one of you, th	nd accurate as possible.	ncome (Official Form 122A-1), if you believe that you are If two married people are filing together, and any of the omplete a separate Form 122A-1 if you believe that this is
Part 1: Ide	ntify the Kind of De	bts You Have		
personal, far		ose." Make sure that you		C. § 101(8) as "incurred by an individual primarily for a h the answer you gave at line 16 of the Voluntary Petition for
	to Form 122A-1; on the omit this supplement with			no presumption of abuse, and sign Part 3. Then
Yes. Go	to Part 2.			
Part 2: Det	ermine Whether Mi	litary Carvias Dravis	siana Apply to Vay	
Part 2. Det	ermine whether will	intary Service Provis	sions Apply to You	
2. Are you a d	isabled veteran (as def	ined in 38 U.S.C. § 374	1(1))?	
☐ No. Go	to line 3.			
	I you incur debts mostly U.S.C. § 101(d)(1); 32 L	•	e duty or while you were p	performing a homeland defense activity?
	No. Go to line 3.			
	Yes. Go to Form 122A- Then submit this s	1; on the top of page 1 cupplement with the sign	of that form, check box 1, 7 aled Form 122A-1.	There is no presumption of abuse, and sign Part 3.
	have you been a Reser mplete Form 122A-1. Do			
Yes. We	ere you called to active d	uty or did you perform a	homeland defense activit	y? 10 U.S.C. § 101(d)(1); 32 U.S.C. § 901(1).
	Complete Form 122A-1.			
☐ Yes.	Check any one of the fo	ollowing categories that	applies:	
	was called to active d 90 days and remain on a		, <b>2001</b> , for at least	If you checked one of the categories to the left, go to Form 122A-1. On the top of page 1 of Form 122A-1,
ę	was called to active d 90 days and was release which is fewer than 540	ed from active duty on _		check box 3, <i>The Means Test does not apply now,</i> and sign Part 3. Then submit this supplement with the signed Form 122A-1. You are not required to fill out the rest of
_	am performing a hom			Official Form 122A-1 during the exclusion period. The exclusion period means the time you are on active duty
<b></b> 1	performed a homelan	d defense activity for a	nt least 90 days,	or are performing a homeland defense activity, and for 540 days afterward. 11 U.S.C. § 707(b)(2)(D)(ii).
	ending on pefore I file this bankrupt	, which is fewer that tcy case.	ın 540 days	If your exclusion period ends before your case is closed,

you may have to file an amended form later.

Fill in this information to identify your case:	Check the appropriate box as directed in lines 40 or 42:
Debtor 1	According to the calculations required by this Statement:
Debtor 2 (Spouse, if filing) First Name Middle Name Last Name	☐ 1. There is no presumption of abuse.
United States Bankruptcy Court for the: District of	2. There is a presumption of abuse.
Case number	ite)
(If known)	☐ Check if this is an amended filing
Official Form 122A–2	
Chapter 7 Means Test Calculation	12/15
To fill out this form, you will need your completed copy of Chapter 7 Sta	ntement of Your Current Monthly Income (Official Form 122A-1).
Be as complete and accurate as possible. If two married people are filing is needed, attach a separate sheet to this form. Include the line number pages, write your name and case number (if known).  Part 1: Determine Your Adjusted Income	g together, both are equally responsible for being accurate. If more space to which the additional information applies. On the top of any additional
Part 1. Determine Tour Adjusted medine	
Copy your total current monthly income	Copy line 11 from Official Form 122A-1 here
2. Did you fill out Column B in Part 1 of Form 122A-1?	
☐ No. Fill in \$0 for the total on line 3.	
☐ Yes. Is your spouse filing with you?	
☐ No. Go to line 3.	
☐ Yes. Fill in \$0 for the total on line 3.	
Adjust your current monthly income by subtracting any part of your shousehold expenses of you or your dependents. Follow these steps:	spouse's income not used to pay for the
On line 11, Column B of Form 122A–1, was any amount of the income you regularly used for the household expenses of you or your dependents?	u reported for your spouse NOT
☐ No. Fill in 0 for the total on line 3.	
☐ Yes. Fill in the information below:	
State each purpose for which the income was used  For example, the income is used to pay your spouse's tax debt or to support people other than you or your dependents	Fill in the amount you are subtracting from your spouse's income
	\$
	\$
	+ \$
Total	\$ Copy total here → -\$
4. Adjust your current monthly income. Subtract the total on line 3 from line	ne 1. \$

Del	ht∩r	1

First Name Middle Name Last Name Case number (if known)

Part 2:

Calculate Your Deductions from Your Income

The Internal Revenue Service (IRS) issues National and Local Standards for certain expense amounts. Use these amounts to answer the questions in lines 6-15. To find the IRS standards, go online using the link specified in the separate instructions for this form. This information may also be available at the bankruptcy clerk's office.

Deduct the expense amounts set out in lines 6-15 regardless of your actual expense. In later parts of the form, you will use some of your actual expenses if they are higher than the standards. Do not deduct any amounts that you subtracted from your spouse's income in line 3 and do not deduct any operating expenses that you subtracted from income in lines 5 and 6 of Form 122A–1.

If your expenses differ from month to month, enter the average expense.

Whenever this part of the form refers to you, it means both you and your spouse if Column B of Form 122A-1 is filled in.

5. The number of people used in determining your deductions from income

Fill in the number of people who could be claimed as exemptions on your federal income tax return, plus the number of any additional dependents whom you support. This number may be different from the number of people in your household.

**National Standards** 

You must use the IRS National Standards to answer the guestions in lines 6-7.

6. **Food, clothing, and other items:** Using the number of people you entered in line 5 and the IRS National Standards, fill in the dollar amount for food, clothing, and other items.

\$			

7. **Out-of-pocket health care allowance:** Using the number of people you entered in line 5 and the IRS National Standards, fill in the dollar amount for out-of-pocket health care. The number of people is split into two categories—people who are under 65 and people who are 65 or older—because older people have a higher IRS allowance for health care costs. If your actual expenses are higher than this IRS amount, you may deduct the additional amount on line 22.

#### People who are under 65 years of age

7a. Out-of-pocket health care allowance per person

\$\_\_\_\_\_

7b. Number of people who are under 65

Χ

7c. Subtotal. Multiply line 7a by line 7b.

Conv	/ here	\$

#### People who are 65 years of age or older

7d. Out-of-pocket health care allowance per person

7e. Number of people who are 65 or older

`----

7f. **Subtotal.** Multiply line 7d by line 7e.

\_ Copy here → +

7g. Total. Add lines 7c and 7f.....

\$_			

Copy total here

Debtor 1				Case number (if known)	
	First Name	Middle Name	Last Name		

Local Standards You must use the IRS Local Standards to answer the questions in lines 8-15.
Based on information from the IRS, the U.S. Trustee Program has divided the IRS Local Standard for housing for bankruptcy purposes into two parts:  Housing and utilities – Insurance and operating expenses Housing and utilities – Mortgage or rent expenses
To answer the questions in lines 8-9, use the U.S. Trustee Program chart.  To find the chart, go online using the link specified in the separate instructions for this form.
This chart may also be available at the bankruptcy clerk's office.
8. Housing and utilities – Insurance and operating expenses: Using the number of people you entered in line 5, fill in the dollar amount listed for your county for insurance and operating expenses. \$
9. Housing and utilities – Mortgage or rent expenses:
9a. Using the number of people you entered in line 5, fill in the dollar amount listed for your county for mortgage or rent expenses
9b. Total average monthly payment for all mortgages and other debts secured by your home.
To calculate the total average monthly payment, add all amounts that are contractually due to each secured creditor in the 60 months after you file for bankruptcy. Then divide by 60.
Name of the creditor  Average monthly payment
\$
\$
<u></u>
<del> + \$</del>
Total average monthly payment \$   Copy here \$   Repeat this amount on line 33a.
9c. Net mortgage or rent expense.
Subtract line 9b (total average monthly payment) from line 9a (mortgage or rent expense). If this amount is less than \$0, enter \$0
10. If you claim that the U.S. Trustee Program's division of the IRS Local Standard for housing is incorrect and affects \$ the calculation of your monthly expenses, fill in any additional amount you claim.
Explainwhy:
wity
11. <b>Local transportation expenses:</b> Check the number of vehicles for which you claim an ownership or operating expense.
O. Go to line 14.
1. Go to line 12.
2 or more. Go to line 12.
12. <b>Vehicle operation expense:</b> Using the IRS Local Standards and the number of vehicles for which you claim the operating expenses, fill in the <i>Operating Costs</i> that apply for your Census region or metropolitan statistical area.

	ach veh dition, y	you may not claim the e		T two vernoics.				
/ehi	cle 1	Describe Vehicle 1:					_	
							_	
За.	Owne	ership or leasing costs us	sing IRS Local Stand	ard		\$	-	
3b.		age monthly payment for ot include costs for lease	•	Vehicle 1.				
	amou	lculate the average mor ints that are contractually you filed for bankruptcy.	y due to each secure	nd on line 13e, add all d creditor in the 60 mon	ths			
	Na	ame of each creditor for \	/ehicle 1	Average monthly payment				
				\$				
			····	+ \$				
					Сору		Repeat this	
		Total average	e monthly payment	\$	here →	<b>-</b> \$	amount on line 33b.	
		Total average ehicle 1 ownership or lea act line 13b from line 13a	ase expense	\$ ss than \$0, enter \$0	here →	- \$ \$		\$
		ehicle 1 ownership or lea	ase expense a. If this amount is les	\$ ss than \$0, enter \$0	here →		Copy net Vehicle 1 expense	\$
/ehi	Subtra	ehicle 1 ownership or lead act line 13b from line 13b from line 13b Describe Vehicle 2:	ase expense a. If this amount is les		here →		Copy net Vehicle 1 expense	\$
<b>/ehi</b> (	Subtra	ehicle 1 ownership or lea act line 13b from line 13a	ase expense a. If this amount is les	ard	here →		Copy net Vehicle 1 expense	\$
<b>/ehi</b> (	cle 2  Owne	ehicle 1 ownership or lead act line 13b from line 13b  Describe Vehicle 2: ership or leasing costs us	ase expense a. If this amount is les sing IRS Local Stand	ard	here →		Copy net Vehicle 1 expense	\$
<b>/ehi</b> (	cle 2  Owne  Avera Do no	ehicle 1 ownership or lead act line 13b from line 13a  Describe Vehicle 2:  ership or leasing costs using monthly payment for	ase expense a. If this amount is less sing IRS Local Stand r all debts secured by ed vehicles.	ard	here →		Copy net Vehicle 1 expense	\$
<b>/ehi</b> (	cle 2  Owne  Avera Do no	ehicle 1 ownership or lead and line 13b from line 13b  Describe Vehicle 2:  ership or leasing costs using monthly payment for lease of include costs for lease	ase expense a. If this amount is less sing IRS Local Stand r all debts secured by ed vehicles.	ard Vehicle 2.  Average monthly	here →		Copy net Vehicle 1 expense	\$
<b>/ehi</b> (	cle 2  Owne  Avera Do no	ehicle 1 ownership or lead and line 13b from line 13b  Describe Vehicle 2:  ership or leasing costs using monthly payment for lease of include costs for lease	ase expense a. If this amount is less sing IRS Local Stand r all debts secured by ed vehicles.	ard Vehicle 2.  Average monthly	here →		Copy net Vehicle 1 expense	\$
<b>/ehi</b> (	cle 2  Owne  Avera Do no	Pehicle 1 ownership or lead act line 13b from line 13a Describe Vehicle 2:  Pership or leasing costs using monthly payment for out include costs for lease ame of each creditor for Venicle 2.	ase expense a. If this amount is less sing IRS Local Stand r all debts secured by ed vehicles.	ard Vehicle 2.  Average monthly	here →		Copy net Vehicle 1 expense	\$

15. **Additional public transportation expense:** If you claimed 1 or more vehicles in line 11 and if you claim that you may also deduct a public transportation expense, you may fill in what you believe is the appropriate expense, but you may not claim more than the IRS Local Standard for *Public Transportation*.

\$

Additional Expense Deductions  These are additional deductions allowed by the Means Test.  Note: Do not include any expense allowances listed in lines 6-24.	
25. Health insurance, disability insurance, and health savings account expenses. The monthly expenses for insurance, disability insurance, and health savings accounts that are reasonably necessary for yourself, your dependents.	
Health insurance \$	
Disability insurance \$	
Health savings account	
Total \$ Copy total here →	\$
Do you actually spend this total amount?	
□ No. How much do you actually spend? □ Yes \$	
26. Continuing contributions to the care of household or family members. The actual monthly expenses the continue to pay for the reasonable and necessary care and support of an elderly, chronically ill, or disabled member of your immediate family who is unable to pay for such expenses. These expenses include contributions to an account of a qualified ABLE program. 26 U.S.C. § 529A(b).	nember of \$
27. <b>Protection against family violence.</b> The reasonably necessary monthly expenses that you incur to maintain of you and your family under the Family Violence Prevention and Services Act or other federal laws that apply	
By law, the court must keep the nature of these expenses confidential.	
28. Additional home energy costs. Your home energy costs are included in your insurance and operating expe	
If you believe that you have home energy costs that are more than the home energy costs included in expens 8, then fill in the excess amount of home energy costs.	ses on line \$
You must give your case trustee documentation of your actual expenses, and you must show that the additional claimed is reasonable and necessary.	
29. Education expenses for dependent children who are younger than 18. The monthly expenses (not more per child) that you pay for your dependent children who are younger than 18 years old to attend a private or pelementary or secondary school.	oublic \$
You must give your case trustee documentation of your actual expenses, and you must explain why the amore reasonable and necessary and not already accounted for in lines 6-23.	unt claimed is
* Subject to adjustment on 4/01/16, and every 3 years after that for cases begun on or after the date of adjust	etment.
30. <b>Additional food and clothing expense.</b> The monthly amount by which your actual food and clothing expense higher than the combined food and clothing allowances in the IRS National Standards. That amount cannot be 5% of the food and clothing allowances in the IRS National Standards.	
To find a chart showing the maximum additional allowance, go online using the link specified in the separate this form. This chart may also be available at the bankruptcy clerk's office.	instructions for
You must show that the additional amount claimed is reasonable and necessary.	
31. <b>Continuing charitable contributions.</b> The amount that you will continue to contribute in the form of cash or instruments to a religious or charitable organization. 26 U.S.C. § 170(c)(1)-(2).	financial + \$
32. Add all of the additional expense deductions.	\$
Add lines 25 through 31.	

			_	
Deductions	tor	Debt	Pay	yment

33. For debts that are secured by an interest in property that you own, including home mortgages, vehicle loans, and other secured debt, fill in lines 33a through 33e.

To calculate the total average monthly payment, add all amounts that are contractually due to each secured creditor in the 60 months after you file for bankruptcy. Then divide by 60.

	Mortgages on your home:			Average monthly payment	
33a.	Copy line 9b here		<b>-</b>	\$	
	Loans on your first two vehicles:				
33b.	Copy line 13b here.		······	\$	
33c.	Copy line 13e here		······	\$	
33d.	List other secured debts:				
	Name of each creditor for other secured debt	Identify property that secures the debt	Does payment include taxes or insurance?		
			☐ No ☐ Yes	\$	
			☐ No ☐ Yes	\$	
			□ No □ Yes	+ \$	
33e. To	otal average monthly payment. Add lines	33a through 33d		\$	

- 34. Are any debts that you listed in line 33 secured by your primary residence, a vehicle, or other property necessary for your support or the support of your dependents?
  - ☐ No. Go to line 35.
  - Yes. State any amount that you must pay to a creditor, in addition to the payments listed in line 33, to keep possession of your property (called the *cure amount*). Next, divide by 60 and fill in the information below.

Name of the creditor	Identify property that secures the debt	Total cure amount		Monthly cure amount
		\$	÷ 60 =	\$
		\$	÷ 60 =	\$
		\$	÷ 60 =	+ \$
			Total	\$Copy total

- 35. Do you owe any priority claims such as a priority tax, child support, or alimony—that are past due as of the filing date of your bankruptcy case? 11 U.S.C. § 507.
  - ☐ No. Go to line 36.
  - Yes. Fill in the total amount of all of these priority claims. Do not include current or ongoing priority claims, such as those you listed in line 19.

Total amount of all past-due priority claims .....

----- ÷ 60 =

\$\_\_\_\_\_

Debtor 1		

Middle Name

Last Name

First Name

Case number (if known)\_\_\_\_\_

For more information, go online using the link for Bankruptcy Basics specified in the separate instructions for this form. Bankruptcy Basics may also be available at the bankruptcy clerk's office.	
□ No. Go to line 37.	
☐ Yes. Fill in the following information.	
Projected monthly plan payment if you were filing under Chapter 13 \$	
Current multiplier for your district as stated on the list issued by the Administrative Office of the United States Courts (for districts in Alabama and North Carolina) or by the Executive Office for United States Trustees (for all other districts).	
To find a list of district multipliers that includes your district, go online using the link specified in the separate instructions for this form. This list may also be available at the bankruptcy clerk's office.	Consulated
Average monthly administrative expense if you were filing under Chapter 13 \$	Copy total here
37. Add all of the deductions for debt payment. Add lines 33e through 36.	s
Total Deductions from Income	
38. Add all of the allowed deductions.	
Copy line 24, All of the expenses allowed under IRS expense allowances\$	
Copy line 32, All of the additional expense deductions \$	
Copy line 37, All of the deductions for debt payment + \$	
Total deductions \$ Copy total here	<b></b> \$
Part 3: Determine Whether There Is a Presumption of Abuse	
39. Calculate monthly disposable income for 60 months	
39. Calculate monthly disposable income for 60 months  39a. Copy line 4, adjusted current monthly income \$	
39a. Copy line 4, adjusted current monthly income \$	
39a. Copy line 4, adjusted current monthly income \$  39b. Copy line 38, Total deductions \$  39c. Monthly disposable income. 11 U.S.C. § 707(b)(2). Subtract line 39b from line 39a.  Copy here → \$	60
39a. Copy line 4, adjusted current monthly income \$  39b. Copy line 38, Total deductions \$  39c. Monthly disposable income. 11 U.S.C. § 707(b)(2). Subtract line 39b from line 39a.  Copy here → \$	60 Copy
39a. Copy line 4, adjusted current monthly income \$  39b. Copy line 38, Total deductions \$  39c. Monthly disposable income. 11 U.S.C. § 707(b)(2). Subtract line 39b from line 39a. \$ \$  For the next 60 months (5 years)	Сору
39a. Copy line 4, adjusted current monthly income \$	Copy here \$
39a. Copy line 4, adjusted current monthly income \$	Copy here → \$
39a. Copy line 4, adjusted current monthly income  39b. Copy line 38, Total deductions \$	Copy here → \$

tor 1	First Name	Middle Name	Last Name	)		Case numb	CI (II KIIOWII)			
<b>41</b> . 41a.	Summary of	Your Assets and	d Liabilities a	ority unsecured of and Certain Statist line 3b on that fo	tical Information	Schedules	······· \$			
							· ———	<u> </u>		
41b.	-	•	-	ed debt. 11 U.S.(		.,.,	\$	Copy here	'	\$
is en	rmine whether to pay ck the box that	25% of your un	ou have lef	t over after subtr onpriority debt.	racting all allov	ed deductions				
	ine 39d is les So to Part 5.	ss than line 41b	o. On the top	of page 1 of this	form, check box	1, There is no p	oresumption of	abuse.		
		ual to or more may fill out Part		<b>b.</b> On the top of p			There is a pre	esumption		
art 4:		ils About Spe	·	,	anoco. mon go					
Do you l reasona	Give Deta have any spe able alternativ	ils About Spe cial circumstar re? 11 U.S.C. §	ecial Circu	mstances			rrent monthly	rincome for wh	ich there	e is no
Do you I reasona	Give Deta have any spe hable alternativ Go to Part 5. Fill in the follo	cial circumstar ve? 11 U.S.C. §	ecial Circunces that junces that junces that junces 707(b)(2)(B	mstances	xpenses or adj	ustments of cu			ich there	is no
Do you I reasona No.	Give Deta  have any spe  able alternativ  Go to Part 5.  Fill in the folk for each item  You must give adjustments r	cial circumstar /e? 11 U.S.C. §  owing informatio . You may include e a detailed exp	nces that ju 707(b)(2)(B on. All figures de expenses	mstances stify additional e ). s should reflect yo	xpenses or adj ur average mon 25. stances that mal	ustments of cu	income adjustr	ment	ich there	e is no
Do you I reasona No. Yes.	Give Deta  have any spe ble alternation Go to Part 5.  Fill in the follof for each item  You must give adjustments re expenses or i	cial circumstar ve? 11 U.S.C. §  owing informatio . You may include e a detailed expensessary and re-	nces that ju 707(b)(2)(B on. All figures de expenses lanation of the easonable. Yents.	mstances  stify additional e ).  s should reflect yo s you listed in line ne special circums you must also give	xpenses or adj ur average mon 25. stances that mal	ustments of cu	or income on of your actu	ment		e is no
Do you I reasona No.	Give Deta  have any spe ble alternation Go to Part 5.  Fill in the follof for each item  You must give adjustments re expenses or i	cial circumstar ve? 11 U.S.C. § owing informatio . You may include e a detailed exp necessary and reincome adjustment	nces that ju 707(b)(2)(B on. All figures de expenses lanation of the easonable. Yents.	mstances  stify additional e ).  s should reflect yo s you listed in line ne special circums you must also give	xpenses or adj ur average mon 25. stances that mal	ustments of cu	or income on of your actu  Averag or inco	ment ual ge monthly expens me adjustment		e is no
Reasona  ☐ No. ☐ Yes.	Give Deta  have any spe ble alternation Go to Part 5.  Fill in the follof for each item  You must give adjustments re expenses or i	cial circumstar ve? 11 U.S.C. § owing informatio . You may include e a detailed exp necessary and reincome adjustment	nces that ju 707(b)(2)(B on. All figures de expenses lanation of the easonable. Yents.	mstances  stify additional e ).  s should reflect yo s you listed in line ne special circums you must also give	xpenses or adj ur average mon 25. stances that mal	ustments of cu	or income on of your actu	ment ual ge monthly expens me adjustment	se	e is no
Do you I reasona No. Yes.	Give Deta  have any spe ble alternation Go to Part 5.  Fill in the follof for each item  You must give adjustments re expenses or i	cial circumstar ve? 11 U.S.C. § owing informatio . You may include e a detailed exp necessary and reincome adjustment	nces that ju 707(b)(2)(B on. All figures de expenses lanation of the easonable. Yents.	mstances  stify additional e ).  s should reflect yo s you listed in line ne special circums you must also give	xpenses or adj ur average mon 25. stances that mal	ustments of cu	or income on of your actu	ment ual ge monthly expens me adjustment	se 	e is no

Official Form 122A–2

Signature of Debtor 1

Date \_\_\_\_\_

Signature of Debtor 2

Date \_\_\_\_

Fill in this information to identify your case:						
Debtor 1	First Name	Middle Name	Last Name			
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name			
United States E	Bankruptcy Court for the: _		District of			
Case number (If known)						

☐ Check if this is an amended filing

## Official Form 122B

## Chapter 11 Statement of Your Current Monthly Income

12/15

You must file this form if you are an individual and are filing for bankruptcy under Chapter 11. If more space is needed, attach a separate sheet to this form. Include the line number to which the additional information applies. On the top of any additional pages, write your name and case number (if known).

nun	inder (II known).						
Pa	art 1: Calculate Your Current Monthly Income	е					
1.	What is your marital and filing status? Check one only.						
	■ Not married. Fill out Column A, lines 2-11.						
	☐ Married and your spouse is filing with you. Fill out	both Colum	ins A and B, I	lines 2-1	1.		
	☐ Married and your spouse is NOT filing with you. F	ill out Colum	nn A, lines 2-	11.			
	Fill in the average monthly income that you received from all sources, derived during the 6 full months before you file this bankruptcy case. 11 U.S.C. § 101(10A). For example, if you are filing on September 15, the 6-month period would be March 1 through August 31. If the amount of your monthly income varied during the 6 months, add the income for all 6 months and divide the total by 6. Fill in the result.  Do not include any income amount more than once. For example, if both spouses own the same rental property, put the income from that property in one column only. If you have nothing to report for any line, write \$0 in the space.						
					Column A Debtor 1	Column B Debtor 2	
2.	Your gross wages, salary, tips, bonuses, overtime, ar payroll deductions).	nd commiss	sions (before	all	\$	\$	
3.	<b>Alimony and maintenance payments.</b> Do not include particular Column B is filled in.	ayments froi	m a spouse if	f	\$	\$	
4.	All amounts from any source which are regularly paid you or your dependents, including child support. Include an unmarried partner, members of your household, your croommates. Include regular contributions from a spouse of Do not include payments you listed on line 3.	ude regular dependents,	contributions , parents, and	from d	\$	\$	
5.	Net income from operating a business, profession, or farm	Debtor 1	Debtor 2				
	Gross receipts (before all deductions)	\$	\$				
	Ordinary and necessary operating expenses	<b>-</b> \$	- \$				
	Net monthly income from a business, profession, or farm	\$	\$	Copy here→	\$	\$	
6.	Net income from rental and other real property	Debtor 1	Debtor 2				
	Gross receipts (before all deductions)	\$	\$				
	Ordinary and necessary operating expenses	- \$	- \$				
	Net monthly income from rental or other real property	\$	\$	Copy here→	\$	\$	

Last Name

		Column A Debtor 1	Column B Debtor 2	
7.	Interest, dividends, and royalties	\$	\$	
8.	Unemployment compensation	\$	\$	
	Do not enter the amount if you contend that the amount received was a benefit under the Social Security Act. Instead, list it here:			
	For you\$			
	For your spouse\$			
9.	<b>Pension or retirement income.</b> Do not include any amount received that was a benefit under the Social Security Act.	\$	\$	
10.	Income from all other sources not listed above. Specify the source and amount Do not include any benefits received under the Social Security Act or payments received as a victim of a war crime, a crime against humanity, or international or domestic terrorism.	t.		
	If necessary, list other sources on a separate page and put the total below.			
		\$	\$	
		\$	\$	
	Total amounts from separate pages, if any.	+ \$	+ \$	
11.	Calculate your total current monthly income. Add lines 2 through 10 for each column. Then add the total for Column A to the total for Column B.	\$	. \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \	\$ Total current monthly income
Pa	art 2: Sign Below			
	By signing here, under penalty of perjury I declare that the information on this states	ment and in any attacl	nments is true and correct.	
	Signature of Debtor 1 Signature of Debtor	or 2		
	Date	/ <u>/YYYY</u>		

Fill in this information to identify your case:				
Debtor 1 _	First Name	Middle Name	Last Name	
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name	
United States Bankruptcy Court for the:			District of (State)	
Case number (If known)				

Check as directed in lines 17 and 21:				
According to the calculations required by this Statement:				
1. Disposable income is not determined under 11 U.S.C. § 1325(b)(3).				
2. Disposable income is determined under 11 U.S.C. § 1325(b)(3).				
☐ 3. The commitment period is 3 years.				
4. The commitment period is 5 years.				

☐ Check if this is an amended filing

## Official Form 122C-1

# Chapter 13 Statement of Your Current Monthly Income and Calculation of Commitment Period

12/15

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for being accurate. If more space is needed, attach a separate sheet to this form. Include the line number to which the additional information applies. On the top of any additional pages, write your name and case number (if known).

Ρ	a	rt	1	

Calculate Your Average Monthly Income

<ol> <li>What is your marital and filing status? Check one only.</li> <li>Not married. Fill out Column A, lines 2-11.</li> </ol>						
☐ Married. Fill out both Columns A and B, lines 2-11.						
Fill in the average monthly income that you received fr bankruptcy case. 11 U.S.C. § 101(10A). For example, if y August 31. If the amount of your monthly income varied du the result. Do not include any income amount more than or from that property in one column only. If you have nothing	rou are filing ring the 6 m nce. For exa	on Septembe onths, add the mple, if both s	er 15, the e income spouses o	6-month period for all 6 months own the same re	would be March 1 throu and divide the total by 6	6. Fill in
				Column A Debtor 1	Column B Debtor 2 or non-filing spouse	
2. Your gross wages, salary, tips, bonuses, overtime, and payroll deductions).	commissio	ons (before all	l	\$	\$	
3. Alimony and maintenance payments. Do not include pay	ments from	a spouse.		\$	\$	
<ol> <li>All amounts from any source which are regularly paid f you or your dependents, including child support. Include an unmarried partner, members of your household, your de roommates. Do not include payments from a spouse. Do no listed on line 3.</li> </ol>	de regular co ependents, p	ontributions fro parents, and		\$	\$	
5. Net income from operating a business, profession, or farm	Debtor 1	Debtor 2				
Gross receipts (before all deductions)	\$	\$				
Ordinary and necessary operating expenses	<b>-</b> \$	<b>-</b> \$				
Net monthly income from a business, profession, or farm	\$	\$	Copy here	\$	\$	
6. Net income from rental and other real property	Debtor 1	Debtor 2				
Gross receipts (before all deductions)	\$	\$				
Ordinary and necessary operating expenses	<b>-</b> \$	<b>-</b> \$				

Net monthly income from rental or other real property

Copy

Case number (if known)\_\_\_\_\_

		Column A Debtor 1	Column B Debtor 2 or non-filing spouse	
7.	Interest, dividends, and royalties	\$	\$	
8.	Unemployment compensation	\$	\$	
	Do not enter the amount if you contend that the amount received was a benefit under the Social Security Act. Instead, list it here: $\Psi$			
	For you\$			
	For your spouse\$			
9.	<b>Pension or retirement income.</b> Do not include any amount received that was a benefit under the Social Security Act.	\$		
10.	Income from all other sources not listed above. Specify the source and amount. Do not include any benefits received under the Social Security Act or payments received as a victim of a war crime, a crime against humanity, or international or domestic terrorism. If necessary, list other sources on a separate page and put the total below.			
		\$	_ \$	
		\$	_ \$	
	Total amounts from separate pages, if any.	+ \$	+ \$	
11.	Calculate your total average monthly income. Add lines 2 through 10 for each column. Then add the total for Column A to the total for Column B.	\$	_ +	= [
				Total average monthly income
Pa	Determine How to Measure Your Deductions from Income			
12.	Copy your total average monthly income from line 11.			\$
13.	Calculate the marital adjustment. Check one:			
	☐ You are not married. Fill in 0 below.			
	☐ You are married and your spouse is filing with you. Fill in 0 below.			
	☐ You are married and your spouse is not filing with you.			
	Fill in the amount of the income listed in line 11, Column B, that was NOT regularl you or your dependents, such as payment of the spouse's tax liability or the spous you or your dependents.			
	Below, specify the basis for excluding this income and the amount of income devo	ted to each purpos	e. If necessary,	
	If this adjustment does not apply, enter 0 below.			
		- \$		
		_ \$	<del></del>	
		_ + \$		
	Total	\$	Copy here	
14.	Your current monthly income. Subtract the total in line 13 from line 12.		[	\$
15.	Calculate your current monthly income for the year. Follow these steps:			
	15a. Copy line 14 here 🗲			\$
	Multiply line 15a by 12 (the number of months in a year).		r	<b>x</b> 12
	15b. The result is your current monthly income for the year for this part of the form			\$

Debtor 1	First Name	Middle Name	Last Name		Case number (# known)	
16. Calculat	te the median	family income th	nat applies to you. F	Follow these steps:		
16a. Fill	in the state in	which you live.	_			
16b. Fill	in the number	of people in your	household.			
То	find a list of ap	plicable median i	ncome amounts, go		specified in the separate erk's office.	······ \$
17. How do	the lines com	pare?				
17a. 🗖					rm, check box 1, <i>Disposable income is not</i> <i>Ir Disposable Income</i> (Official Form 122C-	
17b. 🗖	11 U.S.C. § 1	325(b)(3). <b>Go to</b>	Part 3 and fill out C		box 2, <i>Disposable income is determined u</i> <b>Disposable Income (Official Form 122C-</b> bove.	
Part 3:	Calculate	Your Commitr	nent Period Unde	er 11 U.S.C. § 132	25(b)(4)	
18. <b>Copy vo</b>	ur total averad	ne monthly inco	me from line 11.			
19. <b>Deduct</b> calculati	the marital adj ng the commitn unt from line 13	ustment if it app nent period under	blies. If you are marr 11 U.S.C. § 1325(b	ried, your spouse is n )(4) allows you to ded	not filing with you, and you contend that duct part of your spouse's income, copy	<b>\$</b>
198. 11 11	ie mantai aujus	alment does not a	ppiy, iiii iii o oii iiile	19a		<b>-</b> \$
19b. <b>Su</b>	btract line 19a	from line 18.				\$
20. Calculat	te your curren	t monthly incom	e for the year. Follo	ow these steps:		
20a. Co	py line 19b					<b>¢</b>
N.4	و طفار ۱۹۰۸ ما درا مناطار		ha ia aau)			x 12
IVIU	iitipiy by 12 (the	number of mont	ns in a year).			<b>x</b> 12
20b. The	e result is your	current monthly i	ncome for the year fo	or this part of the form	n.	\$
20c. Cop	y the median fa	amily income for y	our state and size o	f household from line	e 16c	\$
21. <b>How do</b>	the lines com	pare?				
		n line 20c. Unless		by the court, on the to	op of page 1 of this form, check box 3,	
☐ Line	20b is more that	an or equal to line			ourt, on the top of page 1 of this form,	
Part 4:	Sign Below					
	By cianina ha	o under sessite	of porium I declare #	hat the information as	n this statement and in any attachments is	true and correct
		e, under penalty	or perjury r deciare to		X	true and correct.
	Signature of	Debtor 1			Signature of Debtor 2	
	Signature of	Deniol I			Signature of Deptor 2	
	Date				Date	
	MM / [	DD /YYYY			MM / DD / YYYY	
	-		out or file Form 122 n 122C–2 and file it v		e 39 of that form, copy your current monthl	ly income from line 14 above.

Fill in	this informatior	to identify	your case:				
Debtor	1 First Name		Middle Name	Last Name			
Debtor (Spouse	2 , if filing) First Name		Middle Name	Last Name			
	States Bankruptcy	Court for the:		District of			
Case n				(State)			
(If know						☐ Check if th	is is an amended filing
							3
Offic	ial Form	122C-2					
Cha	pter 13	Calcu	_ llation of	Your Dispos	able Inco	ome	12/15
Commi Be as c more s	tment Period (Complete and acopace is needed,	official Form curate as po attach a se	122C–1). ossible. If two ma parate sheet to th	copy of <i>Chapter 13 Staten</i> arried people are filing tog ais form. Include the line r ase number (if known).	ether, both are eq	ually responsible for	being accurate. If
Part 1	: Calculate	e Your Ded	luctions from Y	our Income			
to a inst  Ded som subt spou	nswer the ques ructions for this uct the expense e of your actual e racted from incoluse's income in li- ur expenses diffe	ations in line of form. This amounts set expenses if the me in lines 5 ine 13 of Form er from month	s 6-15. To find the information may out in lines 6-15 re hey are higher than and 6 of Form 122 m 122C-1.	onal and Local Standards e IRS standards, go onlin also be available at the begardless of your actual expenses the standards. Do not inc 2C-1, and do not deduct are the average expense.	e using the link sp ankruptcy clerk's pense. In later parts lude any operating on any amounts that you	pecified in the separa office. s of the form, you will u expenses that you a subtracted from your	<b>te</b> se
5.	Fill in the numb return, plus the	er of people number of a	who could be clain	your deductions from inc ned as exemptions on your ndents whom you support. household.	federal income tax		
	itional andards	You must	t use the IRS Natio	onal Standards to answer th	ne questions in lines	s 6-7.	
6.				umber of people you enterenthing, and other items.	ed in line 5 and the l	IRS National	\$
7.	Standards, fill in categories—ped	n the dollar a ople who are ealth care co	mount for out-of-pounder 65 and peopots. If your actual 6	the number of people you ocket health care. The num ple who are 65 or older—be expenses are higher than th	ber of people is spli cause older people	it into two have a higher IRS	

listed for your county for mortgage or rent expenses.

9b. Total average monthly payment for all mortgages and other debts secured by your home.

To calculate the total average monthly payment, add all amounts that are contractually due to each secured creditor in the 60 months after you file for bankruptcy. Next divide by 60.

Name of the creditor

Average monthly payment Copy Repeat this amount 9b. Total average monthly payment on line 33a. 9c. Net mortgage or rent expense. Subtract line 9b (total average monthly payment) from line 9a (mortgage or Copy here rent expense). If this number is less than \$0, enter \$0.

10. If you claim that the U.S. Trustee Program's division of the IRS Local Standard for housing is incorrect and affects the calculation of your monthly expenses, fill in any additional amount you claim.

Explain why:



14. Public transportation expense: If you claimed 0 vehicles in line 11, using the IRS Local Standards, fill in the Public Transportation expense allowance regardless of whether you use public transportation.

15. Additional public transportation expense: If you claimed 1 or more vehicles in line 11 and if you claim that you may also deduct a public transportation expense, you may fill in what you believe is the appropriate expense, but you may not claim more than the IRS Local Standard for Public Transportation.

First Name

Middle Name

Last Name

	ther Necessary xpenses	In addition to the expen following IRS categorie		above, you are allowed your monthly expenses for the	
16.	self-employment taxes from your pay for thes refund by 12 and subt	s, social security taxes, a e taxes. However, if you	and Medicare taxes. You expect to receive a tale total monthly amou	state and local taxes, such as income taxes, fou may include the monthly amount withheld ax refund, you must divide the expected int that is withheld to pay for taxes.	\$
17.	Involuntary deductio union dues, and uniform		ayroll deductions that	your job requires, such as retirement contributions,	
	Do not include amoun	ts that are not required b	oy your job, such as v	oluntary 401(k) contributions or payroll savings.	\$
18.	together, include payn	nents that you make for	your spouse's term lif		
	life insurance other that	an term.		a non-filing spouse's life insurance, or for any form of	\$
19.		ents: The total monthly sal or child support payr		as required by the order of a court or administrative	\$
	Do not include payme	nts on past due obligation	ons for spousal or chil	d support. You will list these obligations in line 35.	
20.		monthly amount that you	pay for education the	at is either required:	
	<ul><li>as a condition for you</li><li>for your physically of</li></ul>		ependent child if no p	ublic education is available for similar services.	\$
21.		nonthly amount that you nts for any elementary o		ch as babysitting, daycare, nursery, and preschool. ducation.	\$
22.	required for the health		our dependents and t	he monthly amount that you pay for health care that is hat is not reimbursed by insurance or paid by a health I entered in line 7.	
	Payments for health in	surance or health savin	gs accounts should b	e listed only in line 25.	\$
23.	B. Optional telephones and telephone services: The total monthly amount that you pay for telecommunication services for you and your dependents, such as pagers, call waiting, caller identification, special long distance, or business cell phone service, to the extent necessary for your health and welfare or that of your dependents or for the production of income, if it is not reimbursed by your employer.  Do not include payments for basic home telephone, internet or cell phone service. Do not include self-employment expenses, such as those reported on line 5 of Form 122C-1, or any amount you previously deducted.				
24.	Add all of the expens Add lines 6 through 23	ses allowed under the 3.	IRS expense allowar	nces.	\$
	dditional Expense eductions		al deductions allowed de any expense allow	d by the Means Test. ances listed in lines 6-24.	
25.				<b>bunt expenses.</b> The monthly expenses for health e reasonably necessary for yourself, your spouse, or	
	Health insurance		\$		
	Disability insurance		\$		
	Health savings accou	nt	+ \$		
	Total		\$	Copy total here	\$
	Do you actually spend	I this total amount?			
	☐ No. How much do☐ Yes	you actually spend?	\$		
26.	Continuing contribution continue to pay for the your household or me	reasonable and necess	sary care and support family who is unable	embers. The actual monthly expenses that you will of an elderly, chronically ill, or disabled member of to pay for such expenses. These expenses may 6 U.S.C. § 529A(b).	\$
27.	you and your family u		e Prevention and Serv	onthly expenses that you incur to maintain the safety of vices Act or other federal laws that apply.	\$

$\Box$	htor	1	

			Case number (if known)
irot Nome	Middle Nome	Last Name	<del></del>

28. 🗛	dditional home energy costs.	Your home energy costs are	included in your insurance and	I operating expenses on line 8.
-------	------------------------------	----------------------------	--------------------------------	---------------------------------

If you believe that you have home energy costs that are more than the home energy costs included in expenses on line 8, then fill in the excess amount of home energy costs.

•

You must give your case trustee documentation of your actual expenses, and you must show that the additional amount claimed is reasonable and necessary.

29. **Education expenses for dependent children who are younger than 18.** The monthly expenses (not more than \$156.25\* per child) that you pay for your dependent children who are younger than 18 years old to attend a private or public elementary or secondary school.

\_\_\_\_

You must give your case trustee documentation of your actual expenses, and you must explain why the amount claimed is reasonable and necessary and not already accounted for in lines 6-23.

\* Subject to adjustment on 4/01/16, and every 3 years after that for cases begun on or after the date of adjustment.

30. **Additional food and clothing expense.** The monthly amount by which your actual food and clothing expenses are higher than the combined food and clothing allowances in the IRS National Standards. That amount cannot be more than 5% of the food and clothing allowances in the IRS National Standards.

\_\_\_\_

To find a chart showing the maximum additional allowance, go online using the link specified in the separate instructions for this form. This chart may also be available at the bankruptcy clerk's office.

You must show that the additional amount claimed is reasonable and necessary.

31. **Continuing charitable contributions.** The amount that you will continue to contribute in the form of cash or financial instruments to a religious or charitable organization. 11 U.S.C. § 548(d)(3) and (4).

+ \$\_\_\_\_\_

Do not include any amount more than 15% of your gross monthly income.

32. Add all of the additional expense deductions.

Add lines 25 through 31.

\$\_\_\_\_\_

#### **Deductions for Debt Payment**

33. For debts that are secured by an interest in property that you own, including home mortgages, vehicle loans, and other secured debt, fill in lines 33a through 33e.

To calculate the total average monthly payment, add all amounts that are contractually due to each secured creditor in the 60 months after you file for bankruptcy. Then divide by 60.

	Average monthly payment
Mortgages on your home	
33a. Copy line 9b here	\$
Loans on your first two vehicles	
33b. Copy line 13b here. →	\$
33c. Copy line 13e here→	\$

33d. List other secured debts:

	Name of each creditor for other secured debt	Identify property that secures the debt	Does payment include taxes or insurance?		
			□ No □ Yes	\$	
			□ No □ Yes	\$	
			□ No □ Yes	+ \$	
33e.	Total average monthly payment. Add lines	33a through 33d		\$	Copy tot

opy total

t Name	Middle Name	Last Name

34.	Are any debts that you listed in line 33 secured by your primary residence, a vehicle, or other property necessary
	for your support or the support of your dependents?

No. Go to line 35.

☐ Yes. State any amount that you must pay to a creditor, in addition to the payments listed in line 33, to keep possession of your property (called the cure amount). Next, divide by 60 and fill in the information below.

Name of the creditor	Identify property that secures the debt	Total cure amount		Monthly cure amount
		\$	÷ 60 =	\$
		\$	÷ 60 =	\$
		\$	÷ 60 =	+ \$

Total

Copy total here 🛨

35. Do you owe any priority claims—such as a priority tax, child support, or alimony—that are past due as of the filing date of your bankruptcy case? 11 U.S.C. § 507.

☐ No. Go to line 36.

Yes. Fill in the total amount of all of these priority claims. Do not include current or ongoing priority claims, such as those you listed in line 19.

> ÷ 60 Total amount of all past-due priority claims.

36. Projected monthly Chapter 13 plan payment

Current multiplier for your district as stated on the list issued by the Administrative Office of the United States Courts (for districts in Alabama and North Carolina) or by the Executive Office for United States Trustees (for all other districts).

To find a list of district multipliers that includes your district, go online using the link specified in the separate instructions for this form. This list may also be available at the bankruptcy clerk's office.

Average monthly administrative expense

Copy total here 🛨

37. Add all of the deductions for debt payment. Add lines 33e through 36.

#### **Total Deductions from Income**

38. Add all of the allowed deductions.

Copy line 24, All of the expenses allowed under IRS expense allowances......\$\_\_\_\_\_\_\$

Copy line 32, All of the additional expense deductions.....\$

Copy line 37, All of the deductions for debt payment......+\$

Copy total Total deductions ......

Deb	otor 1	First N	ame	Middle	Name	Las	st Name					Cas	e number (	(if known)			 
Par	rt 2:					sable	Income	Under	์ 11 เ	J.S.C. 8	§ 1325	(b)(2)					
	Сору уо	ur to	al curre	nt mo	nthly in	come fr	om line	14 of Fo	orm 1	22C-1, (	Chapter	13					\$ -
	Fill in an children. disability received necessar	The paym	monthly ents for cordance	avera a depe with a	ge of an endent c applicab	/ child s hild, rep e nonba	upport pa	ayments, Part I of	, foste Form	er care p 122C-1	ayments , that yo	s, or ou	\$				
	Fill in all employer specified plans, as	withlin 11	neld from U.S.C.	n wage § 541(	es as cor b)(7) plu	ntribution s all req	ns for qua	alified ret	tirem	ent plans	s, as		\$				
42.	Total of a	all de	duction	s allo	wed und	er 11 U	.S.C. § 7	707(b)(2)	( <b>A</b> ). (	Copy line	e 38 here	e <del> 1</del>	\$				
	Deduction expenses and their special ci	and expe	you hav nses. Yo	e no re ou mus	easonab st give yo	e altern ur case	ative, de trustee	scribe the	e spe	ecial circ	umstand	ces					
	Describe	the s	pecial ci	rcumst	ances				Amo	unt of ex	pense						
									\$_								
									\$_								
							To	otal	+ \$_ \$_			Copy here	+\$		_		
44.	Total adj	ustm	ents. Ad	dd line	s 40 thro	ugh 43.							\$		Copy here	_	\$
45.	Calculate	e you	r month	ıly dis	posable	incom	e under	§ 1325(b	o)(2).	Subtract	t line 44	from line	39.				\$
Pa	rt 3:	Ch	ange ir	n Inco	ome or	Expen	ses										
46.	or are virtopen, fill	tually in the n the	certain t informa first colu	to chai ition be mn, ei	nge after elow. Fo nter line	the dat examp 2 in the	e you file le, if the second o	ed your bands	ankru eporte	uptcy pet ed increa	tition and ased afte	d during t er you file	he time y d your pe	form have o our case w stition, chec when the in	ill be k		
	Form		Line	Reas	on for ch	ange			1	Date of c	hange	Increa decre		Amount	of change		
	122C-	-		_								☐ Inc		\$			
	122C-											☐ Inc		\$	<del></del>		
	☐ 122C-											☐ Inc		\$	<del></del>		
	122C-											☐ Inc		\$			

Debtor 1				Case number (if known)
	First Name	Middle Name	Last Name	
Part 4:	Sign Be	low		
By signing h	ere, under p	enalty of perjury	you declare that the information	on on this statement and in any attachments is true and correct.
×				×
Signature	of Debtor 1			Signature of Debtor 2

Date \_\_\_\_\_

Date \_\_\_\_\_

135

#### **COMMITTEE NOTE**

Official Forms 122A-1, 122A-1Supp, 122A-2, 122B, 122C-1, and 122C-2 are updated to comport with the form numbering style developed as part of the Forms Modernization Project. The forms are derived from Official Forms 22A-1, 22A-1Supp, 22A-2, 22B, 122C-1, and 22C-2.

A statement is added to line 26 of Forms 122A-2 and 122C-2 explaining that contributions to qualified ABLE accounts, as defined in 26 U.S.C. § 529A(b), may be included in the deduction for contributions to the care of household or family members. Authorization of the deduction of such contributions was added to Bankruptcy Code § 707(b)(2)(A)(ii)(II) by the Tax Increase Prevention Act of 2014, Pub. Law No. 113-295.

Official Forms 122A-1, 122B, and 122C-1 are revised to add a workspace column for debtor 2 at questions 5 and 6 on the forms.

Official Form 122B is also revised to remove former Part 2. This portion of the form provided for the exclusion of certain income of a debtor's non-filing spouse; since that income is not required to be reported, its exclusion is unnecessary.

Other stylistic changes were made throughout the forms.

Fill in this information to identify th	e case:			
United States Bankruptcy Court for the	e:			
District of				
Case number (If known):	State) Chapter			☐ Check if this is a amended filing
<u>Official Form 201</u> Voluntary Petitio	n for Non-Individ	duals Fili	ng for Bank	(ruptcy 12/15
If more space is needed, attach a sep number (if known). For more informa				
1. Debtor's name				
2. All other names debtor used in the last 8 years				
Include any assumed names, trade names, and doing business as names				
3. Debtor's federal Employer Identification Number (EIN)				
4. Debtor's address	Principal place of business		Mailing address, if of business	different from principal place
	Number Street		Number Street	
			P.O. Box	
	City Stat	e ZIP Code	City	State ZIP Code
			Location of principal principal place of b	al assets, if different from usiness
	County		Number Street	
			City	State ZIP Code
5. Debtor's website (URL)				
6. Type of debtor	☐ Corporation (including Limited☐ Partnership (excluding LLP)☐ Other. Specify:			Partnership (LLP))

Del	btor Name			Case number (if ki	nown)		
	- Name						
7.	Describe debtor's business	A. Check	one:				
		☐ Health Care Business (as defined in 11 U.S.C. § 101(27A))					
		Single	Asset F	Real Estate (as defined in 11 U.S.C. § 101(51B))			
		Railroa	ad (as d	lefined in 11 U.S.C. § 101(44))			
		☐ Stockb	oroker (	as defined in 11 U.S.C. § 101(53A))			
		☐ Comm	odity B	roker (as defined in 11 U.S.C. § 101(6))			
		Clearing	ng Banl	(as defined in 11 U.S.C. § 781(3))			
		☐ None o	of the a	bove			
		B. Check	all that	apply:			
				ntity (as described in 26 U.S.C. § 501)	onto a bisher (see the fine at its 45 H O O		
		§ 80a-	ment co -3)	empany, including hedge fund or pooled investme	int vehicle (as defined in 15 U.S.C.		
		☐ Invest	ment ac	lvisor (as defined in 15 U.S.C. § 80b-2(a)(11))			
			•	American Industry Classification System) 4-digit	code that best describes debtor.		
		See <u>h</u>	ttp://ww	<u>w.naics.com/search/</u> .			
				· <del>_</del>			
8.	Under which chapter of the	Check on	e:				
	Bankruptcy Code is the debtor filing?	☐ Chapt	er 7				
		☐ Chapt	er 9				
		☐ Chapt	er 11. C	Check all that apply:			
			C	Debtor's aggregate noncontingent liquidated of insiders or affiliates) are less than \$2,490,925	`		
			Г	4/01/16 and every 3 years after that).	and in 44 H C C C 404/F4D) If the		
			•	■ The debtor is a small business debtor as defin debtor is a small business debtor, attach the n of operations, cash-flow statement, and federa	nost recent balance sheet, statement al income tax return or if all of these		
			[	documents do not exist, follow the procedure i  A plan is being filed with this petition.	n 11 U.S.C. § 1116(1)(B).		
			_	☐ Acceptances of the plan were solicited prepeti	tion from one or more classes of		
				creditors, in accordance with 11 U.S.C. § 1126			
			Į.	☐ The debtor is required to file periodic reports ( Securities and Exchange Commission accordi Exchange Act of 1934. File the Attachment to for Bankruptcy under Chapter 11 (Official Forr	ing to § 13 or 15(d) of the Securities  Voluntary Petition for Non-Individuals Filing		
			Ţ	☐ The debtor is a shell company as defined in the 12b-2.	e Securities Exchange Act of 1934 Rule		
		☐ Chapt	er 12	120-2.			
9.	Were prior bankruptcy cases	□ No					
	filed by or against the debtor within the last 8 years?	☐ Yes	District	When	Case number		
	•		Diotriot	When			
	If more than 2 cases, attach a separate list.		District	When	Case number		
10.	Are any bankruptcy cases	☐ No					
	pending or being filed by a	☐ Yes	Debtor		Relationship		
	business partner or an affiliate of the debtor?						
	List all cases. If more than 1, attach a separate list.			mber, if known	MM / DD /YYYY		
	account a coparato not.			•	_		

Del	otor Name		Case number (if known	)
11.	Why is the case filed in this district?	Check all that apply:		
	district:		, principal place of business, or princip date of this petition or for a longer part	
		☐ A bankruptcy case concern	ing debtor's affiliate, general partner, o	or partnership is pending in this district.
12.	Does the debtor own or have possession of any real property or personal property	☐ No☐ Yes. Answer below for each	n property that needs immediate attenti	ion. Attach additional sheets if needed.
	that needs immediate	Why does the proper	ty need immediate attention? (Check	k all that apply.)
	attention?	☐ It poses or is alleg	ed to pose a threat of imminent and ide	entifiable hazard to public health or safety.
		·	l?	•
		☐ It needs to be phys	sically secured or protected from the w	eather.
		attention (for exam	ble goods or assets that could quickly opple, livestock, seasonal goods, meat,	
		assets or other op	tions).	
		Other		
		Where is the propert	y? Number Street	
			Number Street	
			City	State ZIP Code
		Is the property insure	ed?	
		☐ No		
		Yes. Insurance ager	ncy	
		Contact name		
		Phone		
				•
	Statistical and adminis	trative information		
	Statistical and adminis	trative information		
13.	Debtor's estimation of	Check one:		
	available funds	☐ Funds will be available for o	distribution to unsecured creditors.	
		☐ After any administrative exp	penses are paid, no funds will be availa	able for distribution to unsecured creditors.
		<b>1</b> -49	<b>1</b> ,000-5,000	<b>2</b> 5,001-50,000
14.	Estimated number of creditors	50-99	5,001-10,000	50,001-100,000
		☐ 100-199 ☐ 200-999	10,001-25,000	☐ More than 100,000
15.	Estimated assets	□ \$0-\$50,000 □ \$50,001-\$100,000	\$1,000,001-\$10 million \$10,000,001-\$50 million	□ \$500,000,001-\$1 billion □ \$1,000,000,001-\$10 billion
		\$100,001-\$500,000	\$50,000,001-\$30 million	□ \$10,000,000,001-\$10 billion
		■ \$500,001-\$1 million	□ \$100,000,001-\$500 million	☐ More than \$50 billion

Debtor Name		Case number (if kn	own)
16. Estimated liabilities	\$0-\$50,000 \$50,001-\$100,000 \$100,001-\$500,000 \$500,001-\$1 million	□ \$1,000,001-\$10 million □ \$10,000,001-\$50 million □ \$50,000,001-\$100 million □ \$100,000,001-\$500 million	\$500,000,001-\$1 billion \$1,000,000,001-\$10 billion \$10,000,000,001-\$50 billion  More than \$50 billion
Request for Re	elief, Declaration, and Signatures	3	
	aud is a serious crime. Making a false st mprisonment for up to 20 years, or both.		
17. Declaration and sign authorized represent debtor	ature of The debtor requests rel petition.	ief in accordance with the chapter of titl	e 11, United States Code, specified in this
	I have been authorized	to file this petition on behalf of the debt	or.
	I have examined the inf correct.	ormation in this petition and have a rea	sonable belief that the information is true and
	I declare under penalty of p	erjury that the foregoing is true and con	rect.
	Executed on	YYYY	
	*		
	Signature of authorized rep		name
	Title		
18. Signature of attorney	×	Date	
	Signature of attorney for de	ebtor	MM / DD / YYYY
	Printed name		
	Firm name		
	Number Street		
	City	Stat	te ZIP Code
	Contact phone	Ema	ail address
	Bar number	Stat	te

### **COMMITTEE NOTE**

Official Form 201, Voluntary Petition for Non-Individuals Filing for Bankruptcy, replaces Official Form 1, Voluntary Petition, for non-individual debtors. It is renumbered to distinguish it from the forms used by individual debtors and includes formatting and stylistic changes throughout the form.

Official Form 201 is revised as part of the Forms Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. The goals of the Forms Modernization Project include improving the interface between technology and the forms so as to increase efficiency and reducing the need to produce the same information in multiple formats.

The Forms Modernization Project made a preliminary decision that separate forms should be created for individual debtors and for non-individual debtors because separate areas of inquiry apply to each group. The forms for non-individuals do not include questions that pertain only to individuals and use a more open-ended response format. Also, where possible, the forms for non-individuals parallel how businesses commonly keep their financial records.

Official Form 201 been substantially has reformatted and reorganized. References to Exhibits B, C, and D, and the exhibits themselves, have been eliminated because the requested information is now asked in the form or is not applicable to non-individual debtors. Official Form 201A, Attachment to Voluntary Petition for Non-Individuals Filing for Bankruptcy Under Chapter 11, has replaced Exhibit A. The debtor is instructed to file Official Form 201A if the debtor is filing under chapter 11 and is required to file periodic reports with the Securities and Exchange Commission. A checkbox has been added to the form to indicate whether it is an amended filing.

In Question 2, All other names debtor used in the last 8 years, instructions pertaining only to individuals have been deleted, and an instruction to include doing business as names and assumed names has been added. In Question 3, Debtor's federal Employee Identification Number (EIN), references to social security numbers and individual taxpayer I.D. numbers have been deleted. In Question 4, Debtor's address, the order of listing the various addresses for the debtor has been rearranged, and an address for the location of principal assets is required if different from the principal place of business. Also, the form has been revised to include a space for listing the debtor's website in Question 5.

In Question 6, Type of Debtor, options pertaining only to individual debtors have been deleted, and an instruction that the "partnership" option does not include LLPs has been added. Question 7, Describe debtor's business, is revised to include a statutory citation for each business type, to add an option for "none of the above," and to delete the option for "other." A new instruction requires the debtor to indicate if the debtor is an investment company, including a hedge fund or pooled investment vehicle; an investment advisor; or a tax exempt entity. The definition of "tax exempt entity" has been removed and replaced with a statutory citation. Additionally, an instruction has been added to require the debtor to list its North American Industry Classification System 4-digit code. A hyperlink is provided for information on finding the correct code.

In *Question 8, Under which chapter of the Bankruptcy Code is the debtor filing,* several separate boxes have been combined, and the options for Chapter 13 and Chapter 15 have been deleted. More detailed options have been added for Chapter 11. The question regarding the nature of the debtor's debts has been removed.

Question 9, Were prior bankruptcy cases filed by or against the debtor within the last 8 years, has been revised

to instruct the debtor to include prior bankruptcy cases filed against the debtor and to list the district rather than location of the prior filings. In Question 10, Are any bankruptcy cases pending or being filed by a business partner or an affiliate of the debtor, the reference to spouse and the requirement to list the judge in any other cases have been removed.

Question 11, Why is this case filed in this district, has been revised to delete references that pertain only to individuals.

Question 12, Does the debtor own or have possession of any real property or personal property that needs immediate attention, replaces Exhibit C from Official Form 1. The category of "property that needs immediate attention" has been added, as well as options to indicate why the property needs immediate attention. Additionally, the form has been revised to require the debtor to list the location of the property and whether or not the property is insured and, if so, the insurance details.

Statistical and administrative information has been moved to immediately above the signature line, and the reference to exempt property has been removed. The maximum values for "Estimated Assets" and "Estimated Liabilities" have been increased from "more than \$1 billion" to "more than \$50 billion." Request for Relief, Declaration, and Signatures has been reformatted and the signature lines for individual debtors and non-attorney bankruptcy petition preparers have been removed.

[If debtor is required to file periodic reports (e.g., forms 10K and 10Q) with the Securities and Exchange Commission pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934 and is requesting relief under chapter 11 of the Bankruptcy Code, this Exhibit "A" shall be completed and attached to the petition.]

[Caption as in Form 416B]

## Attachment to Voluntary Petition for Non-Individuals Filing for Bankruptcy under Chapter 11

a. Total assets	\$ 
b. Total debts (including debts listed in 2.c., below)	\$ 
c. Debt securities held by more than 500 holders	Approximate number of holders:
secured $\square$ unsecured $\square$ subordinated $\square$ \$ secured $\square$ unsecured $\square$ subordinated $\square$ \$ secured $\square$ unsecured $\square$ subordinated $\square$ \$	
d. Number of shares of preferred stock e. Number of shares common stock	
Comments, if any:	
3. Brief description of debtor's business:	

Fill in this information to identify the case and this filing:	
Debtor Name _	
United States Bankruptcy Court for the:	
Case number (If known):	(State)

## Official Form 202

# Declaration Under Penalty of Perjury for Non-Individual Debtors

12/15

An individual who is authorized to act on behalf of a non-individual debtor, such as a corporation or partnership, must sign and submit this form for the schedules of assets and liabilities, any other document that requires a declaration that is not included in the document, and any amendments of those documents. This form must state the individual's position or relationship to the debtor, the identity of the document, and the date. Bankruptcy Rules 1008 and 9011.

WARNING -- Bankruptcy fraud is a serious crime. Making a false statement, concealing property, or obtaining money or property by fraud in connection with a bankruptcy case can result in fines up to \$500,000 or imprisonment for up to 20 years, or both. 18 U.S.C. §§ 152, 1341, 1519, and 3571.

Declaration and signature

I am the president, another officer, or an authorized agent of the corporation; a member or an authorized agent of the partnership; or another individual serving as a representative of the debtor in this case.

I have examined the information in the documents checked below and I have a reasonable belief that the information is true and correct:

	Schedule A/B: Assets–Real and Personal Pro	operty (Official Form 206A/B)
	Schedule D: Creditors Who Have Claims Sec	cured by Property (Official Form 206D)
	Schedule E/F: Creditors Who Have Unsecure	ed Claims (Official Form 206E/F)
	Schedule G: Executory Contracts and Unexp	rired Leases (Official Form 206G)
	Schedule H: Codebtors (Official Form 206H)	
	Summary of Assets and Liabilities for Non-In-	dividuals (Official Form 206Sum)
	Amended Schedule	
	Chapter 11 or Chapter 9 Cases: List of Credi	tors Who Have the 20 Largest Unsecured Claims and Are Not Insiders (Official Form 204)
	Other document that requires a declaration_	
I de	clare under penalty of perjury that the foregoin	g is true and correct.
Exe	cuted on	<b>x</b>
	MM / DD / YYYY	Signature of individual signing on behalf of debtor
		Printed name

Position or relationship to debtor

#### **COMMITTEE NOTE**

Official Form 202, Declaration Under Penalty of Perjury for Non-Individual Debtors, replaces Official Form 2, Declaration Under Penalty of Perjury on Behalf of a Corporation or Partnership, and the section of Official Form 6 Declaration, Declaration Concerning Debtor's Schedules containing a corporation's or partnership's declaration. It is renumbered to distinguish it from the forms used by individual debtors and includes formatting and stylistic changes throughout the form.

Official Form 202 is revised as part of the Forms Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. The goals of the Forms Modernization Project include improving the interface between technology and the forms so as to increase efficiency and reducing the need to produce the same information in multiple formats.

Official Form 202 has been substantially reformatted and reorganized with elements from both Official Form 2 and the section of Official Form 6 for a corporation or partnership. Instructions have been added. along with warning language regarding bankruptcy fraud. Checkboxes are provided so the declaration will indicate the documents included with the declaration, including the schedules and Official Form 204, Chapter 11 or Chapter 9 Cases: List of Creditors Who Have the 20 Largest Unsecured Claims and Are Not Insiders. If the declaration accompanies another document, a space is provided to include a description of the attached document. The phrase "to the best of my information and belief" has been deleted from the declaration in order to conform to the language of 28 U.S.C. § 1746. See Rule 1008. The form, however, includes a statement that the person signing the declaration has examined the information in the documents subject to the declaration and has "a reasonable belief that the information is true and correct." Finally, the person signing the declaration must indicate his or her position or relationship to the debtor.

Fill in this information to identify the case:
Debtor name
United States Bankruptcy Court for the: District of
Case number (If known):

## Official Form 204

Chapter 11 or Chapter 9 Cases: List of Creditors Who Have the 20 Largest Unsecured Claims and Are Not Insiders

A list of creditors holding the 20 largest unsecured claims must be filed in a Chapter 11 or Chapter 9 case. Include claims which the debtor disputes. Do not include claims by any person or entity who is an *insider*, as defined in 11 U.S.C. § 101(31). Also, do not include claims by secured creditors, unless the unsecured claim resulting from inadequate collateral value places the creditor among the holders of the 20 largest unsecured claims.

	Name of creditor and complete mailing address, including zip code	Name, telephone number, and email address of creditor contact	debts, bank loans, professional	Indicate if claim is contingent, unliquidated, or disputed	claim amount. If total claim amou	ecured claim y unsecured, fill ir claim is partially s nt and deduction t ff to calculate uns	ecured, fill in for value of
					Total claim, if partially secured	Deduction for value of collateral or setoff	Unsecured claim
1							
2							
3							
4							
5							
6							
7							
8							

Debtor		Case number (if known)	
	Name		

Name of creditor and complete mailing address, including zip code	Name, telephone number, and email address of creditor contact	Nature of the claim (for example, trade debts, bank loans, professional services, and government contracts)	Indicate if claim is contingent, unliquidated, or disputed	Amount of unse If the claim is ful claim amount. If total claim amou collateral or seto	ecured claim ly unsecured, fill in claim is partially s nt and deduction ff to calculate uns	n only unsecured ecured, fill in for value of ecured claim.
				Total claim, if partially secured	Deduction for value of collateral or setoff	Unsecured claim
9						
10						
11						
12						
13						
14						
15						
16						
17						
18						
19						
20						

#### **COMMITTEE NOTE**

Official Form 204, Chapter 11 or Chapter 9 Cases: List of Creditors Who Have the 20 Largest Unsecured Claims and Are Not Insiders, replaces Official Form 4, List of Creditors Holding 20 Largest Unsecured Claims, for non-individual debtors. It is renumbered to distinguish it from the forms used by individual debtors and includes formatting and stylistic changes throughout the form.

Official Form 204 is revised as part of the Forms Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. The goals of the Forms Modernization Project include improving the interface between technology and the forms so as to increase efficiency and reducing the need to produce the same information in multiple formats.

The Forms Modernization Project made a preliminary decision that separate forms should be created for individual debtors and for non-individual debtors because separate areas of inquiry apply to each group. The forms for non-individuals do not include questions that pertain only to individuals and use a more open-ended response format. Also, where possible, the forms for non-individuals parallel how businesses commonly keep their financial records

Official Form 204 has been reformatted and reorganized. The instructions have been shortened and revised to include a full cite to the definition of "insider" and a revised explanation of when to include a secured creditor's unsecured claim. The warning regarding the disclosure of a minor child's name has been deleted as a caution has been added to the general instructions for all forms regarding listing a minor child's name.

The heading of the second column of the form has been revised to require the "name, telephone number, and email address of creditor contact," eliminating the need to provide a complete mailing address for the creditor contact. Additional examples of "nature of claim" have been provided in the third column. In the fourth column, "subject to setoff" has been removed as an option.

The fifth column has been revised to include three separate potential entries to be used to list the value of the unsecured claim: the total claim, if partially secured; the deduction for value of collateral or setoff; and unsecured claim. The new instructions for the fifth column contain an explanation that if a claim is a fully unsecured claim, only the final sub-column needs to be completed, and that all of the columns must be completed if a claim is partially secured.

The signature line and the instruction to include a declaration have been deleted from the form.

	Fill in this information to identi	ify the case:			
	United States Bankruptcy Court	for the:			
	Distric	t of (State)			
	Case number (If known):	Chapter			
	Official Form 205				☐ Check if this is an amended filing
_		tition Amainat a N	المملمة	del al	
	nvoluntary Pe	tition Against a N	ion-inaiv	/iduai	12/15
a ( m kn	case against an individual, use or space is needed, attach any lown).	tcy case against a non-individual you a the Involuntary Petition Against an Indi additional sheets to this form. On the	<i>ividual</i> (Official Fo top of any addition	rm 105). Be as complete and ac nal pages, write debtor's name	ccurate as possible. If
4	Chapter of the	Check one:			
1.	Chapter of the Bankruptcy Code	_			
		☐ Chapter 7 ☐ Chapter 11			
D.	ant 2. I don't for the Coleton				
Pa	art 2: Identify the Debtor				
2.	Debtor's name				
3.	Other names you know the debtor has used in the last 8 years				
	Include any assumed names, trade names, or doing business as names.				
4.	Debtor's federal Employer Identification Number (EIN)	Unknown			
		EIN	-		
5.	Debtor's address	Principal place of business		Mailing address, if different	
		Number Street		Number Street	

Official Form 205

City

County

State

ZIP Code

P.O. Box

City

Number

City

ZIP Code

ZIP Code

State

State

Location of principal assets, if different from principal place of business

Street

	Name	
6.	Debtor's website (URL)	
7.	Type of debtor	□ Corporation (including Limited Liability Company (LLC) and Limited Liability Partnership (LLP)) □ Partnership (excluding LLP) □ Other type of debtor. Specify:
8.	Type of debtor's business	Check one:
		<ul> <li>☐ Health Care Business (as defined in 11 U.S.C. § 101(27A))</li> <li>☐ Single Asset Real Estate (as defined in 11 U.S.C. § 101(51B))</li> <li>☐ Railroad (as defined in 11 U.S.C. § 101(44))</li> <li>☐ Stockbroker (as defined in 11 U.S.C. § 101(53A))</li> <li>☐ Commodity Broker (as defined in 11 U.S.C. § 101(6))</li> <li>☐ Clearing Bank (as defined in 11 U.S.C. § 781(3))</li> </ul>
		☐ None of the types of business listed. ☐ Unknown type of business.
9.	To the best of your knowledge, are any bankruptcy cases pending by or against any partner or affiliate of this debtor?	□ No □ Yes. Debtor Relationship District Date filed Case number, if known
		Debtor Relationship  District Date filed Case number, if known
Pa	art 3: Report About the	e Case
10.	Venue	<ul> <li>Check one:</li> <li>Over the last 180 days before the filing of this bankruptcy, the debtor had a domicile, principal place of business, or principal assets in this district longer than in any other district.</li> <li>A bankruptcy case concerning debtor's affiliates, general partner, or partnership is pending in this district.</li> </ul>
11.	. Allegations	Each petitioner is eligible to file this petition under 11 U.S.C. § 303(b).  The debtor may be the subject of an involuntary case under 11 U.S.C. § 303(a).
		<ul> <li>At least one box must be checked:</li> <li>The debtor is generally not paying its debts as they become due, unless they are the subject of a bona fide dispute as to liability or amount.</li> <li>Within 120 days before the filing of this petition, a custodian, other than a trustee, receiver, or an agent appointed or authorized to take charge of less than substantially all of the property of the debtor for the purpose of enforcing a lien against such property, was appointed or took possession.</li> </ul>
12	Has there been a transfer of any claim against the debtor by or to any petitioner?	☐ No ☐ Yes. Attach all documents that evidence the transfer and any statements required under Bankruptcy Rule 1003(a).

Case number (if known)

Debtor

Debtor		Case number (if known)	
Name			
3. Each petitioner's claim	Name of petitioner	Nature of petitioner's claim	Amount of the claim above the value of any lien
			\$
		<del></del>	\$
		Total of petitioners' claims	\$ \$
	rjury set out in Part 4 of the form, fo	er's representative, and the petitioner's attorney bllowed by each additional petitioner's (or repres	
	s a serious crime. Making a false state to 20 years, or both. 18 U.S.C. §§ 15	ement in connection with a bankruptcy case can res 52, 1341, 1519, and 3571.	sult in fines up to
petitioning creditor is a corporation	on, attach the corporate ownership sta	or under the chapter of 11 U.S.C. specified in this pertendent required by Bankruptcy Rule 1010(b). If any fied copy of the order of the court granting recogniti	petitioner is a
I have examined the information	in this document and have a reasonal	ole belief that the information is true and correct.	
Petitioners or Petitioners' Rep	resentative	Attorneys	
Name and mailing address of	petitioner		
Name		Printed name	
Number Street		Firm name, if any	
City	State ZIP Code	Number Street	
Name and mailing address of	petitioner's representative, if any	City	ZIP Code
Name		Contact phone Email	
		Par number	

State

Signature of attorney

MM / DD / YYYY

Date signed

X

Involuntary Petition Against a Non-Individual

Signature of petitioner or representative, including representative's title

MM / DD / YYYY

State

I declare under penalty of perjury that the foregoing is true and correct.

ZIP Code

Executed on

Number Street

City

Debtor		Case number (if known)	
	Name		

Number Street			Printed name		
			Firm name, if any		
			Number Street		
City	State	ZIP Code	Number Street		
Name and mailing address of	petitioner's rep	resentative, if any	City	State	ZIP Code
			Contact phone	Email	
ame			Bar number		
lumber Street					
			State	<del></del>	
Dity	State	ZIP Code			
declare under penalty of perjui	ry that the foregoi	ing is true and correct.	4		
Executed on MM / DD / YYYY			*		
WIWI / DD / ffff			Signature of attorney		
			Date signed		
Signature of petitioner or representa	tive, including repre	sentative's title	MM / DD /	YYYYY	
Name			Printed name		
Number Street			Firm name, if any		
	State	ZIP Code	Firm name, if any  Number Street		
Dity				State	ZIP Code
Sity			Number Street		ZIP Code
ity lame and mailing address of			Number Street  City		
lame and mailing address of			Number Street  City  Contact phone		
Dity  Name and mailing address of  Name			Number Street  City  Contact phone	Email	
lame and mailing address of			Number Street  City  Contact phone  Bar number	Email	
Dity  Name and mailing address of  Name	petitioner's repr	ZIP Code	Number Street  City  Contact phone  Bar number	Email	
lame and mailing address of lame lumber Street  City  declare under penalty of perjuly executed on	petitioner's repr	ZIP Code	Number Street  City  Contact phone  Bar number	Email	
Name and mailing address of Name Street	petitioner's repr	ZIP Code	Number Street  City  Contact phone  Bar number  State	Email	

#### **COMMITTEE NOTE**

Official Form 205, *Involuntary Petition Against a Non-Individual*, replaces Official Form 5, *Involuntary Petition*, for non-individual debtors. It is renumbered to distinguish it from the forms used by individual debtors and includes formatting and stylistic changes throughout the form.

Official Form 205 is revised as part of the Forms Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. The goals of the Forms Modernization Project include improving the interface between technology and the forms so as to increase efficiency and reducing the need to produce the same information in multiple formats.

The Forms Modernization Project made a preliminary decision that separate forms should be created for individual debtors and for non-individual debtors because separate areas of inquiry apply to each group. The forms for non-individuals do not include questions that pertain only to individuals and use a more open-ended response format. Also, where possible, the forms for non-individuals parallel how businesses commonly keep their financial records.

Part 1, *Identify the Chapter of the Bankruptcy Code Under Which Petition is Filed*, has been moved to the beginning of the form.

In Part 2, *Identify the Debtor*, instructions pertaining only to individuals have been deleted, and an instruction to include doing-business-as names and assumed names has been added. The references to social security numbers and individual taxpayer I.D. numbers have been deleted. The order of listing the various addresses for the debtor have been rearranged in Line 5, and an address for the location of principal assets is required if different from the principal place of business.

The form has been revised to include a space for listing the debtor's website in Line 6.

Also in Part 2, the options for type of debtor that pertained only to individuals have been deleted, and an instruction that the "partnership" option does not include LLPs has been added. The options regarding the type of debtor's business have been revised to include a statutory citation for each business type, to add an option for "none of the above," and to delete the option for "other." The question regarding pending bankruptcy cases has been revised to remove the reference to spouse and the requirement to list the judge in any other cases.

In Part 3, Report About the Case, the question regarding venue has been revised in Line 10 to read "[o]ver the last 180 days before the filing of this bankruptcy, the debtor had a domicile, principal place or business, or principal assets in this district longer than in any other district." In the question for Allegations, "each" has been added to the first allegation, the exact citation to the Bankruptcy Code has been provided for the second allegation, and checkboxes have been provided for the last allegation. Also, in Line 12, petitioners must check "yes" or "no" to answer whether there has been any transfer of any claim against the debtor by or to a petitioner.

The information regarding the petitioner's claims has been moved to Part 3, and the portion listing the amount of the claim is amended to ask about the amount of the claim that exceeds the value of the lien, if any.

Part 4, *Request for Relief*, has been amended to include a warning about making a false statement, and the declaration under penalty of perjury has been revised in order to conform to the language of 28 U.S.C. § 1746. *See* Rule 1008. A statement has been added that each petitioner, or the petitioner's representative, has reviewed the information in the petition and has "a reasonable belief that the information is true and correct." A requirement has

been added for each petitioner's mailing address. Also, petitioners' attorneys must provide their email addresses, bar number, and state of bar membership.

Fill in this information to identify the case:		
Debtor name		
United States Bankruptcy Court for the:	District of(State)	
Case number (If known):	(Oldio)	☐ Check if this amended fill

### Official Form 206A/B

# Schedule A/B: Assets — Real and Personal Property

12/15

Disclose all property, real and personal, which the debtor owns or in which the debtor has any other legal, equitable, or future interest. Include all property in which the debtor holds rights and powers exercisable for the debtor's own benefit. Also include assets and properties which have no book value, such as fully depreciated assets or assets that were not capitalized. In Schedule A/B, list any executory contracts or unexpired leases. Also list them on Schedule G: Executory Contracts and Unexpired Leases (Official Form 206G).

Be as complete and accurate as possible. If more space is needed, attach a separate sheet to this form. At the top of any pages added, write the debtor's name and case number (if known). Also identify the form and line number to which the additional information applies. If an additional sheet is attached, include the amounts from the attachment in the total for the pertinent part.

For Part 1 through Part 11, list each asset under the appropriate category or attach separate supporting schedules, such as a fixed asset schedule or depreciation schedule, that gives the details for each asset in a particular category. List each asset only once. In valuing the debtor's interest, do not deduct the value of secured claims. See the instructions to understand the terms used in this form.

Р	rt 1: Cash and cash equivalents			
1.	Does the debtor have any cash or cash equivalents?			
	☐ No. Go to Part 2.			
	Yes. Fill in the information below.			
	All cash or cash equivalents owned or controlled b	v the debter		Current value of debtor's
	All cash of cash equivalents owned of controlled b	y the debtor		interest
2.	Cash on hand			\$
3.	Checking, savings, money market, or financial broke	rage accounts (Identify all)		
	Name of institution (bank or brokerage firm)	Type of account	Last 4 digits of account number	
	3.1			\$
	3.2			\$
4.	Other cash equivalents (Identify all)			
	4.1			\$
	4.2.			\$
5.	Total of Part 1			\$
	Add lines 2 through 4 (including amounts on any addition	nal sheets). Copy the total to I	ne 80.	
Р	art 2: Deposits and prepayments			
6.	Does the debtor have any deposits or prepayments?			
	No. Go to Part 3.			
	Yes. Fill in the information below.			
				Current value of
_				debtor's interest
7.	Deposits, including security deposits and utility depo	osits		
	Description, including name of holder of deposit			
	7.1			\$
	7.2			\$

8.	Prepayments, including pre	epayments on executo	ry contracts, leases, insurance, taxes, a	nd rent		
	Description, including name of h	older of prepayment				
	8.1				\$	
	8.2				\$	
٥	Total of Part 2					
Э.		the total to line 81			\$	
	Add lines / throught o. Copy	the total to line or.				
Pa	art 3: Accounts receive	able				
10	. Does the debtor have any	tal of Part 2.  Id lines 7 through 8. Copy the total to line 81.  3 Accounts receivable  Does the debtor have any accounts receivable?  No. Go to Part 4.  Yes. Fill in the information below.  Current value of debtor's interest  Accounts receivable  1a. 90 days old or less:  1ace amount  1b. Over 90 days old:  1ace amount  1cotal of Part 3  Current value on lines 11a + 11b = line 12. Copy the total to line 82.  1cotal of Part 5.  Yes. Fill in the information below.  Valuation method used for current value of debtor's interest  Autual funds or publicly traded stocks not included in Part 1  ame of fund or stock:  1.1.  Accounts receivable  Current value of debtor's interest  Valuation method used for current value of debtor's interest  Autual funds or publicly traded stocks not included in Part 1  ame of fund or stock:  1.1.  Accounts receivable  Current value of debtor's interest  Valuation method used for current value  Surrent value of debtor's interest  Autual funds or publicly traded stock and interests in incorporated and unincorporated businesses, including any interest in an LLC, partnership, or joint venture				
	No. Go to Part 4.					
	<u></u>	on below.				
					Current value of debtor's	
11	. Accounts receivable					
	11a, 00 days old or loss:		_		¢	
	ria. 90 days old of less	face amount		· <del>7</del>	Φ	
	11b. Over 90 days old:		_ =	· <b>→</b>	\$	
	,	face amount	doubtful or uncollectible accounts			
12	Total of Part 3					
12		+ 11b = line 12. Copy the total to line 82.				
	Carrent value on lines 114	. 115 mic 12. copy til	is total to line oz.			
D,	art 4: Investments					
13		investments?				
	☐ Yes. Fill in the informati	on below.				
11	Mutual funds or publicly t	raded stocks not inclu	ided in Part 1			
-		iaueu stocks not meit	aded in Fait 1			
	44.4				\$	
					Τ	
15				5,		
	including any interest in a	an LLC, parmership, or	r joint venture			
	Name of entity:		% of ownership:			
	15.2		%		\$	
16	Government hands corne	orate honds, and other	r negotiable and non-negotiable			
10	instruments not included		inegonable and non negonable			
	Describe:					
	16.1				\$	
					_	
17	. Total of Part 4	S S S S S S S S S S S S S S S S S S S				
	Add lines 14 through 16. Co	opy the total to line 83.			,	

Part 5:	Inventory, excluding agriculture assets

18.	Does the debtor own any inventory (excluded No. Go to Part 6.  ☐ Yes. Fill in the information below.	ding agriculture assets	s)?		
	General description	Date of the last physical inventory	Net book value of debtor's interest (Where available)	Valuation method used for current value	Current value of debtor's interest
19.	Raw materials				
		MM / DD / YYYY	\$		\$
20.	Work in progress				
		MM / DD / YYYY	\$		\$
21	Finished goods including goods held for				
۷۱.	Finished goods, including goods held for r	resale	•		\$
		MM / DD / YYYY	\$		Ψ
22.	Other inventory or supplies				
		MM / DD / YYYY	\$		\$
23	Total of Part 5				
25.	Add lines 19 through 22. Copy the total to line	e 84.			\$
24.	Is any of the property listed in Part 5 perish No	hable?			
	Yes				
25.	Has any of the property listed in Part 5 bee	en purchased within 20	) davs before the bank	ruptcv was filed?	
	□ No				
	Yes. Book value V	aluation method	Curr	ent value	
26.	Has any of the property listed in Part 5 bee	en appraised by a prof	essional within the las	t year?	
	□ No			-	
	Yes				
Par	t 6: Farming and fishing-related ass	ets (other than title	ed motor vehicles a	nd land)	
27	Does the debtor own or lease any farming	and fishing-related as	sats (ather than titled	motor vehicles and land\2	
21.	□ No. Go to Part 7.	and naming-related as	sets (other than thed	motor vernoles and landy:	
	☐ Yes. Fill in the information below.				
			Not be all value of	Valuation mathed wood	Comment value of debterie
	General description		Net book value of debtor's interest (Where available)	Valuation method used for current value	Current value of debtor's interest
28.	Crops—either planted or harvested				
			\$		\$
29.	Farm animals Examples: Livestock, poultry, f	farm-raised fish			
			\$		\$
30.	Farm machinery and equipment (Other tha	n titled motor vehicles)			
	• • • • • •	,	\$		\$
31	Farm and fishing supplies, chemicals, and	feed	•		
J1.			\$		\$
20	Other ferming and fishing related property				Ψ
32.	Other farming and fishing-related property	_			•
			\$		\$

	□ No □ Yes			
35	Yes  Has any of the property listed in Part 6 been purchased within 20	days before the bank	runtcy was filed?	
00.	No	aujo bololo illo bulli.	aptoy mao mou i	
	☐ Yes. Book value \$ Valuation method	Current value	<b>:</b> \$	
36.	Is a depreciation schedule available for any of the property listed		·	
	□ No			
	☐ Yes			
37.	Has any of the property listed in Part 6 been appraised by a profe	essional within the last	t year?	
	□ No			
	Yes			
Pa	rt 7: Office furniture, fixtures, and equipment; and collect	ctibles		
38.	Does the debtor own or lease any office furniture, fixtures, equip	ment, or collectibles?		
	☐ No. Go to Part 8.			
	☐ Yes. Fill in the information below.			
	General description	Net book value of debtor's interest	Valuation method used for current value	Current value of debtor's interest
		(Where available)	acca for carroin value	
39	Office furniture	,		
00.		\$		\$
40	Office findamen	Ψ		Ψ
40.	Office fixtures			
		\$		\$
41.	Office equipment, including all computer equipment and			
	communication systems equipment and software	•		Φ.
		\$ 		\$
42.	<b>Collectibles</b> <i>Examples:</i> Antiques and figurines; paintings, prints, or of artwork; books, pictures, or other art objects; china and crystal; stamp	ther o, coin,		
	or baseball card collections; other collections, memorabilia, or collections			
	42.1	\$		\$
	42.2			\$
	42.3	\$		\$
43.	Total of Part 7.			\$
	Add lines 39 through 42. Copy the total to line 86.			
44.	Is a depreciation schedule available for any of the property listed	I in Part 7?		
	No No			
	Yes			
45.	Has any of the property listed in Part 7 been appraised by a profe	essional within the last	t year?	
	No No			
	Yes			
	5 - 1 - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 -			161

Pai	t 8: Machinery, equipment, and vehicles			
46.	Does the debtor own or lease any machinery, equipment, or vehic	cles?		
	☐ No. Go to Part 9.			
	Yes. Fill in the information below.			
	General description	Net book value of debtor's interest	Valuation method used for current value	Current value of debtor's interest
	Include year, make, model, and identification numbers (i.e., VIN, HIN, or N-number)	(Where available)	Tor current value	
47.	Automobiles, vans, trucks, motorcycles, trailers, and titled farm v	vehicles		
	47.1	\$		\$
	47.2	\$		\$
	47.3	\$		\$
	47.4	\$		\$
48.	Watercraft, trailers, motors, and related accessories Examples: Bo trailers, motors, floating homes, personal watercraft, and fishing vesses			
	48.1	\$		\$
	48.2	\$		\$
49.	Aircraft and accessories			
	49.1	\$		\$
	49.2	\$		\$
50.	Other machinery, fixtures, and equipment (excluding farm machinery and equipment)			
		\$		\$
51	Total of Part 8.			
01.	Add lines 47 through 50. Copy the total to line 87.			\$
52.	Is a depreciation schedule available for any of the property listed	in Part 8?		
	□ No			
	Yes			
53.	Has any of the property listed in Part 8 been appraised by a profe	essional within the last y	/ear?	
	Yes			

Par	t 9: Real property				
54.	Does the debtor own or lease any real proper  No. Go to Part 10.  Yes. Fill in the information below.	ty?			
55.	Any building, other improved real estate, or la	and which the debtor	owns or in which the	debtor has an interest	
	Description and location of property Include street address or other description such as Assessor Parcel Number (APN), and type of property (for example, acreage, factory, warehouse, apartment or office building), if available.	Nature and extent of debtor's interest in property	Net book value of debtor's interest (Where available)	Valuation method used for current value	Current value of debtor's interest
	55.1	-	\$		\$
	55.2		\$		\$
	55.3		\$		\$
	55.4		\$		\$
			\$		\$
	55.5				
	55.6		\$		\$
56.	<b>Total of Part 9.</b> Add the current value on lines 55.1 through 55.6	and entries from any a	ndditional sheets. Copy	the total to line 88.	\$
	<ul> <li>No</li> <li>Yes</li> <li>Has any of the property listed in Part 9 been a</li> <li>No</li> <li>Yes</li> <li>10: Intangibles and intellectual proper</li> </ul>		sional within the last y	year?	
59.	Does the debtor have any interests in intangil  ☐ No. Go to Part 11.  ☐ Yes. Fill in the information below.	bles or intellectual pr	operty?		
	General description		Net book value of debtor's interest (Where available)	Valuation method used for current value	Current value of debtor's interest
60.	Patents, copyrights, trademarks, and trade se	ecrets	,		
61.	Internet domain names and websites		\$		\$
62.	Licenses, franchises, and royalties		\$ \$		\$
63.	Customer lists, mailing lists, or other compile	ations	\$		\$
64.	Other intangibles, or intellectual property		\$		\$
65.	Goodwill		\$		\$
66.	<b>Total of Part 10.</b> Add lines 60 through 65. Copy the total to line 89	).			\$

Name

Part 12:

Summary

In Part 12 copy all of the totals from the earlier parts of the form.

Type of property	Current value of personal property	Current value of real property
80. Cash, cash equivalents, and financial assets. Copy line 5, Part 1.	\$	
81. Deposits and prepayments. Copy line 9, Part 2.	\$	
32. Accounts receivable. Copy line 12, Part 3.	\$	
83. Investments. Copy line 17, Part 4.	\$	
84. Inventory. Copy line 23, Part 5.	\$	
85. Farming and fishing-related assets. Copy line 33, Part 6.	\$	
86. Office furniture, fixtures, and equipment; and collectibles.  Copy line 43, Part 7.	\$	
87. Machinery, equipment, and vehicles. Copy line 51, Part 8.	\$	
38. <b>Real property.</b> Copy line 56, Part 9	→	\$
39. Intangibles and intellectual property. Copy line 66, Part 10.	\$	
90. <b>All other assets.</b> Copy line 78, Part 11.	+ \$	
91. <b>Total.</b> Add lines 80 through 90 for each column91a.	\$	+ 91b. \$
92. <b>Total of all property on Schedule A/B.</b> Lines 91a + 91b = 92		

	ill in this information to identify the case.			
D U	ebtor name nited States Bankruptcy Court for the: ase number (If known):		-	<b>7</b> or 1 or 10 or
			·	☐ Check if this is an amended filing
_	Official Form 206D			
_		Vho Have Claims Secured b	y Property	12/15
	e as complete and accurate as possible.  Do any creditors have claims secured by deb  No. Check this box and submit page 1 of this  Yes. Fill in all of the information below.	tor's property? s form to the court with debtor's other schedules. Debtor ha	as nothing else to report	on this form.
Pa	List Creditors Who Have Secure	d Claims		
2.	List in alphabetical order all creditors who ha secured claim, list the creditor separately for each	ve secured claims. If a creditor has more than one ch claim.	Column A  Amount of claim  Do not deduct the value of collateral.	Column B Value of collateral that supports this claim
2.1	Creditor's name	Describe debtor's property that is subject to a lien		
	Creditor's mailing address		- \$	\$
		Describe the lien	-	
	Creditor's email address, if known	Is the creditor an insider or related party?  No Yes		
	Date debt was incurred	Is anyone else liable on this claim?		
	number	Yes. Fill out Schedule H: Codebtors (Official Form 206H).  As of the petition filing date, the claim is:		
	No Yes. Specify each creditor, including this creditor, and its relative priority.	Check all that apply.  Contingent Unliquidated Disputed		
2.2	Creditor's name	Describe debtor's property that is subject to a lien		
	Creditor's mailing address		_ \$	\$
		Describe the lien	-	
	Creditor's email address, if known	Is the creditor an insider or related party?  No Yes		
	Date debt was incurred	Is anyone else liable on this claim?		
	Last 4 digits of account number	☐ No ☐ Yes. Fill out <i>Schedule H: Codebtors</i> (Official Form 206H).		
	Do multiple creditors have an interest in the same property?	As of the petition filing date, the claim is: Check all that apply.		
	<ul> <li>No</li> <li>Yes. Have you already specified the relative priority?</li> <li>No. Specify each creditor, including this creditor, and its relative priority.</li> </ul>	☐ Contingent☐ Unliquidated☐ Disputed☐		
	Yes. The relative priority of creditors is specified on lines			
3.	Total of the dollar amounts from Part 1, Colur Page, if any.	nn A, including the amounts from the Additional	\$	

Case	number	(if known)
Ousc	Hullibel	(11 K110W11)

Part 1: Additional Page  Copy this page only if more space is needed (	Continue numbering the lines sequentially from the	Column A  Amount of claim  Do not deduct the value of collateral.	Column B Value of collateral that supports this claim
previous page.			
c Creditor's name	Describe debtor's property that is subject to a lien		
Creditor's mailing address		- \$	\$
	Describe the lien	-	
Creditor's email address, if known	Is the creditor an insider or related party? ☐ No ☐ Yes		
Date debt was incurred Last 4 digits of account	Is anyone else liable on this claim?		
number	Yes. Fill out <i>Schedule H: Codebtors</i> (Official Form 206H).		
Do multiple creditors have an interest in the same property?  No	As of the petition filing date, the claim is:  Check all that apply.  Contingent		
☐ Yes. Have you already specified the relative priority?	☐ Unliquidated ☐ Disputed		
No. Specify each creditor, including this creditor, and its relative priority.			
Yes. The relative priority of creditors is specified on lines	Describe debtor's property that is subject to a lien		
Creditor's mailing address		- \$ -	\$
		-	
	Describe the lien		
Creditor's email address, if known	Is the creditor an insider or related party? ☐ No ☐ Yes		
Date debt was incurred	Is anyone else liable on this claim? ☐ No		
Last 4 digits of account number	Yes. Fill out Schedule H: Codebtors (Official Form 206H).		
Do multiple creditors have an interest in the same property?	As of the petition filing date, the claim is: Check all that apply.		
☐ No	☐ Contingent ☐ Unliquidated		
Yes. Have you already specified the relative priority?	Disputed		
No. Specify each creditor, including this creditor, and its relative priority.			
Yes. The relative priority of creditors is specified on lines			

Part 2:

List Others to Be Notified for a Debt Already Listed in Part 1

List in alphabetical order any others who must be notified for a debt already listed in Part 1. Examples of entities that may be listed are collection agencies, assignees of claims listed above, and attorneys for secured creditors.

Name and address	On which line in Part 1 did you enter the related creditor?	Last 4 digits of account number for this entity
	Line 2	

Fill in this information to identify the case:			
Debtor			
United States Bankruptcy Court for the:	District of		
Case number	(State)		
(If known)			
			Check if this is an amended filing
Official Form 206E/F			amenaea ming
Schedule E/F: Creditors W	/ho Have Unsecured	Claims	12/15
Be as complete and accurate as possible. Use Part unsecured claims. List the other party to any execu on Schedule A/B: Assets - Real and Personal Prope (Official Form 206G). Number the entries in Parts 1 the Additional Page of that Part included in this for	atory contracts or unexpired leases that c erty (Official Form 206A/B) and on Schedu and 2 in the boxes on the left. If more spa	ould result in a claim. Also ule G: Executory Contracts	list executory contracts and Unexpired Leases
Part 1: List All Creditors with PRIORITY Un	secured Claims		
Do any creditors have priority unsecured claims:	? (See 11 U.S.C. § 507).		
☐ No. Go to Part 2.	(11)		
Yes. Go to line 2.			
List in alphabetical order all creditors who have u     3 creditors with priority unsecured claims, fill out and		ity in whole or in part. If the	debtor has more than
		Total claim	Priority amount
.1 Priority creditor's name and mailing address	As of the petition filing date, the claim is: Check all that apply.	\$	\$
	☐ Contingent☐ Unliquidated		
Date or dates debt was incurred	☐ Disputed  Basis for the claim:		
Last 4 digits of account number	Is the claim subject to offset? ☐ No		
Specify Code subsection of PRIORITY unsecured claim: 11 U.S.C. § 507(a) ()	Yes		
.2 Priority creditor's name and mailing address	As of the petition filing date, the claim is: Check all that apply.	\$	\$
	☐ Contingent		
	☐ Unliquidated☐ Disputed		
Date or dates debt was incurred	Basis for the claim:		
Last 4 digits of account number	Is the claim subject to offset? ☐ No		
Specify Code subsection of PRIORITY unsecured claim: 11 U.S.C. § 507(a) ()	☐ Yes		
Priority creditor's name and mailing address	As of the petition filing date, the claim is:  Check all that apply.  Contingent Unliquidated Disputed	\$	\$
Date or dates debt was incurred	Basis for the claim:		
Last 4 digits of account number	Is the claim subject to offset? ☐ No		
Specify Code subsection of PRIORITY unsecured claim: 11 U.S.C. § 507(a) ()	Yes		

Part 1.

Additional Page

	py this page if more space is needed. Continue newious page. If no additional PRIORITY creditors e		Total claim	Priority amount
2	Priority creditor's name and mailing address	As of the petition filing date, the claim is:  Check all that apply.  Contingent  Unliquidated Disputed	\$	\$
	Date or dates debt was incurred	Basis for the claim:		
	Last 4 digits of account number	Is the claim subject to offset? ☐ No ☐ Yes		
2	Priority creditor's name and mailing address	As of the petition filing date, the claim is:  Check all that apply.  Contingent Unliquidated Disputed	\$	\$
	Date or dates debt was incurred	Basis for the claim:		
	Last 4 digits of account number	Is the claim subject to offset?  No Yes		
2	Priority creditor's name and mailing address	As of the petition filing date, the claim is:  Check all that apply.  Contingent Unliquidated Disputed	\$	\$
	Date or dates debt was incurred	Basis for the claim:		
	Last 4 digits of account number	Is the claim subject to offset? ☐ No ☐ Yes		
2	Priority creditor's name and mailing address	As of the petition filing date, the claim is: Check all that apply. Contingent Unliquidated Disputed	\$	\$
	Date or dates debt was incurred	Basis for the claim:		
	Last 4 digits of account number	Is the claim subject to offset? ☐ No ☐ Yes		

Part 2:

#### List All Creditors with NONPRIORITY Unsecured Claims

3.	List in alphabetical order all of the creditors with nonpri- unsecured claims, fill out and attach the Additional Page of I		n 6 creditors with nonpriority
	Ç		Amount of claim
3.1	Nonpriority creditor's name and mailing address	As of the petition filing date, the claim is:  Check all that apply.  Contingent Unliquidated	\$
		Basis for the claim:	
	Date or dates debt was incurred	Is the claim subject to offset?  ☐ No	
	Last 4 digits of account number	☐ Yes	
3.2	Nonpriority creditor's name and mailing address	As of the petition filing date, the claim is:  Check all that apply.  Contingent Unliquidated Disputed	\$
		Basis for the claim:	
	Date or dates debt was incurred	Is the claim subject to offset?	
	Last 4 digits of account number	── No ☐ Yes	
3.3	Nonpriority creditor's name and mailing address	As of the petition filing date, the claim is:  Check all that apply.  Contingent	\$
		☐ Unliquidated ☐ Disputed	
		Basis for the claim:	_
	Date or dates debt was incurred	Is the claim subject to offset?  ☐ No	
	Last 4 digits of account number	☐ Yes	
3.4	Nonpriority creditor's name and mailing address	As of the petition filing date, the claim is: Check all that apply.  Contingent Unliquidated Disputed	\$
		Basis for the claim:	
	Date or dates debt was incurred	Is the claim subject to offset?	
	Last 4 digits of account number	── □ No □ Yes	
3.5	Nonpriority creditor's name and mailing address	As of the petition filing date, the claim is:  Check all that apply.  Contingent Unliquidated Disputed	\$
		Basis for the claim:	
	Date or dates debt was incurred	Is the claim subject to offset?	
	Last 4 digits of account number	──	
3.6	Nonpriority creditor's name and mailing address	As of the petition filing date, the claim is:  Check all that apply.  Contingent Unliquidated Disputed	\$
		Basis for the claim:	
	Date or dates debt was incurred	Is the claim subject to offset? ☐ No	
	Last 4 digits of account number	Yes	

Part 2:

Additional Page

	opy this page only if more space is needed. Continue num evious page. If no additional NONPRIORITY creditors exis		Amount of claim
3	Nonpriority creditor's name and mailing address  Date or dates debt was incurred  Last 4 digits of account number	As of the petition filing date, the claim is:  Check all that apply.  Contingent Unliquidated Disputed Liquidated and neither contingent nor disputed  Basis for the claim:  Is the claim subject to offset?  No Yes	\$
3	Nonpriority creditor's name and mailing address	As of the petition filing date, the claim is:  Check all that apply.  Contingent Unliquidated Disputed	\$
	Date or dates debt was incurred  Last 4 digits of account number	Basis for the claim:  Is the claim subject to offset?  No Yes	_
3	Nonpriority creditor's name and mailing address	As of the petition filing date, the claim is:  Check all that apply.  Contingent Unliquidated Disputed	\$
	Date or dates debt was incurred  Last 4 digits of account number	Basis for the claim:  Is the claim subject to offset?  No Yes	_
3	Nonpriority creditor's name and mailing address	As of the petition filing date, the claim is:  Check all that apply.  Contingent Unliquidated Disputed	\$
	Date or dates debt was incurred  Last 4 digits of account number	Basis for the claim:  Is the claim subject to offset?  No Yes	_
3	Nonpriority creditor's name and mailing address	As of the petition filing date, the claim is:  Check all that apply.  Contingent Unliquidated Disputed	\$
	Date or dates debt was incurred  Last 4 digits of account number	Basis for the claim:  Is the claim subject to offset?  No Yes	_

Name		

Case number (if known)	
------------------------	--

Part 3:

List Others to Be Notified About Unsecured Claims

Name and mailing address		t 4 digits of ount number, i
	Not listed. Explain	
	Not listed. Explain	
	Not listed. Explain	
	Line	
	Not listed. Explain —	
	Line	
	Not listed. Explain —	
	Line	
	Not listed. Explain	
	Line	
	Not listed. Explain	
	Not listed. Explain	
	Line	
	Not listed. Explain —	
	Line	
	Not listed. Explain —	
	Line	
	Not listed. Explain —	
	Line	

Name

Part 3: Additional Page for Others to Be Notified About Unsecured Claims

	Name and mailing address	which line in Part 1 or Part 2 is the ted creditor (if any) listed?	Last 4 digits of account number, if any
4		Not listed. Explain	
4		Not listed. Explain	
4		Not listed. Explain	
4		Not listed. Explain	
4		Not listed. Explain	
4		Not listed. Explain	
4		Not listed. Explain	
4		Not listed. Explain	
4		Not listed. Explain	
4		Not listed. Explain	
4		Not listed. Explain	
4		Not listed. Explain	
4		Not listed. Explain	
4		Not listed. Explain	

Part 4:

Total Amounts of the Priority and Nonpriority Unsecured Claims

5. Add the amounts of priority and nonpriority unsecured claims.

Total of claim amounts

5a. Total claims from Part 1

Name

\$\_\_\_\_\_

5b. Total claims from Part 2

+ \$\_\_\_\_\_

- 5c. Total of Parts 1 and 2
  - Lines 5a + 5b = 5c.

5c.

5b.

5a.

Eill	in this information to identify t	the case:		
			_	
	ed States Bankruptcy Court for the:	(State)		
Case	e number (If known):	Chapter		
Off	icial Form 206G			☐ Check if this is an amended filing
Sc	hedule G: Exec	cutory Contracts and I	Jnexpired Leases	12/15
<b>1.</b>	Does the debtor have any execc  ☐ No. Check this box and file the	cutory contracts or unexpired leases?  his form with the court with the debtor's other sion below even if the contracts or leases are list	chedules. There is nothing else to report on	this form.
2. L	ist all contracts and unexpired	d leases	State the name and mailing address for whom the debtor has an executory contains	
2.1	State what the contract or lease is for and the nature of the debtor's interest			
	State the term remaining			
	List the contract number of any government contract			
2.2	State what the contract or lease is for and the nature of the debtor's interest			
	State the term remaining			
	List the contract number of any government contract			
2.3	State what the contract or lease is for and the nature of the debtor's interest			
	State the term remaining			
	List the contract number of any government contract			
2.4	State what the contract or lease is for and the nature of the debtor's interest			
	State the term remaining			
	List the contract number of any government contract			
2.5	State what the contract or lease is for and the nature of the debtor's interest			
	State the term remaining			
	List the contract number of any government contract			

		ı
		ı
		ı

Additional Page if Debtor Has More Executory Contracts or Unexpired Leases

	Copy this page only if more	space is needed. Continue numbering t	ne lines sequentially from the previous page.
ı	ist all contracts and unexpire	d leases	State the name and mailing address for all other parties with whom the debtor has an executory contract or unexpired lease
2	State what the contract or lease is for and the nature of the debtor's interest		
	State the term remaining List the contract number of any government contract		-
2	State what the contract or lease is for and the nature of the debtor's interest		
	State the term remaining List the contract number of any government contract		
2	State what the contract or lease is for and the nature of the debtor's interest		
	State the term remaining List the contract number of any government contract		-
2	State what the contract or lease is for and the nature of the debtor's interest		
	State the term remaining List the contract number of any government contract		
2	State what the contract or lease is for and the nature of the debtor's interest		
	State the term remaining List the contract number of any government contract		-
2	State what the contract or lease is for and the nature of the debtor's interest		
	State the term remaining List the contract number of any government contract		-
2	State what the contract or lease is for and the nature of the debtor's interest		
	State the term remaining List the contract number of any government contract		

	in this information to i	dentity the case:				
Deb	btor name					
Uni	ited States Bankruptcy Court	for the:	District of			
Cas	se number (If known):			(State)		
						☐ Check if this is an
Off	ficial Form 206	SH				amended filing
Sc	chedule H: C	 Codebtors				12/15
			pace is needed, c	opy the Additional	Page, numbering the entr	
the A	Additional Page to this	page.	, .	.,	3.7	
1.	Does the debtor have a	iny codebtors?				
		and submit this form to the	court with the deb	otor's other schedules	s. Nothing else needs to be	reported on this form.
	☐ Yes					
	In Column 1, list as codebtors all of the people or entities who are also liable for any debts listed by the debtor in the schedules of creditors, Schedules D-G. Include all guarantors and co-obligors. In Column 2, identify the creditor to whom the debt is owed and each schedule on which the creditor is listed. If the codebtor is liable on a debt to more than one creditor, list each creditor separately in Column 2.					
	Column 1: Codebtor				Column 2: Creditor	
	Name	Mailing address			Name	Check all schedules that apply:
2.1						□ D
		Street				□ E/F □ G
					<del></del>	<b>-</b> 0
		City	State	ZIP Code		
2.2		_			<del></del>	D
		Street				□ E/F □ G
		City	State	ZIP Code		
2.3		Street				D
		City	State	ZIP Code		
2.4		City	State	Zii Gode		
		Street				□ D □ E/F
						□ G
		City	State	ZIP Code		
2.5		_				□ D
		Street				 □ E/F □ G
					<del></del>	_ 0
		City	State	ZIP Code		
2.6		Street			<u> </u>	D
		Sueel				□ E/F □ G

178

Official Form 206H Schedule H: Codebtors pag

ZIP Code

State

City

## Additional Page if Debtor Has More Codebtors

	Copy this page only	if more space is nee	eded. Continue num	bering the lines sec	quentially from the previous pa	age.
	Column 1: Codebtor				Column 2: Creditor	
	Name	Mailing address			Name	Check all schedules that apply:
2		Street				□ D
		Street				□ E/F □ G
		City	State	ZIP Code	_	
2						□ D
		Street				□ E/F □ G
		City	State	ZIP Code		
2						□ D
		Street			<del></del>	□ E/F □ G
						_ 3
		City	State	ZIP Code		
2		Street				□ D □ E/F
						□ G
		City	State	ZIP Code		
2						□ D
		Street				□ E/F □ G
		City	State	ZIP Code	<u> </u>	
2						
_		Street			<del></del>	□ D □ E/F
						□ G
		City	State	ZIP Code		
2		Street			<del></del>	□ D
		Street				□ E/F □ G
		City	State	ZIP Code		
2						□ D
		Street				□ E/F □ G
		City	Ctata	7ID Code	<del></del>	
		City	State	ZIP Code		

Fill in this information to identify the case:	
Debtor name	
United States Bankruptcy Court for the: District of (State)	
Case number (If known):	
	☐ Check if this is an amended filing
	amondod ming
Official Form 206Sum	
Summary of Assets and Liabilities for Non-Individuals	12/15
Part 1: Summary of Assets	
Schedule A/B: Assets-Real and Personal Property (Official Form 206A/B)	
1a. Real property:	•
Copy line 88 from Schedule A/B	\$
1b. Total personal property:	\$
Copy line 91A from Schedule A/B	
1c. <b>Total of all property:</b> Copy line 92 from <i>Schedule A/B</i>	\$
Part 2: Summary of Liabilities	
2. Schedule D: Creditors Who Have Claims Secured by Property (Official Form 206D)	\$
Copy the total dollar amount listed in Column A, Amount of claim, from line 3 of Schedule D	
3. Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 206E/F)	
3a. Total claim amounts of priority unsecured claims:	\$
Copy the total claims from Part 1 from line 5a of Schedule E/F	
3b. <b>Total amount of claims of nonpriority amount of unsecured claims:</b> Copy the total of the amount of claims from Part 2 from line 5b of <i>Schedule E/F</i>	+ \$
4. Total liabilities	
Lines 2 + 3a + 3b	\$

#### **COMMITTEE NOTE**

The schedules to be used in cases of non-individual debtors have been revised as part of the Forms Modernization Project, making them easier to read and, as a result, likely to generate more complete and accurate responses. The goals of the Forms Modernization Project include improving the interface between technology and the forms so as to increase efficiency and reduce the need to produce the same information in multiple formats.

The Forms Modernization Project made a preliminary decision that separate forms should be created for individual debtors and for non-individual debtors because separate areas of inquiry apply to each group. The forms for non-individuals eliminate questions that pertain only to individuals and use a more open-ended response Also, where possible, the forms for nonformat. individuals parallel how businesses commonly keep their financial records. The non-individual debtor schedules are also renumbered, starting with the number 206 and followed by the letter or name of the schedule to distinguish them from the versions to be used in individual cases. Each form includes a checkbox to indicate whether it is an amended filing.

Official Form 206Sum, Summary of Assets and Liabilities for Non-Individuals, replaces Official Form 6, Summary of Schedules and Statistical Summary of Certain Liability and Related Data (28 U.S.C. § 159), in cases of non-individual debtors. The form is reformatted and updated with cross-references indicating the line numbers from specific schedules from which the summary information is to be gathered, and the Statistical Summary is deleted because it only applies to individual debtors. In addition, because most filings are now done electronically, the form no longer requires the debtor to indicate which schedules are attached or to state the number of sheets of paper used for the schedules.

Official Form 206A/B, Schedule A/B: Assets – Real and Personal Property, consolidates information about a non-individual debtor's real and personal property into a single form and replaces Official Form 6A - Real Property and Official Form 6B - Personal Property, in cases of non-individual debtors. The layout and categories of property on Official Form 206A/B have changed. Instead of dividing property interests into two categories (real or personal property), the new form uses eleven categories of property types. For each part, the specific items are broken out and debtors are instructed to total the part and list the total on a specific line later in the form.

Part 1: Cash and cash equivalents, includes cash and cash equivalents and a shortened list of examples. All financial assets other than cash or cash equivalents are moved to Part 4: *Investments*. In the section to list checking, savings, money market, or financial brokerage accounts, debtors are instructed to include the name of the institution and the last 4-digits of any account number.

In Part 2: *Deposits and prepayments*, adds prepayments and examples. A requirement has been added to include the name of the holder of any deposit.

Part 3: *Accounts receivable*, has been revised to divide accounts receivable into two categories depending on age and asks for separate values for the two categories.

Part 4: *Investments*, has been expanded and includes more detail.

Part 5: *Inventory, excluding agricultural assets*, has been amended to separate non-agricultural from agricultural assets, and has been expanded to include more detail. Categories of inventory are listed, and debtors must include the last date of physical inventory, the net book value of debtor's interest (if available), the valuation method used for current value, and the current value of debtor's interest. The form has been further amended to

require the debtor to indicate whether the properties listed are perishable, whether any of the property was purchased within 20 days of the bankruptcy filing, and whether any of the property was appraised by a professional within the year prior to the bankruptcy filing.

In Part 6: Farming- and Fishing-Related Assets (other than titled motor vehicles and land), the form has been amended to require more detailed responses and to require the debtor to indicate the net book value of the debtor's interest, the valuation method used for current value, and the current value of debtor's interest. A requirement to list fishing supplies has been added. The form has been further amended to require the debtor to indicate whether the properties listed are perishable, whether any of the property was purchased within 20 days of the bankruptcy filing, whether a depreciation schedule is available for any of the property listed, and whether any of the property was appraised by a professional within the year prior to the bankruptcy filing.

Part 7: Office furniture, fixtures, and equipment; and collectibles, has been amended to combine several categories of assets and to require more detail, including requiring the debtor to indicate the net book value of the debtor's interest, the valuation method used for current value, and the current value of debtor's interest. Examples of collectibles are provided. The form has been further amended to require the debtor to indicate whether a depreciation schedule is available for any property listed and whether any of the property listed was appraised by a professional within the year prior to the bankruptcy filing.

Part 8: *Machinery, equipment, and vehicles*, has been amended to combine several categories of property and to require more detail, including requiring the debtor to indicate the net book value of the debtor's interest, the valuation method used for current value, and the current value of debtor's interest. More examples are provided for each property type. The form has been further amended to

indicate whether a depreciation schedule is available for any property listed and whether any of the property listed was appraised by a professional within the year prior to the bankruptcy filing.

Part 9: *Real property*, includes the elements of Official Form 6A, *Real Property*, and has been amended to expand the required information to include the net book value of the debtor's interest and the valuation method used for current value. Also, an instruction has been added for the description and location of the property. The form has been further amended to indicate whether a depreciation schedule is available for any property listed and whether any of the property listed was appraised by a professional within the year prior to the bankruptcy filing.

Part 10: *Intangibles and intellectual property*, includes amendments to combine several categories of property and to include more property types. The debtor is required to list the net book value of the debtor's interest and the valuation method used for current value. The question regarding personally identifiable information has been revised, and the form has been amended to require the debtor to indicate if there is an amortization schedule or similar schedule available for any property listed and whether any of the property listed was appraised by a professional within the year prior to the bankruptcy filing.

Part 11: All other assets, includes a new category for notes receivable, which requires a description, including the name of the obligor, the face amount, and any uncollectible amount. In addition, the form has been amended to combine tax refunds and net operating losses into a single question and to require more detail, to delete the requirement to list the insurance company name for any interests in insurance policies, to expand the question regarding contingent and unliquidated claims, and to include examples of other property. The form has been further amended to include a question regarding whether

the property listed was appraised by a professional within the year prior to the bankruptcy filing.

Part 12, *Summary*, has been amended to list relevant line numbers for each type of property.

**Official Form 206D**, Schedule D: Creditors Who Have Claims Secured by Property, replaces Official Form 6D, Creditors Holding Secured Claims, for non-individual debtors and has been revised to eliminate instructions that pertain only to individuals. The form has been further amended to instruct debtors that if a creditor has more than one secured claim, to list the creditor separately for each claim; to list the creditor's email address, if known; to indicate if multiple creditors have an interest in the same collateral; to list the order of each creditor's priority interest in the collateral; and to indicate whether the creditor is an insider or related party. The debtor is also instructed to describe the lien and to fill out Schedule H: Codebtors, if anyone else is liable on the claim. Finally, the form has been amended to require the debtor to list the value of the debtor's property that secures the claim.

A new Part 2: List Others to be Notified for a Debt Already Listed in Part 1 has been added, with instructions to list any others who must be notified about the bankruptcy for a debt listed in Part 1 of the form. Examples are provided. The debtor must include the relevant line from Part 1 and the last 4 digits of the account number for the entity.

A new Part 3: *Total Amounts of Claims and the Unsecured Portion of Claims*, has been added.

Official Form 206E/F, Schedule E/F: Creditors Who Have Unsecured Claims, has been amended to combine Official Form 6E, Schedule E – Creditors Holding Unsecured Priority Claims and Official Form 6F, Schedule F – Creditors Holding Unsecured Nonpriority Claims for non-individual debtors. Priority unsecured claims are listed

in Part 1, and nonpriority unsecured claims are listed in Part 2. The instructions have been revised to require the debtor to list the other party to any executory contract or unexpired lease on this schedule and on *Schedule A/B Real and Personal Property* and *Schedule G: Executory Contracts and Unexpired Leases* (Official Forms 206A/B and 206G).

Part 1. List All Creditors with PRIORITY Unsecured Claims, has been revised to delete the requirement to list the amount not entitled to priority and to add requirements to specify the Code section for the priority unsecured claim and whether the claim is subject to offset. requirement was added to Part 2, List All Creditor's with NONPRIORITY Unsecured Claims, to indicate if the claim is subject to offset. The instructions have also been significantly shortened. Part 3, List Others to be Notified About Unsecured Claims, has been added, with instructions to list any others who must be notified for claims listed in Parts 1 and 2. Examples are given. The debtor must include the relevant line from Part 1 or 2 and the last 4 digits of the account number for the entity. A new Part 4: Total Amounts of the Priority and Nonpriority Unsecured Claims has been added.

Official Form 206G, Schedule G: Executory Contracts and Unexpired Leases, replaces Official Form 6G - Executory Contracts and Unexpired Leases for non-individual debtors. The form has been amended to delete the instruction regarding the listing of a minor child's name from the form as a caution is included in the general instructions for all forms regarding listing a minor child's name. A new requirement has been added to state the remaining term for any contract or lease listed.

**Official Form 206H**, *Schedule H: Codebtors*, replaces Official Form 6H – *Codebtors* for non-individual debtors. The form has been amended to delete the instruction regarding the listing of a minor child's name from the form as a caution is included in the general

instructions for all forms regarding listing a minor child's name. A new requirement is added to indicate by checkbox what schedule applies to each codebtor.

Schedules C, Exemptions, I, Income and J, Expenses. There are no Official Forms for Schedules C, I, and J in non-individual debtor cases. There is no need for an Official Form 206C for non-individual debtors because exemptions are inapplicable to non-individual debtors. And, although section 521(a) of the Bankruptcy Code requires all debtors, including non-individual debtors, to provide schedules of income and expenses, uncertainty about the state of the debtor's business on the petition date - whether it is operating or not, for example - makes it difficult to create standard income and expense forms for non-individual debtors. Some bankruptcy courts have adopted local rules and forms for reporting the income and expenses of non-individual debtors, and Director's Procedural Forms 2060I and 2060J, can be used and modified as appropriate if there are no applicable local rules and forms.

**Declaration.** There is no Official Form 206, Declaration. The portion of Official Form 6 Declaration for a declaration on behalf of a corporation or partnership has been replaced by Official Form 202, *Declaration Under Penalty of Perjury for Non-Individual Debtors.* Official Form 202 includes checkboxes for the schedules included in Official Form 206.

Fill in this information to identify the case:	
Debtor name	
United States Bankruptcy Court for the:	District of (State)
Case number (If known):	

lacksquare Check if this is an amended filing

## Official Form 207

#### Statement of Financial Affairs for Non-Individuals Filing for Bankruptcy 12/15

The debtor must answer every question. If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write the debtor's name and case number (if known).

Part	1: Income					
1. <b>Gr</b>	oss revenue from business					
	None					
	Identify the beginning and en may be a calendar year	ding dates of the debtor	's fiscal	l year, which	Sources of revenue Check all that apply	Gross revenue (before deductions and exclusions)
	From the beginning of the fiscal year to filing date:	From	to	Filing date	Operating a business Other	\$
	For prior year:	From MM/DD/YYYY	to	MM / DD / YYYY	Operating a business Other	\$
	For the year before that:	From MM/DD/YYYY	to	MM / DD / YYYY	Operating a business Other	\$
Inc					ne may include interest, dividends, mor rately. Do not include revenue listed in	
					Description of sources of revenue	Gross revenue from each source (before deductions and exclusions)
	From the beginning of the fiscal year to filing date:	From	to	Filing date		\$
	For prior year:	From MM/DD/YYYY	to	MM / DD / YYYY		\$
	For the year before that:	From MM/DD/YYYY	to	MM / DD / YYYY	,	\$

Debtor		Case number (if known)	
	Name	•	

ı	Part 2:	List Certain Transfers Made Before F	iling for Bani	kruptcy	
	List pa days b	in payments or transfers to creditors within 9 ayments or transfers—including expense reimbur before filing this case unless the aggregate value ed on 4/01/16 and every 3 years after that with reference.	rsements—to a	ny creditor, other than regula transferred to that creditor is	less than \$6,225. (This amount may be
	☐ No	one			
		Creditor's name and address	Dates	Total amount or value	Reasons for payment or transfer

	Creditor's name and address		Dates	Total amount or value	Reasons for payment or transfer Check all that apply
3.1.	Creditor's name			\$	Secured debt Unsecured loan repayments
	Street				Suppliers or vendors Services
	City Sta	ate ZIP Code			Other
3.2.	Creditor's name			\$	Secured debt
	Street				<ul><li>☐ Unsecured loan repayments</li><li>☐ Suppliers or vendors</li></ul>
	City Sta	ate ZIP Code			Services Other

#### 4. Payments or other transfers of property made within 1 year before filing this case that benefited any insider

List payments or transfers, including expense reimbursements, made within 1 year before filing this case on debts owed to an insider or guaranteed or cosigned by an insider unless the aggregate value of all property transferred to or for the benefit of the insider is less than \$6,225. (This amount may be adjusted on 4/01/16 and every 3 years after that with respect to cases filed on or after the date of adjustment.) Do not include any payments listed in line 3. *Insiders* include officers, directors, and anyone in control of a corporate debtor and their relatives; general partners of a partnership debtor and their relatives; affiliates of the debtor and insiders of such affiliates; and any managing agent of the debtor. 11 U.S.C. § 101(31).

■ None	
--------	--

	ame and address		Dates	Total amount or value	Reasons for payment or transfer
4.1. Insider's nam	e			\$	
Street					
City	Stat	e ZIP Code			
Relationsh	ip to debtor				
1.2.					
Insider's nam	е			\$	
Street					
City	Stat	e ZIP Code			
Relationsh	ip to debtor				

	oossessions, foreclosu				la de la composición del composición de la composición del composición de la composi		al lace a consultation
					before filing this case, include turned to the seller. Do not		
_		ansieneu	by a deed in he	ed of foreclosure, or i	eturned to the selier. Do not	include property listed in	iiile o.
Ч	None						
	Creditor's name and add	ress		Description of the	property	Date	Value of property
5.1.							
	Creditor's name						\$
						_	
	Street						
	-			-		_	
	City	State	ZIP Code				
5.2.							
	Creditor's name					_	_ \$
	Creditor's name						
	Street					_	
						_	
	City	State	ZIP Code				
	City	State	ZIF Code				
Set	offs						
_	None  Creditor's name and ad	Idress		Description of th	e action creditor took	Date action was taken	Amount
						tuncii	
						tuken	<b>C</b>
	Creditor's name						\$
							\$
	Creditor's name						\$
				Last 4 digits of ac	count number: XXXX		\$
		State	ZIP Code	Last 4 digits of ac	count number: XXXX		\$
rt	Street			Last 4 digits of ac	count number: XXXX		\$
	Street  City  3: Legal Actions or	r Assignn	ments				\$
Leç	City  3: Legal Actions or gal actions, administrate	r Assignn	nents	actions, executions	, attachments, or governme	ental audits	,—————————————————————————————————————
Leç List	City  3: Legal Actions or gal actions, administrate	r Assignn tive proces edings, inv	nents edings, court estigations, arl	actions, executions		ental audits	,—————————————————————————————————————
L <b>eg</b> List	City  3: Legal Actions or gal actions, administrate the legal actions, process involved in any capacity	r Assignn tive proces edings, inv	nents edings, court estigations, arl	actions, executions	, attachments, or governme	ental audits	,—————————————————————————————————————
L <b>eg</b> List	City  3: Legal Actions or gal actions, administrate the legal actions, process involved in any capacity.  None	r Assignn tive proces edings, inv	nents edings, court estigations, art year before fil	actions, executions bitrations, mediations ling this case.	, attachments, or governments, and audits by federal or sta	ental audits te agencies in which the	debtor
Leg List was	City  3: Legal Actions or gal actions, administrate the legal actions, process involved in any capacity	r Assignn tive proces edings, inv	nents edings, court estigations, art year before fil	actions, executions	, attachments, or governme	ental audits te agencies in which the	debtor Status of case
Leg List was	City  3: Legal Actions or gal actions, administrate the legal actions, process involved in any capacity.  None	r Assignn tive proces edings, inv	nents edings, court estigations, art year before fil	actions, executions bitrations, mediations ling this case.	, attachments, or governments, and audits by federal or sta	ental audits te agencies in which the	debtor  Status of case  Pending
Leg List was	City  3: Legal Actions or gal actions, administrate the legal actions, process involved in any capacity.  None	r Assignn tive proces edings, inv	nents edings, court estigations, art year before fil	actions, executions bitrations, mediations ling this case.	, attachments, or governments, and audits by federal or sta	ental audits te agencies in which the	debtor Status of case
Leg List was	City  3: Legal Actions or gal actions, administrate the legal actions, process involved in any capacity.  None	r Assignn tive proces edings, inv	nents edings, court estigations, art year before fil	actions, executions bitrations, mediations ling this case.	, attachments, or governments, and audits by federal or sta	ental audits te agencies in which the	debtor  Status of case  Pending
Leg List was	City  3: Legal Actions or gal actions, administrate the legal actions, process involved in any capacity.  None  Case title	r Assignn tive proces edings, inv	nents edings, court estigations, art year before fil	actions, executions bitrations, mediations ling this case.	, attachments, or governments, and audits by federal or sta	ental audits te agencies in which the	debtor  Status of case Pending On appeal
Leg List was	City  3: Legal Actions or gal actions, administrate the legal actions, process involved in any capacity.  None  Case title	r Assignn tive proces edings, inv	nents edings, court estigations, art year before fil	actions, executions bitrations, mediations ling this case.	court or agency's na	ental audits te agencies in which the	debtor  Status of case Pending On appeal Concluded
Leg List was	City  3: Legal Actions or gal actions, administrate the legal actions, process involved in any capacity.  None  Case title	r Assignn tive proces edings, inv	nents edings, court estigations, art year before fil	actions, executions bitrations, mediations ling this case.	court or agency's na	ental audits te agencies in which the	debtor  Status of case Pending On appeal Concluded
Leg List was	City  3: Legal Actions or gal actions, administrate the legal actions, process involved in any capacity.  None  Case title	r Assignn tive proces edings, inv	nents edings, court estigations, art year before fil	actions, executions bitrations, mediations ling this case.	court or agency's na	ental audits te agencies in which the ame and address	debtor  Status of case Pending On appeal Concluded
Leg List was	City  3: Legal Actions or gal actions, administrate the legal actions, process involved in any capacity.  None  Case title  Case number	r Assignn tive proces edings, inv	nents edings, court estigations, art year before fil	actions, executions bitrations, mediations ling this case.	court or agency's na  Name Street City	ental audits te agencies in which the ame and address	debtor  Status of case Pending On appeal Concluded Pending
Leg List was	City  3: Legal Actions or gal actions, administrate the legal actions, process involved in any capacity.  None  Case title  Case number	r Assignn tive proces edings, inv	nents edings, court estigations, art year before fil	actions, executions bitrations, mediations ling this case.	, attachments, or government, and audits by federal or state of the court or agency's nature of the court of the	ental audits te agencies in which the ame and address	debtor  Status of case Pending On appeal Concluded Pending On appeal
Leg List was	City  3: Legal Actions or gal actions, administrate the legal actions, process involved in any capacity.  None  Case title  Case number	r Assignn tive proces edings, inv	nents edings, court estigations, art year before fil	actions, executions bitrations, mediations ling this case.	court or agency's na  Name Street City	ental audits te agencies in which the ame and address	debtor  Status of case Pending On appeal Concluded Pending
Leg List was	City  3: Legal Actions or gal actions, administrate the legal actions, process involved in any capacity.  None  Case title  Case title	r Assignn tive proces edings, inv	nents edings, court estigations, art year before fil	actions, executions bitrations, mediations ling this case.	, attachments, or government, and audits by federal or state of the court or agency's nature of the court of the	ental audits te agencies in which the ame and address	debtor  Status of case Pending On appeal Concluded Pending On appeal
List was	City  3: Legal Actions or gal actions, administrate the legal actions, process involved in any capacity.  None  Case title  Case title	r Assignn tive proces edings, inv	nents edings, court estigations, art year before fil	actions, executions bitrations, mediations ling this case.	, attachments, or governments, and audits by federal or state of the court or agency's nature of the court of th	ental audits te agencies in which the ame and address	debtor  Status of case Pending On appeal Concluded Pending On appeal

Debtor	Name	Case number (if known)_		
Li ha	ands of a receiver, custodian, or other court-appointed	penefit of creditors during the 120 days before filing this ed officer within 1 year before filing this case.	s case and any prope	erty in the
_	None			
	Custodian's name and address	Description of the property Value		
		\$		
	Custodian's name	Case title Court	name and address	
	Street			
		Name		
	City State ZIP Code	Case number Street		
	•	Sueet		
		Date of order or assignment		
		City	State	ZIP Code
Part	4: Certain Gifts and Charitable Contribu	ıtions		
9.	f the gifts to that recipient is less than \$1,000  None Recipient's name and address  1. Recipient's name Street  City State ZIP Code  Recipient's relationship to debtor  2. Recipient's name Street  City State ZIP Code	Description of the gifts or contributions	Dates given	<b>Value</b> \$
Part	5: Certain Losses			
10. <b>A</b>	Il losses from fire, theft, or other casualty within	1 year before filing this case.		
	None			
	Description of the property lost and how the loss occurred	Amount of payments received for the loss If you have received payments to cover the loss, for example, from insurance, government compensation, or tort liability, list the total received.  List unpaid claims on Official Form 106A/B (Schedule A/B: Assets – Real and Personal Property).	Date of loss	Value of property lost

п	_	L	٠.	

Case number (if known)
------------------------

Part 6:	Certain	Payments	or	Transfer
rait 0.	Certain	rayments	OI	Transier:

Name

List the t		erty made by the debtor or person acting on behalf of the		
	None			
	Who was paid or who received the transfer?	If not money, describe any property transferred	Dates	Total amount or value
11.1.				
	Address			\$
	Street			
	City State ZIP Code			
	Email or website address			
	Who made the payment, if not debtor?			
	Who was paid or who received the transfer?	If not money, describe any property transferred	Dates	Total amount or value
11.2.				\$
	Address			·
	Street			
	City State ZIP Code			
	Email or website address			
	Who made the payment, if not debtor?			
List a se	-settled trusts of which the debtor is a benefici any payments or transfers of property made by the If-settled trust or similar device. not include transfers already listed on this stateme	e debtor or a person acting on behalf of the debtor within	10 years before the	e filing of this case to
	None	nt.		
	Name of trust or device	Describe any property transferred	Dates transfers were made	Total amount or value
				\$
	Trustee			

btor		Case number (if kno	wn)
	Name		
	nsfers not already listed on this statement	e, trade, or any other means—made by the debtor o	or a porson acting on behalf of the debter
with	in 2 years before the filing of this case to another	person, other than property transferred in the ordin- security. Do not include gifts or transfers previously	ary course of business or financial affairs.
	None		
	Who received transfer?	Description of property transferred or payments recordebts paid in exchange	ceived Date transfer Total amount or was made value
13.1.			<b></b> \$
	Address	l	
	Street		
	City State ZIP Code		
	Relationship to debtor		
	Who received transfer?		\$
13.2.			
	Address		
	Street		
	City State ZIP Code		
	Relationship to debtor		
art 7	: Previous Locations		
	vious addresses	years before filing this same and the dates the add	Ironno ware ward
	Does not apply	years before filing this case and the dates the add	resses were used.
	Address	Da	ates of occupancy
14.1.		Fr	om To
	Street		
	City	ate ZIP Code	
4.2.	Street	Fr	om To
	City St	ate ZIP Code	

otor	Name			Case number (if known)	
art 8		Care Bankru	ptcies		
ls th	diagnosing or t	rily engaged in reating injury, o	leformity, or dise	and facilities for: ase, or nent, or obstetric care?	
	No. Go to Part Yes. Fill in the	9. information bel	OW.		
	Facility name	and address		Nature of the business operation, including type of services the debtor provides	If debtor provides meals and housing, number of patients in debtor's care
15.1.	Facility name				
	Street			<b>Location where patient records are maintained</b> (if different from facility address). If electronic, identify any service provider.	How are records kept?
	City	State	ZIP Code		Check all that apply:  □ Electronically □ Paper
	Facility name	and address		Nature of the business operation, including type of services the debtor provides	If debtor provides meal and housing, number of patients in debtor's care
5.2.	Facility name				
	Street			<b>Location where patient records are maintained</b> (if different from facility address). If electronic, identify any service provider.	How are records kept?
	City	State	ZIP Code		Check all that apply:  □ Electronically □ Paper
ırt 9		_	ole Informatio		
Doe		ollect and reta	in personally ic	lentifiable information of customers?	

☐ Yes
years before filing this case, have any employees of the debtor been participants in any ERISA, 401(k), 403(b), or other or profit-sharing plan made available by the debtor as an employee benefit?

☐ No. Go to Part 10.

☐ No

Yes. Does the debtor serve as plan administrator?

☐ No. Go to Part 10.

☐ Yes. Fill in below: Name of plan

 EIN:

Has the plan been terminated?

☐ Yes

Employer identification number of the plan

8.1. Na Str Git 8.2. Na Str Cit	inancial institution name and address  lame  citreet  lame  lame  citreet  lame  citreet	Last 4 digits of account number  XXXX	Type of account  Checking Savings Money marke Brokerage Other Checking Savings Money marke	closed, sold, moved or transferred	Last balance before closing or transfer  \$\$
Str Str Cit	State ZIP Code		Savings  Money market Brokerage Other Checking Savings Money market		
Str Str Cit	State ZIP Code		Savings  Money market Brokerage Other Checking Savings Money market		
Cit  Str  Cit	State ZIP Code	xxxx	Brokerage Other Checking Savings Money market		\$
.2. Na Str Cit	lame street	xxxx	☐ Other ☐ Checking ☐ Savings ☐ Money market		\$
.2. Na Str Cit	lame street	xxxx	☐ Checking ☐ Savings ☐ Money marke		\$
Na Str Cit	itreet	XXXX	☐ Savings ☐ Money marke	et	\$
Na Str Cit	itreet		☐ Savings ☐ Money marke	et	\$ <u></u>
Cit				et	
Safe de	Sity State ZIP Code		☐ Brokerage		
Safe de	State ZIP Code				
			Other		
☐ Non	y safe deposit box or other depository for ne Depository institution name and address	securities, cash, or other values  Names of anyone with acce		ow has or did have within 1 yea	Does debte still have it
Na	lame				☐ Yes
Str	treet				
_		Address			
Cit	State ZIP Code	7.44.555			
ist any	mises storage	ıses within 1 year before filinឲຸ	g this case. Do not i	include facilities that are in a pa	art of a building in
	Facility name and address	Names of anyone with acce	ss to it Desc	cription of the contents	Does debt
	name and address	or anyone with doce	Desc.	onputon of the contents	still have it

City

State

ZIP Code

Address

btor Name		Case number (if known)	
Part 11: Property the Debtor Hold	ls or Controls That the Debtor Does No	ot Own	
1. Property held for another	or controls that another entity owns. Include any		for, or held in
☐ None			
Owner's name and address	Location of the property	Description of the property	Value
Name			\$
Street		<del>-</del>	
City State Z	P Code		
art 12: Details About Environme	ntal Information		
or a similarly harmful substance.	at an environmental law defines as hazardous or dings known, regardless of when they occur dicial or administrative proceeding under an	rred.  y environmental law? Include settlem	nents and orders.
Case title	Court or agency name and address	Nature of the case	Status of case  Pending
Case number	Name	_	On appeal
	Street		Concluded
	City State ZIP Code	_	
	City State ZIP Code	•	
B. Has any governmental unit otherwise environmental law?  No Yes. Provide details below.	notified the debtor that the debtor may be li	iable or potentially liable under or in	violation of an
Site name and address	Governmental unit name and address	Environmental law, if known	Date of notice
Name	Name		
Street	Street	_	
0001	Succe		

City

State

ZIP Code

City

ZIP Code

State

		ed any gov	vernmental	unit of any release of hazardous mater	ial?	
	No Yes. Provide deta	ile bolow				
•						
	Site name and add	dress		Governmental unit name and address	Environmental law, if known	Date of notice
	Name			Name		
	Street			Street		
	City	State	ZIP Code	City State ZIP Code	 e	
	- 7			,		
1	3: Details A	hout the	Dobtoric F	Business or Connections to Any B	ucinocc	
Ц	3: Details A	bout the	Debtor's E	Business of Connections to Any Bi	usiness	
	any business for vude this information			n owner, partner, member, or otherwise a in the Schedules.	person in control within 6 years be	efore filing this case.
	None		·			
	Business name a	nd address		Describe the nature of the business	Employer Identificati	ion number Security number or ITIN.
5.1.					Do not include occidi	decurity number of frills.
					EIN:	-
	Name					
	Name Street				EIN:	ted
					EIN:	ted
		State	ZIP Code		EIN:	ted
	Street			Describe the nature of the business	EIN:  Dates business exist  From	ted
	Street			Describe the nature of the business	EIN:  Dates business exist  From  Employer Identificati	ted
	Street  City  Business name a			Describe the nature of the business	EIN:  Dates business exist  From  Employer Identificati	ted  To
	Street  City  Business name a			Describe the nature of the business	EIN:  Dates business exist  From  Employer Identificati  Do not include Social	ted  To  Son number  Security number or ITIN.
	Street  City  Business name a			Describe the nature of the business	EIN:	ted  To  Son number Security number or ITIN.
	Street  City  Business name a  Name  Street	nd address		Describe the nature of the business	EIN:	ted  To  Son number  Security number or ITIN.
	Street  City  Business name a			Describe the nature of the business	EIN:	ted  To  Son number Security number or ITIN.
	Street  City  Business name a  Name  Street	nd address	ZIP Code	Describe the nature of the business  Describe the nature of the business	EIN:  Dates business exist  From  Employer Identificati Do not include Social :  EIN:  Dates business exist  From  Employer Identificati	ted  To  fon number Security number or ITIN.  ted  To  fon number
	Street  City  Business name at Name Street  City	nd address	ZIP Code		EIN:  Dates business exist  From  Employer Identificati Do not include Social:  EIN:  Dates business exist  From  Employer Identificati Do not include Social:	ion number Security number or ITIN.  Ion number Security number or ITIN.
	Street  City  Business name at Name Street  City	nd address	ZIP Code		EIN:  Dates business exist  From  Employer Identificati Do not include Social and a second and	ion number Security number or ITIN.  To  ion number Security number or ITIN.
	Street  City  Business name a  Name  Street  City  Business name a	nd address	ZIP Code		EIN:  Dates business exist  From  Employer Identificati Do not include Social:  EIN:  Dates business exist  From  Employer Identificati Do not include Social:	ion number Security number or ITIN.  To  To  To  To  To  To number Security number or ITIN.
	Street  City  Business name and Street  City  Business name and Street	nd address	ZIP Code		EIN:  Dates business exist  From  Employer Identificati Do not include Social:  EIN:  Dates business exist  From  Employer Identificati Do not include Social:  EIN:  Dates business exist	ion number Security number or ITIN.  To  ion number Security number or ITIN.

Debtor			Case number (if known)
	Name		

None					
Name and address				Dates of service	
				From	To
Name				-	
Street				-	
City		State	ZIP Code	-	
Name and address				Dates of service	
				From	То
Name				-	
Street				-	
				- -	
City		State	ZIP Code		
	uals who have audited, compars before filing this case.	piled, or reviewed de	ebtor's books of accou	int and records or pre	pared a financial
Name and addres	5			Dates of service	
6b.1.	<del> </del>			From	То
Name					
Street					
City		State	ZIP Code		
Name and addres	s			Dates of service	
Sb.2. Name				From	То
Street					
City		State	ZIP Code		
. List all firms or individ	uals who were in possession	of the debtor's boo	ks of account and rec	ords when this case is	filed.
None	,				
Name and addres	s			If any books of accunavailable, explain	count and records are in why
6c.1.				·	
6c.1. Name				<del></del>	
6c.1.					
6c.1. Name		State	ZIP Code		

Debtor	_					Case numb	er (if known)
	N	lame					
		Name and address					If any books of account and records are unavailable, explain why
26	c.2.						
20	U.Z.	Name					<del></del>
							<del></del>
		Street					
		City	!	State	ZIP	Code	
		all financial institutions, creditons all financial institutions, creditors are all financial filling this case.		ncluding mercan	tile and	trade agend	sies, to whom the debtor issued a financial statement
	<b></b>	None					
		Name and address					
26	d.1.						
20	u. i.	Name					
		Street					
		Sileet					
		City		State	ZIP	Code	
		Name and address					
26	d.2.	Name					
		Street					
		City	;	State	ZIP	Code	
27. <b>Inve</b>	nto vi						
				:- 2 b .f	الديد الله	i0	
		inventories of the debtor's pr	operty been taken with	in 2 years before	illing tr	ils case?	
		Give the details about the two		_			
	res. (	Sive the details about the two	most recent inventorie	S.			
	Nam	e of the person who supervised	d the taking of the inven	tory		Date of	The dollar amount and basis (cost, market, or
						inventory	other basis) of each inventory
							\$
							<b>-</b>
	Nam	e and address of the person wi	ho has possession of in	ventory records			
27.1.							
	Name	2					
	Stree	t					
	City			State ZIP	Code		
	Oily		,	olale ZIP	Coue		

or	Name		_	Case number	(if known)			
	Name of the person who supervis	ed the taking of the inventory		Date of	The dolla	ır amount aı	nd basis (co	ost, market, or
	·			inventory		sis) of each		,
					\$			
	Name and address of the person v	who has possession of inventory	records					
		,						
7.2.								
	Name							
	Street							
	City	State	ZIP Code					
	·							
	the debtor's officers, directors, ble in control of the debtor at th			ers in contr	ol, control	ing sharel	nolders, o	r other
	Name	Address		Posit	ion and nat	ure of any	%	of interest, if an
	Hamo	Addiooo		intere		and on any	70	or mitoroot, ir un
of th					members,	general pa		embers in cor
of th	e debtor, or shareholders in co			positions?	members, tion and nat		Period d	embers in cor uring which or interest was
of th	e debtor, or shareholders in co No 'es. Identify below.	ontrol of the debtor who no lo		positions?	tion and nat		Period d position held	uring which
of th	e debtor, or shareholders in co No 'es. Identify below.	ontrol of the debtor who no lo		positions?	tion and nat		Period d position held From	uring which or interest was To
of th	e debtor, or shareholders in co No 'es. Identify below.	ontrol of the debtor who no lo		positions?	tion and nat		Period d position held From	uring which or interest was To
of th	e debtor, or shareholders in co No 'es. Identify below.	ontrol of the debtor who no lo		positions?	tion and nat		Period d position held From From	uring which or interest was To To
of th	e debtor, or shareholders in co No 'es. Identify below.	ontrol of the debtor who no lo		positions?	tion and nat		Period d position held From From	uring which or interest was To
of th	e debtor, or shareholders in co No 'es. Identify below.	Address	nger hold these	positions?	tion and nat		Period d position held From From	uring which or interest was To To
of th	ne debtor, or shareholders in consolo Ves. Identify below.  Name	Address  awals credited or given to insi	iders	Positions?  Positions is any i	tion and nat	ure of	Period d position held  From  From  From  From	uring which or interest was To To To To
of the	ne debtor, or shareholders in consolo Yes. Identify below.  Name  ments, distributions, or withdra	Address  Address  awals credited or given to insidit the debtor provide an insider	iders with value in any	Positions?  Positions is any i	tion and nat	ure of	Period d position held  From  From  From  From	uring which or interest was To To To To
Paym Withi bonu	ne debtor, or shareholders in color look of the color of	Address  Address  awals credited or given to insidit the debtor provide an insider	iders with value in any	Positions?  Positions is any i	tion and nat	ure of	Period d position held  From  From  From  From	uring which or interest was To To To To
Paym Withi bonu	ne debtor, or shareholders in collowing the debtor, or shareholders in collowing the c	Address  Address  awals credited or given to insidit the debtor provide an insider	iders with value in any	Positions?  Positions is any i	tion and nat	ure of	Period d position held  From  From  From  From	uring which or interest was To To To To
of the	ne debtor, or shareholders in color look of the color of	Address  Address  awals credited or given to insidit the debtor provide an insider	iders with value in any ercised?	positions?  Posiany i  any i  form, including	tion and nat nterest	ure of	Period d position held  From  From  From  ensation, of	uring which or interest was To To To To
Paym Withi bonu	ne debtor, or shareholders in collowing the debtor, or shareholders in collowing the shareholders in collowing the shareholders in collowing the shareholders in 1 year before filing this case, do uses, loans, credits on loans, stockly fes. Identify below.	Address  Address  awals credited or given to insidit the debtor provide an insider	iders with value in any ercised?  Amount descript	positions?  Posiany i  any i  form, including	tion and nat nterest	ure of	Period d position held  From  From  From  ensation, of	uring which or interest was To To To To
Paym Withibonu	ments, distributions, or withdra in 1 year before filing this case, dises, loans, credits on loans, stocklo 'es. Identify below.  Name and address of recipient	Address  Address  awals credited or given to insidit the debtor provide an insider	iders with value in any ercised?  Amount descript	positions?  Posiany i  any i  form, including	tion and nat nterest	ure of	Period d position held  From  From  From  ensation, of	uring which or interest was To To To To  To
Paym Withibonu	ne debtor, or shareholders in collowing the debtor, or shareholders in collowing the shareholders in collowing the shareholders in collowing the shareholders in 1 year before filing this case, do uses, loans, credits on loans, stockly fes. Identify below.	Address  Address  awals credited or given to insidit the debtor provide an insider	iders with value in any ercised?  Amount descript	positions?  Posiany i  any i  form, including	tion and nat nterest	ure of	Period d position held  From  From  From  ensation, of	uring which or interest was To To To To  To
Paym Withibonu	ments, distributions, or withdra in 1 year before filing this case, dises, loans, credits on loans, stocklo 'es. Identify below.  Name and address of recipient	Address  Address  awals credited or given to insidit the debtor provide an insider	iders with value in any ercised?  Amount descript	positions?  Posiany i  any i  form, including	tion and nat nterest	ure of	Period d position held  From  From  From  ensation, of	uring which or interest was To To To To  To
Paym Withi bonu	ne debtor, or shareholders in color of the debtor, or shareholders in color of the debtor, or shareholders in color of the debtor.  Name  ments, distributions, or withdratin 1 year before filing this case, disses, loans, credits on loans, stockly of the debtor.  Name and address of recipient	Address  Address  awals credited or given to insidit the debtor provide an insider	iders with value in any ercised?  Amount descript	positions?  Posiany i  any i  form, including	tion and nat nterest	ure of	Period d position held  From  From  From  ensation, of	uring which or interest was To To To To  To
Payrr Within bonu	ne debtor, or shareholders in color of the debtor, or shareholders in color of the debtor, or shareholders in color of the debtor.  Name  ments, distributions, or withdratin 1 year before filing this case, disses, loans, credits on loans, stockly of the debtor.  Name and address of recipient	Address  Address  awals credited or given to insidit the debtor provide an insider	iders with value in any ercised?  Amount descript	positions?  Posiany i  any i  form, including	tion and nat nterest	ure of	Period d position held  From  From  From  ensation, of	uring which or interest was To To To To  To

		Case number (if known)
	Name	
П	Name and address of recipient	
	·	
2	Name	<del></del>
	Street	
	City State ZIP Code	<del></del>
	Relationship to debtor	
	in 6 years before filing this case, has the debtor been a memb	per of any consolidated group for tax purposes?
ו וב	lo 'es. Identify below.	
	Name of the parent corporation	Francisco Identification combanation and
	Name of the parent corporation	Employer Identification number of the parent corporation
		EIN:
	Name of the pension fund	Employer Identification number of the pension fund
		EIN:
1	4: Signature and Declaration	
		statement, concealing property, or obtaining money or property by fraud in con
	or imprisonment for up to 20 years, or both. 18 U.S.C. §§ 152, 1341, 1519, and 3571.	
		airs and any attachments and have a reasonable belief that the information is tr
	I declare under penalty of perjury that the foregoing is true and co	rrect.
	Executed onMM / DD / YYYY	
	IVIIVI / DD / IIIII	
X		Participant
•	Signature of individual signing on behalf of the debtor	Printed name
	Position or relationship to debtor	<del></del>
۸	additional pages to Statement of Einancial Affaire for Non-In-	dividuals Filing for Rankruptov (Official Form 207) attached?
Are	additional pages to Statement of Financial Affairs for Non-In No	dividuals Filing for Bankruptcy (Official Form 207) attached?

#### **COMMITTEE NOTE**

Official Form 207, Statement of Financial Affairs for Non-Individuals Filing for Bankruptcy, replaces Official Form 7, Statement of Financial Affairs, for non-individual debtors. It is renumbered to distinguish it from the forms used by individual debtors and includes formatting and stylistic changes throughout the form.

Official Form 207 is revised as part of the Forms Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. The goals of the Forms Modernization Project include improving the interface between technology and the forms so as to increase efficiency and reducing the need to produce the same information in multiple formats.

The Forms Modernization Project made a preliminary decision that separate forms should be created for individual debtors and for non-individual debtors because separate areas of inquiry apply to each group. The forms for non-individuals do not include questions that pertain only to individuals and use a more open-ended response format. Also, where possible, the forms for non-individuals parallel how businesses commonly keep their financial records.

The form is derived from Official Form 7, Statement of Financial Affairs, and has been substantially reorganized. The form is divided into 14 sections grouping similar questions together. Many of the instructions have been shortened, and questions and instructions pertaining to individual debtors have been deleted. The instructions at the beginning of the form have been shortened, and the definitions deleted or moved to other parts of the form.

In Part 1, *Income*, the questions regarding gross revenue from business and non-business revenue have been consolidated, and checkboxes have been added to indicate the source of revenue. Also, the debtor is instructed to include revenue only once.

In Part 2, List Certain Transfers Made Before Filing for Bankruptcy, information that pertains only to individuals has been eliminated, and the questions related to payments made in the 90 days prior to bankruptcy, payments made to insiders within one year prior to bankruptey, repossessions, and setoffs have been Instructions have been added to include consolidated. expense reimbursements in answer to the questions regarding payments and to exclude regular employee compensation from the question regarding payments within 90 days. A dollar limitation has been added to the instructions for the question regarding payments to insiders. Checkboxes have been added to both questions to provide a reason for the payment, and the explanation that the dollar limitation changes every three years has been moved to the instructions from the footnotes. "Amount still owing" has been removed, and a definition of "insider" has been added along with a statutory citation to the question regarding insiders. Partnerships have been added to examples of The question regarding setoffs includes a "insiders." revised definition and has been revised to require that the debtor provide a description of the creditor's actions and the last four digits of any account number.

In Part 3, *Legal Actions or Assignments*, several questions have been consolidated, instructions pertaining only to individuals have been removed, and additional examples have been added. Checkboxes have been added to indicate the status of the legal action. The requirement to list the terms of any assignment or settlement has been removed.

In Part 4, *Certain Gifts and Charitable Contributions*, instructions pertaining only to individuals have been removed, and the reporting threshold has been changed to \$1,000 per recipient. The look-back period has been increased from one to two years.

Part 5, *Certain Losses*, has been revised to expand the types of payments for losses, and an instruction has

been added to list unpaid claims on Official Form 206A/B (Schedule A/B: Assets – Real and Personal Property). Portions of the instructions that pertain only to individuals have been removed. Losses due to gambling have been excluded from this part.

In Part 6, Certain Payments or Transfers, the questions regarding payments related to bankruptcy, payments to self-settled trusts, and other payments or transfers have been consolidated. Instructions and questions that relate only to individuals have been eliminated. An instruction has been added to include payments related to restructuring, and the email or website of the person who received the money or transfer is added as a requirement. In response to the question regarding self-settled trusts and other transfers not already listed, debtors are instructed to include payments or transfers of property made by a person acting on behalf of the debtor. A requirement has been added to the question regarding self-settled trusts to list the name of the trustee. The relationship to the debtor must be included for all transfers not already listed, as well as any debts paid in exchange. There is a reminder added not to include transfers already listed.

Part 7, *Previous Locations*, has been revised in the instructions, and information pertaining only to individuals has been deleted.

Part 8, *Health Care Bankruptcies*, is new. Part 8 requires additional information if the debtor is primarily engaged in offering services and facilities for diagnosing or treating injury, deformity, or disease or providing any surgical, psychiatric, drug treatment, or obstetric care. This part has been added to comply with the special requirements imposed by the Bankruptcy Abuse Prevention and Consumer Protection Act of 2005.

Part 9, *Personally Identifiable Information*, is also new and includes questions about pension and profit sharing plans and adds a question about whether the debtor

collects and retains personally identifiable information of customers. Questions are added about whether the debtor is the plan administrator of any pension or profit sharing plan and if any such plan is terminated. Similar to Part 8, this part has been added to comply with the special requirements imposed by the Bankruptcy Abuse Prevention and Consumer Protection Act of 2005.

In Part 10, Certain Financial Accounts, Safe Deposit Boxes, and Storage Units, money market accounts have been added to the examples provided for the question regarding financial accounts, and checkboxes have been added to indicate the type of account. The requirement of the date of surrender of any safe deposit box has been removed. A question has been added about whether the debtor has property kept in storage units or warehouses within one year of filing, and the debtor must provide the facility name and address, the name and address of anyone with access to the facility, the description of the contents, and whether the debtor still has the storage unit or warehouse. Facilities that are in a part of a building in which the debtor does business are excluded.

In Part 11, *Property the Debtor Holds or Controls That the Debtor Does Not Own*, an instruction has been added to include any property borrowed from, being stored for, or held in trust, and to exclude leased or rented property.

Part 12, *Details About Environmental Information*, has been revised to include new definitions of "Environmental law," "Site," and "Hazardous materials." An instruction to report all notices, releases, and proceedings known, regardless of when they occurred, has been added.

In Part 13, *Details About the Debtor's Business or Connections to Any Business*, questions regarding various business issues have been consolidated, and instructions that pertain only to individuals have been eliminated. The

five-percent ownership limitation has been eliminated. The phrase "kept or supervised the keeping of books or account and records" has been replaced with "maintained the debtor's books and records." The instructions for the question regarding auditing or preparation of financial records have been revised to add compiling and reviewing the debtor's books of account and records. A requirement has been added to explain if the debtor's books of account and records are unavailable. The questions regarding current and former officers, directors, managing members, general partners, members in control, or controlling shareholders have combined the formerly separate corporate and partnership questions. The question regarding former officers and partners has been changed to add the requirement of indicating the start and end dates for The instruction for withdrawals from a each listing. partnership or distribution by a corporation has been changed to add salary, other compensation, and draws to the list of examples.

In Part 14, Signature and Declaration, the declaration under penalty of perjury has been revised in order to conform to the language of 28 U.S.C. § 1746. See Rule 1008. A statement has been added that the individual signing on behalf of the debtor has reviewed the information in the Statement of Financial Affairs and any attachments and has "a reasonable belief that the information is true and correct." The signature boxes for bankruptcy petition preparers have been eliminated, and checkboxes for the debtor to indicate whether additional pages are attached to the form have been added.

Information	to identify the case:				
Debtor 1 _	First Name	Middle Name	Last Name	Last 4 digits of Social Security number or ITIN	
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name	Last 4 digits of Social Security number or ITIN	
	Bankruptcy Court for the:		District of (State)	[Date case filed for chapter 7	MM / DD / YYYY OR
Case number:				[Date case filed in chapter  Date case converted to chapter 7	MM / DD / YYYY

### Official Form 309A (For Individuals or Joint Debtors)

# Notice of Chapter 7 Bankruptcy Case — No Proof of Claim Deadline 12/15

For the debtors listed above, a case has been filed under chapter 7 of the Bankruptcy Code. An order for relief has been entered.

This notice has important information about the case for creditors, debtors, and trustees, including information about the meeting of creditors and deadlines. Read both pages carefully.

The filing of the case imposed an automatic stay against most collection activities. This means that creditors generally may not take action to collect debts from the debtors or the debtors' property. For example, while the stay is in effect, creditors cannot sue, garnish wages, assert a deficiency, repossess property, or otherwise try to collect from the debtors. Creditors cannot demand repayment from debtors by mail, phone, or otherwise. Creditors who violate the stay can be required to pay actual and punitive damages and attorney's fees. Under certain circumstances, the stay may be limited to 30 days or not exist at all, although debtors can ask the court to extend or impose a stay.

The debtors are seeking a discharge. Creditors who assert that the debtors are not entitled to a discharge of any debts or who want to have a particular debt excepted from discharge may be required to file a complaint in the bankruptcy clerk's office within the deadlines specified in this notice. (See line 9 for more information.)

To protect your rights, consult an attorney. All documents filed in the case may be inspected at the bankruptcy clerk's office at the address listed below or through PACER (Public Access to Court Electronic Records at <a href="https://www.pacer.gov">www.pacer.gov</a>).

The staff of the bankruptcy clerk's office cannot give legal advice.

To help creditors correctly identify debtors, debtors submit full Social Security or Individual Taxpayer Identification Numbers, which may appear on a version of this notice. However, the full numbers must not appear on any document filed with the court.

Do not file this notice with any proof of claim or other filing in the case. Do not include more than the last four digits of a Social Security or Individual Taxpayer Identification Number in any document, including attachments, that you file with the court.

		About Debtor 1:	About Debtor 2:	
1.	Debtor's full name			
2.	All other names used in the last 8 years			
3.	Address		If Debtor 2 lives at	a different address:
4.	Debtor's attorney		Contact phone	
••	Name and address		Email	
5.	Bankruptcy trustee		Contact phone	
	Name and address		Email	

De	btor Name	Case	e number (if known)	
6.	Bankruptcy clerk's office Documents in this case may be filed at this address. You may inspect all records filed in this case at this office or online at <a href="https://www.pacer.gov">www.pacer.gov</a> .		Hours open  Contact phone	
7.	Meeting of creditors  Debtors must attend the meeting to be questioned under oath. In a joint case, both spouses must attend.  Creditors may attend, but are not required to do so.	Date Time  The meeting may be continued or adjourned to a later date. If so, the date will be on the court docket.	Location:	
8.	Presumption of abuse If the presumption of abuse arises, you may have the right to file a motion to dismiss the case under 11 U.S.C. § 707(b). Debtors may rebut the presumption by showing special circumstances.	[The presumption of abuse does not arise.] [The presumption of abuse arises.] [Insufficient information has been filed to permit the clerk If more complete information is filed and shows that the process of the complete information is filed and shows that the process of the complete information is filed and shows that the process of the complete information is filed and shows that the process of the complete information is filed and shows that the process of the complete information is filed and shows that the process of the complete information is filed and shows that the process of the complete information is filed and shows that the process of the complete information is filed and shows that the process of the complete information is filed and shows that the process of the complete information is filed and shows that the process of the complete information is filed and shows that the process of the complete information is filed and shows that the process of the complete information is filed and shows that the process of the complete information is filed and shows the complete information is filed and s		•
9.	Deadlines The bankruptcy clerk's office must receive these documents and any required filing fee by the following deadlines.	File by the deadline to object to discharge or to challenge whether certain debts are dischargeable:  You must file a complaint:  if you assert that the debtor is not entitled to receive a discharge of any debts under any of the subdivisions of 11 U.S.C. § 727(a)(2) through (7), or  if you want to have a debt excepted from discharge under 11 U.S.C. § 523(a)(2), (4), or (6).  You must file a motion if you assert that the discharge should be denied under § 727(a)(8) or (9).	Filing deadline:	
		Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.	Filing deadline:	30 days after the <i>conclusion</i> of the meeting of creditors
10.	Proof of claim Please do not file a proof of claim unless you receive a notice to do so.	No property appears to be available to pay creditors. T If it later appears that assets are available to pay credit that you may file a proof of claim and stating the deadli	tors, the clerk will ser	
11.	Creditors with a foreign address	If you are a creditor receiving a notice mailed to a forei extend the deadlines in this notice. Consult an attorney have any questions about your rights in this case.		
12.	Exempt property	The law allows debtors to keep certain property as exe distributed to creditors. Debtors must file a list of prope the bankruptcy clerk's office or online at www.pacer.go exemption that the debtors claim, you may file an objection by the deadline to object to exemptions in line	erty claimed as exem v. If you believe that ction. The bankruptcy	pt. You may inspect that list at the law does not authorize an

Information	to identify the case:				
Debtor 1	First Name	Middle Name	Last Name	Last 4 digits of Social Security number or ITIN	
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name	Last 4 digits of Social Security number or ITIN	
United States	Bankruptcy Court for the: _		District of (State)	[Date case filed for chapter 7	MM / DD / YYYY OR
Case number:				[Date case filed in chapter  Date case converted to chapter 7	MM / DD / YYYY

### Official Form 309B (For Individuals or Joint Debtors)

# Notice of Chapter 7 Bankruptcy Case — Proof of Claim Deadline Set 12/15

For the debtors listed above, a case has been filed under chapter 7 of the Bankruptcy Code. An order for relief has been entered.

This notice has important information about the case for creditors, debtors, and trustees, including information about the meeting of creditors and deadlines. Read both pages carefully.

The filing of the case imposed an automatic stay against most collection activities. This means that creditors generally may not take action to collect debts from the debtors or the debtors' property. For example, while the stay is in effect, creditors cannot sue, garnish wages, assert a deficiency, repossess property, or otherwise try to collect from the debtors. Creditors cannot demand repayment from debtors by mail, phone, or otherwise. Creditors who violate the stay can be required to pay actual and punitive damages and attorney's fees. Under certain circumstances, the stay may be limited to 30 days or not exist at all, although debtors can ask the court to extend or impose a stay.

The debtors are seeking a discharge. Creditors who assert that the debtors are not entitled to a discharge of any debts or who want to have a particular debt excepted from discharge may be required to file a complaint in the bankruptcy clerk's office within the deadlines specified in this notice. (See line 9 for more information.)

To protect your rights, consult an attorney. All documents filed in the case may be inspected at the bankruptcy clerk's office at the address listed below or through PACER (Public Access to Court Electronic Records at <a href="https://www.pacer.gov">www.pacer.gov</a>).

The staff of the bankruptcy clerk's office cannot give legal advice.

To help creditors correctly identify debtors, debtors submit full Social Security or Individual Taxpayer Identification Numbers, which may appear on a version of this notice. However, the full numbers must not appear on any document filed with the court.

Do not file this notice with any proof of claim or other filing in the case. Do not include more than the last four digits of a Social Security or Individual Taxpayer Identification Number in any document, including attachments, that you file with the court.

		About Debtor 1:	About Debtor 2:
1.	Debtor's full name		
2.	All other names used in the last 8 years		
3.	Address		If Debtor 2 lives at a different address:
4.	<b>Debtor's attorney</b> Name and address		Contact phone
5.	Bankruptcy trustee Name and address		Contact phone Email



	Debtor	Case number	PF (if known)
	Name		
6.	Bankruptcy clerk's office Documents in this case may be filed at this address. You may inspect all records filed in this case at this office or online at <a href="https://www.pacer.gov">www.pacer.gov</a> .		Hours open  Contact phone
7.	Meeting of creditors  Debtors must attend the meeting to be questioned under oath. In a joint case, both spouses must attend. Creditors may attend, but are not required to do so.	at	Location:
8.	Presumption of abuse If the presumption of abuse arises, you may have the right to file a motion to dismiss the case under 11 U.S.C. § 707(b). Debtors may rebut the presumption by showing special circumstances.	[The presumption of abuse does not arise.] [The presumption of abuse arises.] [Insufficient information has been filed to permit the clerk to dete more complete information is filed and shows that the presumption	·
9.	Deadlines The bankruptcy clerk's office must receive these documents and any required filing fee by the following deadlines.	File by the deadline to object to discharge or to challenge whether certain debts are dischargeable:  You must file a complaint:  if you assert that the debtor is not entitled to receive a discharge of any debts under any of the subdivisions of 11 U.S.C. § 727(a)(2) through (7), or  if you want to have a debt excepted from discharge under 11 U.S.C. § 523(a)(2), (4), or (6).  You must file a motion:  if you assert that the discharge should be denied under § 727(a)(8) or (9).	Filing deadline:
		Deadline for all creditors to file a proof of claim (except governmental units):  Deadline for governmental units to file a proof of claim:	Filing deadline:
		Deadlines for filing proof of claim:  A proof of claim is a signed statement describing a creditor's cla at <a href="https://www.uscourts.gov">www.uscourts.gov</a> or any bankruptcy clerk's office. If you do not be paid on your claim. To be paid, you must file a proof of claim debtor filed.  Secured creditors retain rights in their collateral regardless of who claim submits the creditor to the jurisdiction of the bankruptcy context of the proof of claim may surrestight to a jury trial.	not file a proof of claim by the deadline, you might aim even if your claim is listed in the schedules that nether they file a proof of claim. Filing a proof of ourt, with consequences a lawyer can explain. For
		Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.	<b>Filing deadline:</b> 30 days after the <i>conclusion</i> of the meeting of creditors
10.	Creditors with a foreign address	If you are a creditor receiving a notice mailed to a foreign address extend the deadlines in this notice. Consult an attorney familiar of questions about your rights in this case.	
11.	Liquidation of the debtor's property and payment of creditors' claims	The bankruptcy trustee listed on the front of this notice will collect If the trustee can collect enough money, creditors may be paid s specified by the Bankruptcy Code. To ensure you receive any shas described above.	ome or all of the debts owed to them in the order nare of that money, you must file a proof of claim
12.	Exempt property	The law allows debtors to keep certain property as exempt. Fully to creditors. Debtors must file a list of property claimed as exempterk's office or online at <a href="https://www.pacer.gov">www.pacer.gov</a> . If you believe that the lebtors claim, you may file an objection. The bankruptcy clerk's to object to exemptions in line 9.	pt. You may inspect that list at the bankruptcy law does not authorize an exemption that the

Information to identify the case:			
Debtor Name		EIN	-
United States Bankruptcy Court for the: Die	strict of(State)	[Date case filed for chapter 7	
Case number:		[Date case filed in chapter	MM / DD / YYYY OR  MM / DD / YYYY
		Date case converted to chapter 7	MM / DD / YYYY

### Official Form 309C (For Corporations or Partnerships)

## Notice of Chapter 7 Bankruptcy Case — No Proof of Claim Deadline 12/15

For the debtor listed above, a case has been filed under chapter 7 of the Bankruptcy Code. An order for relief has been entered.

This notice has important information about the case for creditors, debtors, and trustees, including information about the meeting of creditors and deadlines.

The filing of the case imposed an automatic stay against most collection activities. This means that creditors generally may not take action to collect debts from the debtor or the debtor's property. For example, while the stay is in effect, creditors cannot sue, assert a deficiency, repossess property, or otherwise try to collect from the debtor. Creditors cannot demand repayment from debtors by mail, phone, or otherwise. Creditors who violate the stay can be required to pay actual and punitive damages and attorney's fees.

To protect your rights, consult an attorney. All documents filed in the case may be inspected at the bankruptcy clerk's office at the address listed below or through PACER (Public Access to Court Electronic Records at <a href="https://www.pacer.gov">www.pacer.gov</a>).

The staff of the bankruptcy clerk's office cannot give legal advice.

Do not file this notice with any proof of claim or other filing in the case.

1.	Debtor's full name				
2.	All other names used in the last 8 years				
3.	Address				
4.	Debtor's attorney			Contact phone	
	Name and address			Email	
5.	Bankruptcy trustee			Contact phone	
	Name and address			Email	
6.	Bankruptcy clerk's office Documents in this case may be			Hours open	
	filed at this address.			Contact phone	
	You may inspect all records filed in this case at this office or online at <a href="https://www.pacer.gov">www.pacer.gov</a> .				
7.	Meeting of creditors				
	The debtor's representative must	at Date	Time	Location:	
	attend the meeting to be questioned under oath.	The meeting may be co		d to a later	
	Creditors may attend, but are not required to do so.	date. If so, the date will			
8.	Proof of claim	No property appears to	be available to pay	creditors. Therefore, please do no	t file a proof of claim now.
	Please do not file a proof of claim unless you receive a notice to do so.	If it later appears that as that you may file a proof		o pay creditors, the clerk will send g the deadline.	you another notice telling you
9.	Creditors with a foreign address		this notice. Consult	ed to a foreign address, you may fi an attorney familiar with United St	

Information to identify the case:			
Debtor Name		EIN	-
United States Bankruptcy Court for the:	District of(State)	[Date case filed for chapter 7	
Case number:		[Date case filed in chapter	MM / DD / YYYY OR  MM / DD / YYYY
		Date case converted to chapter 7	MM / DD / YYYY

## Official Form 309D (For Corporations or Partnerships)

# Notice of Chapter 7 Bankruptcy Case — Proof of Claim Deadline Set 12/15

For the debtor listed above, a case has been filed under chapter 7 of the Bankruptcy Code. An order for relief has been entered.

This notice has important information about the case for creditors, debtors, and trustees, including information about the meeting of creditors and deadlines. Read both pages carefully.

The filing of the case imposed an automatic stay against most collection activities. This means that creditors generally may not take action to collect debts from the debtor or the debtor's property. For example, while the stay is in effect, creditors cannot sue, assert a deficiency, repossess property, or otherwise try to collect from the debtor. Creditors cannot demand repayment from debtors by mail, phone, or otherwise. Creditors who violate the stay can be required to pay actual and punitive damages and attorney's fees.

To protect your rights, consult an attorney. All documents filed in the case may be inspected at the bankruptcy clerk's office at the address listed below or through PACER (Public Access to Court Electronic Records at <a href="www.pacer.gov"><u>www.pacer.gov</u></a>).

The staff of the bankruptcy clerk's office cannot give legal advice.

Do not file this notice with any proof of claim or other filing in the case.

1.	Debtor's full name			
2.	All other names used in the last 8 years			
3.	Address			
4.	<b>Debtor's attorney</b> Name and address		Contact phone Email	
5.	Bankruptcy trustee Name and address		Contact phone Email	
6.	Bankruptcy clerk's office Documents in this case may be filed at this address. You may inspect all records filed in this case at this office or online at <a href="https://www.pacer.gov">www.pacer.gov</a> .		Hours open Contact phone	
7.	Meeting of creditors The debtor's representative must attend the meeting to be questioned under oath. Creditors may attend, but are not required to do so.	at Date Time  The meeting may be continued or adjourned to a later date. If so, the date will be on the court docket.	Location:	

Debtor	Case number (if known)	
Denioi	 Case Humber (# known)	

8.	<b>Deadlines</b> The bankruptcy clerk's office must receive proofs of claim by the following deadlines.	Deadline for all creditors to file a proof of claim (except governmental units):  Deadline for governmental units to file a  Filing deadline:  Filing deadline:	
		proof of claim:	
		A proof of claim is a signed statement describing a creditor's claim. A proof of claim form may be obtained at <a href="https://www.uscourts.gov">www.uscourts.gov</a> or any bankruptcy clerk's office. If you do not file a proof of claim by the deadline, you might not be paid on your claim. To be paid, you must file a proof of claim even if your claim is listed in the schedules that the debtor filed.	
		Secured creditors retain rights in their collateral regardless of whether they file a proof of claim. Filing a proof of claim submits the creditor to the jurisdiction of the bankruptcy court, with consequences a lawyer can explain. For example, a secured creditor who files a proof of claim may surrender important nonmonetary rights, including the right to a jury trial.	
9.	Creditors with a foreign address	If you are a creditor receiving a notice mailed to a foreign address, you may file a motion asking the court to extend the deadlines in this notice. Consult an attorney familiar with United States bankruptcy law if you have any questions about your rights in this case.	
10.	Liquidation of the debtor's property and payment of creditors' claims	The bankruptcy trustee listed on the front of this notice will collect and sell the debtor's property. If the trustee can collect enough money, creditors may be paid some or all of the debts owed to them, in the order specified by the Bankruptcy Code. To ensure you receive any share of that money, you must file a proof of claim, as	

described above.

Information to identify the case:					
Debtor 1	First Name	Middle Name	Last Name	Last 4 digits of Social Security number or ITIN	·
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name	Last 4 digits of Social Security number or ITIN	·
(,g)				EIN	
United States	Bankruptcy Court for the:		District of (State)	[Date case filed for chapter 11	MM / DD / YYYY] OR
Case number:			<del></del>	[Date case filed in chapter	MM / DD / YYYY
				Date case converted to chapter 11	MM / DD / YYYY

### Official Form 309E (For Individuals or Joint Debtors)

# Notice of Chapter 11 Bankruptcy Case

12/15

For the debtors listed above, a case has been filed under chapter 11 of the Bankruptcy Code. An order for relief has been entered.

This notice has important information about the case for creditors and debtors, including information about the meeting of creditors and deadlines. Read both pages carefully.

The filing of the case imposed an automatic stay against most collection activities. This means that creditors generally may not take action to collect debts from the debtors or the debtors' property. For example, while the stay is in effect, creditors cannot sue, garnish wages, assert a deficiency, repossess property, or otherwise try to collect from the debtors. Creditors cannot demand repayment from debtors by mail, phone, or otherwise. Creditors who violate the stay can be required to pay actual and punitive damages and attorney's fees. Under certain circumstances, the stay may be limited to 30 days or not exist at all, although debtors can ask the court to extend or impose a stay.

Confirmation of a chapter 11 plan may result in a discharge of debt. Creditors who assert that the debtors are not entitled to a discharge of any debts or who want to have a particular debt excepted from discharge may be required to file a complaint in the bankruptcy clerk's office within the deadlines specified in this notice. (See line 10 below for more information.)

To protect your rights, consult an attorney. All documents filed in the case may be inspected at the bankruptcy clerk's office at the address listed below or through PACER (Public Access to Court Electronic Records at <a href="https://www.pacer.gov">www.pacer.gov</a>).

The staff of the bankruptcy clerk's office cannot give legal advice.

To help creditors correctly identify debtors, debtors submit full Social Security or Individual Taxpayer Identification Numbers, which may appear on a version of this notice. However, the full numbers must not appear on any document filed with the court.

Do not file this notice with any proof of claim or other filing in the case. Do not include more than the last four digits of a Social Security or Individual Taxpayer Identification Number in any document, including attachments, that you file with the court.

		About Debtor 1:	About Debtor 2:
1.	Debtor's full name		
2.	All other names used in the last 8 years		
3.	Address		If Debtor 2 lives at a different address:
4.	Debtor's attorney		Contact phone
	Name and address		Email
5.	Bankruptcy clerk's office Documents in this case may be		Hours open
	filed at this address.		Contact phone
	You may inspect all records filed in this case at this office or online at <a href="https://www.pacer.gov">www.pacer.gov</a> .		

6.	Meeting of creditors  Debtors must attend the meeting to be questioned under oath. In a joint case, both spouses must attend.  Creditors may attend, but are not required to do so.	at	Location:		
7.	Deadlines The bankruptcy clerk's office must receive these documents	File by the deadline to object to discharge or to challenge whether certain debts are dischargeable:	First date set for hearing on confirmation of plan. The court will send you a notice of that date later.		
	and any required filing fee by the following deadlines.	You must file a complaint:  if you assert that the debtor is not entitled to receive a discharge of any debts under 11 U.S.C. § 1141(d)(3) or	Filing deadline for dischargeability complaints:		
		if you want to have a debt excepted from discharge under 11 U.S.C. § 523(a)(2), (4), or (6).			
		Deadline for filing proof of claim:	[Not yet set. If a deadline is set, the court will send you another notice.] or		
			[date, if set by the court)]		
		A proof of claim is a signed statement describing a creditor's claim. A proof of claim form may be obtained at <a href="https://www.uscourts.gov">www.uscourts.gov</a> or any bankruptcy clerk's office.			
		Your claim will be allowed in the amount scheduled unless:			
		<ul> <li>your claim is designated as disputed, contingent, or unliquidated;</li> <li>you file a proof of claim in a different amount; or</li> <li>you receive another notice.</li> </ul>			
		If your claim is not scheduled or if your claim is designated as <i>disputed</i> , <i>contingent</i> , or <i>unliquidated</i> , you must file a proof of claim or you might not be paid on your claim and you might be unable to vote on a plan. You may file a proof of claim even if your claim is scheduled.			
		You may review the schedules at the bankruptcy clerk's office or	online at <u>www.pacer.gov</u> .		
		Secured creditors retain rights in their collateral regardless of whe claim submits a creditor to the jurisdiction of the bankruptcy court example, a secured creditor who files a proof of claim may surrenthe right to a jury trial.	, with consequences a lawyer can explain. For		
		Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt.  If you believe that the law does not authorize an exemption claimed, you may file an objection.	Filing deadline: 30 days after the conclusion of the meeting of creditors		
8.	Creditors with a foreign address	If you are a creditor receiving mailed notice at a foreign address, you may file a motion asking the court to extend the deadlines in this notice. Consult an attorney familiar with United States bankruptcy law if you have any questions about your rights in this case.			
9.	Filing a Chapter 11 bankruptcy case	Chapter 11 allows debtors to reorganize or liquidate according to a plan. A plan is not effective unless the court confirms it. You may receive a copy of the plan and a disclosure statement telling you about the plan, and you may have the opportunity to vote on the plan. You will receive notice of the date of the confirmation hearing, and you may object to confirmation of the plan and attend the confirmation hearing. Unless a trustee is serving, the debtor will remain in possession of the property and may continue to operate the debtor's business.			
10.	Discharge of debts	Confirmation of a chapter 11 plan may result in a discharge of debts, which may include all or part of a debt. See 11 U.S.C. § 1141(d). However, unless the court orders otherwise, the debts will not be discharged until all payments under the plan are made. A discharge means that creditors may never try to collect the debt from the debtors personally except as provided in the plan. If you believe that a particular debt owed to you should be excepted from the discharge under 11 U.S.C. § 523 (a)(2), (4), or (6), you must file a complaint and pay the filing fee in the bankruptcy clerk's office by the deadline. If you believe that the debtors are not entitled to a discharge of any of their debts under 11 U.S.C. § 1141 (d)(3), you must file a complaint and pay the filing fee in the clerk's office by the first date set for the hearing on confirmation of the plan. The court will send you another notice telling you of that date.			
11.	Exempt property	The law allows debtors to keep certain property as exempt. Fully exempt property will not be sold and distributed to creditors, even if the case is converted to chapter 7. Debtors must file a list of property claimed as exempt. You may inspect that list at the bankruptcy clerk's office or online at <a href="https://www.pacer.gov">www.pacer.gov</a> . If you believe that the law does not authorize an exemption that the debtors claim, you may file an objection. The bankruptcy clerk's office must receive the objection by the deadline to object to exemptions in line 7.			

Information to identify the case:			
Debtor Name		EIN	-
United States Bankruptcy Court for the:	District of(State)	[Date case filed for chapter 11	MM / DD / YYYY OR
Case number:		[Date case filed in chapter	MM / DD / YYYY
		Date case converted to chapter 11	MM / DD / YYYY

### Official Form 309F (For Corporations or Partnerships)

# Notice of Chapter 11 Bankruptcy Case

12/15

For the debtor listed above, a case has been filed under chapter 11 of the Bankruptcy Code. An order for relief has been entered.

This notice has important information about the case for creditors, debtors, and trustees, including information about the meeting of creditors and deadlines. Read both pages carefully.

The filing of the case imposed an automatic stay against most collection activities. This means that creditors generally may not take action to collect debts from the debtor or the debtor's property. For example, while the stay is in effect, creditors cannot sue, assert a deficiency, repossess property, or otherwise try to collect from the debtor. Creditors cannot demand repayment from the debtor by mail, phone, or otherwise. Creditors who violate the stay can be required to pay actual and punitive damages and attorney's fees.

Confirmation of a chapter 11 plan may result in a discharge of debt. A creditor who wants to have a particular debt excepted from discharge may be required to file a complaint in the bankruptcy clerk's office within the deadline specified in this notice. (See line 11 below for more information.)

To protect your rights, consult an attorney. All documents filed in the case may be inspected at the bankruptcy clerk's office at the address listed below or through PACER (Public Access to Court Electronic Records at <a href="https://www.pacer.gov">www.pacer.gov</a>).

The staff of the bankruptcy clerk's office cannot give legal advice.

Do not file this notice with any proof of claim or other filing in the case.

1.	Debtor's full name			
2.	All other names used in the last 8 years			
3.	Address			
_	Debtor's attorney		Contact phone	
4.	Name and address		Email	
5.	Bankruptcy clerk's office		Hours open	
	Documents in this case may be		•	
	filed at this address.		Contact phone	
	You may inspect all records filed in this case at this office or			
	online at www.pacer.gov.			
_	Mosting of oraditors			
6.	Meeting of creditors	at	Location:	
	The debtor's representative must attend the meeting to be questioned under oath.	Date Time	Location.	
	Creditors may attend, but are not required to do so.	The meeting may be continued or adjourned to a late date. If so, the date will be on the court docket.	er	

Debtor		Case number (if known	
	Name		

7.	Proof of claim deadline	Deadline for filing proof of claim:	[Not yet set. If a deadline is set, the court will send you another notice.] or
			[date, if set by the court)]
		A proof of claim is a signed statement describing a creditor at <a href="www.uscourts.gov">www.uscourts.gov</a> or any bankruptcy clerk's office.	r's claim. A proof of claim form may be obtained
		Your claim will be allowed in the amount scheduled unless	:
		<ul> <li>your claim is designated as disputed, contingent, or unli</li> <li>you file a proof of claim in a different amount; or</li> <li>you receive another notice.</li> </ul>	iquidated;
		If your claim is not scheduled or if your claim is designated a proof of claim or you might not be paid on your claim and a proof of claim even if your claim is scheduled.	
		You may review the schedules at the bankruptcy clerk's of	fice or online at www.pacer.gov.
		Secured creditors retain rights in their collateral regardless claim submits a creditor to the jurisdiction of the bankrupto example, a secured creditor who files a proof of claim may the right to a jury trial.	y court, with consequences a lawyer can explain. For
8.	Exception to discharge deadline	You must start a judicial proceeding by filing a complaint if discharge under 11 U.S.C. § 1141(d)(6)(A).	you want to have a debt excepted from
	The bankruptcy clerk's office must receive a complaint and any required filing fee by the following deadline.	Deadline for filing the complaint:	
9.	Creditors with a foreign address	If you are a creditor receiving notice mailed to a foreign ad extend the deadlines in this notice. Consult an attorney far any questions about your rights in this case.	· · · · · · · · · · · · · · · · · · ·
10.	Filing a Chapter 11 bankruptcy case	Chapter 11 allows debtors to reorganize or liquidate according to the plan and a disciplant may have the opportunity to vote on the plan. You will receive you may object to confirmation of the plan and attend the odebtor will remain in possession of the property and may of	losure statement telling you about the plan, and you eive notice of the date of the confirmation hearing, and confirmation hearing. Unless a trustee is serving, the
11.	Discharge of debts	Confirmation of a chapter 11 plan may result in a discharge See 11 U.S.C. § 1141(d). A discharge means that creditor except as provided in the plan. If you want to have a particunder 11 U.S.C. § 1141(d)(6)(A), you must start a judicial fee in the bankruptcy clerk's office by the deadline.	s may never try to collect the debt from the debtor cular debt owed to you excepted from the discharge

Information	to identify the case:					
Debtor 1	First Name	Middle Name	Last Name		Last 4 digits of Social Security number or ITIN	
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name		Last 4 digits of Social Security number or ITIN	
United States E	Bankruptcy Court for the: _		District o	(State)	[Date case filed for chapter 12	MM / DD / YYYY OR
Case number:				-	[Date case filed in chapter  Date case converted to chapter 12	MM / DD / YYYY
					Date case semisited to onapter 12	MM / DD / YYYY

## Official Form 309G (For Individuals or Joint Debtors)

## Notice of Chapter 12 Bankruptcy Case

12/15

For the debtors listed above, a case has been filed under chapter 12 of the Bankruptcy Code. An order for relief has been entered.

This notice has important information about the case for creditors, debtors, and trustees, including information about the meeting of creditors and deadlines. Read both pages carefully.

The filing of the case imposed an automatic stay against most collection activities. This means that creditors generally may not take action to collect debts from the debtors, from the debtors' property, or from certain codebtors. For example, while the stay is in effect, creditors cannot sue, garnish wages, assert a deficiency, repossess property, or otherwise try to collect from the debtors. Creditors cannot demand repayment from debtors by mail, phone, or otherwise. Creditors who violate the stay can be required to pay actual and punitive damages and attorney's fees.

Confirmation of a chapter 12 plan may result in a discharge of debt. Creditors who want to have a particular debt excepted from discharge may be required to file a complaint in the bankruptcy clerk's office within the deadline specified in this notice. (See line 13 below for more information.)

To protect your rights, consult an attorney. All documents filed in the case may be inspected at the bankruptcy clerk's office at the address listed below or through PACER (Public Access to Court Electronic Records at <a href="https://www.pacer.gov">www.pacer.gov</a>).

The staff of the bankruptcy clerk's office cannot give legal advice.

To help creditors correctly identify debtors, debtors submit full Social Security or Individual Taxpayer Identification Numbers, which may appear on a version of this notice. However, the full numbers must not appear on any document filed with the court.

Do not file this notice with any proof of claim or other filing in the case. Do not include more than the last four digits of a Social Security or Individual Taxpayer Identification Number in any document, including attachments, that you file with the court.

	About Debtor 1:	About Debtor 2:
1.	Debtor's full name	
2.	All other names used in the last 8 years	
3.	Address	If Debtor 2 lives at a different address:
4.	Debtor's attorney	Contact phone
	Name and address	Email
5.	Bankruptcy trustee	Contact phone
	Name and address	Email
6.	Bankruptcy clerk's office  Documents in this case may be filed at this address.	Hours open
	You may inspect all records filed in this case at this office or online at <a href="https://www.pacer.gov">www.pacer.gov</a> .	Contact phone

For more information, see page 2

7. Meeting of creditors  Debtors must attend the meeting to be questioned under oath. In a joint case, both spouses must attend.  Creditors may attend, but are not required to do so.  Creditors may attend, but are not required to do so.  Creditors may attend, but are not required to do so.  Date Time  Location:  Date Time  The meeting may be continued or adjourned to a later date. If so, the date will be on the court docket.	
Debtors must attend the meeting to be questioned under oath. In a joint case, both spouses must attend.  Creditors may attend, but are not required to do so.  Time  Time  Time  Time  Totation:  Time  Totation:  Time  Totation:  Totation:  Time  So, the date will be on the court docket.	
meeting to be questioned under oath. In a joint case, both spouses must attend.  Creditors may attend, but are not required to do so.  Date Time  Time  The meeting may be continued or adjourned to a later date. If so, the date will be on the court docket.	
spouses must attend. The meeting may be continued or adjourned to a later date. If Creditors may attend, but are not required to do so.	
required to do so.	
8. Deadlines Deadline to file a complaint to challenge Filing deadline:	
The bankruptcy clerk's office dischargeability of certain debts:  must receive these documents	
and any required filing fee by the following deadlines.  You must start a judicial proceeding by filing a complaint if you want to have a debt excepted from discharge under 11 U.S.C. § 523(a)(2), (4), or (6).	
Deadline for all creditors to file a proof of claim Filing deadline: (except governmental units):	
Deadline for governmental units to file a proof of Filing deadline:	
Deadlines for filing proof of claim:	
A proof of claim is a signed statement describing a creditor's claim. A proof of claim form ma at <a href="https://www.uscourts.gov">www.uscourts.gov</a> or any bankruptcy clerk's office.	may be obtained
If you do not file a proof of claim by the deadline, you might not be paid on your claim. To be proof of claim even if your claim is listed in the schedules that the debtor filed.	be paid, you must file a
Secured creditors retain rights in their collateral regardless of whether they file a proof of cla claim submits the creditor to the jurisdiction of the bankruptcy court, with consequences a la	
example, a secured creditor who files a proof of claim may surrender important nonmonetary right to a jury trial.	tary rights, including the
Deadline to object to exemptions: Filing deadline: 30 da	days after the neeting
The law permits debtors to keep certain property as exempt. of cre	creditors
If you believe that the law does not authorize an exemption claimed, you may file an objection.	
9. Filing of plan [The debtor has filed a plan. The plan or a summary of the plan is enclosed. The hearing on	
held on: at Location: Date Time ]	<del></del>
Or [The debtor has filed a plan. The plan or a summary of the plan and notice of confirmation	ation hearing will be
sent separately.]	<b>3</b>
Or [The debtor has not filed a plan as of this date. A copy of the plan or summary and a noti confirmation will be sent separately.]	notice of the hearing on
Commination will be sent separately.]	
If you are a creditor receiving a notice mailed to a foreign address, you may file a motion asl extend the deadlines in this notice. Consult an attorney familiar with United States bankrupt questions about your rights in this case.	
11. Filing a Chapter 12 Chapter 12 allows family farmers and family fishermen to reorganize according to a plan. A	
bankruptcy case  unless the court confirms it. You may receive a copy of the plan. You may object to confirmation hearing. The debtor will remain in possession of the property and may operate the business unless the court orders otherwise.	
12. Discharge of debts  Confirmation of a chapter 12 plan may result in a discharge of debts, which may include all of the court orders of the discharge will not be effective until all payments under	
Unless the court orders otherwise, the discharge will not be effective until all payments unde A discharge means that you may never try to collect the debt from the debtor except as prov	
you want to have a particular debt excepted under 11 U.S.C. § 523(a)(2), (4), or (6), you mu proceeding by filing a complaint and paying the filing fee in the clerk's office by the deadline	must start a judicial
The law allows debtors to keep certain property as exempt. Fully exempt property will not be	
to creditors, even if the case is converted to chapter 7. Debtors must file a list of property clamay inspect that list at the bankruptcy clerk's office. If you believe that the law does not authorate that the debtors claim, you may file an objection. The bankruptcy clerk's office must receive deadline to object to exemptions in line 8.	uthorize an exemption

Case number (if known)\_

Debtor

Information to identify the case:			
Debtor Name		EIN	_
United States Bankruptcy Court for the:	District of(State)	[Date case filed for chapter 12	
Case number:	<del></del>	[Date case filed in chapter	MM / DD / YYYY OR  MM / DD / YYYY
		Date case converted to chapter 12	MM / DD / YYYY

## Official Form 309H (For Corporations or Partnerships)

## Notice of Chapter 12 Bankruptcy Case

12/15

For the debtor listed above, a case has been filed under chapter 12 of the Bankruptcy Code. An order for relief has been entered.

This notice has important information about the case for creditors, debtors, and trustees, including information about the meeting of creditors and deadlines. Read both pages carefully.

The filing of the case imposed an automatic stay against most collection activities. This means that creditors generally may not take action to collect debts from the debtor, the debtor's property, or certain codebtors. For example, while the stay is in effect, creditors cannot sue, assert a deficiency, repossess property, or otherwise try to collect from the debtor. Creditors cannot demand repayment from the debtor by mail, phone, or otherwise. Creditors who violate the stay can be required to pay actual and punitive damages and attorney's fees.

Confirmation of a chapter 12 plan may result in the discharge of debt. Creditors who want to have a particular debt excepted from discharge may be required to file a complaint in the bankruptcy clerk's office within the deadline specified in this notice. (See line 13 below for more information.)

To protect your rights, consult an attorney. All documents filed in the case may be inspected at the bankruptcy clerk's office at the address listed below or through PACER (Public Access to Court Electronic Records at <a href="https://www.pacer.gov">www.pacer.gov</a>).

The staff of the bankruptcy clerk's office cannot give legal advice.

Do not file this notice with any proof of claim or other filing in the case.

1.	Debtor's full name		
2.	All other names used in the last 8 years		
3.	Address		
4.	Debtor's attorney Name and address	Contact phone Email	
5.	Bankruptcy clerk's office  Documents in this case may be filed at this address.  You may inspect all records filed in this case at this office or online at <a href="https://www.pacer.gov">www.pacer.gov</a> .	Hours open Contact phone	
6.	Bankruptcy trustee Name and address	Contact phone Email	

For more information, see page 2

De	btor	Case number (if known)
	Name	
7.	Meeting of creditors The debtor's representative must attend the meeting to be	atLocation:
	questioned under oath.	The meeting may be continued or adjourned to a later date. If
	Creditors may attend, but are not required to do so.	so, the date will be on the court docket.
8.	Exception to discharge deadline The bankruptcy clerk's office must receive a complaint and any required filing fee by the following deadline.	You must start a judicial proceeding by filing a complaint if you want to have a debt excepted from discharge under 11 U.S.C. § 523(a)(2), (4), or (6).
9.	Filing of plan	[The debtor has filed a plan. The plan or a summary of the plan is enclosed. The hearing on confirmation will be held on: at Location:
		Date Time ]
		Or [The debtor has filed a plan. The plan or a summary of the plan and notice of confirmation hearing will be sent separately.]
		Or [The debtor has not filed a plan as of this date. A copy of the plan or summary and a notice of the hearing on confirmation will be sent separately.]
10.	Deadlines	Deadline for all creditors to file a proof of claim Filing deadline: (except governmental units):
		Deadline for governmental units to file a proof of claim:
		A proof of claim is a signed statement describing a creditor's claim. A proof of claim form may be obtained at <a href="https://www.uscourts.gov">www.uscourts.gov</a> or any bankruptcy clerk's office.  If you do not file a proof of claim by the deadline, you might not be paid on your claim. To be paid, you must file
		a proof of claim even if your claim is listed in the schedules that the debtor filed.
		Secured creditors retain rights in their collateral regardless of whether they file a proof of claim. Filing a proof of claim submits the creditor to the jurisdiction of the bankruptcy court, with consequences a lawyer can explain. For example, a secured creditor who files a proof of claim may surrender important nonmonetary rights, including the right to a jury trial.
11.	Creditors with a foreign address	If you are a creditor receiving a notice mailed to a foreign address, you may file a motion asking the court to extend the deadlines in this notice. Consult an attorney familiar with United States bankruptcy law if you have any questions about your rights in this case.
12.	Filing a chapter 12 bankruptcy case	Chapter 12 allows family farmers and family fishermen to reorganize according to a plan. A plan is not effective unless the court confirms it. You may receive a copy of the plan. You may object to confirmation of the plan and attend the confirmation hearing. The debtor will remain in possession of the property and may continue to operate the business.
13.	Discharge of debts	Confirmation of a chapter 12 plan may result in a discharge of debts, which may include all or part of your debt.  Unless the court orders otherwise, the discharge will not be effective until all payments under the plan are made.  A discharge means that you may never try to collect the debt from the debtor except as provided in the plan.
		If you want to have a particular debt excepted from discharge under 11 U.S.C. § 523(a)(2), (4), or (6), you must start a judicial proceeding by filing a complaint and paying the filing fee in the bankruptcy clerk's office by the deadline.

Information	to identify the case:				
Debtor 1	First Name	Middle Name	Last Name	Last 4 digits of Social Security number or ITIN  EIN  -	
Debtor 2	) <del></del>			Last 4 digits of Social Security number or ITIN	
(Spouse, if filing	) First Name	Middle Name	Last Name	EIN	
United States	Bankruptcy Court for the:		District of(State)	[Date case filed for chapter 13	
Case number:	: <u></u>		<del></del>	[Date case filed in chapter	MM / DD / YYYY OR  MM / DD / YYYY
				Date case converted to chapter 13	MM / DD / YYYY

#### Official Form 309I

## Notice of Chapter 13 Bankruptcy Case

12/15

For the debtors listed above, a case has been filed under chapter 13 of the Bankruptcy Code. An order for relief has been entered

This notice has important information about the case for creditors, debtors, and trustees, including information about the meeting of creditors and deadlines. Read both pages carefully.

The filing of the case imposed an automatic stay against most collection activities. This means that creditors generally may not take action to collect debts from the debtors, the debtors' property, and certain codebtors. For example, while the stay is in effect, creditors cannot sue, garnish wages, assert a deficiency, repossess property, or otherwise try to collect from the debtors. Creditors cannot demand repayment from debtors by mail, phone, or otherwise. Creditors who violate the stay can be required to pay actual and punitive damages and attorney's fees. Under certain circumstances, the stay may be limited to 30 days or not exist at all, although debtors can ask the court to extend or impose a stay.

Confirmation of a chapter 13 plan may result in a discharge. Creditors who assert that the debtors are not entitled to a discharge under 11 U.S.C. § 1328(f) must file a motion objecting to discharge in the bankruptcy clerk's office within the deadline specified in this notice. Creditors who want to have their debt excepted from discharge may be required to file a complaint in the bankruptcy clerk's office by the same deadline. (See line 13 below for more information.)

To protect your rights, consult an attorney. All documents filed in the case may be inspected at the bankruptcy clerk's office at the address listed below or through PACER (Public Access to Court Electronic Records at www.pacer.gov).

The staff of the bankruptcy clerk's office cannot give legal advice.

To help creditors correctly identify debtors, debtors submit full Social Security or Individual Taxpayer Identification Numbers, which may appear on a version of this notice. However, the full numbers must not appear on any document filed with the court.

Do not file this notice with any proof of claim or other filing in the case. Do not include more than the last four digits of a Social Security or Individual Taxpayer Identification Number in any document, including attachments, that you file with the court.

	About Debtor 1:	About Debtor 2:
1.	Debtor's full name	
2.	All other names used in the last 8 years	
3.	Address	If Debtor 2 lives at a different address:
4.	Debtor's attorney Name and address	Contact phone Email
5.	Bankruptcy trustee Name and address	Contact phone
6.	Bankruptcy clerk's office  Documents in this case may be filed at this address.	Hours open  Contact phone
	You may inspect all records filed in this case at this office or online at <a href="https://www.pacer.gov">www.pacer.gov</a> .	

For more information, see page 2

De	btor Name	Case number	er (if known)
	Name		
7.	Meeting of creditors		
•	Debtors must attend the meeting	at	Location:
	to be questioned under oath. In	Date Time	
	a joint case, both spouses must		
	attend.	The meeting may be continued or adjourned to a later date. If so, the date will be on the court docket.	
	Creditors may attend, but are not required to do so.	so, the date will be on the court docket.	
	<u> </u>		
8.	Deadlines	Deadline to file a complaint to challenge	Filing deadline:
	The bankruptcy clerk's office	dischargeability of certain debts:	
	must receive these documents and any required filing fee by the	You must file:	
	following deadlines.	a motion if you assert that the debtors are not entitled to	
		receive a discharge under U.S.C. § 1328(f), or	
		a complaint if you want to have a particular debt excepted	
		from discharge under 11 U.S.C. § 523(a)(2) or (4).	
		Deadline for all creditors to file a proof of claim	Filing deadline:
		(except governmental units):	
		,	Filing doubling.
		Deadline for governmental units to file a proof of claim:	Filing deadline:
		Ciaiii.	
		Deadlines for filing proof of claim:	
		A proof of claim is a signed statement describing a creditor's cla	•
		at www.uscourts.gov or any bankruptcy clerk's office. If you do	
		might not be paid on your claim. To be paid, you must file a pro- schedules that the debtor filed.	of claim even if your claim is listed in the
		Secured creditors retain rights in their collateral regardless of w	hether they file a proof of claim.
		Filing a proof of claim submits the creditor to the jurisdiction of the	he bankruptcy court, with consequences a
		lawyer can explain. For example, a secured creditor who files a	proof of claim may surrender important
		nonmonetary rights, including the right to a jury trial.	
			Filing deadline: 30 days after the
		Deadline to object to exemptions:	Filing deadline: 30 days after the conclusion of the meeting
		Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If	- · · · · · · · · · · · · · · · · · · ·
		Deadline to object to exemptions:	conclusion of the meeting
•	Filing of plan	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.	conclusion of the meeting of creditors
9.	Filing of plan	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be
9.	Filing of plan	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be
9.	Filing of plan	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:
9.	Filing of plan	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan held on: at Local	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:  an and notice of confirmation hearing will be
9.	Filing of plan	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan held on: at Local	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:  an and notice of confirmation hearing will be
9.	Filing of plan	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan held on: at Local	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:  an and notice of confirmation hearing will be
	Filing of plan  Creditors with a foreign	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan held on: at Local Date Time ]  Or [The debtor has filed a plan. The plan or a summary of the plan esent separately.]  Or [The debtor has not filed a plan as of this date. A copy of the confirmation will be sent separately.]  If you are a creditor receiving a notice mailed to a foreign addre	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:  an and notice of confirmation hearing will be plan or summary and a notice of the hearing on ss, you may file a motion asking the court to
		Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan held on: at Local	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:  an and notice of confirmation hearing will be plan or summary and a notice of the hearing on ss, you may file a motion asking the court to
	Creditors with a foreign	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan held on: at Local Date Time ]  Or [The debtor has filed a plan. The plan or a summary of the plan esent separately.]  Or [The debtor has not filed a plan as of this date. A copy of the confirmation will be sent separately.]  If you are a creditor receiving a notice mailed to a foreign addre	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:  an and notice of confirmation hearing will be plan or summary and a notice of the hearing on ss, you may file a motion asking the court to
10.	Creditors with a foreign address	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan held on: at Local	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:  an and notice of confirmation hearing will be plan or summary and a notice of the hearing on ss, you may file a motion asking the court to with United States bankruptcy law if you have any
10.	Creditors with a foreign	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan held on: at Local Date Time ]  Or [The debtor has filed a plan. The plan or a summary of the plan sent separately.]  Or [The debtor has not filed a plan as of this date. A copy of the confirmation will be sent separately.]  If you are a creditor receiving a notice mailed to a foreign addre extend the deadline in this notice. Consult an attorney familiar viguestions about your rights in this case.	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:  an and notice of confirmation hearing will be plan or summary and a notice of the hearing on ss, you may file a motion asking the court to with United States bankruptcy law if you have any below a specified amount to adjust debts
10.	Creditors with a foreign address Filing a chapter 13	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan held on:  Date  Time ]  Or [The debtor has filed a plan. The plan or a summary of the plan sent separately.]  Or [The debtor has filed a plan. The plan or a summary of the plan to sent separately.]  If you are a creditor receiving a notice mailed to a foreign addreextend the deadline in this notice. Consult an attorney familiar of questions about your rights in this case.  Chapter 13 allows an individual with regular income and debts to according to a plan. A plan is not effective unless the court confiplan and appear at the confirmation hearing. A copy of the plan	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:  an and notice of confirmation hearing will be plan or summary and a notice of the hearing on ss, you may file a motion asking the court to with United States bankruptcy law if you have any below a specified amount to adjust debts irms it. You may object to confirmation of the [is included with this notice] or [will be sent to
10.	Creditors with a foreign address Filing a chapter 13	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan held on:  Date  Time ]  Or [The debtor has filed a plan. The plan or a summary of the plan sent separately.]  Or [The debtor has filed a plan. The plan or a summary of the plan to sent separately.]  If you are a creditor receiving a notice mailed to a foreign addreextend the deadline in this notice. Consult an attorney familiar of questions about your rights in this case.  Chapter 13 allows an individual with regular income and debts to according to a plan. A plan is not effective unless the court confiplan and appear at the confirmation hearing. A copy of the plan you later], and [the confirmation hearing will be held on the date	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:  an and notice of confirmation hearing will be plan or summary and a notice of the hearing on ss, you may file a motion asking the court to with United States bankruptcy law if you have any below a specified amount to adjust debts irms it. You may object to confirmation of the [is included with this notice] or [will be sent to shown in line 9 of this notice] or [the court will]
10.	Creditors with a foreign address Filing a chapter 13	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan held on:  Date  Time ]  Or [The debtor has filed a plan. The plan or a summary of the plan sent separately.]  Or [The debtor has filed a plan. The plan or a summary of the plan sent separately.]  If you are a creditor receiving a notice mailed to a foreign addreextend the deadline in this notice. Consult an attorney familiar of questions about your rights in this case.  Chapter 13 allows an individual with regular income and debts to according to a plan. A plan is not effective unless the court confiplan and appear at the confirmation hearing. A copy of the plan you later], and [the confirmation hearing will be held on the dates send you a notice of the confirmation hearing]. The debtor will reserve the confirmation hearing.	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:  an and notice of confirmation hearing will be plan or summary and a notice of the hearing on ss, you may file a motion asking the court to with United States bankruptcy law if you have any below a specified amount to adjust debts irms it. You may object to confirmation of the [is included with this notice] or [will be sent to shown in line 9 of this notice] or [the court will emain in possession of the property and may
10.	Creditors with a foreign address Filing a chapter 13	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan held on:  Date  Time ]  Or [The debtor has filed a plan. The plan or a summary of the plan sent separately.]  Or [The debtor has not filed a plan as of this date. A copy of the confirmation will be sent separately.]  If you are a creditor receiving a notice mailed to a foreign addreextend the deadline in this notice. Consult an attorney familiar valuestions about your rights in this case.  Chapter 13 allows an individual with regular income and debts to according to a plan. A plan is not effective unless the court confiplan and appear at the confirmation hearing. A copy of the plan you later], and [the confirmation hearing will be held on the date send you a notice of the confirmation hearing]. The debtor will recontinue to operate the business, if any, unless the court orders.	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:  an and notice of confirmation hearing will be plan or summary and a notice of the hearing on ss, you may file a motion asking the court to with United States bankruptcy law if you have any selow a specified amount to adjust debts irms it. You may object to confirmation of the [is included with this notice] or [will be sent to a shown in line 9 of this notice] or [the court will emain in possession of the property and may notherwise.
10.	Creditors with a foreign address Filing a chapter 13	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan held on:  Date  Time ]  Or [The debtor has filed a plan. The plan or a summary of the plan sent separately.]  Or [The debtor has not filed a plan as of this date. A copy of the confirmation will be sent separately.]  If you are a creditor receiving a notice mailed to a foreign addreextend the deadline in this notice. Consult an attorney familiar of questions about your rights in this case.  Chapter 13 allows an individual with regular income and debts according to a plan. A plan is not effective unless the court confiplan and appear at the confirmation hearing. A copy of the plan you later], and [the confirmation hearing will be held on the date send you a notice of the confirmation hearing]. The debtor will recontinue to operate the business, if any, unless the court orders.	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:  an and notice of confirmation hearing will be plan or summary and a notice of the hearing on ss, you may file a motion asking the court to with United States bankruptcy law if you have any below a specified amount to adjust debts irms it. You may object to confirmation of the [is included with this notice] or [will be sent to be shown in line 9 of this notice] or [the court will emain in possession of the property and may notherwise.
10.	Creditors with a foreign address  Filing a chapter 13 bankruptcy case	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan held on:  Date  Time ]  Or [The debtor has filed a plan. The plan or a summary of the plan sent separately.]  Or [The debtor has not filed a plan as of this date. A copy of the confirmation will be sent separately.]  If you are a creditor receiving a notice mailed to a foreign addreextend the deadline in this notice. Consult an attorney familiar valuestions about your rights in this case.  Chapter 13 allows an individual with regular income and debts to according to a plan. A plan is not effective unless the court confiplan and appear at the confirmation hearing. A copy of the plan you later], and [the confirmation hearing will be held on the date send you a notice of the confirmation hearing]. The debtor will recontinue to operate the business, if any, unless the court orders.	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:  an and notice of confirmation hearing will be plan or summary and a notice of the hearing on ss, you may file a motion asking the court to with United States bankruptcy law if you have any below a specified amount to adjust debts irms it. You may object to confirmation of the [is included with this notice] or [will be sent to be shown in line 9 of this notice] or [the court will emain in possession of the property and may be otherwise.  It was a specified amount to adjust debts irms it. You may object to confirmation of the property and may be otherwise.
10.	Creditors with a foreign address  Filing a chapter 13 bankruptcy case	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan held on:  Date  Time ]  Or [The debtor has filed a plan. The plan or a summary of the plan sent separately.]  Or [The debtor has not filed a plan as of this date. A copy of the confirmation will be sent separately.]  If you are a creditor receiving a notice mailed to a foreign addreextend the deadline in this notice. Consult an attorney familiar of questions about your rights in this case.  Chapter 13 allows an individual with regular income and debts to according to a plan. A plan is not effective unless the court confiplan and appear at the confirmation hearing. A copy of the plan you later], and [the confirmation hearing will be held on the date send you a notice of the confirmation hearing]. The debtor will recontinue to operate the business, if any, unless the court orders. The law allows debtors to keep certain property as exempt. Full to creditors, even if the case is converted to chapter 7. Debtors may inspect that list at the bankruptcy clerk's office or online at not authorize an exemption that debtors claimed, you may file a	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:  an and notice of confirmation hearing will be plan or summary and a notice of the hearing on ss, you may file a motion asking the court to with United States bankruptcy law if you have any below a specified amount to adjust debts irms it. You may object to confirmation of the [is included with this notice] or [will be sent to e shown in line 9 of this notice] or [the court will emain in possession of the property and may otherwise.  If you believe that the law does n objection by the deadline.
10.	Creditors with a foreign address  Filing a chapter 13 bankruptcy case  Exempt property	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan held on:  Date  Time ]  Or [The debtor has filed a plan. The plan or a summary of the plan sent separately.]  Or [The debtor has not filed a plan as of this date. A copy of the confirmation will be sent separately.]  If you are a creditor receiving a notice mailed to a foreign addreextend the deadline in this notice. Consult an attorney familiar of questions about your rights in this case.  Chapter 13 allows an individual with regular income and debts to according to a plan. A plan is not effective unless the court confiplan and appear at the confirmation hearing. A copy of the plan you later], and [the confirmation hearing will be held on the date send you a notice of the confirmation hearing]. The debtor will recontinue to operate the business, if any, unless the court orders. The law allows debtors to keep certain property as exempt. Full to creditors, even if the case is converted to chapter 7. Debtors may inspect that list at the bankruptcy clerk's office or online at not authorize an exemption that debtors claimed, you may file a Confirmation of a chapter 13 plan may result in a discharge of decoration.	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:  an and notice of confirmation hearing will be plan or summary and a notice of the hearing on ss, you may file a motion asking the court to with United States bankruptcy law if you have any below a specified amount to adjust debts irms it. You may object to confirmation of the [is included with this notice] or [will be sent to e shown in line 9 of this notice] or [the court will emain in possession of the property and may otherwise.  If you were property will not be sold and distributed must file a list of property claimed as exempt. You www.pacer.gov. If you believe that the law does n objection by the deadline.
10.	Creditors with a foreign address  Filing a chapter 13 bankruptcy case	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan held on: at Local Date Time ]  Or [The debtor has filed a plan. The plan or a summary of the plan sent separately.]  Or [The debtor has not filed a plan as of this date. A copy of the confirmation will be sent separately.]  If you are a creditor receiving a notice mailed to a foreign addre extend the deadline in this notice. Consult an attorney familiar viguestions about your rights in this case.  Chapter 13 allows an individual with regular income and debts to according to a plan. A plan is not effective unless the court confiplan and appear at the confirmation hearing. A copy of the plan you later], and [the confirmation hearing will be held on the date send you a notice of the confirmation hearing]. The debtor will recontinue to operate the business, if any, unless the court orders. The law allows debtors to keep certain property as exempt. Full to creditors, even if the case is converted to chapter 7. Debtors may inspect that list at the bankruptcy clerk's office or online at not authorize an exemption that debtors claimed, you may file a Confirmation of a chapter 13 plan may result in a discharge of definition of the court orders otherwise, the debts will not be court orders.	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:  an and notice of confirmation hearing will be plan or summary and a notice of the hearing on ss, you may file a motion asking the court to with United States bankruptcy law if you have any below a specified amount to adjust debts irms it. You may object to confirmation of the [is included with this notice] or [will be sent to e shown in line 9 of this notice] or [the court will emain in possession of the property and may otherwise.  If you believe that the law does no objection by the deadline.  ebts, which may include all or part of a debt. e discharged until all payments under the plan
10.	Creditors with a foreign address  Filing a chapter 13 bankruptcy case  Exempt property	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan held on: at Local Date Time ]  Or [The debtor has filed a plan. The plan or a summary of the plan sent separately.]  Or [The debtor has not filed a plan as of this date. A copy of the confirmation will be sent separately.]  If you are a creditor receiving a notice mailed to a foreign addre extend the deadline in this notice. Consult an attorney familiar vigurestions about your rights in this case.  Chapter 13 allows an individual with regular income and debts to according to a plan. A plan is not effective unless the court confiplan and appear at the confirmation hearing. A copy of the plan you later], and [the confirmation hearing will be held on the date send you a notice of the confirmation hearing]. The debtor will recontinue to operate the business, if any, unless the court orders. The law allows debtors to keep certain property as exempt. Full to creditors, even if the case is converted to chapter 7. Debtors may inspect that list at the bankruptcy clerk's office or online at not authorize an exemption that debtors claimed, you may file at Confirmation of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan ma	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:  an and notice of confirmation hearing will be plan or summary and a notice of the hearing on ss, you may file a motion asking the court to with United States bankruptcy law if you have any below a specified amount to adjust debts irms it. You may object to confirmation of the [is included with this notice] or [will be sent to e shown in line 9 of this notice] or [the court will emain in possession of the property and may eitherwise.  If you believe that the law does no objection by the deadline.  ebts, which may include all or part of a debt. e discharged until all payments under the plan allect the debt from the debtors personally except
10.	Creditors with a foreign address  Filing a chapter 13 bankruptcy case  Exempt property	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan held on: at Local Date Time ]  Or [The debtor has filed a plan. The plan or a summary of the plan sent separately.]  Or [The debtor has not filed a plan as of this date. A copy of the confirmation will be sent separately.]  If you are a creditor receiving a notice mailed to a foreign addre extend the deadline in this notice. Consult an attorney familiar violation about your rights in this case.  Chapter 13 allows an individual with regular income and debts to according to a plan. A plan is not effective unless the court confiplan and appear at the confirmation hearing. A copy of the plan you later], and [the confirmation hearing will be held on the date send you a notice of the confirmation hearing]. The debtor will recontinue to operate the business, if any, unless the court orders. The law allows debtors to keep certain property as exempt. Full to creditors, even if the case is converted to chapter 7. Debtors may inspect that list at the bankruptcy clerk's office or online at not authorize an exemption that debtors claimed, you may file a Confirmation of a chapter 13 plan may result in a discharge of definition of the plan. If you want to have a particular debt exercises and the plan. If you want to have a particular debt exercises and the plan. If you want to have a particular debt exercises are provided in the plan. If you want to have a particular debt exercises are provided in the plan. If you want to have a particular debt exercises are provided in the plan. If you want to have a particular debt exercises are provided in the plan. If you want to have a particular debt exercises are provided in the plan. If you want to have a particular debt exercises are provided in the plan. If you want to have a particular debt exercises are pro	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:  an and notice of confirmation hearing will be plan or summary and a notice of the hearing on ss, you may file a motion asking the court to with United States bankruptcy law if you have any below a specified amount to adjust debts irms it. You may object to confirmation of the [is included with this notice] or [will be sent to e shown in line 9 of this notice] or [the court will emain in possession of the property and may eitherwise.  If you believe that the law does no objection by the deadline.  Www.pacer.gov. If you believe that the law does no objection by the deadline.  Wet the debt from the debtors personally except the dept from discharge under
10.	Creditors with a foreign address  Filing a chapter 13 bankruptcy case  Exempt property	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan held on: at Local Date Time ]  Or [The debtor has filed a plan. The plan or a summary of the plan sent separately.]  Or [The debtor has not filed a plan as of this date. A copy of the confirmation will be sent separately.]  If you are a creditor receiving a notice mailed to a foreign addre extend the deadline in this notice. Consult an attorney familiar vigurestions about your rights in this case.  Chapter 13 allows an individual with regular income and debts to according to a plan. A plan is not effective unless the court confiplan and appear at the confirmation hearing. A copy of the plan you later], and [the confirmation hearing will be held on the date send you a notice of the confirmation hearing]. The debtor will recontinue to operate the business, if any, unless the court orders. The law allows debtors to keep certain property as exempt. Full to creditors, even if the case is converted to chapter 7. Debtors may inspect that list at the bankruptcy clerk's office or online at not authorize an exemption that debtors claimed, you may file at Confirmation of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan may result in a discharge of definition of a chapter 13 plan ma	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:  an and notice of confirmation hearing will be plan or summary and a notice of the hearing on ss, you may file a motion asking the court to with United States bankruptcy law if you have any below a specified amount to adjust debts irms it. You may object to confirmation of the [is included with this notice] or [will be sent to shown in line 9 of this notice] or [the court will emain in possession of the property and may otherwise.  If you believe that the law does mobjection by the deadline.  The shown in the debt from the debtors personally except the filing fee in the bankruptcy clerk's office by
10.	Creditors with a foreign address  Filing a chapter 13 bankruptcy case  Exempt property	Deadline to object to exemptions:  The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.  [The debtor has filed a plan. The plan or a summary of the plan held on: at Local Date Time ]  Or [The debtor has filed a plan. The plan or a summary of the plan sent separately.]  Or [The debtor has not filed a plan as of this date. A copy of the confirmation will be sent separately.]  If you are a creditor receiving a notice mailed to a foreign addre extend the deadline in this notice. Consult an attorney familiar violation about your rights in this case.  Chapter 13 allows an individual with regular income and debts to according to a plan. A plan is not effective unless the court confiplan and appear at the confirmation hearing. A copy of the plan you later], and [the confirmation hearing will be held on the date send you a notice of the confirmation hearing]. The debtor will recontinue to operate the business, if any, unless the court orders. The law allows debtors to keep certain property as exempt. Full to creditors, even if the case is converted to chapter 7. Debtors may inspect that list at the bankruptcy clerk's office or online at not authorize an exemption that debtors claimed, you may file a Confirmation of a chapter 13 plan may result in a discharge of definition of the plan. If you want to have a particular debt exempted to case in the plan. If you want to have a particular debt exempted to case in the plan. If you want to have a particular debt exempted to case in the plan. If you want to have a particular debt exempted to case in the plan. If you want to have a particular debt exempted to case in the plan. If you want to have a particular debt exempted to case in the plan. If you want to have a particular debt exempted to case in the plan. If you want to have a particular debt exempted to case in the plan. If you want to have a particular debt exempted to case in the plan. If you wan	conclusion of the meeting of creditors  is enclosed. The hearing on confirmation will be tion:  an and notice of confirmation hearing will be plan or summary and a notice of the hearing on ss, you may file a motion asking the court to with United States bankruptcy law if you have any below a specified amount to adjust debts irms it. You may object to confirmation of the [is included with this notice] or [will be sent to shown in line 9 of this notice] or [the court will emain in possession of the property and may otherwise.  If you believe that the law does nobjection by the deadline.  The discharged until all payments under the plan of the filing fee in the bankruptcy clerk's office by discharge of any of their debts under 11 U.S.C.

Official Forms 309A-I, collectively the Bankruptcy Case Commencement Notices, have been revised as part of the Forms Modernization Project to make them easier to read and understand. The notices, derived from Official Forms 9A-I are renumbered and stylistic changes have been made.

References to the limitations on the automatic stay imposed by 11 U.S.C. § 362(c)(3) and (4) in some repeat bankruptcy filings by individuals have been deleted from the three versions of the notice for cases filed by corporations and partnerships. Email addresses for the debtor's attorney and the trustee have been added to the form.

The parties are informed that they may review papers filed in the case through the judiciary's PACER system (Public Access to Court Electronic Records) as well as at the bankruptcy clerk's office.

The lettering scheme for the versions of Official Form 309 track the versions of Official Form 9 used in different types of bankruptcy cases with following exceptions. Official Forms 9E(Alt.) and 9F(Alt.) have been eliminated by including alternative language in Official Forms 309E and 309F to be used if the court sets a deadline for filing claims at the start of the chapter 11 case. In addition, the B and C versions have been reversed in order. That is, Official Form 9C has been designated 309B and Official Form 9B as 309C. This groups together the notices for chapter 7 individual debtors and for non-individual debtors. Finally, as a result of the reformatting, Official Form 309C has been reduced to a single page.

The four versions of the form for chapter 7 cases have been renamed to state whether the notice specifies a deadline for filing proofs of claim, rather than whether the case is an "asset" or "no-asset" case.

## Order and Notice for Hearing on Disclosure Statement

To the	e debtor, its creditors, and other parties in interest:
	closure statement and a plan under chapter 11 [or chapter 9] of the Bankruptcy Code having been filed by
	on,
IT IS (	ORDERED and notice is hereby given, that:
1.	The hearing to consider the approval of the disclosure statement shall be held at:
-	on, at, o'clockm.
2.	is fixed as the last day for filing and serving in accordance with Fed. R. Bankr. P. 3017(a) written objections to the disclosure statement.
3.	Within days after entry of this order, the disclosure statement and plan shall be distributed in accordance with Fed. R. Bankr. P. 3017(a).
4.	Requests for copies of the disclosure statement and plan shall be mailed to the debtor in possession [or trustee or debtor or] at the following mailing address:
	[]·
	By the court:
	MM / DD / YYYY United States Bankruptcy Judge

Official Form 312, Order and Notice for Hearing on Disclosure Statement replaces Official Form 12, Order and Notice for Hearing on Disclosure Statement. It is renumbered as part of the Forms Modernization Project, and includes stylistic changes throughout the form.

# Order Approving Disclosure Statement and Fixing Time for Filing Acceptances or Rejections of Plan, Combined with Notice Thereof

	losure statement under chapter 11	• •	~	-
on			, on	], referring to a plan under [if appropriate, and by
	cation filed on ng been determined after hearing o ation:	<b>-</b> '	sclosure statement [or	statements] contain[s] adequate
IT IS (	<b>DRDERED</b> , and notice is hereby giv	en, that:		
A.	The disclosure statement filed by and by			
В.	is fingler plans [or plans] referred to above.	xed as the last day	for filing written acce	otances or rejections of the
C.	Within days after the end thereof approved by the court, [and any, dated, approving [or statements], and a ballot conform (Official Form 314) shall be mailed and shall be transmitted to the University of the univer	d [if appropriate] a the disclosure state rming to <i>Ballot for a</i> I to creditors, equity	summary approved by ement [or statements] Accepting or Rejecting y security holders, and	the court of its opinion, if I, the disclosure statement I Plan of Reorganization I other parties in interest,
D.	If acceptances are filed for more the indicated.	nan one plan, prefe	rences among the pla	ns so accepted may be
E.	[If appropriate]i	s fixed for the hear	ing on confirmation of	the plan [or plans].
F.	[If appropriate] Bankr. P. 3020(b)(1) written object			ing pursuant to Fed. R.
	MM / DD / YYYY		By the court:	nited States Bankruptcy Judge

Official Form 313, Order Approving Disclosure Statement and Fixing Time for Filing Acceptances or Rejections of Plan, Combined with Notice Thereof replaces Official Form 13, Order Approving Disclosure Statement and Fixing Time for Filing Acceptances or Rejections of Plan, Combined with Notice Thereof. It is renumbered as part of the Forms Modernization Project, and includes stylistic changes throughout the form.

Official Form 314 (12/15)

[Caption as in 416A]

## Class [ ] Ballot for Accepting or Rejecting Plan of Reorganization

[Proponent] filed a plan of reorganization dated [Date] (the *Plan*) for the Debtor in this case. The Court has [conditionally] approved a disclosure statement with respect to the Plan (the *Disclosure Statement*). The Disclosure Statement provides information to assist you in deciding how to vote your ballot. If you do not have a Disclosure Statement, you may obtain a copy from [name, address, telephone number and telecopy number of proponent/proponent's attorney.]

Court approval of the disclosure statement does not indicate approval of the Plan by the Court.

You should review the Disclosure Statement and the Plan before you vote. You may wish to seek legal advice concerning the Plan and your classification and treatment under the Plan. Your [claim] [equity interest] has been placed in class [] under the Plan. If you hold claims or equity interests in more than one class, you will receive a ballot for each class in which you are entitled to vote.

If your ballot is not received by [name and address of proponent's attorney or other appropriate address] on or before [date], and such deadline is not extended, your vote will not count as either an acceptance or rejection of the Plan.

If the Plan is confirmed by the Bankruptcy Court, it will be binding on you whether or not you vote.

Acceptance or Rejection of the Plan

[At this point the ballot should provide for voting by the particular class of creditors or equity holders receiving the ballot using one of the following alternatives;]

[If the voter is the holder of a secured, priority, or unsecured nonpriority claim:]

The undersigned, the holder of a Class [] claim against the Debtor in the unpaid amount of Dollars (\$

[or, if the voter is the holder of a bond, debenture, or other debt security:]

The undersigned, the holder of a Class [] claim against the Debtor, consisting of Dollars (\$ ) principal amount of [describe bond, debenture, or other debt security] of the Debtor (For purposes of this Ballot, it is not necessary and you should not adjust the principal amount for any accrued or unmatured interest.)

[or, if the voter is the holder of an equity interest:]

The undersigned, the holder of Class [] equity interest in the Debtor, consisting of \_\_\_\_\_ shares or other interests of [describe equity interest] in the Debtor

[In each case, the following language should be included:]							
Check one box only							
Accepts the							
Rejects the p	lan						
Dated:							
Print or type name:							
Signature:			Title (if corporation or partnership)				
Address:							

#### Return this ballot to:

[Name and address of proponent's attorney or other appropriate address]

Official Form 314, *Ballot for Accepting or Rejecting Plan* replaces Official Form 14, *Ballot for Accepting or Rejecting Plan*. It is renumbered as part of the Forms Modernization Project, and includes stylistic changes throughout the form.

[Caption as in 416A]		
Order Confirming Plan		
The plan under chapter 11 of the Bankruptcy Code filed by	/	on
[if applicable, as modified by a m	nodification filed on,]	or a
summary thereof, having been transmitted to creditors and	d equity security holders; and	
It having been determined after hearing on notice that the 11 U.S.C. § 1129(a) [or, <i>if appropriate</i> , 11 U.S.C. § 1129(b)	·	
IT IS ORDERED that:		
The plan filed by	, on,	
[If appropriate, include dates and any other pertinent detail the plan provides for an injunction against conduct not other information required by Rule 3020.]	· -	
A copy of the confirmed plan is attached.		
	By the court:	
MM / DD / YYYY	United States Bankruptcy J	ludge

Official Form 315 (12/15)

Official Form 315, *Order Confirming Plan* replaces Official Form 15, *Order Confirming Plan*. It is renumbered as part of the Forms Modernization Project, and includes stylistic changes throughout the form.

Information to	identify the ca	se:						
Debtor 1	First Name	Middle Name	Last Name	Last 4 d	igits of Social Security number or ITIN			
	First Name	Middle Name	Last Name	EIN _	<sup>-</sup>			
Debtor 2	First Name	Middle Name	Last Name	Last 4 di	gits of Social Security number or ITIN			
(Spouse, if filing)	riist name	Middle Name	Last Name	EIN _				
United States E	Bankruptcy Cour	t for the:	District of(State)					
Case number:			(3.3.3.)					
Order of	Order of Discharge							
		<u> </u>	C. § 727 is granted to:					
			_		]			
[include all names used by each debtor, including trade names, within the 8 years prior to the filing of the petition]								
	MM / DD / YY	YY	Ву	the court:	United States Bankruptcy Judge			

Explanation of Bankruptcy Discharge in a Chapter 7 Case

This order does not close or dismiss the case, and it does not determine how much money, if any, the trustee will pay creditors.

#### Creditors cannot collect discharged debts

This order means that no one may make any attempt to collect a discharged debt from the debtors personally. For example, creditors cannot sue, garnish wages, assert a deficiency, or otherwise try to collect from the debtors personally on discharged debts. Creditors cannot contact the debtors by mail, phone, or otherwise in any attempt to collect the debt personally. Creditors who violate this order can be required to pay debtors damages and attorney's fees.

However, a creditor with a lien may enforce a claim against the debtors' property subject to that lien unless the lien was avoided or eliminated. For example, a creditor may have the right to foreclose a home mortgage or repossess an automobile.

This order does not prevent debtors from paying any debt voluntarily or from paying reaffirmed debts according to the reaffirmation agreement.

11 U.S.C. § 524(c), (f).

#### Most debts are discharged

Most debts are covered by the discharge, but not all. Generally, a discharge removes the debtors' personal liability for debts owed before the debtors' bankruptcy case was filed.

Also, if this case began under a different chapter of the Bankruptcy Code and was later converted to chapter 7, debts owed before the conversion are discharged.

In a case involving community property: Special rules protect certain community property owned by the debtor's spouse, even if that spouse did not file a bankruptcy case.

For more information, see page 2 ▶

#### Some debts are not discharged

Examples of debts that are not discharged are:

debts that are domestic support obligations;

debts for most student loans;

debts for most taxes:

debts that the bankruptcy court has decided or will decide are not discharged in this bankruptcy case;

debts for most fines, penalties, forfeitures, or criminal restitution obligations;

some debts which the debtors did not properly list;

debts for certain types of loans owed to pension, profit sharing, stock bonus, or retirement plans; and

debts for death or personal injury caused by operating a vehicle while intoxicated.

Also, debts covered by a valid reaffirmation agreement are not discharged.

In addition, this discharge does not stop creditors from collecting from anyone else who is also liable on the debt, such as an insurance company or a person who cosigned or guaranteed a loan.

This information is only a general summary of the bankruptcy discharge; some exceptions exist. Because the law is complicated, you should consult an attorney to determine the exact effect of the discharge in this case.

Official Form 318, *Order of Discharge*, is revised and renumbered as part of the Forms Modernization Project. The form is used to issue a discharge in chapter 7 cases filed by individuals or joint debtors. It replaces former Official Form 18, *Discharge of Debtor*, Director's Procedural Form 18J, *Discharge of Joint Debtors*, and Director's Procedural Form 18JO, *Discharge of One Joint Debtor*.

To make the discharge order and the explanation of it easier to read and understand, legal terms are explained more fully or replaced with commonly understood terms, and the form is reformatted.

Reaffirmed debts are explained more fully, and readers are informed that a discharge will not stop creditors from collecting debts from any property in which they have a valid lien. In addition, readers are advised that the discharge does not stop creditors from collecting from anyone else who is liable on the debt, such as a cosigner on the loan or an insurance company.

Director's Procedural Forms 18J and 18JO are no longer needed because Form 318 specifies the names of the debtors, or debtor, to whom the discharge is issued. Any alternate names of the debtor or debtors appear in the order not in the information box at the top of the form.

Fill in this information to identify the case:		
United States Bankrupto	cy Court for the:	
	_ District of	
Case number (If known): _		Chapter 15

## Official Form 401

#### Chapter 15 Petition for Recognition of a Foreign Proceeding 12/15

If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write debtor's name and case number (if known)					
1.	Debtor's name				
2.	Debtor's unique identifier	For non-individual debtors:			
		Federal Employer Identification Number (EIN)			
		Other Describe identifier			
		For individual debtors:			
		□ Social Security number: xxx - xx			
		☐ Individual Taxpayer Identification number (ITIN): 9 xx − xx −			
		Other Describe identifier			
3.	Name of foreign representative(s)				
4.	Foreign proceeding in which appointment of the foreign representative(s) occurred				
5.	Nature of the foreign proceeding	Check one:			
	p	Foreign main proceeding			
		Foreign nonmain proceeding  Foreign main proceeding, or in the alternative foreign nonmain proceeding			
6.	Evidence of the foreign proceeding	A certified copy, translated into English, of the decision commencing the foreign proceeding and appointing the foreign representative is attached.			
		A certificate, translated into English, from the foreign court, affirming the existence of the foreign proceeding and of the appointment of the foreign representative, is attached.			
		Other evidence of the existence of the foreign proceeding and of the appointment of the foreign representative is described below, and relevant documentation, translated into English, is attached.			
7.	Is this the only foreign	No. (Attach a statement identifying each country in which a foreign proceeding by, regarding, or against the			
	proceeding with respect to the debtor known to the foreign representative(s)?	debtor is pending.)  Yes			

Debtor Name	Cas	e number (# known)			
8. Others entitled to notice	Attach a list containing the names and addresses of:				
	(i) all persons or bodies authorized to administer for	reign proceedings of the debtor,			
	<ul><li>(ii) all parties to litigation pending in the United State petition, and</li></ul>	es in which the debtor is a party at the time of filing of this			
	(iii) all entities against whom provisional relief is bein	g sought under § 1519 of the Bankruptcy Code.			
9. Addresses	Country where the debtor has the center of its main interests:	Debtor's registered office:			
		Number Street			
		P.O. Box			
		City State/Province/Region ZIP/Postal Code			
		Country			
	Individual debtor's habitual residence:  Number Street	Address of foreign representative(s):  Number Street			
	P.O. Box	P.O. Box			
	1.0. 500	1.0.00			
	City State/Province/Region ZIP/Postal Code	City State/Province/Region ZIP/Postal Code			
	Country	Country			
10. Debtor's website (URL)					
11. Type of debtor	Check one:				
	☐ Non-individual ( <i>check one</i> ):				
	Corporation. Attach a corporate ownership described in Fed. R. Bankr. P. 7007.1.	p statement containing the information			
	☐ Partnership				
	Other. Specify:				

Debtor Name	Case	e number (if known)			
12. Why is venue proper in this district?	Check one:  Debtor's principal place of business or principal as  Debtor does not have a place of business or asset action or proceeding in a federal or state court is p	assets in the United States, but the following			
	If neither box is checked, venue is consistent with of the parties, having regard to the relief sought by	the interests of justice and the convenience the foreign representative, because:			
13. Signature of foreign representative(s)	I request relief in accordance with chapter 15 of ti	itle 11, United States Code.			
	I am the foreign representative of a debtor in a for relief sought in this petition, and I am authorized to				
	I have examined the information in this petition ar information is true and correct.	nd have a reasonable belief that the			
	I declare under penalty of perjury that the foregoing is true and correct,				
	×				
	Signature of foreign representative	Printed name			
	Executed on				
	<b>x</b>	_			
	Signature of foreign representative	Printed name			
	Executed on				
14. Signature of attorney	Signature of Attorney for foreign representative	Date MM / DD / YYYY			
	Printed name				
	Firm name				
	Number Street				
	City	State ZIP Code			
	Contact phone	Email address			
	Bar number	State			

Official Form 401 is required for any petition seeking recognition of a foreign proceeding under chapter 15 of the Bankruptcy Code. The form, which applies to foreign proceedings involving individual and non-individual debtors, consolidates information formerly included on Official Form 1 (Voluntary Petition). The petition must be signed by the foreign representative, under penalty of perjury, and by the foreign representative's attorney.

The petition requires disclosure of the foreign proceeding in which the foreign representative has been appointed (Line 4) and whether it is a foreign main proceeding or foreign nonmain proceeding (Line 5). If the foreign representative seeks recognition of the foreign proceeding as a foreign main proceeding or, in the alternative, a foreign nonmain proceeding, that request should be indicated in Line 5. Each country where any additional foreign proceeding known to the foreign representative is pending must be disclosed on Line 7. See Bankruptcy Rule 1004.2. Evidence of the foreign proceeding and of the foreign representative's appointment must accompany the petition. See 11 U.S.C. § 1515(b). These documents must be translated into English in accordance with 11 U.S.C. § 1515(d). The foreign representative must also attach a list of persons or bodies entitled to notice. See Bankruptcy Rule 2002(q).

The petition calls for information about the debtor, including the debtor's name (Line 1), other unique identifying information, if available (Line 2), and center of main interest (Line 9). The type of debtor is also requested (Line 11).

The foreign representative must indicate the basis for venue in the district by selecting an appropriate checkbox and, if necessary, providing additional information, such as a statement explaining why venue in the district is appropriate (Line 12). See 28 U.S.C. § 1410.

Fill in this information to identify the case:						
Debtor 1						
Debtor 2 (Spouse, if filing)						
United States Bankruptcy Court for the:	District of (State)					
Case number						

#### Official Form 410

## Proof of Claim

12/15

Read the instructions before filling out this form. This form is for making a claim for payment in a bankruptcy case. Do not use this form to make a request for payment of an administrative expense. Make such a request according to 11 U.S.C. § 503.

**Filers must leave out or redact** information that is entitled to privacy on this form or on any attached documents. Attach redacted copies of any documents that support the claim, such as promissory notes, purchase orders, invoices, itemized statements of running accounts, contracts, judgments, mortgages, and security agreements. **Do not send original documents**; they may be destroyed after scanning. If the documents are not available, explain in an attachment.

A person who files a fraudulent claim could be fined up to \$500,000, imprisoned for up to 5 years, or both. 18 U.S.C. §§ 152, 157, and 3571.

Fill in all the information about the claim as of the date the case was filed. That date is on the notice of bankruptcy (Form 309) that you received.

Who is the current creditor?		Name of the current creditor (the person or entity to be paid for this claim)  Other names the creditor used with the debtor						
Has this claim been acquired from someone else?	□ No □ Yes. From whom?							
Where should notices and payments to the creditor be sent?	Where should notices t	to the creditor be	sent?	Where should p different)	Where should payments to the creditor be sent? (if different)			
Federal Rule of Bankruptcy Procedure (FRBP) 2002(g)	Name			Name				
, , ,	Number Street			Number Stre	eet			
	City	State	ZIP Code	City	State	ZIP Code		
	Contact phone	· · · · · · · · · · · · · · · · · · ·		Contact phone		<del></del>		
	Contact email		_	Contact email		<del></del>		
	Uniform claim identifier for e	lectronic payments ir	n chapter 13 (if you u	se one):				
Does this claim amend one already filed?	☐ No☐ Yes. Claim number	on court claims reç	gistry (if known) _		Filed on	) / YYYY		
Do you know if anyone else has filed a proof of claim for this claim?	☐ No ☐ Yes. Who made the	earlier filing?						

	Do you have any number you use to identify the debtor?  No Yes. Last 4 digits of the debtor's account or any number you use to identify the debtor:				
7.	How much is the claim?	\$ Does this amount include interest or other charges?			
		Yes. Attach statement itemizing interest, fees, expenses, or other charges required by Bankruptcy Rule 3001(c)(2)(A).			
3.	What is the basis of the claim?	Examples: Goods sold, money loaned, lease, services performed, personal injury or wrongful death, or credit card.			
	oldiii.	Attach redacted copies of any documents supporting the claim required by Bankruptcy Rule 3001(c).			
		Limit disclosing information that is entitled to privacy, such as health care information.			
9.	Is all or part of the claim secured?	□ No □ Yes. The claim is secured by a lien on property.  Nature of property:			
		Real estate. If the claim is secured by the debtor's principal residence, file a <i>Mortgage Proof of Claim</i>			
		Attachment (Official Form 410-A) with this Proof of Claim.			
		☐ Motor vehicle ☐ Other. Describe:			
		Basis for perfection:			
		Attach redacted copies of documents, if any, that show evidence of perfection of a security interest (for example, a mortgage, lien, certificate of title, financing statement, or other document that shows the lien has been filed or recorded.)			
		Value of property: \$			
		Amount of the claim that is secured: \$			
		Amount of the claim that is unsecured: \$(The sum of the secured and unsecured amounts should match the amount in line 7.			
		Amount necessary to cure any default as of the date of the petition: \$			
		Annual Internal Pate (when account filed)			
		Annual Interest Rate (when case was filed)%  ☐ Fixed			
		☐ Variable			
10	Is this claim based on a	□ No			
	lease?	Yes. Amount necessary to cure any default as of the date of the petition.			
11	Is this claim subject to a	□ No			
	right of setoff?	☐ Yes. Identify the property:			
		- 100. Idolikii ji kio piopolity.			

12. Is all or part of the claim entitled to priority under	☐ No					
11 U.S.C. § 507(a)?	☐ Yes. Check	Amount entitled to priority				
A claim may be partly priority and partly		c support obligations (includ C. § 507(a)(1)(A) or (a)(1)(B	\$			
nonpriority. For example, in some categories, the law limits the amount	Up to \$2 personal	ses for \$				
entitled to priority.	bankrupt	salaries, or commissions (utcy petition is filed or the del C. § 507(a)(4).	re the \$			
		penalties owed to governm	nental units. 11 U.S.C. § 5	07(a)(8).	\$	
	☐ Contribu	tions to an employee benef	it plan. 11 U.S.C. § 507(a)	)(5).	\$	
	_	pecify subsection of 11 U.S		,	\$	
					on or after the date of adjustment.	
	, unounto di				on or alter the date of adjustment.	
Part 3: Sign Below						
The person completing	Check the approp	oriate box:				
this proof of claim must sign and date it.	☐ I am the cre	ditor.				
FRBP 9011(b).	☐ I am the cre	ditor's attorney or authorize	d agent.			
If you file this claim	☐ I am the trustee, or the debtor, or their authorized agent. Bankruptcy Rule 3004.					
electronically, FRBP	☐ I am a guarantor, surety, endorser, or other codebtor. Bankruptcy Rule 3005.					
5005(a)(2) authorizes courts to establish local rules	Tama gadrano, saisty, shasissi, or saist seassist. Earning to the season.					
specifying what a signature	I understand that an authorized signature on this <i>Proof of Claim</i> serves as an acknowledgment that when calculating the					
is.		aim, the creditor gave the debtor credit for any payments received toward the debt.				
A person who files a						
fraudulent claim could be fined up to \$500,000,	I have examined the information in this <i>Proof of Claim</i> and have a reasonable belief that the information is true					
imprisoned for up to 5	and correct.					
years, or both. 18 U.S.C. §§ 152, 157, and	I declare under penalty of perjury that the foregoing is true and correct.					
3571.						
	Executed on date	MM / DD / YYYY				
	Signature					
	Drint the name	of the person who is comp	oloting and cigning this	oloimi		
	Fillit the name t	or the person who is comp	neung and signing this t	viaiiii.		
	Name	<del></del>				
		First name	Middle name	Last	name	
	Title					
	Company	Identify the corporate service	r as the company if the author	rized agent is a servi	icer.	
gowe and a second secon						
	Address	Number Street				
		City		State ZIP (	Code	
	Contact phone			Email		

If you file a claim secured by a security interest in the debtor's principal residence, you must use this form as an attachment to your proof of claim. See separate instructions.

Part 1: Mortgage and Case Information				Part 2: Total Debt Calculation			Par	Part 3: Arrearage as of Date of the Petition				Part 4: Monthly Mortgage Payment				
Case number:					Principal balance:			Prin	Principal & interest due:				Principal & interest:			
Debtor 1:				Interest due:			Pre	petition fe	es due:			Monthly escrow:				
Debtor 2:				Fees, costs of	lue:			row deficion anced:	ency for fun	ds 		Private mortgage insurance:				
Last 4 digits to identify:					Escrow deficiency for funds advanced:			Proj	ected esc	row shortag	e:		Total monthly payment:			
Cred	itor:				Less total fur	nds on hand: -	-	Les	s funds or	hand:						<u>'</u>
Servi	icer:				Total debt:			Tota	al prepetiti	on arrearag	e:					
	d accrual/daily le interest/oth					_							!			
Part 5	5 : Loan Payn			irst Date of I	Default											
		Account											After Amount Received or Incurred			
<b>A.</b> Date	B. Contractual payment amount	C. Funds received	<b>D.</b> Amount incurred	E. Description	F. Contractual due date	G. Prin, int & esc past due balance	to	I. Amount to interest	to	K. Amount to fees or charges	L. Unapplied funds		N. Accrued interest balance	O. Escrow balance	P. Fees / Charges balance	
															244	

Case number:	
Debtor 1:	

		Account A	Activity				How Funds Were Applied/Amount Incurred						Balance After Amount Received or Incurred				
A.	В.	C.	D.	E.	F.	G.	H.	I.	J.	K.	L.	М.	N.	Ο.	P.	Q.	
Date	Contractual payment amount	Funds received	Amount incurred	Description	Contractual due date	Prin, int & esc past due balance	to	Amount to interest	to	to fees or	Unapplied funds	Principal balance	Accrued interest balance	Escrow balance	Fees / Charges balance	funds	

Fill in this information to identify the case:	
Debtor 1	
Debtor 2 (Spouse, if filing)	
United States Bankruptcy Court for the: District of(State)	
Case number	
Official Form 410S1	
Notice of Mortgage Payment Cha	ange 12/15
If the debtor's plan provides for payment of postpetition contractual instal debtor's principal residence, you must use this form to give notice of any as a supplement to your proof of claim at least 21 days before the new pay	changes in the installment payment amount. File this form
Name of creditor:	Court claim no. (if known):
Last 4 digits of any number you use to	Date of payment change:
identify the debtor's account:	Must be at least 21 days after date// of this notice
	New total payment:  Principal, interest, and escrow, if any
Part 1: Escrow Account Payment Adjustment	
Will there be a change in the debtor's escrow account payment	nt2
□ No □ Yes. Attach a copy of the escrow account statement prepared in a form the basis for the change. If a statement is not attached, explain when the basis for the change.	consistent with applicable nonbankruptcy law. Describe
Current escrow payment: \$	New escrow payment: \$
Part 2: Mortgage Payment Adjustment	
2. Will the debtor's principal and interest payment change based variable-rate account?	on an adjustment to the interest rate on the debtor's
<ul><li>□ No</li><li>□ Yes. Attach a copy of the rate change notice prepared in a form consis</li></ul>	tent with applicable nonbankruptcy law. If a notice is not
attached, explain why:	
Current interest rate:%	New interest rate:%
Current principal and interest payment: \$	New principal and interest payment: \$
Part 3: Other Payment Change	
3. Will there be a change in the debtor's mortgage payment for a	reason not listed above?
<ul> <li>□ No</li> <li>□ Yes. Attach a copy of any documents describing the basis for the change (Court approval may be required before the payment change can</li> </ul>	
Reason for change:	
Current mortgage payment: \$	New mortgage payment: \$

Debtor 1						Case i	number (if known)
	First Name	Middle Name	Last Name				
Part 4: S	Sign Here						
The persor telephone		this Notice mu	st sign it. Siç	gn and prin	t your nam	e and yo	our title, if any, and state your address and
Check the a	ppropriate bo	X.					
☐ I am	the creditor.						
☐ I am	the creditor	s authorized age	nt.				
					ovided in t	his clai	im is true and correct to the best of my
knowleag	e, informati	on, and reasor	iabie beliet.	•			
<b>x</b>						Date	
Signature							
Print:						Title	
1 11116.	First Name	Midd	le Name	Last Name		THE	
Company							
Company							
Address							
	Number	Street					
	City			State	ZIP Code		
Contact phone	e ()_		_			Email	

Debtor 1 Debtor 2 Debtor 3 Debtor 3 Debtor 4 Debtor 4 Debtor 5 Destor 6 Destor 7 Destor 7 Destor 7 Destor 8 Destor 1 Destor 9 Destor 1 Destor 2 Notice of Postpetition Mortgage Fees, Expenses, and Charges 12/15 If the debtor's pincipal residence, you must use this form to give notice of any frees, expenses, and charges incurred after the bankruptcy filling that you assert are recoverable against the debtor or against the debtor's principal residence.  File this form as a supplement to your proof of claim. See Bankruptcy Rule 3002.1.  Name of creditor:  Court claim no. (If known):  Last 4 digits of any number you use to identify the debtor's account:  Does this notice supplement a prior notice of postpetition fees, expenses, and charges?  No See Debtor of the last notice:  Ves. Debtor of the last notice:  Description Desc		_	
Debtor 2 (Souses, Elling)  United States Bankruptcy Court for the:	Fill in this information to identify the case:		
United States Bankruptcy Court for the:	Debtor 1		
United States Bankruptcy Court for the:			
Official Form 410S2 Notice of Postpetition Mortgage Fees, Expenses, and Charges 12/15 If the debtor's plan provides for payment of postpetition contractual installments on your claim secured by a security interest in the debtor's principal residence, you must use this form to give notice of any fees, expenses, and charges incurred after the bankruptcy filing that you assert are recoverable against the debtor or against the debtor's principal residence.  Name of creditor:  Court claim no. (if known):  Last 4 digits of any number you use to identify the debtor's account:  Does this notice supplement a prior notice of postpetition fees, expenses, and charges?  No Pays. Date of the last notice:  Itemize the fees, expenses, and charges incurred on the debtor's mortgage account after the petition was filed. Do not include any escrow account disbursements or any amounts previously itemized in a notice filed in this case or ruled on by the bankruptcy court.  Description  Dates incurred  1. Late charges  1. Late charges  2. Non-sufficient funds (NSF) fees  3. Attorney fees  4. Filing fees and court costs  5. Bankruptcy/Proof of claim fees  6. Appraisal/Broker's price opinion fees  7. Property inspection fees  8. Tax advances (non-escrow)  9. Insurance advances (non-escrow)  10. Property preservation expenses. Specify:  11. Other. Specify:  12. Other. Specify:  13. Other. Specify:  14. Other. Specify:  15. Other. Specify:  16. Other. Specify:  17. Other. Specify:  18. Other. Specify:  19. Other. Specify:		P. C. C.	
Official Form 410S2  Notice of Postpetition Mortgage Fees, Expenses, and Charges 12/15  If the debtor's plan provides for payment of postpetition contractual installments on your claim secured by a security interest in the debtor's principal residence, you must use this form to give notice of any fees, expenses, and charges incurred after the bankruptcy filling that you assent are recoverable against the debtor's principal residence.  File this form as a supplement to your proof of claim. See Bankruptcy Rule 3002.1.  Name of creditor:  Court claim no. (if known):  Last 4 digits of any number you use to identify the debtor's account:  Does this notice supplement a prior notice of postpetition fees, expenses, and charges?  No Yes. Date of the last notice:  Litemize Postpetition Fees, Expenses, and Charges  Itemize the fees, expenses, and charges incurred on the debtor's mortgage account after the petition was filed. Do not include any escrow account disbursements or any amounts previously itemized in a notice filed in this case or ruled on by the bankruptcy court.  Description  Dates incurred  Amount  1. Late charges  2. Non-sufficient funds (NSF) fees  3. Attorney fees  4. Filing fees and court costs  5. Bankruptcy/Proof of claim fees  6. Appraisa/Broker's price opinion fees  7. Property inspection fees  8. Tax advances (non-escrow)  9. Insurance advances (non-escrow)  10. Property preservation expenses. Specify:  11. Other. Specify:  12. Other. Specify:  13. Other. Specify:  14. Other. Specify:  15. Other. Specify:  16. Other. Specify:  17. Other. Specify:  18. Other. Specify:  18. Other. Specify:  19. Other. Specify:		District of(State)	
Notice of Postpetition Mortgage Fees, Expenses, and Charges  If the debtor's plan provides for payment of postpetition contractual installments on your claim secured by a security interest in the debtor's principal residence, you must use this form to give notice of any fees, expenses, and charges incurred after the bankruptcy filling that you assert are recoverable against the debtor or against the debtor's principal residence.  File this form as a supplement to your proof of claim. See Bankruptcy Rule 3002.1.  Name of creditor:  Court claim no. (if known):  Last 4 digits of any number you use to identify the debtor's account:  Doos this notice supplement a prior notice of postpetition fees, expenses, and charges?  No Yes. Date of the last notice:  Part 1:  Itemize Postpetition Fees, Expenses, and Charges  Itemize the fees, expenses, and charges incurred on the debtor's mortgage account after the petition was filed. Do not include any escrow account disbursements or any amounts previously itemized in a notice filed in this case or ruled on by the bankruptcy court.  Description  Dates incurred  Amount  1. Late charges  (1) \$  2. Non-sufficient funds (NSF) fees  2. Non-sufficient funds (NSF) fees  3. Attomey fees  4. Filling fees and court costs  5. Bankruptcy/Poof of claim fees  6. Appraisal/Broker's price opinion fees  7. Property inspection fees  7. Property inspection fees  8. Tax advances (non-escrow)  9. Insurance advances (non-escrow)  10. Property preservation expenses. Specify:  11. Other. Specify:  12. Other. Specify:  13. Other. Specify:  14. Other. Specify:  15. Other. Specify:  16. Other. Specify:  17. Other. Specify:  18. Other. Specify:  19. Other	Case number		
Notice of Postpetition Mortgage Fees, Expenses, and Charges  If the debtor's plan provides for payment of postpetition contractual installments on your claim secured by a security interest in the debtor's principal residence, you must use this form to give notice of any fees, expenses, and charges incurred after the bankruptcy filling that you assert are recoverable against the debtor or against the debtor's principal residence.  File this form as a supplement to your proof of claim. See Bankruptcy Rule 3002.1.  Name of creditor:  Court claim no. (if known):  Last 4 digits of any number you use to identify the debtor's account:  Doos this notice supplement a prior notice of postpetition fees, expenses, and charges?  No Yes. Date of the last notice:  Part 1:  Itemize Postpetition Fees, Expenses, and Charges  Itemize the fees, expenses, and charges incurred on the debtor's mortgage account after the petition was filed. Do not include any escrow account disbursements or any amounts previously itemized in a notice filed in this case or ruled on by the bankruptcy court.  Description  Dates incurred  Amount  1. Late charges  (1) \$  2. Non-sufficient funds (NSF) fees  2. Non-sufficient funds (NSF) fees  3. Attomey fees  4. Filling fees and court costs  5. Bankruptcy/Poof of claim fees  6. Appraisal/Broker's price opinion fees  7. Property inspection fees  7. Property inspection fees  8. Tax advances (non-escrow)  9. Insurance advances (non-escrow)  10. Property preservation expenses. Specify:  11. Other. Specify:  12. Other. Specify:  13. Other. Specify:  14. Other. Specify:  15. Other. Specify:  16. Other. Specify:  17. Other. Specify:  18. Other. Specify:  19. Other			
If the debtor's plan provides for payment of postpetition contractual installments on your claim secured by a security interest in the debtor's principal residence, you must use this form to give notice of any fees, expenses, and charges incurred after the bankruptcy filling that you assert are recoverable against the debtor or against the debtor's principal residence.    Name of creditor:	Official Form 410S2		
debtor's principal residence, you must use this form to give notice of any fees, expenses, and charges incurred after the bankruptcy filing that you assert are recoverable against the debtor or against the debtor's principal residence.    Name of creditor:	Notice of Postpetition Morto	gage Fees, Expenses, an	d Charges 12/15
Name of creditor: Court claim no. (if known):  Last 4 digits of any number you use to identify the debtor's account:  Does this notice supplement a prior notice of postpetition fees, expenses, and charges?  No Yes. Date of the last notice://  Itemize Postpetition Fees, Expenses, and Charges  Itemize the fees, expenses, and charges incurred on the debtor's mortgage account after the petition was filed. Do not include any escrow account disbursements or any amounts previously itemized in a notice filed in this case or ruled on by the bankruptcy court.  Description Dates incurred Amount  1. Late charges (1) \$  2. Non-sufficient funds (NSF) fees (2) \$  3. Attorney fees (3) \$  4. Filing fees and court costs (4) \$  5. Bankruptcy/Proof of claim fees (5) \$  6. Appraisal/Broker's price opinion fees (6) \$  7. Property inspection fees (6) \$  8. Tax advances (non-escrow) (8) \$  9. Insurance advances (non-escrow) (9) \$  10. Property preservation expenses. Specify:  11. Other. Specify:  12. Other. Specify:  13. Other. Specify:  14. Other. Specify:  15. Other. Specify:  16. Appraisal/Broker's price opinion fees (9) \$  17. Other. Specify:  18. Other. Specify:  19. Other. Specify:	debtor's principal residence, you must use this form to g	give notice of any fees, expenses, and charges in	
Last 4 digits of any number you use to identify the debtor's account:  Does this notice supplement a prior notice of postpetition fees, expenses, and charges?  No Yes. Date of the last notice://  Part 1: Itemize Postpetition Fees, Expenses, and Charges  Itemize the fees, expenses, and charges incurred on the debtor's mortgage account after the petition was filed. Do not include any escrow account disbursements or any amounts previously itemized in a notice filed in this case or ruled on by the bankruptcy court.  Description Dates incurred Amount  1. Late charges (1) \$ 2. Non-sufficient funds (NSF) fees (2) \$ 3. Attorney fees (3) \$ 4. Filing fees and court costs (4) \$ 5. Bankruptcy/Proof of claim fees (5) \$ 6. Appraisal/Broker's price opinion fees (6) \$ 7. Property inspection fees (6) \$ 8. Tax advances (non-escrow) (8) \$ 9. Insurance advances (non-escrow) (9) \$ 10. Property preservation expenses. Specify:	File this form as a supplement to your proof of claim. Se	e Bankruptcy Rule 3002.1.	
Last 4 digits of any number you use to identify the debtor's account:  Does this notice supplement a prior notice of postpetition fees, expenses, and charges?  No Yes. Date of the last notice://  Part 1: Itemize Postpetition Fees, Expenses, and Charges  Itemize the fees, expenses, and charges incurred on the debtor's mortgage account after the petition was filed. Do not include any escrow account disbursements or any amounts previously itemized in a notice filed in this case or ruled on by the bankruptcy court.  Description Dates incurred Amount  1. Late charges (1) \$ 2. Non-sufficient funds (NSF) fees (2) \$ 3. Attorney fees (3) \$ 4. Filing fees and court costs (4) \$ 5. Bankruptcy/Proof of claim fees (5) \$ 6. Appraisal/Broker's price opinion fees (6) \$ 7. Property inspection fees (6) \$ 8. Tax advances (non-escrow) (8) \$ 9. Insurance advances (non-escrow) (9) \$ 10. Property preservation expenses. Specify:			
Does this notice supplement a prior notice of postpetition fees, expenses, and charges?    No	Name of creditor:	Court claim no. (if kno	own):
expenses, and charges?  No Yes. Date of the last notice://  Itemize Postpetition Fees, Expenses, and Charges  Itemize the fees, expenses, and charges incurred on the debtor's mortgage account after the petition was filed. Do not include any escrow account disbursements or any amounts previously itemized in a notice filed in this case or ruled on by the bankruptcy court.  Description Dates incurred Amount  1. Late charges			
Part 1: Itemize Postpetition Fees, Expenses, and Charges  Itemize the fees, expenses, and charges incurred on the debtor's mortgage account after the petition was filed. Do not include any escrow account disbursements or any amounts previously itemized in a notice filed in this case or ruled on by the bankruptcy court.  Description Dates incurred Amount  1. Late charges (1) \$ 2. Non-sufficient funds (NSF) fees (2) \$ 3. Attorney fees (3) \$ 4. Filling fees and court costs (4) \$ 5. Bankruptcy/Proof of claim fees (5) \$ 6. Appraisal/Broker's price opinion fees (6) \$ 7. Property inspection fees (6) \$ 8. Tax advances (non-escrow) (8) \$ 9. Insurance advances (non-escrow) (9) \$ 10. Property preservation expenses. Specify: (10) \$ 11. Other. Specify: (11) \$ 12. Other. Specify: (12) \$ 13. Other. Specify: (13) \$ 13. Other. Specify: (13) \$ 14. Other. Specify: (13) \$ 15. Other. Specify: (13) \$ 15. Other. Specify: (13) \$ 16. Other. Specify: (13) \$ 17. Other. Specify: (14) \$ 18. Other. Specify: (15) \$ 19. Other. Specify: (15) \$ 19. Other. Specify: (16) \$ 19. Other. Specify: (17) \$ 10. Other. Specify: (18) \$ 11. Other. Specify: (19) \$ 12. Other. Specify: (19) \$ 13. Other. Specify: (19) \$ 14. Other. Specify: (19) \$ 15. Other. Specify: (19) \$ 16. Other. Specify: (19) \$ 17. Other. Specify: (19) \$ 18. Other. Specify: (19) \$ 19. Other. Specify:		stpetition fees,	
Itemize Postpetition Fees, Expenses, and Charges	□ No		
ltemize the fees, expenses, and charges incurred on the debtor's mortgage account after the petition was filed. Do not include any escrow account disbursements or any amounts previously itemized in a notice filed in this case or ruled on by the bankruptcy court.    Description   Dates incurred   Amount	Yes. Date of the last notice://		
ltemize the fees, expenses, and charges incurred on the debtor's mortgage account after the petition was filed. Do not include any escrow account disbursements or any amounts previously itemized in a notice filed in this case or ruled on by the bankruptcy court.    Description   Dates incurred   Amount	Don't de l'Annaire Destructive France France	va di Chianna	
Description         Dates incurred         Amount           1. Late charges         (1) \$	Itemize Postpetition Fees, Expenses, a	and Charges	
1. Late charges       (1) \$         2. Non-sufficient funds (NSF) fees       (2) \$         3. Attorney fees       (3) \$         4. Filing fees and court costs       (4) \$         5. Bankruptcy/Proof of claim fees       (5) \$         6. Appraisal/Broker's price opinion fees       (6) \$         7. Property inspection fees       (7) \$         8. Tax advances (non-escrow)       (8) \$         9. Insurance advances (non-escrow)       (9) \$         10. Property preservation expenses. Specify:			
2. Non-sufficient funds (NSF) fees       (2) \$         3. Attorney fees       (3) \$         4. Filing fees and court costs       (4) \$         5. Bankruptcy/Proof of claim fees       (5) \$         6. Appraisal/Broker's price opinion fees       (6) \$         7. Property inspection fees       (7) \$         8. Tax advances (non-escrow)       (8) \$         9. Insurance advances (non-escrow)       (9) \$         10. Property preservation expenses. Specify:       (10) \$         11. Other. Specify:       (11) \$         12. Other. Specify:       (12) \$         13. Other. Specify:       (13) \$	Description	Dates incurred	Amount
2. Non-sufficient funds (NSF) fees       (2) \$         3. Attorney fees       (3) \$         4. Filing fees and court costs       (4) \$         5. Bankruptcy/Proof of claim fees       (5) \$         6. Appraisal/Broker's price opinion fees       (6) \$         7. Property inspection fees       (7) \$         8. Tax advances (non-escrow)       (8) \$         9. Insurance advances (non-escrow)       (9) \$         10. Property preservation expenses. Specify:       (10) \$         11. Other. Specify:       (11) \$         12. Other. Specify:       (12) \$         13. Other. Specify:       (13) \$	1. Late charges		(1) \$
3. Attorney fees       (3) \$         4. Filling fees and court costs       (4) \$         5. Bankruptcy/Proof of claim fees       (5) \$         6. Appraisal/Broker's price opinion fees       (6) \$         7. Property inspection fees       (7) \$         8. Tax advances (non-escrow)       (8) \$         9. Insurance advances (non-escrow)       (9) \$         10. Property preservation expenses. Specify:       (10) \$         11. Other. Specify:       (11) \$         12. Other. Specify:       (12) \$         13. Other. Specify:       (13) \$	, and the second se		
4. Filing fees and court costs       (4) \$         5. Bankruptcy/Proof of claim fees       (5) \$         6. Appraisal/Broker's price opinion fees       (6) \$         7. Property inspection fees       (7) \$         8. Tax advances (non-escrow)       (8) \$         9. Insurance advances (non-escrow)       (9) \$         10. Property preservation expenses. Specify:       (10) \$         11. Other. Specify:       (11) \$         12. Other. Specify:       (12) \$         13. Other. Specify:       (13) \$			(3) \$
6. Appraisal/Broker's price opinion fees       (6) \$         7. Property inspection fees       (7) \$         8. Tax advances (non-escrow)       (8) \$         9. Insurance advances (non-escrow)       (9) \$         10. Property preservation expenses. Specify:       (10) \$         11. Other. Specify:       (11) \$         12. Other. Specify:       (12) \$         13. Other. Specify:       (13) \$	Filing fees and court costs		_
6. Appraisal/Broker's price opinion fees       (6) \$         7. Property inspection fees       (7) \$         8. Tax advances (non-escrow)       (8) \$         9. Insurance advances (non-escrow)       (9) \$         10. Property preservation expenses. Specify:       (10) \$         11. Other. Specify:       (11) \$         12. Other. Specify:       (12) \$         13. Other. Specify:       (13) \$	Bankruptcy/Proof of claim fees		
8. Tax advances (non-escrow)       (8) \$         9. Insurance advances (non-escrow)       (9) \$         10. Property preservation expenses. Specify:       (10) \$         11. Other. Specify:       (11) \$         12. Other. Specify:       (12) \$         13. Other. Specify:       (13) \$			
8. Tax advances (non-escrow)       (8) \$         9. Insurance advances (non-escrow)       (9) \$         10. Property preservation expenses. Specify:       (10) \$         11. Other. Specify:       (11) \$         12. Other. Specify:       (12) \$         13. Other. Specify:       (13) \$	7. Property inspection fees		
9. Insurance advances (non-escrow)       (9) \$         10. Property preservation expenses. Specify:       (10) \$         11. Other. Specify:       (11) \$         12. Other. Specify:       (12) \$         13. Other. Specify:       (13) \$			
10. Property preservation expenses. Specify:       (10) \$         11. Other. Specify:       (11) \$         12. Other. Specify:       (12) \$         13. Other. Specify:       (13) \$			
11. Other. Specify:       (11) \$         12. Other. Specify:       (12) \$         13. Other. Specify:       (13) \$	,		
12. Other. Specify:       (12) \$         13. Other. Specify:       (13) \$			<b>6</b>
13. Other. Specify:			

The debtor or trustee may challenge whether the fees, expenses, and charges you listed are required to be paid. See 11 U.S.C.  $\S$  1322(b)(5) and Bankruptcy Rule 3002.1.

Debtor 1				Case number (if known)
	Eiret Name	Middle Name	Last Name	

Part 2: Si	gn Here									
The persor telephone		Notice must sign it.	Sign and pri	int your nan	ame and your title, if any, and state your address and					
Check the a	ppropriate box.									
☐ I am the	☐ I am the creditor.									
☐ I am the	creditor's author	ized agent.								
I declare u of my kno	I declare under penalty of perjury that the information provided in this claim is true and correct to the best of my knowledge, information, and reasonable belief.									
<b>)</b>	ξ				Date / /					
	Signature									
Print:	First Name	At duly Norma	Lock Name		Title					
	First Name	Middle Name	Last Name							
Company										
Address										
Addicas	Number	Street								
	City		State	ZIP Code	-					
Contact phone	( )				Email					

Official Form 410, *Proof of Claim*, applies in all cases. Form 410 replaces Official Form 10, Proof of Claim. It is renumbered to distinguish it from the forms used by debtors for case opening, and includes stylistic changes throughout the form. It is revised as part of the Forms Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. Because the goals of the Forms Modernization Project include improving the interface between technology and the forms so as to increase efficiency and reduce the need to produce the same information in multiple formats, many of the open-ended questions and multiple-part instructions have been replaced with more specific questions.

Official Form 410 has been substantially reorganized. A new question has been added at line 10 that solicits information about claims based on leases.

Official Form 410A, Mortgage Proof of Claim Attachment, is revised in its content and format. Rather than requiring a home mortgage claimant to fill in blanks with itemized information about the principal, interest, and fees due as of the petition date and the amount necessary to cure a prepetition default, the form now requires the claimant to provide a loan history that reveals when payments were received, how they were applied, when fees and charges were incurred, and when escrow charges were satisfied. Because completion of the revised form can be automated, it will permit claimants to comply with Rule 3001(c)(2)(C) with efficiency and accuracy. Attachment of a loan history with a home mortgage proof of claim will also provide transparency about the basis for the claimant's calculation of the claim and arrearage amount.

The loan history should begin with the first date on which the borrower failed to make a payment in accordance with the terms of the note and mortgage, unless the note was subsequently brought current with no principal, interest, fees, escrow payments, or other charges immediately payable.

Official Forms 410S1 and 410S2, *Notice of Mortgage Payment Change* and *Notice of Postpetition Mortgage Fees, Expenses, and Charges*, are revised as part of the Forms Modernization Project. There are formatting changes throughout the forms.

## Form 416A. CAPTION (FULL)

# United States Bankruptcy Court

Case No.
Chapter

[Designation of Character of Paper]

Official Form 416A, *Caption*, applies on all forms where prescribed. Form 416A replaces Official Form 16A, *Caption*. It is renumbered as part of the Forms Modernization Project.

# Form 416B CAPTION (Short Title)

(May be used if 11 U.S.C. § 342(c) is not applicable)

# United States Bankruptcy Court

	District Of	
In re	,	Case No.
	Debtor	
		Chapter

[Designation of Character of Paper]

Official Form 416B, *Caption*, applies on all forms where prescribed. Form 416B replaces Official Form 16B, *Caption*. It is renumbered as part of the Forms Modernization Project.

# Form 416D Caption for Use in Adversary Proceeding

# United States Bankruptcy Court

			District Of		
In re					Case No.
	Debtor				Chapter
	Plaintiff				
	Defenda	ant		,	Adv. Proc. No.

COMPLAINT [or other designation]

[If in a Notice of Appeal and Statement of Election (see Form 417A) or other notice filed and served by the debtor, the caption must be altered to include the debtor's address and Employer's Tax Identification Numbers(s) or last four digits of Social Security Number(s) as in Form 416A]

Official Form 416D, Caption for Use in Adversary Proceeding Other Than for a Complaint Filed by a Debtor, applies on all forms where prescribed. Form 416D replaces Official Form 16D, Caption for Use in Adversary Proceeding Other Than for a Complaint Filed by a Debtor. It is renumbered as part of the Forms Modernization Project.

[Caption as in Form 416A, 416B, or 416D, as appropriate]

## **NOTICE OF APPEAL AND STATEMENT OF ELECTION**

# Part 1: Identify the appellant(s)

1.	Name(	Name(s) of appellant(s):		
Position of appellant(s) in the adversary proceeding or bankruptcy case that is the appeal:			roceeding or bankruptcy case that is the subject of this	
	For ap	peals in an adversary	proceeding.	For appeals in a bankruptcy case and not in an adversary proceeding.
	☐ Def			Debtor
	☐ Oth	er (describe)		
				☐ Trustee
				Other (describe)
Part 2	2: Ider	ntify the subject o	f this appe	<u>eal</u>
1.	Descri	be the judgment, order	, or decree a	ppealed from:
2.	State t	he date on which the ju	udgment, ord	er, or decree was entered:
Part 3	3: Iden	tify the other part	ies to the	<u>appeal</u>
			-	r, or decree appealed from and the names, addresses, additional pages if necessary):
1.	Party:		Attorney: _	<del></del>
			-	
			-	
2	Darty:		Attorney:	
۷.	raity.		Allomey	<del></del>
			_	
			_	<del>-</del>

# Part 4: Optional election to have appeal heard by District Court (applicable only in certain districts)

If a Bankruptcy Appellate Panel is available in this judicial district, the Bankruptcy Appellate Panel will
hear this appeal unless, pursuant to 28 U.S.C. § 158(c)(1), a party elects to have the appeal heard by the
United States District Court. If an appellant filing this notice wishes to have the appeal heard by the
United States District Court, check below. Do not check the box if the appellant wishes the Bankruptcy
Appellate Panel to hear the appeal.

Appellant(s) elect to have the appeal heard by the United States District Court rather than by
the Bankruptcy Appellate Panel.

## Part 5: Sign below

	Date:	
Signature of attorney for appellant(s) (or appellant(s) if not represented by an attorney)		
Name, address, and telephone number of attorney (or appellant(s) if not represented by an attorney):		

Fee waiver notice: If appellant is a child support creditor or its representative and appellant has filed the form specified in  $\S$  304(g) of the Bankruptcy Reform Act of 1994, no fee is required.

The form number is updated to comport with the form numbering style developed as part of the Forms Modernization Project. Other stylistic changes were made throughout the form.

[Caption as in Form 416A, 416B, or 416D, as appropriate]

# OPTIONAL APPELLEE STATEMENT OF ELECTION TO PROCEED IN DISTRICT COURT

This form should be filed only if all of the following are true:

- this appeal is pending in a district served by a Bankruptcy Appellate Panel,
- the appellant(s) did not elect in the Notice of Appeal to proceed in the District Court rather than in the Bankruptcy Appellate Panel,
- no other appellee has filed a statement of election to proceed in the district court, and
- you elect to proceed in the District Court.

1. Name(s) of appellee(s):

## Part 1: Identify the appellee(s) electing to proceed in the District Court

<del></del>	<del></del>			
<ol><li>Position of appellee(s) in the adversary pro appeal:</li></ol>	Position of appellee(s) in the adversary proceeding or bankruptcy case that is the subject of this appeal:			
For appeals in an adversary proceeding.  Plaintiff	For appeals in a bankruptcy case and not in an adversary proceeding.			
Defendant	☐ Debtor			
Other (describe)	☐ Creditor			
	☐ Trustee			
	Other (describe)			
	rd by the District Court (applicable only in			
<u>certain districts)</u>				
I (we) elect to have the appeal heard by the Un Appellate Panel.	ited States District Court rather than by the Bankruptcy			
Part 3: Sign below				
	Date:			
Signature of attorney for appellee(s) (or appellee(s) if not represented by an attorney)				
Name, address, and telephone number of attorney				
(or appellee(s) if not represented by an attorney):				

The form number is updated to comport with the form numbering style developed as part of the Forms Modernization Project. Other stylistic changes were made throughout the form.

[This certification must be appended to your brief if the length of your brief is calculated by maximum number of words or lines of text rather than number of pages.]

# Certificate of Compliance With Rule 8015(a)(7)(B) or 8016(d)(2)

This br	ief complies with the type-volume limitation of Rule 8015(a)(7)(B) or 8016(d)(2) because:			
	this brief contains [state the number of] words, excluding the parts of the brief exempted by Rule 8015(a)(7)(B)(iii) or 8016(d)(2)(D), or			
	this brief uses a monospaced typeface having no more than 10½ characters per inch and contains [state the number of] lines of text, excluding the parts of the brief exempted by Rule 8015(a)(7)(B)(iii) or 8016(d)(2)(D).			
Signati	Date:ure			
Print name of person signing certificate of compliance:				

The form number is updated to comport with the form numbering style developed as part of the Forms Modernization Project. Other stylistic changes were made throughout the form.

Fill in this information to identify the case:				
Debtor 1	First Name	Middle Name	Last Name	
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name	
United States E	Bankruptcy Court for the:		District of(State)	
Case number (If known)			<del></del>	

#### Official Form 423

## Certification About a Financial Management Course

12/15

If you are an individual, you must take an approved course about personal financial management if:

- you filed for bankruptcy under chapter 7 or 13, or
- you filed for bankruptcy under chapter 11 and § 1141 (d)(3) does not apply.

In a joint case, each debtor must take the course. 11 U.S.C. §§ 727(a)(11) and 1328(g).

After you finish the course, the provider will give you a certificate. The provider may notify the court that you have completed the course. If the provider does not ify the court, you need not file this form. If the provider does not notify the court, then Debtor 1 and Debtor 2 must each file this form with the certificate number before your debts will be discharged.

- If you filed under chapter 7 and you need to file this form, file it within 60 days after the first date set for the meeting of creditors under § 341 of the Bankruptcy Code.
- If you filed under chapter 11 or 13 and you need to file this form, file it before you make the last payment that your plan requires or before you file a motion for a discharge under § 1141(d)(5)(B) or § 1328(b) of the Bankruptcy Code. Fed. R. Bankr. P. 1007(c).

In some cases, the court can waive the requirement to take the financial management course. To have the requirement waived, you must file a motion with the court and obtain a court order.

Part 1: Tell the Court About the Required Course You must check one: ☐ I completed an approved course in personal financial management: Date I took the course MM / DD / YYYY Name of approved provider Certificate number I am not required to complete a course in personal financial management because the court has granted my motion for a waiver of the requirement based on (check one): ☐ Incapacity. I have a mental illness or a mental deficiency that makes me incapable of realizing or making rational decisions Disability. My physical disability causes me to be unable to complete a course in personal financial management in person. by phone, or through the internet, even after I reasonably tried to do so. Active duty. I am currently on active military duty in a military combat zone. Residence. I live in a district in which the United States trustee (or bankruptcy administrator) has determined that the approved instructional courses cannot adequately meet my needs. Part 2: Sign Here I certify that the information I have provided is true and correct.

Signature of debtor named on certificate

Printed name of debtor

MM / DD / YYYY

Official Form 423, Certification About a Financial Management Course, is revised as part of the Forms Modernization Project. The form replaces former Official Form 23, Debtor's Certification of Completion of Postpetition Instructional Course Concerning Personal Financial Management. Form 423 is renumbered to distinguish it from the forms used by non-individual debtors, such as corporations and partnerships.

To make Form 423 easier to understand, legal terms are explained more fully or replaced with commonly understood terms, and the form is reformatted. Part 1, *Tell the Court About the Required Course*, provides definitions for "incapacity" and "disability," rather than providing statutory citations.

A statement is added that, in some cases, the court can waive the requirement to complete the financial management course. To have the requirement waived, the debtor must file a motion with the court and obtain a court order

[Caption as described in Fed. R. Bankr. P. 7010 or 9004(b), as applicable]

# **Certification to Court of Appeals by All Parties** A notice of appeal having been filed in the above-styled matter on \_\_\_\_\_[Date], \_\_\_\_\_ \_\_\_\_\_, and \_\_\_\_\_, [names of all the appellants and all the appellees, if any], who are all the appellants [and all the appellees] hereby certify to the court under 28 U.S.C. § 158(d)(2)(A) that a circumstance specified in 28 U.S.C. § 158(d)(2) exists as stated below. Leave to appeal in this matter: is required under 28 U.S.C. § 158(a) is not required under 28 U.S.C. § 158(a). [If from a final judgment, order, or decree] This certification arises in an appeal from a final judgment, order, or decree of the United States Bankruptcy Court for the \_\_\_\_\_ District of \_\_\_\_\_ entered on \_\_\_\_\_[Date]. [If from an interlocutory order or decree] This certification arises in an appeal from an interlocutory order or decree, and the parties hereby request leave to appeal as required by 28 U.S.C. § 158(a). [The certification shall contain one or more of the following statements, as is appropriate to the circumstances.] The judgment, order, or decree involves a question of law as to which there is no controlling decision of the court of appeals for this circuit or of the Supreme Court of the United States, or involves a matter of public importance. Or The judgment, order, or decree involves a question of law requiring resolution of conflicting decisions. Or

An immediate appeal from the judgment, order, or decree may materially advance the progress of the case or

proceeding in which the appeal is taken.

# Official Form 424 continued (12/15)

Date:

[The parties may include or attach the information specified in Rule 8001.]

Signed:	ed: [If there are more than two signatories, all must sign and provide the information requested below. additional signed sheets if needed.]		
	Attorneys for Appellant and Appellee (or Appellant and Appellee, if not represented by an attorney):		
	Printed names of signers:		
	Addresses:		
	Telephone numbers:	(	()

Official Form 424, Certification to Court of Appeal by All Parties replaces Official Form 24, Certification to Court of Appeal by All Parties. It is revised as part of the Forms Modernization Project, and includes stylistic changes throughout the form.

Fill in this information to identify your case:			
Debtor 1 First Name Middle Name Last Name			
Debtor 2 (Spouse, if filing)		Middle Name	Last Name
United States Bankruptcy Court for the:			District of
Case number(If known)			(State)

# Official Form 427

# Cover Sheet for Reaffirmation Agreement

12/15

Anyone who is a party to a reaffirmation agreement may fill out and file this form. Fill it out completely, attach it to the reaffirmation agreement, and file the documents within the time set under Bankruptcy Rule 4008.

P	art 1: Explain the Re	payment Terms of the Reaffirmation Agreement	
1.	Who is the creditor?		
		Name of the creditor	
2.	How much is the debt?	On the date that the bankruptcy case is filed \$	
		To be paid under the reaffirmation agreement \$	
		\$ per month for months (if fixed into	erest rate)
3.	What is the Annual Percentage Rate (APR) of interest? (See	Before the bankruptcy case was filed	%
	Bankruptcy Code § 524(k)(3)(E).)	Under the reaffirmation agreement	%
4.	Does collateral secure the debt?	☐ No ☐ Yes. Describe the collateral.	
		Current market value \$	
5.	Does the creditor assert that the debt is nondischargeable?	☐ No☐ Yes. Attach an explanation of the nature of the debt	and the basis for contending that the debt is nondischargeable.
6.	Using information from Schedule I: Your Income	Income and expenses reported on Schedules I and J	Income and expenses stated on the reaffirmation agreement
	(Official Form 106I) and Schedule J: Your Expenses (Official Form 106J), fill in the amounts.	6a. Combined monthly income from \$ line 12 of Schedule I	6e. Monthly income from all sources \$ after payroll deductions
		6b. Monthly expenses from line 22c of \$	6f. Monthly expenses — \$
		6c. Monthly payments on all reaffirmed debts not listed on Schedule J	6g. Monthly payments on all reaffirmed debts not included in monthly expenses
		6d. Scheduled net monthly income \$	6h. Present net monthly income \$
		Subtract lines 6b and 6c from 6a.	Subtract lines 6f and 6g from 6e.
		If the total is less than 0, put the number in brackets.	If the total is less than 0, put the number in brackets.

7.	Are the income amounts on lines 6a and 6e different?	☐ No ☐ Yes.	Explain why they are different and	I complete line 10			_
8.	Are the expense amounts on lines 6b and 6f different?	□ No □ Yes.	Explain why they are different and	complete line 10			
9.	Is the net monthly income in line 6h less than 0?	□ No □ Yes.	A presumption of hardship arises (unless the creditor is a credit union).  Explain how the debtor will make monthly payments on the reaffirmed debt and pay other living expenses.  Complete line 10.				
							-
10.	Debtor's certification about lines 7-9		I certify that each explanation on	lines 7-9 is true and correct.			
	If any answer on lines 7-9 is Yes, the debtor must sign here.		×	×			
	If all the answers on lines 7-9 are No, go to line 11.		Signature of Debtor 1	Signature of Debt	or 2 (	Spouse Only in a Joint Case)	
11.	Did an attorney represent the debtor in negotiating the reaffirmation agreement?	☐ Yes.	□ No □ Yes. Has the attorney executed a declaration or an affidavit to support the reaffirmation agreement? □ No □ Yes				
Р	art 2: Sign Here						
	/hoever fills out this form lust sign here.	I certify the	nat the attached agreement is a treentified on this Cover Sheet for I	rue and correct copy of the reaffirma Reaffirmation Agreement.	tion	agreement between the	
		×			Date		
		Signatu	ure			MM / DD / YYYY	
		Printed	d Name				
		_	sk one:				
		_	Debtor or Debtor's Attorney Creditor or Creditor's Attorney				
		_	orealion of orealions Allotties				

Case number (if known)

Debtor 1

First Name

Middle Name

Last Name

Official Form 427, Cover Sheet for Reaffirmation Agreement, is revised and renumbered as part of the Forms Modernization Project. The form replaces former Official Form 27, Reaffirmation Agreement Cover Sheet. To make it easier to understand, the form is reformatted, and legal terms are explained more fully or replaced with commonly understood terms.

The calculation of the debtor's net monthly income is expanded to include the debtor's net monthly income at the time the bankruptcy petition is filed, as well as the debtor's net monthly income at the time of the reaffirmation agreement. Rather than requiring filers to state their relationship to the case, checkboxes are provided for the debtor or the debtor's attorney and for the creditor or the creditor's attorney.

Line 3 of the form has been changed to clarify the requirement to disclose an annual percentage rate of interest. Section 524(k)(3)(E) of the Bankruptcy Code defines the "Annual Percentage Rate" to be disclosed in connection with a reaffirmation agreement. Line 3 of the form now includes a reference to that Code provision, which in appropriate circumstances permits disclosure of the simple interest rate as the Annual Percentage Rate.